OPTIMISING

STORM-WATER DRAINAGE

NETWORKS

Thesis submitted in accordance with the requirements of the University of Liverpool for the degree of Doctor of Philosophy by

GODFREY ALAN WALTERS, M.A. (Cantab), C.Eng.

December 1981.

BEST COPY

AVAILABLE

Poor text in the original thesis.

Ph.D. Thesis. G.A. WALTERS, 1981

OPTIMISING STORM WATER DRAINAGE NETWORKS

SUMMARY

This thesis examines ways in which the design of storm water drainage networks can be optimised and proposes, develops and tests some such methods.

The introduction is followed by a résumé of current design practice and an examination of previous work on the drainage optimisation problem. Methods of estimating the construction cost of a drainage network are detailed and functions proposed for modelling these costs.

The optimisation problem may logically be split into two areas, namely optimising fixed plan networks and optimising variable plan networks. The former involves the simultaneous selection of gradients and diameters for a network of pipes fixed in plan. A new Dynamic Programming model is proposed for this, having several advantages over previously published methods.

The main area of innovation is, however, in optimising variable plan networks. The general plan optimisation problem is seen to be far too complex for solution. However, taking the special case of road drainage networks, two possible modes of optimisation are defined. These are, firstly, the positioning of an unknown number of manholes along a drain running between two fixed manholes, and secondly, the positioning of an unknown number of cross-drains along a road carriageway. Both modes include the simultaneous choice of pipe gradients and diameters.

Models for these modes are proposed, with practical computer programs being developed and tested. Both models use a novel form of Dynamic Programming conceived and developed during this research.

The thesis ends with a brief outline of a Dynamic Programming solution to a rather different variable plan problem, followed by suggestions of areas for further study and conclusions of both a specific and a general nature.

Acknowledgement

I would like to thank all those who have helped and encouraged me during the course of this research. In particular I would like to express my gratitude to the following:

Dr. A.B.Templeman, Senior Lecturer in Civil Engineering at the University of Liverpool, for his thoughtful guidance and supervision.

The Highway Engineering Computer Branch of the Department of Transport for financing the project.

The Department of Engineering Science at the University of Exeter for assistance in the production of this thesis.

Kate, Andrew and Sally for being a loving family in trying circumstances.

CONTENTS

		page
Summary		
Acknowledgement	8	
Notation		
Chapter 1	The objectives	1
Chapter 2	Designing a drainage network	4.
Chapter 3	Related research	15
Chapter 4	Costing a drainage network	23
Chapter 5	The fixed plan optimisation model - MANFIX	32
Chapter 6	Variable plan optimisation	92
Chapter 7	The variable manhole position model - MANVAR	102
Chapter 8	The variable cross-drain position model - CROSSVAR	148
Chapter 9	A further variable plan optimisation problem	187
Chapter 10	Conclusions and areas for further study	192
References		198
Appendix A	Cost calculations based on Spons Architects and	203
	Builders price book 1977	
Appendix B	Cost calculations based on Farrar (ref. 1)	206
Appendix C	Fortran coding for program DPO	210
Appendix D	Fortran coding for program ASSEMB	221
Appendix E	Fortran coding for program MOD	225
Appendix F	Fortran coding for program SORT	239
Appendix G	Fortran coding for program MODEX	243

.

:

LIST OF FIGURES

Number	Title	Page
2.1	Variation of velocity with flow for partially full pipe	11
4.1	A typical drainage element	27
4.2	Cost functions	29
5.1	Tree of a typical drainage network	35
5.2a	N -stage serial system	43
5.20	Section along non-branching drainage run	
- 5.2c	Single stage of a serial drainage system	
5.2d	N -stage serial drainage system	
5•3	Convergence of serial systems	46
5.4a	The basic serial system	4 8
5•4b	System with diameter constrains : non-serial	
5.4c	System with diameter constrains : serial	
5.4d	System with Rational method	
5.5a	Establishing upper bound on pipe level	54
5.50	The design of a stage	
5.6	Design of a stage for basic N -stage serial system	57
5•7	Tracing the solution back through basic N-stage serial system	59
5.8	Drop manholes	60
5•9	Establishing quasi-input state costs, drop manholes	62
5.10	Establishing quasi-input state costs, branched system	65
5.11	Tracing the optimal solution back through a junction	. 66
5.12	Establishing quasi-input state costs, diameter constrained	68
5•13a	Minimum cover design	70
5.13Ъ	Alternative designs for two pipe system	
5.14	Establishing bounds on DP procedure with diameter constraint	72
5.15	Design of a stage with diameter constraint	74
5.16	Tracing solution back : with diameter constraint	75
5.17	The Rational method of design	76
5.18	Subcatchment areas	78
5.19	Proposed MANFIX computer program	82
5.20	Optimal design by DDDP	84
5.21	Optimised design for housing estate at Peterborough (area A)	87
5.22	Optimised design for housing estate at Peterborough (area B)	88
5.23	Optimised design for housing estate at Peterborough : sections	89

Number	Title	Page
6.1	Variable plan modes	94
6.2	Typical highway storm drainage network	98
6.3	Layout of highway drainage networks	100
7.1a	Serial system for skeleton network	108
7 . 1b	Serial system for non-branching run	
7.1c	Ranges of manhole positions	
7.2a	Stages for modified serial system	112
7.23	Stages for a non-branching drainage run	
7.2c	Modified serial system for drainage design	
73	ISDP applied to variable manhole position problem	115
7•4	Trace back for the variable manhole position problem	116
7.5	Establishing bounds on pipe level	119
7.6	Test network 2	122
7.7	Sensitivity of network cost to optimising parameters: DPO	124
7.8	Implementing the MANVAR model	126
7.9	Flow chart for program ASSEMB	128
7.10	Flow chart for program MOD	131
7.11	Skeleton plan for network 3	135
7.12a	A solution using MOD	1 36
7 . 12b	A minimum cover solution	
7.13	Construction cost vs execution time	137
7.14	Sensitivity of network cost to optimising parameters: MOD	139
715a	Network costs for combination of optimising parameters	140
7 . 15b	Sensitivity of network costs to time of entry	
7.16	Design example	142
7.17	Structure of DAPHOP	147
8.1a	Typical network	150
8.1Ъ	Position of single cross-drain	
8.1c	Fibonacci search : network 3	
8.2	Fibonacci search over 88 internal points	153
8.3	Optimal position of a single cross-drain	155
8.4	Polytope search for two cross-drains	158
8.5a	Typical highway network	161
8.50	Discrete cross-drain positions	
8.5c	Typical stage	

8.5d Modified serial system

Number	Title	Page
8.6	Design flows with variable network	165
8.7a	Cross-drain set with branches joining	166
8.7Ъ	Cross-drain sets with a common base	
8.8	Implementing the CROSSVAR model	170
8.9	Flow chart for program SORT	171
8.10	Flow chart for program MODEX - main program	173
8.11	Flow chart for program MODEX - subroutine XDSET	174
8.12a	Network 4	178
8.12b	Network 5	
8.13	Cross-drain positions for network 4 using MODEX	179
8.14	Sensitivity of network cost to cross-drain resolution	184
8 .15a	MODEX solution for network 5	185
8 . 15b	MODEX solution for network 3	
9.1	Connecting sources to single main drain	189
9. 2	Flow chart for MULTICON	1 91
	·	

A	catchment area
b	trench width
с	construction cost for a drainage network
Cb	construction cost for a branch of the network
Ce	construction cost for an element of a network
Co	construction cost for the outfall manhole
C1	cost coefficient - pipe supply
C2	cost coefficient - wheeled excavator
С3	cost coefficient - labour
C4	cost coefficient - granular material
D	pipe internal diameter
DA	diameter of pipe entering manhole
Dmin	smallest permissible pipe diameter
Dus	diameter of largest pipe entering a manhole
đ	decision in a serial system
<u>e</u>	direction vector in optimal search procedure
F1	cost factor for excavation
G	correction matrix in optimal search procedure
a	gradient vector in optimal search procedure
h	drop across drop-manhole
I	rainfall intensity
L	distance between manholes
Lb	length of a branch
Lmax	maximum manhole spacing
Lmin	minimum manhole spacing
N	return period of rainfall
Q	design discharge for pipe
Qf	full flow discharge for pipe
RZ	range of levels for pipe
r	return in a serial system
S	state in a serial system
SP	spacing of possible positions for intermediate manholes
S	pipe slope
smax	maximum pipe slope
smin	minimum pipe slope
t	duration of storm event
tc	time to concentration
te	time of entry
v	velocity of flow in a pipe

Vf	velocity at full flow
Vmax	maximum velocity
Vmin	minimum velocity
W	step length in optimal search procedure
xd	distance from fixed manhole to downstream intermediate manhole
Xu	distance from fixed manhole to upstream intermediate manhole
x	distance from base cross-drain to cross-drain
x	position vector in optimal search procedure
Y	depth of cover over pipe crown
Yav	average depth of cover along pipe
Ymax	maximum depth of cover
Ymin	minimum depth of cover
Yu	depth of cover at upper end of pipe
Y	trench depth
Z	level of pipe
ZA	level of pipe entering manhole
ZB	level of pipe leaving manhole
Zđ	level of pipe at lower end of an element
Zu	level of pipe at upper end of an element
Zus	level of lowest pipe entering a manhole
5	correction vector in optimal search procedure

CHAPTER 1

THE OBJECTIVES

1.1 The Design Problem

1.2 The Research Objectives

- 1 -

CHAPTER 1 The Objectives

1.1 The Design Problem

Large sums of money, in excess of £100m (ref. 1) are spent annually on storm drainage networks in Britain alone. In broad terms storm drainage optimisation aims to ensure that the best value for money is obtained from this investment.

Ideally this requires that cost-benefit analyses be performed for all drainage schemes (see ref. 2) to ensure that the greatest benefit results, but in practice this is seldom done explicitly.

Instead drainage schemes are generally designed to a set of criteria chosen on the basis of experience. Such criteria give an informal balance between cost and public acceptability. It is of interest to note that this form of cost-benefit analysis is implicit in any engineering code of practice or set of design criteria.

The problem facing the designer is thus reduced to that of choosing a drainage scheme to meet all the design criteria whilst satisfying any constraints imposed by local conditions. The wider question of whether the design criteria are optimally suited to his particular problem is generally beyond his terms of reference, although he may occasionally use his "engineering judgment" to modify design criteria locally.

However, even with this reduced design problem, the designer is still left with, in general, an infinite number of possible solutions, all of which meet the design criteria whilst satisfying the constraints. Assuming that all these solutions have the same benefit, the scheme which involves least cost is the best, or optimal, solution.

1.2 The Research Objectives

It is the problem of finding the least cost solution for a drainage design problem, given a set of criteria and constraints, with which the research is concerned. The question of cost is discussed in Chapter 4, and is taken to be the cost of construction of the drainage network expressed in monetary terms. Given sufficient detailed information it could include future maintenance and running costs, but as these are often

- 2 -

estimated to be a fixed proportion of the initial capital cost, their explicit inclusion does not seem justified.

Attention is limited to the system of underground pipes and manholes forming the storm-water drainage network. Excluded are all aspects of water quality and treatment and all effects on natural and artificial water courses downstream of the network outfall. Also excluded is any discussion of flow of storm-water overland before entering the pipe network or of detention storage within or outside the network.

Much of the research relates directly to road drainage as becomes apparent in the discussion of optimal plan layout (Chapter 6). However, an attempt has been made to retain generality wherever possible so that results and conclusions are in many cases relevant to any storm-water or indeed foul sewerage network.

The possibility of optimising drainage design has only arisen since the advent of cheap and readily available electronic computers. Most medium and large design offices have computing facilities available and indeed much analysis of drainage designs is already performed by computer.

The objectives of this research therefore include an investigation of existing methods for storm drainage optimisation, the development of further practical methods for use on a computer, and the implementation and testing of such methods.

The bulk of the research in fact concentrates on optimising the plan layout of particular types of drainage network with practical computer programs being written and tested for these applications.

- 3 -

CHAPTER 2

DESIGNING A DRAINAGE NETWORK

2.1	Principles of Storm Drainage
2.2	Present Practice in Storm Water Drainage Design
2.3	System Constraints
2.3.1	Permissible Pipe Depth
2.3.2	Permissible Pipe Slope
2.3.3	Permissible Flow Velocity
2.3.4	Discharge
2.3.5	Pipe Level Continuity at Manholes
2.3.6	Pipe Diameter Continuity at Manholes
2.3.7	Pipe Diameter
2.3.8	Distance between Manholes
2.4	Glossary of Drainage Terms

- 4 -

.

Chapter 2 Designing a Drainage Network

2.1 Principles of Storm drainage

Storm drainage is provided to reduce nuisance and flood damage from incident rainfall. Flood damage may occur on natural catchments due to prolonged heavy rainfall, but man's influence greatly increases the problem. By changing moorland and forest into well drained agricultural land both the percentage of rain that flows off the land (the percentage runoff) and the speed at which this happens increases. Short, severe storms, which on natural catchments would perhaps be partially absorbed with the remaining runoff spread over a long period, may cause flooding of channels and fields on agricultural land.

The problem becomes far more severe in the urban catchment. High proportions of paved and otherwise largely impermeable areas, such as house-roofs, roads, carparks, footpaths, industrial yards, lead to large percentage runoffs occurring shortly after the rainfall. A very short storm, say a 10 minute cloudburst producing a total of 15 mm of rain, which would be insignificant in the countryside could cause severe local flooding in a town.

For this reason extensive storm drainage networks have been and continue to be built throughout urban areas. Traditionally the philosophy has been to remove the incident rainfall from surfaced areas as quickly as possible. Incidentally, however, thought is now being given to ways of temporarily detaining the runoff as near to the source as possible as a means of economising on the storm drainage network downstream. By slowing down the drainage flows the flood peak further down the system is considerably reduced. This allows the use of smaller pipes, or otherwise inadequate existing sewers, and can prevent damage to the natural watercourses into which storm sewers eventually flow.

Three types of urban sewer exist. There is the foul sewer taking sewage from domestic, industrial and commercial premises to a sewage treatment works or straight out to sea or even into an estuary or river.

- 5 -

There is the storm sewer taking only rainfall. This generally drains to the nearest convenient natural watercourse, but, if it originates from a road or industrial premises, it may lead to some form of treatment works or ponded storage before the water is released. The third type of sewer, rarely installed nowadays, is the combined storm and foul sewer, taking both sewage and rain-water. This is generally provided with storm overflows allowing water to flow out of the network into watercourses when excess flows develop due to storms.

This research concentrates largely on storm sewers, although many of the methods are also applicable to foul sewage networks. Both types may be regarded as "tree-like" networks with the base of the tree at the network outfall. Once the flow has entered the network at a branch it must follow one path and cannot diverge from that path. For this reason combined sewers with storm overflows operational cannot be classified in the same way.

Storm water drainage for new roads is an area of special interest in optimising drainage layout, (See Ch. 6). The design principles are however identical to normal urban storm water drainage.

2.2 Present Practice in Storm Water Drainage Design

It is the optimal design of storm-water drainage systems consisting of tree-like networks of pipes between manholes with which this research is concerned. This section examines how such systems are at present designed.

There are four logical stages:-

- a) Identifying the correct design parameters.
- b) Specifying the plan layout of the network.
- c) Designing the gradients and diameters of the pipes.
- d) Detailed specification of drain and manhole construction.

(a) and (d) are outside the scope of the present research, which thus concentrates on optimising the plan layout, pipe gradients and diameters.

- 6 -

The choice of correct design parameters is usually established by reference to the relevant national Codes of Practice (refs. 3 & 4) or locally based design guidelines. Such parameters would include a measure of the acceptable risk of flooding (generally given as the average period of occurence, or Return Period, of a storm giving flows equal to or greater than those designed for), the minimum acceptable velocity of flow in a pipe, the minimum acceptable cover over the crown of the pipe, and the maximum allowable distance between manholes.

Such Codes of Practice or guidelines would also cover such standard practices as

- (a) keeping drains straight and at constant gradient and diameter between manholes,
- (b) having a manhole at every pipe junction (except for gully connections),
- (c) establishing flow capacity and flow velocity by assuming that pipes flow just full, (i.e. with no surcharge pressure), and by using an acceptable flow formula (e.g. Colebrook -White equations).

The second stage, that of specifying the plan layout of the network gives the designer considerable freedom of choice. He must use good judgement and experience to select from an infinite number of possible layouts one that is reasonably efficient and economical. If he wishes to do so he may select several networks and compare designs based on each. If he has sufficient information he may indeed estimate the likely construction cost of each and select the cheapest, thus performing a very basic optimisation, but this is rarely done.

With the layout specified and the position of all manholes fixed in plan, the designer must now specify the gradients and diameters of all pipes in the network. For this he needs to know the design flow for each pipe. Most drainage design in the U.K. is performed using either the Rational or the T.R.R.L. method for establishing design flows (see ref. 5). The Rational method is explained further in section 5.11.2. Very briefly, it enables the designer to calculate a flow for a pipe which is dependent on the total catchment area for the pipe,

- 7 -

the average percentage runoff, and the time taken for flow to reach the downstream end of the pipe from the most remote part of the catchment.

The designer may now assign a gradient and diameter to each pipe such that its capacity is greater than or equal to its design flow. For an individual pipe there are likely to be several possible solutions. For example a large pipe at a shallow gradient will convey the same flow as a small pipe at a steep gradient. The number of different possible solutions for a network of pipes soon becomes very large indeed. For example with just 10 pipes in the network and with a choice of 3 different diameters for each pipe, 3^{10} or 59049 different possible solutions exist, assuming no design criterion or other constraint is violated.

Standard practice, however, is for the designer to place the pipe as close to the ground surface as is permissible. This could be governed by a minimum cover criterion or by a minimum velocity of flow constraint. The smallest pipe diameter is then chosen that will provide the required flow capacity. This procedure is based on the assumption that the shallowest solution is the cheapest, an erroneous supposition which can lead to designs considerably more expensive than necessary as will be shown in subsequent chapters.

The last part of the design process is the detailed design and specification for the drains and manholes. Although these may influence costs considerably, it is not the author's intention to investigate this part of the design process, except to say that in general the "detailed design" consists of the selection of appropriate standard designs from local or national guidelines.

2.3 System Constraints

The nature of the constraints on the design of a storm drainage network can fundamentally affect the optimisation procedure adopted. It is worth while here considering in some detail each of these possible constraints. They are as follows:

- 8 -

- a) Permissible pipe depth
- b) Permissible pipe slope
- c) Permissible flow velocity
- d) Discharge
- e) Pipe level continuity at manhole
- f) Pipe diameter continuity at manhole
- g) Pipe diameter
- h) Distance between manholes

2.3.1. Permissible Pipe Depth $Ymin \leq Y \leq Ymax$

In all designs a minimum depth of cover is required. This varies depending on the use of the land under which the pipes are to be laid. The current code of practice for sewerage (ref. 3) in the U.K. gives values of Ymin = 0.9 m under fields and gardens and 1.2 m under roads.

Sometimes a maximum depth of cover may be specified. Generally, however, the costs of deep excavation, and the extra requirements of stronger pipes, better bedding or concrete surrounds act to limit the depth. Cost functions can always be provided to reflect these practical costs. Hence, in theory, no strict upper limit need be placed on Y, and Ymax can often be omitted as a constraint.

2.3.2 Permissible pipe slope smin < s < smax

These constraints may sometimes be specified. Since flow is unsurcharged gravity flow, s > 0, but this is a necessary condition of constraint (d) and so need not be specified separately here.

The constraint smin \leq s is generally used where the designer considers it impracticable to lay pipes at slopes less than smin. For example, if the gradient is too small inaccuracies in laying could cause pipes to slope in the wrong direction with possible silting up at low flow conditions and trapping of air and partial surcharging at full-flow conditions. Similarly s \leq smax is a practical condition associated with pipe laying on steep slopes. Trouble can be caused with flexible jointed pipes on steep ground as these can slide down the slope if there is insufficient friction in the pipe bedding, particularly when the trench is being backfilled.

2.3.3. Permissible flow velocity V min <V <V max

There is generally some form of restriction on the velocity of flow in the pipe. This is usually in the form of a restriction on the velocity of flow (Vf) that would occur in the pipe flowing just full, but sometimes it is on the actual flow velocity (V) in the pipe at the design discharge (Q). Assuming that the pipe is being used reasonably efficiently with Q/Qf > 0.25, the full flow velocity will approximate to the design flow velocity as shown in Fig. 2.1.

Restricting the velocity to be above a minimum value is to prevent deposition of solids along the pipe invert, the minimum value generally being taken as 0.7 m/s (ref 3). A maximum flow velocity is to prevent excessive scour on the pipe walls. This can, of course, vary with differing pipe materials, but is often taken to be about 6.0 m/s. Recent experience suggests that this upper limit on velocity is not as important as was once thought (ref 3).

2.3.4. Discharge $Q \leq Qf$

Each pipe in the system must be capable of discharging the design flow at that point without surcharging. The maximum unsurcharged flow down a pipe of given gradient and diameter D occurs when the pipe is flowing with a depth equal to about 0.94 D and is about 1.08 x Qf where Qf is the discharge in the pipe when it flows just full.

For practical purposes however, the maximum discharge is assumed to be Qf. A combination of pipe gradient and diameter must be chosen such that $Qf \ge$ design flow Q.

Q may be explicitly defined at the start of the design as in the case of conventional foul sewerage, or may depend on the pipe network upstream of the point being considered as in the case of storm sewers designed to the Rational (LLoyd- Davies) method (ref. 5), the Transport and Road Research Laboratory (TRRL) method (refs. 5, 6), and most other methods in common use.

2.3.5. Pipe level continuity at manholes Zu < Zus

The outgoing pipe from a manhole must be able to drain completely all the incoming pipes. Hence the outgoing pipe invert level (Zu) must be no higher than the lowest invert level of the incoming pipes (Zus).

- 10 -



VARIATION OF VELOCITY WITH FLOW FOR PARTIALLY FULL PIPE

FIGURE 2.1

Moreover, if the outgoing pipe is flowing full and an incoming pipe is of smaller diameter, the incoming pipe will be submerged and hence surcharged unless the soffit of the incoming pipe is at or above the soffit of the outgoing pipe. This leads to the commonly adopted criterion that $Zu \leq Zus$, where Zu and Zus refer to soffit levels. Sometimes, however, designs are done to the alternative criterion $Zu \leq Zus$ where Zu and Zus are invert levels.

Strictly, both criteria are required if pipe diameters are allowed to reduce across a manhole in a downstream sense (see 2.3.6.). A full statement of the constraint then becomes: the downstream pipe soffit level must not exceed any upstream pipe soffit level, and the downstream pipe invert level must not exceed any upstream pipe invert level.

2.3.6. Pipe diameter continuity at manholes D > Dus

It is common practice to require that the diameter, D, of the outgoing pipe leaving a manhole is at least as big as the diameter, Dus, of any incoming pipe. There is no logical argument for this restriction on the grounds of pipe capacity, as a steep outgoing pipe could have a greater capacity than a larger incoming pipe at a flatter gradient.

It could however be argued that a reduction in pipe diameter at a manhole would increase the likelihood of blockages particularly in a foul or combined system.

2.3.7. Pipe Diameter D must be discrete, available diameter > Dmin

Pipes are only available in discrete diameters. The sizes obtainable depend on the pipe material selected and on the pipe manufacturer. Some guidance can be obtained from the British Standard preferred diameters.

For clayware (ref. 7) these are as follows:

75 mm, 100 mm, 150 mm and then in 75 mm increments to 900 mm. For asbestos-cement (ref. 8) they are in 25 mm increments from

100 to 250 mm, then 300 mm to 1050 mm in increments of 75 mm. For unreinforced concrete pipes (ref. 9) they are 100 mm, then 150 mm to 600 mm in increments of 75 mm.

For prestressed concrete pipes (ref. 10) they are

450 mm to 1200 mm in 75 mm increments, then to 3000 mm in 150 mm increments.

-12-

For pitch-fibre (ref. 11) they are

100 mm to 225 mm in 25 mm increments.

Finally in uPVC (refs. 12 and 13) they are

110 mm, 160 mm, 200 mm, 250 mm, 315 mm 400 mm, 500 mm and 630 mm.

To prevent blockages, there is likely to be a limit to the smallest pipe size permitted for a drain. The current Building Drainage code in the U.K. (ref. 4) restricts drains to be 100 mm or over in diameter. Surface water drains for roads normally have D min = 150 mm.

2.3.8. Distance between Manholes

Where manholes are not required closer together for other reasons they should be spaced at distances not exceeding L max. This is to enable maintenance to be carried out, such as clearing blockages by rodding. L max is usually specified as a figure between 100 and 150 m.

2.4 Glossary of Drainage Terms

Pipe	Either: A pipeline of constant diameter and gradient
	joining two manholes. Pipes are normally straight,
	but for road drainage are sometimes curved in
	plan being at a constant offset from a curving
	road centreline.
	Or: A component of a pipeline.
Manhole	An access chamber provided for maintenance, being for the purposes of this research a real manhole,
	a catchpit, an outfall or a rodding eye.
Diameter	The internal pipe diameter, (or pipe bore).
Invert	The lowest part of the internal pipe cross-section.
Soffit	The highest part of the internal pipe cross-section.
Crown	The highest part of the external pipe cross-section.
	In this research crown and soffit levels are
	considered identical.
French drain	A perforated pipe in a trench backfilled with sand
	or gravel, thus allowing water to enter the trench

<u>Carrier drain</u> A pipe that does not accept water anywhere along its length.

- 13 -

and flow into the pipe.

<u>Gully</u> A grated inlet provided at a low point in a paved area, through which flow is led to a drain.

- <u>Carriageway drain</u> A road drain running parallel to the road centreline and collecting water from the carriageway, either through gullies or as a French Drain.
- <u>Cross-drain</u> A road drain running across a road carriageway. A cross-drain is invariably a carrier drain.
- Outfall The point at which flow leaves the drainage network. For the purposes of this research it may be a manhole belonging to another drainage network or may be a true outfall into an open watercourse, the sea or a treatment works.
- <u>Cover</u> The vertical distance between the crown of a pipe and the ground level.

CHAPTER 3

.

RELATED RESEARCH

3.1		Introduction
3.2		Optimising a Fixed Plan Drainage Network
	3.2.1.	Linear Programming
	3.2.2.	Non-linear Programming
	3.2.3.	Geometric Programming
	3.2.4.	Dynamic Programming
	3.2.5.	Discrete Differential Dynamic Programming
3.3		Optimising the Plan Layout of a Drainage Network
	3.3.1.	Optimal Layout only
	3.3.2.	Combined Layout, Gradient and Diameter
		Optimisation.
3.4		Summary

- 15 -

Chapter 3 Related Research

3.1. Introduction

Previous research into optimal design of storm-water drainage networks can be divided conveniently into two categories:

- a) optimal choice of diameter and vertical alignment of pipes for a network which is fixed in plan.
- b) optimal plan layout of a network.

The former category has received steady attention over the last 15 years and this is summarised in section 3.2.

The latter category has been less well covered with only occasional publications. Work in this area is summarised in section 3.3.

3.2 Optimising a Fixed Plan Drainage Network

To date five techniques of optimisation have been used by various authors in an attempt to find a robust and rigorous method of minimising the cost of a fixed plan drainage network.

These five techniques will be dealt with in turn. They are

- a) Linear Programming (LP)
- b) Non-linear Programming (NLP)
- c) Geometric Programming (GP)
- d) Dynamic Programming (DP)
- e) Discrete Differential Dynamic Programming (DDDP)

Historically, an interest in optimising drainage networks stems from the work of Haith (ref. 14) who used DP in 1966 to optimise sewer and drainage system vertical alignment. DP itself was originated by Bellman (ref 15) in 1957.

Attempts were made to use standard LP techniques and as sophisticated NLP algorithms became available these were also tried.

Recent work has reverted to better DP approaches, with DDDP being developed as an alternative to conventional DP.

- 16 -

3.2.1. Linear Programming

The theory and practical application of LP has been well developed over many years. Thus there are powerful algorithms available for the solution of any optimisation problem that can be linearised.

It is therefore tempting to linearise the drainage network problem, this being the approach adopted by several researchers.

Naturally problems occur with the non-linear nature of the function to be minimised (the objective function) and with the non-linear nature of some constraints on the function (see 2.3). Less obviously, the availability of pipes only in discrete sizes, too, causes trouble.

Dajani, Gemmell and Morlok (ref. 16) split the non-linear objective function into linear segments and developed sets of linear constraints to replace non-linear constraining functions. Later, Dajani and Hasit (ref. 17) adopted mixed integer equations as constraints to handle discrete pipe diameters.

General studies on optimisation of drainage networks by Yletyinen (ref. 18) and by Dobschutz (ref. 19) led them to adopt LP approaches. Again, more recent work by Iman, McCorquodale and Bewtra (Ref. 20), the principal aim of which was to incorporate flood damage costs into the cost functions, adopted LP as the means of optimisation.

3.2.2. Non-linear Programming

With the rapid advance of computers the feasibility of non-linear programming algorithms to deal with large numbers of variables has been widely investigated. Many large scale optimisation problems can now be tackled by NLP algorithms but there are still difficult areas. Principally these occur with discrete functions, discontinuities and non-linear constraints. As these are all features of drainage network optimisation (see section 5.3), it is clear that NLP cannot at present provide a complete answer.

Lemieux, Zech and Delarue (ref. 21) used Rosen's projected gradient method (ref. 22) to optimise a drainage network assuming that the objective function was convex and linearly constrained and that pipes

- 17 -

were available in a continuous range of diameters. The solution was subsequently adapted to include commercial pipe sizes.

Price (Ref. 23) used a quasi-Newton algorithm and developed a method whereby the pipes in a network were adjusted to commerical sizes in a step-by-step approach. This also enabled network dependent design flows to be used. Essentially the method involved optimising the full network using approximate flows and continuous pipe diameters. The furthest upstream pipes were then altered to the nearest commercial diameters, pipe flows were simulated and the network downstream of these pipes optimised again. By repeating the process the optimum solution for the whole network was found. Price found the method insufficiently robust and generally inferior to a DDDP method that he also used. (see 3.2.5.).

3.2.3. Geometric Programming

A somewhat different approach is that of geometric programming (ref. 24) which is described in section 5.3.3.

Wilson (ref. 25) attempted to develop a general purpose tool for optimisation in the building industry using a GP computer model. He used sewer networks as an example to test his model with limited success, the discrete nature of available pipe diameters being a considerable problem. He concluded that the GP technique was "too powerful" for the drainage application and developed instead a tailor-made DP method.

3.2.4. Dynamic Programming

Dynamic programming using discrete values of pipe level is the basis of the present author's current research and is described in detail in Chapter 5.

DP has been applied to fixed plan drainage networks by Haith (ref. 14), Meredith (ref. 26), Merrit and Bogan (ref. 27), Wilson (ref. 25), Walsh and Brown (ref. 28) and recently by Froise and Burges (ref. 29) who incorporate storage elements into the network. Liang (ref. 30) applied DP to a general conduit network.

- 18 -

The above authors have produced models with varying degrees of success and validity. Most conclude that DP is a very effective approach to the drainage optimisation problem due to the serial nature of a drainage network and due to the ability of DP to handle discrete, non-linearly constrained discontinuous functions.

One of the problems with DP is the necessity to define the range of levels within which the pipe must lie at each manhole position. If this range is large and the spacing of discrete levels within it small, then a large number of discrete levels must be considered at each manhole. This leads to large computer storage and execution times.

The present author demonstrates that this can easily be avoided (see Chapter 5), but this apparent requirement for large computer resources led to DP being superseded by DDDP (see 3.2.5.).

Two other points were largely ignored by previous authors. Firstly, the fact that design flows are dependent on the network (see 5.5.5) and that if pipe diameters are constrained not to decrease in a downstream direction, this fundamentally affects the DP method (see 5.5.4 and ref. 31).

3.2.5. Discrete Differential Dynamic Programming

DDDP was developed from DP as a means of reducing computer storage and execution time requirements. Basically DDDP is an iterative DP approach and is described in detail in section 5.13.

It was first introduced into the field of water resources by Heidari, Chow, Kokotovic and Meredith of the University of Illinois (ref 32) and later developed at Illinois by Mays and Yen (ref. 33 and 34) for use with drainage design.

Yen, Tang and Mays produced a model incorporating Rational method design (ref. 35) and introduced the risk of flood damage into the cost function (ref. 36).

Mays and Wenzel (ref. 37) restructured the DDDP by redefining the basic stage in the serial system. The concept of isonodal lines (see 3.3 and ref. 38) was introduced, these being lines joining points an equal number of manholes upstream from the outfall. A stage then becomes the design of the network between isonodal lines. This was claimed to be more

- 19 -

efficient than previous DDDP approaches.

Nopgomol and Askew (ref. 39) further developed DDDP or, as they called it, Incremental Dynamic Programming, within the general water resources context. They developed Multilevel Incremental Dynamic Programming to enable problems of higher dimensionality to be tackled by DDDP than were previously feasible.

Chow, Maidement and Tauxe (ref 40) compared the execution times for DP and DDDP programs used for drainage network design.

Price (ref. 23) adapted a DDDP method to allow for network dependent design flows.

3.3 Optimising the Plan Layout of a Drainage Network

Little research has been reported on optimising the plan layout of drainage networks. The problem is less well defined than the optimisation of fixed plan networks, there being many modes in which plan optimisation could occur (see section 6.1).

Published papers concentrate on particular aspects of layout optimisation, or on particulr types of network and not on a solution to the general problem.

Research can be split roughly into two categories:

- a) finding the optimal layout with pipe diameters and gradients fixed (and therefore suboptimal).
- b) optimising layout, pipe sizes and gradients for a special type of network.

3.3.1. Optimal layout only

Liebman (ref. 41) used a simple search procedure which attempted to improve an initially selected trial layout. All pipes had to be the same pre-determined size and were at predetermined slopes. The search consisted of changing one branch of the network at a time, the change being retained if the network cost was decreased. Flows in the system were fixed for each branch.

- 20 -

Barlow (ref. 42) proposed a heuristic method for establishing the route for major trunk sewers and then shortest-path-through-many-points and shortest-spanning-tree techniques to establish the complete layout.

Lowsley (ref. 43) proposed an implicit enumeration procedure based on defining a trunk sewer. Pipe sizes were fixed and the layout optimised for minimum excavation and pipe costs.

3.3.2. Combined layout, gradient and diameter optimisation

In his work on optimisation in the building industry, Wilson (ref. 25) attempted briefly to apply Geometric Programming to a particular drainage layout optimisation problem. He met with little success due to the large numbers of equality constraints, the problems of coincident manhole positions, and the generally large number of variables and constraints in all but the simplest of problems.

Argaman, Shamir and Spivak (ref. 38) proposed an interesting DP model for a particular type of network. The network consisted of a rectangular mesh of pipes which were defined as either local pipes or main pipes. Local pipes originated from a manhole, but had no connection from it. Hence they did not drain the manhole. Main pipes lead from a manhole, thus draining it. The network was a tree, hence only one main pipe could leave a manhole. Both main and local pipes collected water along their lengths.

Isonodal lines were defined as joining nodes an equal number of manholes upstream of the outfall. The layout optimisation consists of determining which pipes were main and which pipes were local and was performed using DP between isonodal lines.

Even with this special network layout, and with a procedure which was not entirely rigorous, the computational resources required for this method made it impractical.

Mays, Wenzel and Liebman (refs. 33, 44) used DP and DDDP with the concept of isonodal lines to develop a two phase screening model for

- 21 -

practical optimal drainage layout design. The networks used are similar to the type studied by Argaman. Mays states that the method may not find the true global optimum due to the necessity of adopting a somewhat non-rigorous procedure.

3.4 Conclusion

Certain general conclusions can be drawn from the above summary of published research.

For the fixed plan drainage network problem, only those methods involving the use of DP or DDDP have met with any success, and none of these is entirely satisfactory (see Chapter 5). Methods involving LP, NLP or GP cannot deal with the discrete non-linear and discontinuous nature of the problem. Their use involves either oversimplification of the problem or adoption of a sub-optimal procedure.

For the variable plan problem, no rigorous procedure has been published for even the simplest of cases.

CHAPTER 4

COSTING A DRAINAGE NETWORK

4.1	Introduction
4.2	Cost of Measured Work
4.3	Cost Model
4.3.1.	Cost of a Network
4.3.2.	Cost of an Element
4.3.3.	Cost Functions
4.3.4.	Alternative Cost Functions

- 23 -

.

CHAPTER 4

Costing A Drainage Network

4.1 Introduction

A prerequisite of minimum cost design is the ability to cost a design, or at least to compare the relative costs of one design with another.

The cost of a drainage scheme from the point of view of the scheme's promoter would include such items as

- (a) acquisition of land or easements,
- (b) design and supervision costs,
- (c) future likely maintenance and replacement costs
- (d) the final contract costs.

The tendered contract price is the contractor's estimate of the cost of the job plus his profit and consists of

- (a) cost of measured work
- (b) lump sum items such as setting up temporary site buildings, insurance, temporary works and mobilising plant and labour
- (c) profit and head office costs.

In optimising the design, it is the cost of the measured work that one attempts to minimise. On small schemes the measured work may well represent less than half the total cost. However, by minimising the cost of measured work, savings may also be made on some other items, such as maintenance and replacement costs and insurance, but these savings will not generally be directly proportional.

It would be unwise to compare two completely different schemes purely on the basis of the cost of measured work. However, the nature of drainage optimisation is that all schemes compared are generally similar with only slight differences in pipe slopes, diameters and manhole positions. Hence a comparison on the basis of the cost of measured work is usually valid.

4.2 Cost of measured work

Traditionally the prices of measured work, as presented in tender

- 24 -

documents are given as rates per unit for the various items of construction multiplied by the estimated quantity for those items as given in the Bill of Quantities. The total price for measured work is then the sum of the prices calculated for all items. The actual cost of the measured work is found at the end of the contract by measuring all items as constructed and multiplying by the appropriate rates. This assumes that there is no great difference between the quantities as estimated in the Bill of Quantities and the final measurement.

If there is a significant difference, the contractor may have grounds for a claim for extra payment. For example, if the total length laid of a certain large diameter pipe has been reduced from say 200m to 20m, the contractor could argue that the cost of setting up the pipelaying operation is not now being met by the rate quoted in his tender, and that had he known that a much smaller length was to be laid, he would have put in a much higher rate.

Of course, the reverse situation could arise, with the contractor making an unexpected windfall from an increase in quantity of a highly priced item, and on balance the two effects will tend to cancel out.

Returning to the design stage and the problem of comparing the costs of different schemes, one should ideally have a costing model that gives an increased rate per unit for low total quantities of that unit.

This however would be very difficult to achieve due to lack of sufficient data and the variations in individual contractors' working methods. Also the quantities involved in drainage tend to be of sufficient size for this effect to be generally negligible.

4.3 Cost Model

4.3.1 Cost of a Network

In building up a useable and realistic cost model for use in the optimising process two basic assumptions are made:

- The cost of a scheme = The sum of the independent costs of individual parts of the scheme.
- (2) The rates used to calculate costs of individual parts of the scheme are independent of the quantities involved.

It is useful here to define a typical element in a drainage scheme. This can be taken as a length of pipeline between manholes, together with all the associated excavation and backfill, and the manhole immediately upstream of the pipeline. As can be seen from Fig. 5.1, in a network of n pipes where no two pipes have the same upstream manhole, there are n + i manholes (including an outfall manhole) and n elements.

Hence the cost of the total network (C) equals the sum of the element costs (Ce) plus the cost of the outfall (Co)

i.e. $C = \sum_{1}^{n} Ce + Co$ 4.1

4.3.2. Cost of an Element

A typical element is shown in Fig 4.1. Various parameters can be used to define the element for costing purposes. These must include the pipe diameter, pipe type and bedding type, and must also include some measure of the total volume of excavation and the depth of excavation. The upstream manhole diameter and depth are also required, as is some information as to soil type, dewatering requirements, whether a road surface has to be broken up and removed, the degree of reinstatement required, restriction of access to the work, and frequency of other services crossing the trenches.

Farrar (ref. 45) has collected data based on observations of site operations in the UK for laying sewer pipes of up to 600mm. From this he has derived a simple costing procedure, involving most of the above parameters, and hence generally applicable.

A rather less detailed approach can be used based on annually published cost data from the building industry (ref. 46) and a third approach would be to study the prices tendered by contractors for past drainage schemes.

- 26 -


A TYPICAL DRAINAGE ELEMENT

FIGURE 4.1

This last approach is, however, rather unsatisfactory for the following reasons:

- (1) The breakdown of prices is not very detailed.
- (2) The prices quoted do not necessarily reflect the actual costs to the contractor.

4.3.3. Cost Functions

When previous authors on drainage optimisation have quoted the cost function they have used, it has ten been in a generalised form (see Table 4.1). Without knowing the values of the constants in these functions they are of little practical use.

SOURCE	COST/UNIT LENGTH	<u>NOTES</u>
Lemieux, Zech and Delarue. (ref. 21)	$a + bD^n + eV$.	e = unit cost of excavation
Meredith (ref. 26)	10.98D + 0.8H - 5.98	Cost in dollars D, H in feet H is depth to invert.
Dajani and Hasit (ref. 17)	$a + bD^2 + cH^2$	D, H in feet H is depth of excavation.
Barlow (ref. 42)	aV + bD ⁿ	
Wilson (ref. 25)	0.73D + 0.243H - 0.088	H is depth to soffit. D, H in feet.

TABLE 4.1 - PUBLISHED COST FUNCTIONS FOR DRAINAGE

General Notes: a, b, c, n are unspecified constants. D = pipe diameter V = volume of excavation per unit length

Two authors however quote the specific form of their cost functions. These are included in Table 4.1 and are illustrated for two pipe sizes in a dimensionless form in Fig. 4.2. Also illustrated are costs taken from work done by the Hydraulics Research Station (ref. 47) and from a report by the Local Government Operational Research Unit (ref. 2).



For the purposes of this research the author developed a set of cost functions for discrete pipe sizes, based on Spon's Architects and Builders Price Book (ref. 46) and prices quoted by pipe manufacturers. Details of the calculations are given in Appendix A.

In producing these cost functions, the following assumptions were made:

- Structural design of the pipe and bedding was to be in accordance with Department of the Environment recommendations (ref. 48) and was to be for pipes laid in a road carriageway.
- (2) The cheapest satisfactory combination of pipe type and pipe bedding was to be used for a given pipe depth and diameter.
- (3) Average soil conditions applied throughout, with no hard rock or exceptional dewatering requirements.
- (4) There was no breaking up of road surface or reinstatement required.
- (5) There was adequate working room for excavation.
- (6) All excavation was by machine, there being no necessity for hand excavation.
- (7) Surplus fill material could be disposed of on site.

These conditions are generally consistent with drainage schemes for new roads. The main exception would be requirement (3), as variable soil conditions and high water tables could be encountered in cuttings.

The cost functions developed give a rate per unit length for the finished pipeline, and for a given pipe diameter, depend only on the depth of cover (Y) over the pipe. These functions are based on prices in March 1977 and are as follows:

Pipe Diameter (mm)	<u>Cost (£ per m)</u>
150	2.8 + 4.1 Y
225	5.7 + 4.1 Y
300	8.9 + 4.1 Y
375	12.3 + 4.4 Y
450	15.9 + 4.7 Y
525	19.7 + 5.0 Y
600	23.7 + 5.3 Y

As these functions are linear with depth, it is reasonable to take the cost of a pipeline between manholes = $L \times f$ (Yav), where L is the distance between the centres of the manholes and Yav is the average cover along the length of the pipe. The cost of the upstream manhole depends on the depth of the manhole, measured from ground level to the lowest pipe invert, and on the manhole diameter. In turn the manhole diameter is determined by the biggest pipe entering or leaving the manhole. Assuming that pipe diameters cannot decrease down the pipe network (see 2.3.6), the largest pipe must be the pipe leaving the manhole. Now as the lowest invert is that of the outgoing pipe, the manhole cost is determined by the diameter and invert level of the outgoing pipe. But since depth to soffit = depth to invert - diameter of pipe, the cost of the upstream manhole of an element can be taken as f(D, Yu) where Yu = depth of cover atupstream end of the pipe. Hence the total cost of an element is a function of pipe length, pipe diameter, average depth of cover, and depth of cover at the upstream manhole i.e. Ce = f(L, D, Yav, Yu).

The cost of an element as used for this study is thus given below:

Pipe Diameter (mm)	Element Cost (£)
150	(2.8 + 4.1 Yav)L + 30 + 70 Yu
225	(5.7 + 4.1 Yav)L + 30 + 70 Yu
300	(8.9 + 4.1 Yav)L + 30 + 75 Yu
375	(12.3 + 4.4 Yav)L + 30 + 80 Yu
450	(15.9 + 4.7 Yav)L + 30 + 85 Yu
525	(19.7 + 5.0 Yav)L + 30 + 90 Yu
600	(23.7 + 5.3 Yav)L + 30 + 95 Yu

4.3.4. Alternative Cost Function

A more comprehensive set of cost equations was developed based on the work of Farrar (ref. 45) and is included in the final commercial drainage design computer program resulting from this research. Details of these functions are given in Appendix B.

The two sets of cost functions developed for use in this research are compared in Fig. 4.2 with previously published information, using two typical pipe diameters.

CHAPTER 5

THE FIXED PLAN OPTIMISATION MODEL - MANFIX

5.1		Introduction
5.2		Problem Definition
5.3		Optimising the Objective Function
	5.3.1.	The Objective Function
	5.3.2.	The Polytope or Simplex Method
	5.3.3.	Using A Smooth Continuous Objective Function
5.4		A Serial System
5.5		Drainage as a Serial System
	5.5.1.	Introduction
	5.5.2.	The Basic System
	5.5.3.	A Branching System
	5.5.4.	Non-decreasing Pipe Diameter
	5.5.5.	Design Flows that are not Pre-determined
	5.5.6.	Summary
5.6		Optimising a Serial System by Dynamic Programming
	5.6.1.	Introduction
	5.6.2.	Principle of Optimality
	5.6.3.	State Vector Space
	5.6.4.	Dynamic Programming Using Discrete Values
5.7		Optimising a Simple Fixed Plan Drainage Run by Dynamic Programming
	5.7.1.	Upper Bound on State Variable
	5.7.2.	Lower Bound on State Variable
	5.7.3.	Establishing Discrete Values of Level
	5.7.4.	Establishing a Feasible Design for an Element
	5.7.5.	Cost at Each Discrete Input State
	5.7.6.	Cost at Each Discrete Output State '
	5.7.7.	Overall Minimum Cost
	5.7.8.	Optimal Solution
5.8		Inclusion of Drop-Manholes
	5.8.1.	Introduction
	5.8.2.	Defining the Quasi-Input State
	5.8.3.	Tracing Back

5.9		Optimising a Branched Drainage Network
	5.9.1.	Introduction
	5.9.2.	Procedure
	5.9.3.	Tracing Back Through a Branch Junction
5.10		Inclusion of the Constraint on Decreasing Pipe Diameter
	5.10.1.	Introduction
	5.10.2.	Procedure
	5.10.3.	Defining the Range of Diameters and their Discrete Values
	5.10.4.	Organising the Computation
5.11		Dependence of Flows on the Network Design
	5.11.1.	Introduction
	5.11.2.	The Rational or Lloyd-Davies Method
	5.11.3.	Other Methods of Calculating Storm Water Flows
	5.11.4.	The three Dimensional State Vector Approach
	5.11.5.	An Approximate Approach
5.12		The Final Fixed Plan Model - MANFIX
5.13		The Use of Discrete Differential Dynamic Programming
5.14		Experience and Results
	5.14.1.	Introduction
	5.14.2.	Experience of the Model
	5.14.3.	Results
5.15		Cost of Using MANFIX
5.16		Conclusion on the Fixed Layout Model - MANFIX

- 33 -

•. • •

•

Chapter 5

The Fixed Plan Optimisation Model

5.1 Introduction

Whilst the main area of research for the present project was in the field of variable plan networks, it was an essential prerequisite to examine published work on fixed plan models. During this examination it was found that there were shortcomings (ref. 31) in all published methods, and that there was no one approach that seemed entirely satisfactory. Of the methods that were available, Discrete Differential Dynamic Programming (DDDP) seemed to have gained most acceptance and this is discussed in Section 5.13.

During the development of the variable plan models it became clear that a simple fixed plan Dynamic Programming (DP) model, essentially a subset of the variable plan model, could be of interest.

Although a separate computer program for the fixed plan model was not written, the model is presented in this chapter both for completeness and as an introduction to the more complicated variable plan problem. Results, conclusions and the choice of parameters are based on computer runs using the variable plan models (see Chapter 6) on fixed plan problems.

5.2 Problem Definition

Fixed Plan Optimisation represents the most basic level of improvement over current design methods, and is the simplest of the drainage network optimisation problems considered. It is also applicable to virtually all storm drainage networks, and with minor modifications, to foul sewer networks.

The designer specifies the plan layout of the pipes and the positions of all manholes. One tree of a typical network is shown in Figure 5.1. The problem is to find admissible pipe diameters and levels for every pipe in the system so that the total construction cost for the system is as small as possible, whilst all the technological and physical constraints imposed on the system are met.

As an example, consider a network of n pipes between (n + 1) manholes fixed in plan.

- 34 -



•



TREE OF A TYPICAL DRAINAGE NETWORK

FIGURE 5.1

The design of an element i (Figure 4.1) can be defined in terms of the pipe diameter D_i , pipe level at the upstream end Zu_i , and pipe level at the downstream end Zd_i . In general, given Zu_i and Zd_i , the smallest, and hence cheapest, pipe size that will carry, the required flow and satisfy the design constraints, will be chosen. Hence the pipe diameter D_i is dependent on Zu_i and Zd_i and need not be considered as an independent variable. There are thus 2n variables in the problem.

The cost of constructing the pipe element $Ce_i = f(D, Yav, Yu)$ (see 4.3.3), where Yav and D are functions of Zu_i and Zd_i , and Yu is a function of Zu_i . Thus Ce_i is a function of Zu_i and Zd_i for a given design flow and set of ground levels. Hence the problem becomes one of minimising C where

 $C = Ce_{1}(Zu_{1}, Zd_{1}) + Ce_{2}(Zu_{2}, Zd_{2}) + \dots + Ce_{n}(Zu_{i}, Zd_{i}) + \dots + \dots + Ce_{n}(Zu_{n}, Zd_{n})$

subject to the following constraints (see 2.3)

```
Ymin\leqY\leqYmaxsmin\leqs\leqsmaxVmin\leqV\leqVmaxQ\leqQfZu\leqZusD\geqDusDa discrete, available, diameter
```

The problem could indeed be further simplified by specifying that all pipes at a manhole must have the same level. The problem would then reduce to that of finding a pipe level at each of the (n + 1) manholes. There would thus be only (n + 1) variables.

The problem would then become:

Minimise C, where $C = Ce_1(Zu_1, Zd_1) + Ce_2(Zu_2, Zd_2) + ------B$ $+ Ce_1(Zu_1, Zd_1) +Ce_n(Zu_n, Zd_n)$

- 36 -

with the (n - 1) equalities $Zd_j = Zu_k$ (j = 1, n - 1), (where k depends on the connectivity of the network) and the following constraints:

```
Ymin\leqY\leqYmaxsmin\leqs\leqsmaxVmin\leqV\leqVmaxQ\leqQfD\geqDusDa discrete, available, diameter
```

The disadvantage of this approach is that it is far too restrictive for practical drainage networks. Branches joining a main run will, for example, generally join at a much higher level. To restrict them to joining at the main pipe level could incur severe financial penalties.

3

Hence it is better to consider only the general form of the problem as in expression A.

5.3 Optimising the objective function

5.3.1 The objective function

The expression that is to be minimised, expression A, is known as the objective function and is here a non-linear function of 2n variables, where n, the number of pipes in the network, is unlikely to be less than 10 and could be as many as several hundred.

Consider the cost of a typical element $Ce_i(Zu_i, Zd_i)$. For $Zu_i = constant$, consider the range of values of Zd_i . Assuming available pipe diameters are in discrete sizes, there will be discrete values of diameter D_i for different parts of the range Zd_i . Hence there will be jumps in the cost of the element where the required diameter goes from one size to the next. The cost function for an element is thus discontinuous and therefore also non-differentiable.

Even if diameters were available in a continuous range of sizes the cost function for an element could still be discontinuous. This would occur if the cost function truly represented the cost of the different site practices involved in excavating pipe trenches to various depths.

- 37 -

For example, trench supports are not normally required for trench depths up to 1.5m, but trenches deeper than this must be supported for safety. Similar discontinuities with increasing depth could arise from the use of differing classes of pipe, types of bedding design or widths of trench.

The constraints on the optimisation are in the form of both linear and non-linear inequalities (see 2.3). For example, constraints on depth and pipe slope are linear inequalities, whereas those on flow velocities and discharge are non-linear inequalities.

Hence the objective function is a non-linearly constrained multivariable non-linear discontinuous function. There are at present no suitable mathematical techniques available for the general solution of this type of optimisation problem.

5.3.2. The polytope or simplex method

For problems involving a very small number of variables, say less than about 10, a polytope algorithm could possibly be used.

Essentially the polytope technique applied to a problem with m variables involves the following procedure.

- (a) Define the feasible zone of the m dimensional space within which the solution must lie.
- (b) Define (m + 1) points within that space, preferably equally spaced, and evaluate the function at these points.
- (c) Identify the worst (most expensive) points.
- (d) Reflect the worst point through the centroid of the other points to obtain a new point.
- (e) Evaluate the function at the new point, identify the new worst point and repeat from step (d).

The polygon may be expanded or contracted according to various rules. Other rules may also limit the deformity of the polygon and specify the procedure to adopt at constraint boundaries. The process continues until the polygon is reduced to a predetermined size and further iterations produce negligible improvement. Unfortunately there are doubts as to its ability to find the optimal solution for even a moderate number of variables. In addition the process is relatively slow, requiring a large number of function evaluations.

2

Although the technique is known to be very robust, there could well be difficulties encountered in using it with an objective function such as expression A which has a number of large discontinuities corresponding to discrete values of pipe diameter.

5.3.3. Using a smooth continuous objective function

If one ignores the problem of discontinuities outlined in 5.3.1. and treats the function as smoothly continuous, a range of possible solution_techniques emerge, depending on whether first and second derivatives of the function can be evaluated.

Consider first a problem in which there are no constraints. If first derivatives cannot be evaluated, even though they uniquely exist at all points, the minimum could be found by a linear search method using only function evaluations.

All such methods follow the general iteration $\underline{x}_{i+1} = \underline{x}_{i} + w_{i} \cdot \underline{e}_{i}$ where \underline{x}_{i+1} is the improved position and \underline{x}_{i} is the old position of the vector x which defines the position of the search, w_{i} is a step length and \underline{e}_{i} is the direction of the step.

The simplest of all such methods uses each axial direction in turn as the search direction \underline{e}_i , with the step w_i being determined by a linear search along that one direction. The current best point then moves parallel to each axis in turn.

Various algorithms have been developed by Hooke & Jeaves (ref. 49), Rosenbrock (ref. 50), Davies Swann & Campey (ref. 51) and others as improvements to the basic method.

As an alternative to linear search methods, a gradient method could be adopted by using information about the first and sometimes the second derivatives as well as the function values to help determine the direction of search e_i .

- 39 -

The derivatives can be obtained either analytically if suitable formulae are available, or numerically from evaluations of the objective function, although this latter course has the disadvantage of extra function evaluations and possible problems with arithmetic calculation of very small quantities.

The most basic approach is to follow the line of steepest descent until the minimum value of the objective function along that line is reached, whereon a new direction is established and the process is repeated.

When second derivatives are available a far more powerful class of methods known as Newton methods may be used. Around its minimum value, the objective function can be assumed to be approximately quadratic, and for such a function it can be shown by Taylor expansion that the correction $\underline{\delta}$ for which $x + \underline{\delta}$ minimises the function can be written as $\underline{\delta} = -G^{-1}\underline{g}$ where \underline{g} is the gradient vector and G is the matrix with elements Gjk = $\underline{\partial^2 f}$

Hence the iteration becomes $x_{i+1} = x_i - G'g$

Modifications to the basic Newton method involving the use of only first derivatives and only function evaluations have been made, notably by Davidson (ref. 52) and Fletcher and Powell (ref. 53).

These Quasi Newton Algorithms tend to be the most efficient in terms of function evaluations, although if computer storage is critical a conjugate gradient method (ref. 54) may be more suitable.

All the algorithms so far mentioned are for the general unconstrained problem and in particular the storm drainage problem has both non-linear and linear inequality constraints.

One approach taken to such problems is to create penalty functions corresponding to the constraint boundaries so that the value of the function rises rapidly at the constraint thereby prohibiting the minimum value from being beyond the constraint boundary.

- 40 -

The function may then be minimised as an unconstrained problem. However severe problems can occur due to ill conditioning at the boundaries and generally several unconstrained problems have to be solved with varying values of penalty functions to obtain the true optimal solution.

Another approach is again to convert the problem to an unconstrained one, but this time by creating an augmented Lagrangian function (ref. 55). Alternatively the non-linearly constrained problem may be transformed into an equivalent linearly constrained exercise.

A rather different approach is to modify the search direction to avoid entering a non-feasible zone. Such techniques are known as projected gradient techniques. Essentially if the proposed search direction contravenes a constraint, a new direction is adopted, being the projection of the original onto the tangent plane of the constraint.

Yet another approach is that of Geometric Programming (ref. 24). The objective function must be expressed as a posynomial, (a function which is the sum of positive polynomial terms) and constraints should also be posynomial expressions. The method is based on the general geometric inequality theorem, which states that the arithmetic mean of a set of positive terms is always greater than their geometric mean, with equality when all the terms are equal.

Whilst the methods outlined above will, at least in theory, provide optimal solutions for a continuous smooth objective function there may still be severe problems due to lack of robustness particularly with complicated constraints.

The main problem, however, remains: The actual objective function is discontinuous. This could be avoided by allowing pipes to be of any size and ignoring practical discontinuities in the cost function corresponding to site practice or design. After obtaining the optimal solution, the pipe diameters must then be converted in some way to commercially available sizes. Attempts at doing this have met with only limited success (refs. 23, 25).

The conclusion therefore must be that there is no suitable technique for solving the general problem of which storm drainage optimisation is a particular example. It is therefore logical to examine whether storm drainage optimisation is in any useful way different from the

- 41 -

general problem.

5.4 A Serial System

It is convenient here to introduce the concept of a serial system, for which a powerful alternative optimising approach is available.

The essence of such a system is that a quantity S, called the state, passes in one direction through a sequence of stages at each of which it is modified in value by decisions which produce returns. This is illustrated in fig. 5.2(a).

The quantity S has an initial value S_0 which is the input state to stage 1. In stage 1 decisions d_1 are made which change the value of S_0 to S_1 - the output state from stage 1 - and produce a stage return r_1 . S_1 is then the input state for stage 2 at which decisions d_2 are made, producing stage returns r_2 and changing the value of S from S_1 to S_2 . This process of making decisions at each stage which change the value of the state and produce stage returns continues until all N stages have been traversed and the state has a final value S_N .

The serial system must contain no loops. At any particular stage, say stage k, the only information known about the system is the input state S_{k-1} and the details within stage k. Hence the decision made, d_k , the return r_k and the output state S_k can only be influenced by the input state S_{k-1} and not by how that state was achieved (i.e. not by decisions d_1 to d_{k-1}).

5.5 Drainage as a Serial System

5.5.1. Introduction

The design of a drainage network may, with care, be treated as a serial system.

First consider the simplest case. This is a non-branching length of sewer consisting of N pipes between N+1 manholes as shown in Figure 5.2(b). The constraints listed in Chapter 2 section 3 will, in general, apply to the design. These are summarised below for convenience.

- 42 -



)

(b) SECTION ALONG NON-BRANCHING DRAINAGE RUN



(c) SINGLE STAGE OF A SERIAL DRAINAGE SYSTEM



(d) N-STAGE SERIAL DRAINAGE SYSTEM

FIGURE 5.2

a) $Y\min \leq Y \leq Y\max$ b) $s\min \leq s \leq s\max$ c) $V\min \leq V \leq V\max$ d) $Q \leq Q$ full e) $Zu \leq Zus$ f) $D \geq Dus$

g) D is a discrete, available diameter

Constraints a, b, c, e and g present no problems to the concept of drainage as a serial system. Zu and Zus refer to the pipe soffit levels although there is no theoretical argument against using the invert or pipe centre line as the reference for the pipe levels.

3

Constraint (d) raises the question of the design flow Q. For simplicity first assume that all design flows are known before the design starts. This is generally not the case for storm-water drainage and is a question which will be considered later (see section 5.5.5).

Constraint (f) fundamentally changes the nature of the system and so the system will be considered with or without this constraint. Initially, consider the simpler case of the constraint not applying.

5.5.2. The basic system

Consider a single stage of the system as shown in Figure 5.2 (c). This consists of a pipe together with its upstream manhole. The complete system consists of N such stages starting from stage 1 at the upstream end of the sewer and ending in stage N which has a downstream manhole as well as the usual upstream one. Let the input state to stage K be the soffit level of the pipe entering the upstream manhole Zus_K . One can now make a decision on pipe levels and diameter for this stage based on the input state and design flow such that all constraints are satisfied. There will in general be many possible decisions. The choice of pipe levels and diameter will incur a return for the stage which is here considered to be the construction cost of the element, and will produce an output state forms the input state to the next, (K + 1)th, stage.

The serial nature of the system is shown diagrammatically in Figure 5.2(d). Note that $Zus_{K} \equiv Zd_{K-1}$ and that the input state is not the

- 44 -

level of the pipe leaving the upstream manhole but merely the highest level at which that pipe could be set. Hence the decision on pipe levels can involve a change in level or 'drop' across a manhole. For an isolated drainage run, the input state to stage 1, Zd₀ and the output state from stage N, Zd_N are not required, but where the run forms part of a larger network they will be used, as outlined in the following section.

5.5.3. A branching system

Having shown that a simple non-branching sewer can be treated as a serial system, it is now necessary to consider a branching system.

Drainage systems are typically arranged as tree-like networks as shown in Figure 5.1. There are no loops, at least not in newly designed networks, although old existing systems often have cross connections and diverging flows.

As it is the design of new networks under consideration it is reasonable to assume that there are no loops and that sewers never diverge, but always converge. The convergence of two serial systems is illustrated in Figure 5.3(a). The only complication is that the input to stage K, has to be determined from both the ouput state from stage A_{K-1} and from the output state from stage B_N . This is done by redefining the input state as the soffit level of the lowest pipe entering the upstream manhole. It is in fact rather more convenient to rearrange the serial system slightly as shown in Figure 5.3(b). Instead of one sewer joining a main sewer, we now have two sewers leading into a third sewer. These are exactly equivalent but the latter arrangement is easier to handle computationally.

5.5.4 Non-decreasing pipe diameter

As mentioned in section 5.5.1., constraint (f) fundamentally changes the nature of the serial system. This fact has not generally been recognised by previous authors (e.g. refs. 28, 34, 44) and has led to the use of incorrect algorithms (ref. 31).

Consider first the basic system as described in 5.5.2. and illustrated in Figure 5.4(a). At stage K, information about the upstream system (stages 1 to K-1) is conveyed purely by the state variable Z. A decision

```
- 45 -
```



2

••

<u>(a)</u>



<u>(b)</u>

CONVERGENCE OF SERIAL SYSTEMS

FIGURE 5.3

as to the design of stage K is made on the basis of the information available at that stage, i.e. the input state Z, the design flow Q and the constraints. None of these constraints are affected in any way by the design or conditions outside stage K.

Next consider the introduction of constraint (f), i.e. pipe diameter must not be less than the upstream pipe diameter. Where previous authors have used this constraint, they have treated the system serially as described above and have used the constraint as just another condition on the selection of suitable levels and diameter for each stage as it is designed. This, however, destroys one of the essential features of a serial system, that there should be no loops. The decisions at a stage are no longer made purely on information available at that stage. They now use a constraint which is affected by the design of the previous stage. This is illustrated in Figure 5.4(b).

This problem can however be handled correctly with a certain amount of rearrangement. Instead of a single state variable Z, an additional variable, D, the upstream pipe diameter can be introduced. The state is now defined by the values of Z and D. Where several pipes enter a manhole, D is defined as the diameter of the largest of these pipes. A decision at stage K can now once again be made using only the information available at that stage, i.e. the input state (Z, D) the design flow Q and the full set of constraints. None of the constraints now refer to information not available either as input to that stage or as information within the stage.

The new serial system is illustrated in Figure 5.4(c).

5.5.5. Design flows that are not pre-determined

It is normal practice in the design of stormwater drainage networks for the flow at points in the network to be dependent on the design of the network upstream. This is true both for the simple Rational (Lloyd-Davies) method of design and for more sophisticated procedures using routing techniques, e.g. The Transport and Road Research Laboratory Hydrograph method, (refs. 5, 6).

- 47 -



FIGURE 5.4

- 48 -

2

As an illustration consider the Rational method. Essentially the flow at a point in a stormwater network is dependent on the total equivalent impermeable catchment area upstream of that point and on the maximum time taken by stormwater to reach that point from any point upstream. This time is known as the time to concentration. The greater the time to concentration, the less the design rainfall rate and hence the less the design flow. Whilst the impermeable area is fixed and may be determined before the start of any design, the time to concentration depends on the diameter, slope and roughness of all pipes upstream of the point considered. Hence the design flows cannot be predetermined.

To treat this situation as a true serial system, one has to use a third state variable, the time to concentration. The flow at stage K may then be determined from the stage input and decisions made strictly internally for that stage.

Such a serial system is shown in Figure 5.4(d).

As will be shown later, this concept is of limited usefulness (see section 5.11).

5.5.6. Summary

Design of drainage may be treated as a serial system with one, two, or three state variables depending on the nature of the constraints and the design flows.

Converging, tree-like networks present no problems to the concept of serial systems.

5.6 Optimising a Serial System by Dynamic Programming

5.6.1. Introduction

In 1957 Richard Bellman wrote a book entitled Dynamic Programming (ref. 15). This text introduced a novel mathematical approach to the problem of optimising multi-stage decision processes, and the name Dynamic Programming has been retained for the general approach he devised.

5.6.2. Principle of Optimality

Bellman stated in his principle of optimality that "An optimal policy has the property that whatever the initial state and initial decisions are, the remaining decisions must constitute an optimal policy with regard to the state resulting from the first decision". This is intuitively obvious and as Bellman argues, a proof by contradiction is immediate. Applying this principle recursively, an alternative and equivalent statement can be made, namely "any subarc of an optimal path is itself optimal".

It is this principle that allows the decomposition of a multivariable serial optimisation problem into the successive optimisation of problems with small numbers of variables.

5.6.3. State vector space

The input to a stage can be considered as a vector, with one dimension for each state variable. A state variable may be a continuous function, or may only exist at discrete values. Hence the state vector may be continuous, discontinuous or discrete in nature.

An optimisation problem concerns finding the set of values for the state vector at each stage such that the total return from the system is maximised or minimised. As minimisation is merely maximising a negative return, the argument may be restricted to maximisation.

It is in the nature of Dynamic Programming that the optimal values of the state vector are not known until all the stages have been considered. At each stage, however, a range of values of the state vector is considered. This range must be predefined, and must include the final optimal value of the vector. Thus an allowable state vector space is defined at each stage prior to the optimisation process.

5.6.4. Dynamic programming using discrete values

There are many different ways of optimising serial systems using Dynamic Programming. All use a similar approach, and the way described below is the most useful for the drainage network problem. The state vector is considered to be discrete valued, whether or not this corresponds to a physical reality.

The vector space is predetermined at each stage, and so are all the individual discrete values of the vector within that space. Hence for a particular stage K, there is a set of possible values for the input state vector, and a set of possible values for the output state vector.

Assume that for each discrete value of the input state vector at stage K, the total optimal return for stages 1 to (K-1) is known. Consider a particular discrete value of the output state vector. The optimal way of arriving at that output state must now be obtained. This is done as follows:-

- (1) consider a discrete input state.
- (2) optimise the design from the discrete input state to the discrete output state by making the decisions which maximise the individual stage return whilst conforming to any design constraints. This may in itself be a complicated optimisation problem or may be trivial as in the case of drainage networks. There may indeed be no feasible solution, in which case a very large negative return can be assigned to this combination of input and output states.
- (3) Add the stage return to the total optimal return for stages1 to (K-1) for the discrete input state considered.
- (4) If this total return is greater than for any previous way of arriving at the same output state, the value of the return is retained, as are references to the decisions that led to it, and any previously stored values for this output state are discarded. If the return is less, then the value of the return and the details of the design are ignored.
- (5) If there are any discrete input states that have not yet been considered, return to (1).

Hence the optimal return has been obtained for a discrete value of the output state vector, and the stage decisions which led to this optimal return are known.

- 51 -

This process must now be repeated for each discrete value of the output state vector. Hence we end up with a set of values for the optimal return for each discrete value of the output state vector at stage K. These can be used to form a set of optimal returns for each discrete value of the input state vector at stage K+1.

The process may now be repeated for stages (K+1) to N of the N stage system.

On completion of the Nth stage, the returns for each discrete output state can be examined and the maximum return selected. This is then the value of the optimum return from the system.

In itself this is of little use. What is required is the set of decisions at each stage that led to this optimum return. This can be established by tracing the optimal solution back through the system as follows:

- Identify the output state corresponding to the optimal return at output from stage N.
- (2) Identify the decision for stage N that led to this output state, together with the corresponding input state, (a set of such data having previously been stored for each output state).
- (3) For the particular input state identified, identify the corresponding output state for the previous stage.
- (4) Identify the decisions for this stage that led to this output state, together with the corresponding input state.
- (5) Repeat (3) and (4) until the first stage is reached, whereupon a complete set of optimal decisions for the system will have been identified.

5.7 Optimising a simple fixed plan drainage run by Dynamic Programming

The simplest drainage network is that of a single pipe run with manholes at fixed positions along it. The sizes and slopes of all the pipes for the optimum design may be obtained in the following manner. Consider first the case of a single state variable, the pipe soffit level at a manhole. Flows are assumed to be independent of the pipe network design, and pipe diameters are not constrained, thus being free to decrease in diameter downstream. Assume also for simplicity that drops in level across manholes are not permitted (see 5.8). A stage is as defined in 5.5.2. The input state is the pipe soffit level at the upstream end of the pipe. The output state is the pipe soffit level at the downstream end of the pipe. The output from state K equals the input for stage (K+1).

It is now necessary to consider the range of permissible states at each stage. For a typical drainage problem the range to consider is not at all obvious. On the one hand the range chosen must be sufficient to guarantee the inclusion of the global optimum, and yet not so large as to incur severe computational penalties. The problem may be circumvented by adopting a rather different D.P. approach called Discrete Differential Dynamic Programming or D.D.D.P. (ref. 32) as described in section 5.13. As D.D.D.P. is unsuitable for variable plan problems which lead on from the problem at present under consideration, it is necessary to find a rational basis for defining the range of states using the more conventional D.P. approach.

5.7.1. Upper bound on state variable

There will generally be a restriction on depth of cover, (constraint (a) of section 2.3), such that the pipe soffit level must be less than the ground level minus the depth of cover.

Hence there is some sort of upper limit on the range of levels that should be considered at each stage. Considering the pipe run shown in Figure 5.5 (a), one can see that although this would form a reasonable bound for pipe lengths (1)+(2) and (2)+(3), it would not be realistic for the rest of the run. Obviously the level downstream of A cannot exceed the level at A. If there is a minimum gradient specified (constraint (b) of section 2.3) this may be applied downstream of A to give a modified upper limit.

If there is no minimum gradient specified, the upper limit may still be restricted by a constraint on minimum velocity of flow in the pipe. However, by choosing a very large diameter pipe, the pipe slope to

- 53 -



(a) ESTABLISHING UPPER BOUND ON PIPE LEVEL



(b) THE DESIGN OF A STAGE

FIGURE 5.5

achieve a minimum velocity restriction will approach zero. So unless there is a restriction of maximum pipe size, the minimum velocity constraint does not in itself form a restriction on maximum pipe levels.

To summarise, the upper bound pipe level at a general point, P, is the lesserof:

- (Upper bound pipe level at any point upstream)
 m x (distance from that point), where m is maximum of (zero, specified minimum gradient, minimum gradient to provide specified minimum velocity with largest available pipe).

5.7.2. Lower bound on state variable

There may be a constraint on the maximum depth of cover (constraint (a) of section 2.3) which will give a lower bound on pipe level.

Whether or not this limit exists, experience of practical designs shows that it is reasonable to consider a lower bound at a fixed depth below the upper bound as determined in the previous section, giving a zone of fixed depth within which the optimal solution should lie. The selection of the correct depth to ensure optimality is a matter of judgment and experience and will be considered later (See 5.14.2).

The only other way in which the lower level could be limited is if a minimum outfall level is specified, but this would rarely form a practical limit for most of the network.

5.7.3. Establishing discrete values of level

Having specified the upper and lower limits on the state variable, pipe soffit level, at every manhole in the system, it is now necessary to define the discrete values of level that the variable may take.

To guarantee a true optimal solution, it is necessary to specify an infinite number of discrete values. However, in most cases a close approximation to the optimal solution may be obtained by adopting only a few discrete values. The choice of the number adopted is again one of judgment and experience (see 5.14.2), and is a balance between the marginal cost savings on the network designed and the extra running

- 55 -

costs of the computer program.

The arrangement of a typical stage in the network is shown in Figure 5.5(b).

5.7.4. Establishing a feasible design for an element

Given a discrete downstream level and a discrete upstream level, the design for an element then consists of selecting the pipe diameter that will give the least construction cost for the element whilst satisfying all the constraints listed in section 2.3. In practice it is assumed that element costs increase with increased pipe diameter hence the smallest feasible diameter is chosen.

For many combinations of upstream and downstream level, there may be no feasible pipe diameter.

5.7.5. Cost at each discrete input state

In dealing with a typical stage K, it is assumed that the minimum cost of arriving at each discrete input state is known, i.e. for each input state the optimal set of decisions for stages 1 to (K-1) and the returns (costs) resulting from them have already been determined.

5.7.6. Cost at each discrete output state

The problem now becomes that of determining the decisions in stage K that produce the minimum cost of arrival at each output state from stage K, where the minimum cost of arrival is the sum of the cost of arrival at the input to stage K plus the cost of the decisions taken in stage K to get from the input state to the particular output state.

For a particular output state, each input state is taken in turn. For that particular input state the smallest pipe is selected that will meet all the constraints listed in 2.3. The stage cost for this solution is added to the cost at the input state to give a cost at the output state.

When all input states have been examined, the overall cheapest cost of arrival at the output state is identified. This cost and the stage decisions that led to it are retained, all other costs and decisions relating to that output state being abandoned.

- 56 -



2

DESIGN OF A STAGE, FOR BASIC N-STAGE SERIAL SYSTEM



.

Hence the minimum cost of arrival at each output state is established. This is illustrated in Figure 5.6.

5.7.7. Overall Minimum Cost

It has now been shown that, given a set of minimum total costs for the input states to stage K, it is possible to obtain a set of minimum costs for the input states to stage (K+1).

As the costs for the input states to stage 1 are known, being generally zero, the process can be applied recursively along the serial system to obtain the set of minimum costs for the last (Nth) stage. This set of costs can then be examined and the cheapest will be the overall cheapest solution for the serial system.

5.7.8. Optimal Solution

In itself this is of little value. It is the decisions that led to the minimum cost solution that are important, hence a trace-back as described in section 5.6.4 is performed to establish the pipe levels and diameters used. This is illustrated in Figure 5.7.

5.8. Inclusion of Drop-Manholes

5.8.1. Introduction

A drop-manhole is one in which there is a change in level between the incoming and outgoing pipes. Such structures are required where ground levels change rapidly. Maximum slope or maximum velocity restrictions (see 2.3) may cause the outgoing pipe to be lower than the incoming for there to be any feasible solution (see Figure 5.8(a).) Alternatively if there is an obstruction it may be more economical to drop levels across a manhole (see Figure 5.8(b)). Small changes in level can normally be accommodated without incurring extra costs but large changes may well necessitate different and more expensive forms of manhole construction. Typical of these are the Back-Drop manholes described in the British Standard code of practice for Building Drainage (ref. 4).

- 58 -



TRACING THE SOLUTION BACK THROUGH BASIC N-STAGE SERIAL SYSTEM

FIGURE 5.7







- 60 -

5.8.2 Defining the Quasi-Input State

The main theoretical difficulty in dealing with drops across manholes is that the input state, defined as the lowest pipe level entering the upstream manhole, is no longer the level of the outgoing pipe.

This difficulty can be overcome by one of two methods

- (a) consider the manhole itself as a stage with the input state corresponding to the incoming pipe level and the output state corresponding to the outgoing pipe level, the decisions being the drop across the stage and the return being the cost of the manhole. This approach has been adopted by some previous authors (ref.23, 34) but further modifications are required when dealing with converging networks and on balance this approach was considered unnecessary.
- (b) Set up a "quasi input state", corresponding to the level of the outgoing pipe, this being the pipe level at the upstream end of the stage under consideration. This approach was developed for and used in the current research. Referring to Figure 5.8(a), the maximum level of the pipe leaving a manhole is determined by the level of the ground at the downstream end of the pipe and by the maximum permissible pipe slope. It is thus sensible to use this as an upper bound limit on the quasi input state. From this upper bound, a lower bound limit can be deduced with experience. Hence m discrete values of the quasi input-state may be determined. For each of these it is necessary to know the total optimal upstream cost.

Referring to Figure 5.8(c), consider a typical quasi-input state j, level ZB_j. Any output state, k, level ZA_k from the previous stage combined with a suitable value of drop, h, could give rise to this state, provided $ZA_k \ge ZB_j$. The optimal upstream cost associated with state j is then the least of (cost to output state k + cost of drop from ZA_k to ZB_j) for all k such that ZA_k $\ge ZB_j$. This procedure is shown in the flow chart of Figure 5.9.

This procedure can be incorporated into the D.P. method already detailed to achieve the overall optimum cost for the network.



ESTABLISHING QUASI-INPUT STATE COSTS, ALLOWING FOR DROP MANHOLES

FIGURE 5.9
5.8.3. Tracing Back

It now remains to ensure that one can perform a trace back up the system to obtain the set of optimal decisions that led to the overall optimum cost. This would conventionally require there to be m references stored one for each discrete input state, labelling the output state corresponding to the optimal upstream cost.

However, as a computationally easier alternative, one can omit the m references and re-establish the optimal upstream state for the single value of quasi-input state specified by the trace-back. This is a similar procedure to that described in the last part of section 5.8.2, and illustrated in Figure 5.9, except that it is only one quasi-input state that is considered.

As the trace-back is performed only once, the extra computation involved is negligible, and the savings made in data handling and storage can be significant.

5.9 Optimising a branched drainage network

5.9.1. Introduction

In 5.5.3. it was shown that a converging system such as the typical tree-like drainage network could be treated as several serial systems linked together. Hence there should be no difficulty implementing a D.P. approach to optimise such a system and this is indeed the case.

5.9.2 Procedure

For a tree-like network the order of design should be such that when a particular branch is being designed, all the branches upstream of it should already have been designed.

Take a typical branch of the network, consisting of several lengths of pipe between manholes, with several branches joining the most upstream manhole (e.g. branch AB of Figure 5.1).

Assume that for each upstream branch a set of optimal costs has been established for each discrete output state on the most downstream stage. In general the ranges of discrete output states and the range of permissible pipe levels at the upstream end of the typical branch under consideration will all be different.

Consider a set of quasi-input states for the most upstream stage of the branch. The optimal cost for each quasi-input state may now be obtained by combining the costs of the output states of all upstream branches in a suitable way, adding in the cost of a drop manhole if this is required.

A flow chart for the above process is shown in Figure 5.10.

Having established a set of costs corresponding to the quasi-input states, the design process may then proceed in the normal way, resulting in a set of costs for each output state of the most downstream stage in the branch.

5.9.3 Tracing back through a branch junction

There remains the problem of tracing back the overall optimum solution through the junction. The optimum quasi-input state will have been identified. Adopting a procedure similar to that described in 5.8.3., the optimum output state may be obtained for each branch in turn. This is illustrated in the flow chart of Figure 5.11.

5.10. Inclusion of the constraint on decreasing pipe diameters

5.10.1. Introduction

A common requirement in drainage network design is that pipe diameters should never decrease in a downstream direction, i.e. the pipe leaving a manhole must be at least as big as the largest pipe entering (constraint f, section 2.3).

It was shown in section 5.5.4. that if the network is to be represented as a serial system, it becomes necessary to introduce a second state variable D, the pipe diameter.

5.10.2. Procedure

Thus the output state from a stage is a two dimensional vector (Z, D), where Z is the pipe soffit level and D is the pipe diameter.

- 64 -



- Note j = quasi-input state number: 1,2,3,..., j max
 k = output state number : 1,2,3,..., k max
 - L = upstream branch reference:1,2,3,..., L max

ESTABLISHING QUASI-INPUT STATE COSTS FOR A BRANCHED SYSTEM

FIGURE 5.10

- 65 -



ι

L = upstream branch reference: 1,2,3,...,L max

TRACING THE OPTIMAL SOLUTION BACK THROUGH A JUNCTION

The input state for the next stage is then, strictly, the level and diameter (ZA, DA) of the pipe entering the upstream manhole. Where branches converge this becomes the minimum pipe level and maximum pipe diameter of all pipes entering the upstream manhole.

As in sections 5.8 and 5.9, it is convenient to define a quasi-input state, here consisting of pipe level and diameter (Z, D) of the pipe at the upstream end of the current stage, i.e. the level and diameter of the pipe on exit from the upstream manhole. It is then necessary to find for each state (Z, D) the least cost of arriving at (Z, D) from any input state (ZA, DA) such that $ZA \ge Z$ and $DA \le D$ allowing for the cost of any drop manhole feature associated with the value of (ZA - Z).

This procedure is similar to that described in 5.8.2. and is detailed in the flow chart of Figure 5.12.

5.10.3. Defining the range of diameters and their discrete values The D.P. method adopted requires that the range of values of diameter D should be defined at every stage in the system, and D should adopt

discrete values at these stages. The latter condition is automatically met by the fact that pipes are

only available in discrete increments of size. (see 2.3, constraint (g)).

The actual diameters available may depend on the pipe material and manufacturer. Hence it is necessary for the designer to identify the range of pipes that are available to him, and the pipes he wishes to use on each particular length of drain.

For example, assume a particular network consists of lengths of French drain and Carrier Drains (as defined in Section 1.3). The designer may choose to use perforated clay pipes of diameters 100 and 150 mm and porous concrete pipes of diameters 228, 309 and 380 mm for the French drains, with Carrier drains of Asbestos Cement selected from the range 300mm to 600mm in increments of 75mm.

One convenient way of dealing with these pipe variations is to specify a pipe class for every length of drain in the system and separately specify the pipes that are available in each class.

- 67 -



Note: j = quasi-input state number k = output state number

ESTABLISHING QUASI-INPUT STATE COSTS FOR A DIAMETER CONSTRAINED SYSTEM

An example of a typical pipe class is shown in Table 5.1. Note that other pipe properties can conveniently be attributed to each pipe in this way.

A TYPICAL PIPE CLASS

TABLE 5.1

No	Diameter (mm)	<u>Material</u>	Roughness (mm)	Min. Velocity (m/s)	Max. Velocity (m/s)
1	100	Perf. Clay	0.5	0.7	10.0
2	150	19	(1	97	64
3	228	Porous Conc.	1.0	0.7	6.0
4	309	"	**	11	11
5	380	11	88	18	18

PIPE CLASS A (for French Drains)

The overall available range of discrete diameters has now been established for a particular pipe length. This could therefore be used as the range of the state variable D. Such a procedure is, however, likely to be vary inefficient where the pipe class contains more than a few diameters.

In some way it is necessary to establish upper and lower bounds on the size of pipe to enable realistic ranges of diameter to be taken.

Consider a single length of pipe between manholes A and B (Figure 5.13 (a)). Design the pipe first to the minimum possible depth of cover. This will involve either minimum cover or minimum gradient or minimum velocity constraints (see 2.3). Then consider any other design. This will necessarily be at or below the level of the first solution. Use of a smaller pipe diameter at a steeper slope may give a cheaper solution. However, using a larger pipe must give a more expensive solution as the extra pipe cost cannot be compensated by reducing the trench excavation. Hence the optimal solution cannot involve a pipe diameter greater than that for a minimum cover solution.

- 69-





(a) MINIMUM COVER DESIGN

M/HA



(b) ALTERNATIVE DESIGNS FOR TWO PIPE SYSTEM

It would be very convenient if such an argument could be extended to cover a network of pipes rather than just one single pipe. Unfortunately this is not theoretically justified as can be seen from Figure 5.13(b). In this case the combined costs of the two pipes AB, BC, of diameters 150mm and 300mm could be less than for the two 225mm pipes at minimum cover. However results show (see 7.10.3) that in practice the optimal solution almost always has a diameter less than or equal to the 'minimum cover' solution. This condition is likely to apply to any network for designs involving sensible methods of costing the pipe elements and for reasonable ranges of pipe diameters and hence forms a realistic method of obtaining an upper bound on the diameter.

These same results show a second important feature. This is that the optimal solution tends to be confined within one or two increments of diameter of the minimum cover solution. This gives a method of establishing a lower bound on the pipe diameter.

It can now be seen that bounds on both level and diameter may be achieved by performing a minimum cover design and using the levels and diameters so produced to define the upper limits on the state variables for a Dynamic Programming process. The lower limits may be taken for level as a fixed distance below the upper limit and for diameter as a fixed number of increments below the upper limit.

A flow chart to illustrate this process is given in Figure 5.14.

5.10.4. Organising the computation

The method of computation for an individual stage is similar to that for just one state variable. Essentially every output state (defined by the vector (Z, D)) is considered in turn, with each quasi-input state taken as a possible source of the optimal solution. A series of checks are made on the feasibility of this design.

If there are m discrete levels and n discrete pipe diameters, there are m x n states and hence $m^2 \times n^2$ 'designs' to consider. This is a large increase over the single state variable case. There is, however, one great computational simplification. The pipe design

- 71 -



ESTABLISHING BOUNDS ON D.P. PROCEDURE WITH DIAMETER CONSTRAINT

is now completely defined as both pipe diameter and levels are specified. Previously (see 5.7.4.) for the single state variable case it was necessary to design the element by finding the smallest suitable diameter. Using the Colebrook-White equation the diameter cannot be obtained explicitly for a given slope and discharge, hence some procedure using enumeration or iteration was necessary. That included in the flow chart of Fig. 5.6 is one simple possibility. Hence, although two state variables are used, computational effort is not greatly increased.

For each output state feasible solutions are compared in cost with the cheapest being retained. The design of a stage is summarised in the flow chart of Fig 5.15.

Tracing back the final optimal solution presents no new difficulties and is organised similarly to those processes described in 5.6.4 and 5.8.3. A flow chart of the trace back procedure is shown in Fig. 5.16.

5.11 Dependence of flows on the network design

5.11.1 Introduction

In most methods of design for stormwater networks the design flow at a point in the system is dependent on the size, slope and roughness of some or all of the pipes upstream of that point.

As described in section 5.5.5 this leads to a three dimensional state vector for a true serial representation and hence a rigorous D.P. approach.

5.11.2 The Rational or Lloyd-Davies method

The most common method of calculating flows for small stormwater drainage networks is the Rational or Lloyd-Davies method (ref. 5). This will be used to demonstrate the application of both the rigorous and an approximate approach to the problem of network dependent design flows.

A flow chart showing the Rational method is shown in rig. 5.17. The essential feature is that the design flow in a pipe in the network depends on the time it takes for water to flow from the most remote point upstream of that pipe to the downstream end of the pipe. This time, (the time to concentration), consists of the time it takes the water to enter the pipe network (the time of entry) plus the time taken to flow down the pipes to the downstream end of the pipe under consideration, assuming that the pipes are flowing full (the time of flow).

The Rational method design philosophy assumes that the rainfall can be treated as having a constant intensity during a storm event. Generally the shorter the length of storm the higher is the

- 73 -



FIGURE 5.15



TRACING THE SOLUTION BACK THROUGH N-STAGE SERIAL SYSTEM WITH DIAMETER CONSTRAINT



rainfall intensity. It is assumed that the rainfall is evenly distributed over the catchment area.

Consider two subcatchments 1 and 2 draining into a common pipe AB (Fig. 5.18). Consider a storm of length t such that tc(2) < t < tc(1), where tc(1), tc(2) are the times to concentration at point B for flow from subcatchments 1 and 2.

As t > tc(2) all of subcatchment 2 contributes to the flow at B, but as t < tc(1) only part of subcatchment 1 contributes to the flow at B. It is assumed that the design flow at B increases with increasing values of t until a critical situation is met when t = tc(1). At this stage both of the subcatchment areas contribute in full to the flow at B. However if t is increased beyond this, the rainfall intensity is reduced, and hence the design flow decreases.

So, in general, for a particular pipe in the network a length of storm is selected equal to the time to concentration to the downstream end of the pipe.

Statistical rainfall data has been compiled which can either be used directly for a given location in Great Britain (Ref. 5) or formulae based on this data can be used to obtain an average rainfall intensity for a given return period and length of storm, the return period being the average period of time between events that exceed the chosen event.

One common formula used in Britain is the Bilham formula with the Holland modification (Ref.5). This is given below:

Bilham:I = $\frac{60}{t}$ (Nt x 202.26) $\frac{1/3.55}{2.54}$ -----(C)where I is rainfall intensity in mm/hrN is return period in yearst is length of storm in minutesHolland modification :for I > 33.0 mm/hrIn $\left[\frac{15240}{NIt} \left(\frac{It}{1524} + 0.1 \right)^{-3.55} \right] = 1 - 0.0314 I ---- (D)$

These formulae have been used throughout this research although there is no theoretical reason why tabular data should not be used.

Examination of equations C and D show that the rainfall intensity I is only given explicitly for values of $I \leq 33.0$ mm/hr. The Holland modification has to be solved iteratively for values of $I \geq 33.0$ mm/hr with the Bilham formula giving a reasonably close initial value for the iteration.

Having obtained the rainfall intensity the design flow at a point in the network is then (the rainfall intensity) x (the catchment area upstream of that point).

- 77 -



SUBCATCHMENT AREAS

5.11.3 Other methods of calculating stormwater flows

Various other methods of calculating stormwater design flows have been proposed and used. These are conveniently summarised in Ref. 1. In all these methods the design flow at a point in the network is in some way dependent on the design of that part of the network upstream of the point.

The most prevelant of these alternative methods is the Transport and Road Research Laboratory Hydrograph method, this being widely used in the U.K. for the design of large drainage networks. A full description of the method is given elsewhere (Mefs. 5, 6) but briefly it involves the use of a time varying rainfall intensity and takes into account the storage or routing effects of the pipes through which the water flows. The main effect is that an increase in pipe djameter creates greater storage within the pipe, which in turn diminishes the peak of the time varying flow out of the pipe. Hence the design flow at a point in the system is dependent on the upstream pipes, although the precise nature of the dependence is much more difficult to establish than with the Rational method. To devise a rigorous Dynamic Programming approach for such a design system would be very difficult and totally impracticable.

5.11.4 The three-dimensional state vector approach

As described in section 5.5.5, using the Rational method of design, drainage may be considered as a true serial system by using three state variables. So, in theory, a rigorous DP approach could be devised. It is, however, of interest to consider the computational effort involved in such a strategy, bearing in mind also that such a method would only be relevant to the Rational design philosophy which is likely to be superseded.

DP is generally considered to be efficient when there are one or two state variables. More state variables incur severe computational penalties as can be seen from the general approach used for all detailed computations. For a stage this consists of selecting each discrete output state and considering all possible ways of arriving at that state from each input state.

If there are ,say, i(n) discrete values of the n dimensions defining a state then there are $(i(1) \times i(2) \times i(3) \times \dots i(n))^2$ designs to consider for each stage with $(i(1) \times i(2) \times \dots i(n))$ values of cost and an equal number of trace-back references to store.

- 79 -

For example, if $i(1) = i(2) = \dots i(n) = 10$ for n = 1 number of designs = 100 for n = 2 number of designs = 10 000 for n = 3 number of designs = 1 000 000 for n = 4 number of designs = 100 000 000

Experience has shown that values of i(1) and i(2) could be reduced to 7 and 3 respectively for level and diameter state variables. Using these values the number of designs for incorporation of a third state variable becomes $7^2 \times 3^2 \times (i(3))^2 = 441(i(3))^2$. A suitable value of i(3) is unlikely to be less than 7. This would give at least $441 \times 7^2 = 21\ 609\ designs\ per\ stage$.

Although a method using 20 000 designs per stage may be possible for the fixed plan problem, it is certainly not desirable, and as a basis for a variable plan model it can quickly be set aside as impracticable.

5.11.5 An approximate approach

Having shown that a completely rigorous approach is impracticable, it is now necessary to examine the practicability of a reasonable but non-rigorous method.

The development of such a method was actually performed for a variable plan model of which the fixed plan model under discussion is a special case. The details of the development will thus be presented in Chapter 7, "The variable manhole position model". For completeness, however, a summary of the development is given here, as far as it is applicable to the simpler fixed plan network.

The first step taken was to assume that all flows were fixed, ie. did not depend on the pipe network upstream. The method of fixing the flows was less obvious.

The initial approach was to calculate a time of flow for all pipes using a uniform flow velocity (eg. 1.5 m/s). Hence a rainfall and design flow could be calculated for each pipe. When the DP design was complete a comparison of the actual flow velocities and hence actual flows could be made with the assumed values.

It was soon seen that this led to unacceptably large discrepancies. However an iterative approach based on this was a logical development. The flow velocities resulting from the new design were thus used to calculate new times to concentration and design flows for the next DP design. The iterations continued until the

- 00 -

variation in flow from one iteration to the next was within acceptable limits. This usually occured within four or five iterations.

The method was seen to be rather clumsy and somewhat prone to problems of convergence (see 7.9.2).

A more satisfactory approach was to define the initial design flows as being equal to the design flows for a minimum cover design. Very rapid convergence then ensued. As a minimum cover design was already used to establish the limits on pipe level and diameter (see 5.10.3), the minimum cover design flows were readily available.

It was additionally found that in general the diameters of pipes designed by the first DP design did not subsequently change in further iterations. The pipe slopes merely altered to accomodate changes in design flows.

5.12 The final fixed plan model - MANFIX

The observation that the designed pipe diameters did not change after the first iteration provided a very useful method of truncating the iterative procedure.

Instead of allowing the iteration to proceed until flows were acceptably stable, an exact explicit solution could be obtained by taking the diameters produced by the first iteration as fixed and then performing a normal Rational method design to obtain pipe slopes, and incidentally design flows.

Although a separate computer program was never written for the fixed plan model, results using the variable plan program on fixed plan examples showed the method to be sound.

For convenience the fixed plan model will be referred to as the MANFIX (manholes fixed) model. For completeness a flow chart for a proposed MANFIX computer program is shown in Fig. 5.19.

One very important feature of MANFIX is that it is not necessarily restricted to the use of the Rational method of design. In principle any design method could be used to establish design flows and bounds on the state variables based on a minimum cover condition. These flows and bounds can then be used in the core of the program the DP process - to determine pipe diameters only. These diameters canthen be used in the selected design method to determine pipe slopes.



PROPOSED MANFIX COMPUTER PROGRAM

5.13 The use of Discrete Differential Dynamic Programming

DDDP (ref. 32) was referred to in Section 5.1 as being the most economic existing approach for the fixed plan optimisation problem. As such it is worthwhile discussing the method and comparing it with the MANFIX model.

Stormwater drainage using DDDP has been described in some detail elsewhere (refs. 23,34). Hence only the principles will be presented here.

A simple drainage run between three manholes is illustrated in Fig. 5.20. The DDDP approach to optimising the design of such a run is as follows :

(a) Specify a "trial trajectory", this being an initial guess at the longitudinal profile of the pipeline between the manholes.

(b) Specify an initial "band width", this being the width of a "corridor" centred on the trial trajectory, giving the limits within which the pipe profile may lie.

(c) Select a small number (3 or 5) of discrete depths at each manhole equally spaced across the corridor.

(d) Use conventional DP to select the optimum profile using the discrete depths.

(e) Use the optimum profile as the trial trajectory for another iteration using the same band width. This forms the primary iteration.

(f) When the optimum profile coincides with the trial trajectory, decrease the band width and repeat the process. This is the secondary iteration.

(g) Continue decreasing the band width until the required accuracy is obtained.

DDDP is claimed to be much more efficient computationally than DP (ref 40), due to the small number of discrete levels considered in the corridor for any one iteration. Hence if the possible range of levels at the manholes is large the potential saving over DP could be remarkable. For example, if the pipe levels were required to an accuracy of 0.01m and there was a possible range of levels of , say, 3m at each manhole, a conventional DP approach would require 300 discrete levels at a manhole, with $300^2 = 90\ 000$ possible designs to consider at each stage. Using 3 discrete levels in DDDP there are $3^2 = 9$ designs per stage per iteration. If it takes 3 primary iterations to achieve a stable trajectory, there are then 3 x 9 = 27 designs per secondary iteration. To reduce a 3m band width to 0.01m with a reduction by, say, a factor of 0.7 at each secondary iteration,

- 83 -





OPTIMAL DESIGN BY D.D.D.P.

requires n secondary iterations, where $3 \ge 0.7^n = 0.01$. Hence n = 16, and the total number of designs per stage = $16 \ge 27 = 432$, compared to 90 000 for a conventional DP design.

However, it has been shown that by using MANFIX the possible range of levels can be reduced substantially by first performing a minimum cover design. Further, the required accuracy for pipe levels can be very coarse, with only the optimal pipe diameters being required, the pipe gradients being obtained from a conventional design procedure using the optimal diameters.

Hence MANFIX will be more comparable to the DDDP approach than will conventional DP methods.

For example with MANFIX, if the range of levels is restricted to 0.9m and the levels are required to 0.1m, there are 10 discrete levels and $10^2 = 100$ designs per stage.

Similar values for a DDDP approach give $3^2 \times 3 \times n$ designs per stage where $1 \times 0.7^n = 0.1$, giving n = 6 and number of designs = 162.

It would appear that a DDDP approach is possibly less efficient than a carefully prepared DP approach such as MANFIX, though it may be more likely to find a true optimal solution in unusual circumstances.

One disadvantage with DDDP is that a computer program is necessarily more complex than for DP. For the purpose of the present research the main disadvantage is that the concepts of a trial trajectory and a decreasing band width are incompatible with the variable plan problem.

An additional consideration is that storm drainage design by DDDP has only been presented using a one dimensional state vector. Hence the constraint on non-decreasing pipe diameters (see 2.3), if required, is handled incorrectly in published material (ref. 31). It would be possible to have a DDDP approach to storm drainage using a two dimensional state vector, but this has not yet been tried. Also, for a complete approach, some method of approximating stormwater design flows is needed, perhaps similar to that in MANFIX.

5.14 Experience and results

5.14.1 Introduction

As the main research was concerned with variable plan networks a computer program for MANFIX was not written. Hence experience relates to the development and use of the variable plan programs and will be presented fully in Chapter 7. Hesults are those using variable plan programs for fixed plan examples.

- 85 -

5.14.2 Experience of the model

The following general comments can be made :

(a) DP optimised solutions generally followed closely to minimum cover solutions and were often identical for many of the upper branches in a network. Where they differed it was rarely by more than 0.5m in depth or by more than two increments in pipe diameter.

(b) a solution with a cost close to the true minimum could be obtained by using a coarse DP grid with discrete levels at spacings of between 0.1 and 0.15m.

(c) for standard pipe diameter increments (75mm) and a sensible minimum gradient (1 in 250), near optimal solutions could be expected with confidence using six discrete levels over a range of 0.7m and considering just three diameters for each pipe.

5.14.3 Results

All the optimally designed networks showed cost savings of between 5% and 15% over networks designed to a minimum cover solution. This is consistent with the findings of other authors. (refs. 27, 28, 29)

A set of results for the stormwater drainage of a small housing estate is given in Figs. 5.21, 5.22 and 5.23. The original design against which the optimal design is compared does contain some inconsistencies and cannot be regarded as a perfect minimum cover design. It is however considered typical of present practice. Qualitatively the results show a feature typical of optimised designs compared to traditional designs. The pipe diameters are unaltered at the upstream ends, the main savings being on reduced diameters towards the outfalls. Where depths of cover are increased it is only necessary to do so very slightly to accommodate the increase in gradients.

In fact at both outfalls into existing sewers the optimised networks give invert levels slightly higher than the original scheme, due to smaller pipes being required at the same or slightly increased depths of cover.

From a practical viewpoint the optimal scheme is preferable in that less of the pipe network is at minimum gradient. Hence there would be less trouble from siltation and blockage. Increases in depth of cover tend to be minimal and hence would not add significantly to access problems in cases of failure or new connections.

- 86 -



FIGURE 5.21

- 87 -



;



quantitatively the DP optimised scheme represents a saving in construction cost of about 12% over the original scheme.

5.15 Cost of using MANFIX

Clearly the extra computing costs involved in using an optimisation program should not exceed the likely savings on construction costs. If all designs led to actual constructed schemes it would be reasonable to allow the cost of optimising to approach the likely savings over a non-optimised scheme. However, for a number of reasons this may not be acceptable. These include :

(a) Allowance for non-productive and superseded design runs: the cost of these runs must be paid for from the savings on productive runs.

(b) High computer costs imply a requirement for large computers or long run times. If these facilities are not immediately available designs may be delayed for up to several days by awaiting turnround on multi-user machines.

It is difficult to estimate the actual cost of using MANFIX as present results were obtained from a variable plan program. However, as an example, the resources used in the design of the housing estate networks were 22 secs. execution time on an ICL 1906S computer with a core store requirement of 54K. In 1979 this cost about £3. This is insignificant compared to the saving on construction cost of the scheme, which is calculated as £2300 at March 1977 prices.

It is clear from this that the computer costs can be a small fraction of the likely savings. Indeed performing a design using an optimising computer program may well be cheaper than manually designing the scheme, which is the usual design office practice.

5.16 <u>Conclusions on the fixed layout model - MANFIX</u>

It has been shown that an effective and simple Dynamic Programming model can be used for the optimised design of stormwater drainage networks. The efficiency of such a model is likely to be greater than that of a DDDP model or any other existing fixed plan optimising model.

The novel features of the model include the following: (a) Establishing upper bounds on levels and diameters, and an estimate of design flows, by performing a conventional minimum cover design before the optimising process.

(b) Limiting the DP to a coarse grid over a narrow band, and using only the pipe diameters thus obtained.

- 90 -

(c) Using a second state variable, pipe diameter, to handle the constraint on non-decreasing pipe diameters rigorously.

(d) Performing a final conventional design using the pipe diameters obtained from the DP optimisation, thus producing the correct design flows and pipe gradients.

(e) The ability to use, in theory, any design method (Rational, TRRL, etc), since the core of the model, the DP program, is unaffected by the design method.

The cost of design using such a model need be little more and may in fact be less than current design costs.

MANFIX requires only limited computer resources and could be tailored for use on a mini-computer within a design office as well as being a fully supported design program on a main-frame computer.

The type of design produced by MANFIX is in general sensible and preferable in at least one respect other than cost to the minimum cover design often produced manually. Minimum cover designs very often have most of the network at minimum gradients or flow velocities. MANFIX produces designs with more of the network at steeper slopes, thus reducing possible trouble with siltation and blockages. Some trench depths are increased but the increases are often minimal and very rarely exceed 0.5m. Indeed due to reduced pipe diameters at the downstream end of the network, trench depths can often be decreased.

CHAPTER 6

VARIABLE PLAN OPTIMISATION

- 6.1 Introduction
- 6.2 Highway Storm Drainage Networks
- 6.3 Potential for Optimisation

CHAPTER 6 Variable Plan Optimisation

6.1 Introduction

The main object of this research was to investigate the possibility of optimising the plan layout of storm water drainage networks for roads and to produce a practical working computer program for optimal drainage design if this was indeed feasible.

In a general storm drainage network, plan optimisation could be performed in many different modes. The simplest mode can be defined as follows: given a network of pipes and manholes some of which are fixed in position, find the optimum position of all other manholes together with the optimum gradients and diameters for all pipes. Such a network is shown in figure 6.1a. Some of the manholes are fixed in position (i.e. A, B, C, D), others are variable (i.e. E, F). For this example the problem is then to find the plan coordinates of manholes E and F together with the slopes and diameters of all pipes.

A second mode of optimisation is where the connectivity, or in other words the basic network layout, is unspecified. In figure 6.1b, for example, the flow from A may go either to E or to C, the choice being part of the optimisation.

A third mode of optimisation is where the number of manholes is unspecified. For example in figure 6.1c variable position manhole G may or may not exist in the optimal solution.

As a general case could combine all three such modes of optimisation it is clear that for anything but very small networks the complexities of general variable plan optimisation become formidable. Any attempt to create a general variable plan optimisation model would at present be completely futile.

- 93 -





The procedure adopted was therefore to examine the characteristics of several particular types of drainage network and to examine ways in which variable plan optimisation might be achieved.

As the research was primarily concerned with road drainage, the type of network normally designed for new major roads was initially considered, and is dealt with in the following two chapters. In addition one further type of network is specified and examined in chapter 9. This is the case of joining several sources of flow to a single main drainage run.

For the road drainage type of network two variable plan models were developed, MANVAR (variable manholes) and CROSSVAR (variable cross-drains), computer programs for these being written and tested. A model for the final type of network considered is proposed but a program has not been written or tested.

A fully documented and tested commercial version of MANVAR has been written for the Highway Engineering Computer Branch of the Department of the Environment, and will be released soon as an optional mode of operation for their current Drainage Design Program DAPHNE (ref.**56**).

6.2 Highway Storm Drainage Networks

An essential element in most modern highway design is the provision of a drainage system to remove incident rainfall. The road profile is used to direct run-off to the road edges or possibly to the central reservation in the case of a dual carriageway. Occasionally, especially on minor roads in rural areas, the designer will allow run-off to pass over a grass verge and into an open ditch. In general, however, piped drains are provided running roughly parallel to the road edges. The run-off may flow straight into these as in the case of a "French drain", this being a gravel filled trench with a perforated or porous pipe to

- 95 -

collect the water at the bottom of the drain. Alternatively water may first be collected by open channels formed by a kerb and the crossfall on the road, thence passing through gullies sited in the channel into conventional closed pipes.

Generally kerbs and gullies are used throughout urban areas, and in rural areas where the road is on an embankment.

French drains are used in cuttings in rural areas and also along the central reservation of motorways and dual carriageway roads. They often have the additional duty of keeping the road foundation drained. This purpose is however ignored in this research as the flows involved are minimal compared with stormwater flows.

For convenience French drains and "gully -fed" drains will be referred to as "carriageway drains", as their primary duty is to collect the run-off from carriageways. Carriageway drains are generally either laid at a constant offset from the road centreline, thus being curved in plan where the road is curved, or laid in straight lines between manholes which are at a uniform offset from the road centreline.

At intervals it is generally necessary to convey the water from carriageway drains across the road. This is done by the provision of drains consisting of conventional closed pipes in trenches that are very carefully backfilled and compacted and almost always run directly across the carriageways. These will be referred to as "cross-drains".

Cross-drains, as well as being constructed to a higher specification, are often designed for more severe storm events than the rest of the drainage network. This is a sensible precaution as access in the event of failure is very expensive and overloading in a severe storm could lead to dangerous flooding of the road carriageways.

- 96 -

A third type of drain exists which will be referred to as a "carrier drain". Carrier drains convey water from carriageway or cross-drains to the outfalls of the network. Water can only enter carrier drains at manholes. Carrier drains and carriageway drains may sometimes share the same trench.

It is general practice to provide manholes for maintenance purposes at all drain junctions, changes in pipe size or changes in pipe gradient and at intervals along all drains subject to maximum spacing restrictions. Cross-drains will not have any intermediate manholes except one in the central reservation where such a reservation exists.

A manhole will usually be placed at the head of a drainage run, but sometimes a "rodding eye" will be provided instead, thus giving a cheaper form of access. "Rodding eyes" can be considered as cheap manholes for the purpose of the present research.

Figure 6.2 shows a typical dual-carriageway storm-water drainage system, consisting of carriageway drains, cross-drains, carrier drains and outfalls. The drains form two tree-like networks.

Design of the networks conventionally start with the designer drawing a plan of the pipework layout, specifying the position of all manholes and calculating the catchment areas for all pipes. Values of runoff coefficient (runoff/rainfall) are specified for the various parts of the catchment (e.g. carriageway, verge, hard shoulder). The storm severity is selected by the choice of a return period. As highway pipes usually have diameters of less than 600 mm, Road Note 35 (ref. 5) allows design flows to be calculated by the Rational (Lloyd-Davies) method. The designer proceeds with the sequential design of all pipes in the network, starting with those at the upstream ends and then working downstream. In general the designer will place all pipes at minimum possible cover and select the pipe diameters that will convey the required flow at the resultant gradients.

- 97 -



TYPICAL HIGHWAY STORM DRAINAGE NETWORK

FIGURE 6.2
6.3 Potential for Optimisation

It has been shown in chapter 5 that, given the position of all manholes, the diameters and slopes of all pipes may be optimally designed using MANFIX.

What scope is there, however, for altering the position of manholes to improve the design still further? To answer this question, it is desirable to examine the procedure adopted by the designer in positioning the manholes. He must first decide what type of drain is necessary and over what length it is required. In addition he must choose the offset of carriageway drains from the road centreline. These decisions can be regarded as fixed and invariable in any optimisation. For example in Fig.6.3(a), there must be drains between A and D, B and E, C and F, and also, therefore, manholes at A, B, C, D, E and F.

There remains a certain flexibility about how these drains are connected to an outfall.

There can for example, be one or more crossdrains with one outfall as in Fig. 6.3(b) and (c).

Other schemes could involve several outfalls.

In practice the number and position of the outfalls will be largely governed by factors other than minimum cost design. Water authority requirements and availability of land being two important constraints in the U.K.

Again there will almost always be a cross-drain at the lowest point along the length of carriageway under consideration. Hence it is reasonable to assume that in figure 6.3 (d) manholes A, B, C, D, E, F, G, H, I, J, K are all effectively fixed in plan.

- 99 -



The designer then has

to decide first of all if any additional cross-drains are required and if so, where. Secondly he must decide on the number and position of all intermediate manholes, remembering that manholes must be provided at changes in pipe diameter and slope and at intervals not greater than a certain maximum spacing for maintenance requirements.

These decisions are essentially of an economic nature as the functional efficiency of the network is not likely to be altered by such decisions and yet the network cost may well be. Yet these decisions are usually made on engineering judgement and experience. Unfortunately they may well be taken by engineers with limited experience and consequently little foundation for engineering judgement. Even experienced engineers would have little, if any, quantitative evidence of costs on which to base their judgement. Cost comparisons of alternative layouts are rarely, if ever, performed.

These decisions, on the number and position of cross-drains and intermediate manholes, therefore appeared to be prime candidates for computer based optimisation methods.

A model for optimising the number and position of intermediate manholes is presented in chapter 7, whilst chapter 8 presents a model for additionally optimising the number and position of cross-drains.

CHAPTER 7

THE VARIABLE MANHOLE POSITION MODEL - MANVAR

- 7.1. Introduction.
- 7.2. Defining the Problem.
- 7.3. The Design Flow.
- 7.4. Method of Approach to the Optimisation Problem.
- 7.5. A Dynamic Programming Approach.
 - 7.5.1. Introduction.
 - 7.5.2. The Basic Skeleton Serial System.
 - 7.5.3. The Design of a Run by Normal D.P.
- 7.6. Indeterminate Stage Dynamic Programming.
 - 7.6.1. Introduction.
 - 7.6.2. A Modified Serial System.
 - 7.6.3. Intermediate Manholes and the Modified Serial System.
 - 7.6.4. Applying I.S.D.P. to the Variable Manhole Problem.
 - 7.6.5. Efficiency of I.S.D.P. for the Variable Manhole Problem.
- 7.7. The Set of Discrete Possible Manhole Positions.
- 7.8. Establishing the Ranges of Value for the State Variables.
- 7.9. Dependence of Flows on Network Design.
 - 7.9.1. Introduction.
 - 7.9.2. An Approximate Approach.
- 7.10. Experience and Results of Using Preliminary Program DPO.
 - 7.10.1. Introduction.
 - 7.10.2. The Test Networks.
 - 7.10.3. General Results.
 - 7.10.4. Selection of Optimising Parameters.
- 7.11. The Variable Manhole Position Model MANVAR.
 - 7.11.1. Introduction.
 - 7.11.2. Structure of MANVAR.
 - 7.11.3. Implementation of MANVAR.

- 7.12. Program ASSEMB.
 - 7.12.1. Introduction.
 - 7.12.2. The Program.
 - 7.12.3. Input.
 - 7.12.4. Output.
 - 7.12.5. Use of ASSEMB.
- 7.13. Program MOD.
 - 7.13.1. Introduction.
 - 7.13.2. The Program.
 - 7.13.3. Input.
 - 7.13.4. Output.

7.14. Results from Using ASSEMB and MOD.

- 7.14.1. Introduction.
- 7.14.2. Checks on the consistency of MOD and DPO.
- 7.14.3. The Effect of Varying the Optimising Parameters.
- 7.14.4. Varying the Design Parameters.
- 7.14.5. Tests on Other Networks.
- 7.15. Conclusions from Using MOD.
- 7.16. A Commercial Program.
 - 7.16.1. Introduction.
 - 7.16.2. The Existing Program DAPHNE.
 - 7.16.3. The Optimising Version DAPHOP.
- 7.17. Conclusions on the MANVAR Model.

7.1 Introduction

An extensive literature search in the fields of drainage and optimisation failed to unearth any published material of relevance to the problem of positioning an unknown number of manholes along a drainage run.

Previous research in the optimisation of variable plan drainage networks has generally concentrated on problems of connectivity (refs. 33,38,44) rather than the problem of a variable plan position for a manhole. A review of multivariable optimisation techniques showed that they generally deal with problems in which the number of variables is known. Here the number of manholes, and hence the number of variables, is initially unknown.

7.2 Defining the Problem

For a typical tree-like network the problem is to find the number of intermediate manholes along each non-branching run, together with their positions, together with the diameters and levels of all pipes, such that the total construction cost of the network is as small as possible whilst all the technological and physical constraints imposed on the system are met.

One of the constraints given in section 2.3 is a condition that manholes should not be spaced at more than a given distance apart, Lmax, along each run.

As an example consider a network of m pipe runs between (m + 1) fixed manholes.

Consider a typical pipe run I, (I = 1 to m). Pipe run I consists of an unknown number of manholes, N(I). Define an element as a pipe with its upstream manhole (see Fig. 4.1).

The design of an element (I,J) can then be defined in terms of the pipe diameter D(I,J), upstream and downstream pipe levels Zu(I,J), Zd(I,J), and upstream and downstream distances Xu(I,J), Xd(I,J) from a fixed manhole.

In general given Zu and Zd, Xu and Xd, the smallest and hence cheapest pipe size that will carry the required flow and satisfy the flow constraints will be chosen. Hence the pipe diameter D(I,J) is dependent on Zu(I,J), Zd(I,J), Xu, Xd and need not be considered as an independent variable.

There are then $4 \times N(I)$ variables for a typical branch where N(I) is itself an additional variable, and with the constraints that Xd(I,J) = Xu(I,J+1) and Xu(I,1) = 0, Xd(I,N(I)) = length of branch, Lb(I).

The cost of constructing a pipe element is a function of element length, pipe diameter, average depth and upstream depth. In terms of the independent variables the cost of an element Ce(I,J) = $f\{Zu(I,J), Zd(I,J), Xu(I,J), Xd(I,J)\}$ Let the cost of constructing a pipe run be Cb(I). Hence the problem becomes one of minimising C where

 $C = \sum_{i=1}^{m} Cb(i)$ I=1 m N(i) $= \sum_{i=1}^{m} \sum_{j=1}^{n} Ce(i,j)$ I=1 J=1

where

$$Ce(I,J) = f(Zu(I,J), Zd(I,J), Xu(I,J), Xd(I,J))$$

and N(I) are variable parameters in the minimisation, subject to a set of constraints of the form:

Xd(I,J)	= Xu(I,J+1)	JI = 1,m		
		J = 1, N(I) - 1		
Xu(I,1)	= 0	۱ ۱		
Xd(1,N(1)) = L(I)			
and constraints	0 ≤ Xd(I,J) - Xu(I,J)≤	Lmax		
and constraints	Ymin < Y < Ymax			
	smin < s < smax			
	Vmin < V < Vmax			
	Q ≤ Qf			
	Zu s Zus			
	D ≥ Dus			
	D a discrete, availab	le, diameter.		

7.3 The Design Flow

The design flow Q for an element will in general depend on the total catchment area, A, for the element and, if the Rational method of design is used, on the time to concentration, tc, (i.e. time taken for runoff to reach the downstream end of the element from the furthest upstream point).

Hence Q = f(A,tc)

The catchment area in general increases with distance along a run.

Hence A = f(Xd)

The time to concentration, tc, depends on the diameters, slopes and lengths of all pipes upstream of the element and on the diameter, slope and length of the pipe in the element.

If an approximation can be made for the time to concentration such that it is independent of the pipe diameters and gradients upstream and merely dependent on position, then tc = f(Xd).

Hence as Q = f(A,tc)

Q = (f(Xd))

7.4 Method of Approach to the optimisation problem

The main difficulties involved in forming an optimisation model for this application are listed below.

- a) Unknown number of variables.
- b) Non-linear, non-differentiable objective function.
- c) Discrete values of pipe diameter.

Difficulty (a) could be partially overcome by assuming that a sufficiently large number of intermediate manholes exist thus giving a definite number of variables and allowing solutions in which many of these manholes are coincident. This is very inefficient and the presence of singularities in the solution would present great problems to any known multi-variable optimisation algorithm.

Even if this first difficulty could be overcome, for the reasons discussed in section 5.3 there would still be formidable problems in producing a robust, economic and effective model based on conventional multi-variable optimisation algorithms. As Dynamic Programming had been shown to be effective in fixedplan drainage optimisation it was decided to investigate whether it was also suitable for the problem of variable manhole positions.

7.5 A Dynamic Programming approach.

7.5.1 Introduction

As has been shown in Chapter 5 Dynamic Programming is very efficient when dealing with serial systems, and can cope well with discrete valued variables and discontinuous objective functions. Fixed plan drainage networks were shown in section 5.5. to be serial systems suitable for D.P. The questions in dealing with the variable manhole position problem are whether some or all of the network can be considered a serial system, and whether such a serial system is amenable to D.P.

Throughout this chapter it is assumed that diameters may not decrease in a downstream direction (See 2.3.(g)). Hence pipe diameter is a necessary state variable (See 5.5.4).

7.5.2 The basic skelton serial system

If the fixed manhole positions are assumed to define a basic skeleton of drainage runs, (e.g. Fig. 6.3(d)), the design of each run could then be considered as a stage in a serial system. This is evident by direct comparison with the fixed plan serial system described in section 5.5.

Such a serial system is illustrated in Figure 7.1(a).

An individual stage involves decisions on the number and position of intermediate manholes along a run together with decisions on the diameter and slopes of all pipes.

If a method of producing an optimal design for an individual run can be found, it follows from the nature of serial systems that the optimal solution for the whole network can be established by D.P.

7.5.3 The design of a run by normal D.P.

The same principal difficulties exist in trying to optimise the design of an individual run as do for the problem of optimising the complete network (see 7.4). Admittedly the scale of the problem is



(c) RANGES OF MANHOLE POSITIONS

FIGURE 7.1

much reduced. Even so conventional multi-variable optimisation is unlikely to be a fruitful line of approach.

Hence D.P. was again investigated carefully as the most likely way of achieving satisfactory results.

The first approach was to assume that there were a fixed number of stages in the design of a run. Each stage consisted of the design of a pipe with its upstream manhole, the design consisting of the length, diameter and levels of the pipe. To allow the number of manholes to be variable, a stage could consist of a pipe of zero length, in which case the stage return (cost) would be zero. Such a system would need to have pipe level, pipe diameter and position of manhole as state variables, and is shown diagrammatically in Fig. 7.1(b).

Assume that manholes have a maximum spacing, and also a minimum spacing. The possible positions of successive intermediate manholes is then as shown in Fig. 7.1(c).

If the minimum spacing = Lmin , the maximum spacing = Lmax and the length of run = Lb , then the possible range of position for the Nth intermediate manhole is the lesser of

 $N \times (Lmax - Lmin)$ and $(Lb - N \times Lmin)$

and the total number of possible stages is the integer value of (Lb/Lmin).

It can be seen that, if the total length of run is large compared to the minimum spacing, the possible range of position for a manhole could itself be large.

For highway drainage a typical run length could exceed 1 km, with a minimum spacing of, say, 30 m and a maximum of, say, 150 m. This would give 33 stages and a range of 790 m for the position of the seventh manhole.

The number of stages could probably be substantially reduced without affecting the solution in almost all cases. Indeed, with experience, the range of position for the manholes could probably be reduced somewhat.

It seems therefore that a practicable solution may be possible using the above approach. The main disadvantages are

- 109 -

- (a) three state variables
- (b) large range of values for the manhole position state variable
- (c) number of stages has to be pre-determined. Hence some redundant stages are inevitable.

To illustrate these problems consider briefly the typical highway case outlined above.

Assuming that discrete values are adopted for the state variables in the D.P. process (see Section 5.6.4), the state variables level, diameter and position may have *l*, m and n discrete values respectively.

There are thus $(l \times m \times n)^2$ designs to consider at each stage. If there are N stages and l, m and n are constant for each stage, there are N × $(l \times m \times n)^2$ elemental designs for a complete run.

Assume values of l and m are the same as the typical values adopted in the MANFIX model, i.e. l = 7, m = 3. For a run length of 1 km assume that the number of stages can in practice be reduced to 10 and the maximum range of positions to, say, 450 m. Then if the discrete values of the position state vector are taken at 30 m intervals, n = 16 and the total number of designs = $10 \times (7 \times 3 \times 16)^2$ = 1,128,960.

Hence, although the method could well be successful in achieving near optimal results in most cases, it seems likely that the computer time required may be unrealistically large.

7.6 Indeterminate Stage Dynamic Programming

7.6.1 Introduction

In an attempt to improve on the DP approach of section 7.5.3, a new concept in DP was developed. This will be called, for convenience, Indeterminate Stage Dynamic Programming (ISDP).

As the name suggests the stages are not predetermined either in number or position but result from the DP optimisation.

When applied to the intermediate manhole problem an elegent and effective method results.

7.6.2 A modified serial system

A modified serial system was adopted in which the input state to a stage results from the output from one of a range of possible previous stages. This is best explained by a simple example.

Consider a set of stages a, b, c, d etc. as shown in Figure 7.2(a). Consider stage d. Allow the input to stage d to be the output from any of stages a, b, or c with the actual choice being one of the decisions D. Depending on the decision D, stage c may or may not be redundant, or both b and c may be redundant.

Likewise stage c could have either stage a or stage b as input.

Hence the following are all possible serial systems; a b c d, a b d, a c d, a d, with the actual serial system adopted being dependent on the decisions made at c and d. Note that there could be 2, 3 or 4 stages in the final serial system.

7.6.3 Intermediate manholes and the modified serial system

Consider a run along which an unknown number of intermediate manholes are to be placed, with fixed manholes Y and Z at the upstream and downstream ends.

It is possible to define a modified serial system as described in 7.6.2 in the following way.

Define a set of possible discrete intermediate manhole positions a, b, c etc. along the run (Fig. 7.2(b)). Let each of these correspond to the downstream end of a stage in the modified serial system. The input to one of these stages is then the output from one of the upstream stages. Figure 7.2(c) shows the modified serial system.

7.6.4 Applying ISDP to the variable manhole problem

The dynamic programming is now performed in a standard way except that instead of considering the input state for a stage as being any one of the output states from the one stage immediately upstream, the input state must now be considered as any one of the output states from any feasible previous stage. A previous stage can be infeasible if the distance between the stages is greater than the maximum manhole spacing or smaller than the minimum spacing if this is specified.

As an example consider the case of possible intermediate manholes a, b, c etc. along fixed run YZ (see Fig. 7.2(b)).

- 111 -



Assume that ranges of the state variables, level and diameter, have been established at each possible intermediate manhole position along the run, and that discrete values of these variables have been specified.

Stage a has the fixed manhole Y as its upstream manhole and ends at the possible intermediate manhole position a. In general there will be a set of input states for stage a corresponding to discrete values of the input state variables, together with a set of costs corresponding to these input states. If Y is an upstream end of the network, the input states will still exist but the associated costs will be zero.

Then, for a particular output state from stage a, select the way of arriving at that state from any input state such that the total upstream cost is least whilst satisfying all the constraints.

Repeat this for every output state, thus obtaining a set of minimum total upstream costs at the output from stage a, and a set of references to identify the input state corresponding to that minimum cost.

Now consider stage b. This stage may either have the fixed manhole Y, or the possible intermediate manhole a as its upstream manhole provided that the distance from Y to b is less than the maximum spacing and that the distance from a to b is greater than the minimum spacing. Hence for a particular output state from stage b, select the way of arriving at that state from any input state either at manhole Y or at manhole a such that the total upstream cost is least, whilst satisfying all the constraints.

Repeat this for every output state, thus obtaining a set of minimum total upstream costs at the output from stage b, and a set of references to identify the upstream manhole and input state corresponding to that minimum cost.

Similarly Y, a and b can be considered as feasible upstream manholes for stage c provided the maximum or minimum spacing constraints are not violated. If, say, the distance from Y to c is greater than the maximum spacing then Y is not considered as a feasible upstream manhole for this (or subsequent) stages, and stage c is optimised using only manholes a and b.

- 113 -

This process is continued until the final fixed manhole Z is reached, this last stage being treated in an identical way to give a set of minimum costs and a set of upstream references. The process is illustrated in Figure 7.3.

If Z is the outfall to the network the costs at Z may now be examined and the output state giving the least cost selected. This gives the origin for the trace back up the run YZ. If, however, the network continues downstream of the fixed manhole Z, the trace-back origin for YZ will be obtained as part of the trace back over the whole network.

The trace back up run YZ then proceeds as follows. The upstream reference for the origin will give the upstream manhole and output state for the optimal solution. This manhole position and output state will in turn have an upstream reference to another manhole position and output state. Hence the trace back up the branch will eventually lead to the fixed manhole Y. This is illustrated in Figure 7.4.

In this way the positions of the manholes, pipe diameters and pipe levels will have been simultaneously chosen to give the least cost solution.

7.6.5 Efficiency of ISDP for the variable manhole problem

As a comparison with the DP approach proposed in 7.5.3 consider the number of designs required for the same 1 km run using consistent parameters.

Hence take intermediate manholes at 30 m spacing, with, say, a maximum manhole spacing of 150 m and a minimum of 30 m. Take 7 discrete levels and 3 discrete diameters.

There are a total of 34 manholes, giving 33 stages. Each stage has a maximum of 5 possible upstream manholes, with $7 \times 3 = 21$ input states per manhole.

Hence the maximum number of designs per stage = $21 \times 21 \times 5$ and the maximum number of elemental designs for the run

> = 21 × 21 × 5 × 33 = 72,765

This compares with 1,128,960 for the DP approach.

- 114 -



- Notes 1) possible manhole positions are numbered in downstream direction from fixed manhole 1 to fixed manhole n max
 - 2) m max = number of discrete states at each manhole

ISDP APPLIED TO VARIABLE MANHOLE POSITION PROBLEM

FIGURE 7.3



TRACE-BACK FOR THE VARIABLE MANHOLE POSITION PROBLEM

FIGURE 7.4

7.7 The set of discrete possible manhole positions

One of the key concepts involved in the ISDP approach to the intermediate manhole problem is the establishment of a set of discrete possible positions for the intermediate manholes.

To obtain a true optimal solution the intermediate manholes should not be constrained to a set of discrete positions. Hence in theory an infinite number of discrete manhole positions is required to achieve an optimal solution. Obviously the number has to be limited in practice, and this is in keeping with the discrete values adopted for the continuous state variable, pipe level. For practical highway drainage there is another justification for using a set of discrete possible positions for the manholes, this being the preference of highway engineers to the placing of manholes at convenient chainages along a road. For example a designer may well wish to have all manholes at chainages which are multiples of 10 m. Establishing a set of possible manhole positions at all such chainages along a length of carriageway drain would then give a practicable and elegant solution to the problem.

One further advantage of this approach is that manholes may be excluded from certain parts of a run by simply not specifying any manhole positions along that part. This may be necessary at, for example, bridges, culverts and road junctions.

7.8 Establishing the ranges of value for the state variables

It was assumed in section 7.6.4 that a set of discrete levels and diameters had been established at each possible intermediate manhole position as an essential prerequisite of the dynamic programming method.

In Chapter 5, the fixed plan model, this was achieved by producing a minimum cover design, (see section 5.10.3) and using this as the upper limit of both pipe level and pipe diameter. The lower limit was then a fixed distance below the upper limit for level and a fixed number of pipe diameters below the upper limit for diameters.

It would be very useful if the same approach could be used for the variable manhole problem. However it is not immediately obvious how to perform a minimum cover design when the manhole positions are undetermined.

- 117 -

If an arbitrary set of manhole positions is assumed and a minimum cover design is performed, the resulting information on maximum level and diameter will only be relevant to the manhole positions used and not to all the other positions which were possible but not selected. For example in Figure 7.5(a) if A.B.C...G are possible manhole positions, and A and G are used to establish a minimum cover design, upper bound pipe levels at B,C,D,E,F are all lower than they could be in the true optimal solution.

It is, therefore, necessary to consider a minimum cover design based on the inclusion of manholes at all possible manhole positions. Such a design is of course unrealistic but does nonetheless provide a useful upper limit on level at each possible manhole position. Experience shows that it also provides a satisfactory upper limit on diameter.

From these upper limits, lower limits of level and diameter at manholes can be established from experience (see section 7.10).

It should be noted that the range of levels at a possible manhole position as defined by the upper and lower limits applies only to a solution with a manhole at that position. Hence the final optimal solution is not constrained within a range of levels along each pipe length, only within ranges of levels at each final manhole position (see Figure 7.5(b)).

The number of discrete values and diameters used in the DP are chosen by experience (see section 7.10).

7.9 Dependence of flows on network design

7.9.1 Introduction

It was noted in section 5.11 that design flows are usually dependent on the pipe network upstream of the point under consideration, and for a rigorous DP approach using the most common, the Rational, design method three state variables are required, the third being the time to concentration. So far in this chapter it has been assumed that design flows are fixed.

It was shown in section 5.11.5 that an approximate method could be used with success for the fixed manhole case. Such a method is now required for the intermediate manhole problem before a full optimisation model can be formulated and the development of such a method is given in the following sections.

- 118 -



(a) USING SELECTED MANHOLE POSITIONS



(b) USING ALL MANHOLE POSITIONS

ESTABLISHING BOUNDS ON PIPE LEVELS FIGURE 7.5

7.9.2 An Approximate Approach

The simplest approximate approach is to assume that all design flows are fixed at some initial value and do not thereafter vary as part of the design process.

The method of fixing the initial values is rather less obvious and may be dependent on the design method. Here the Rational method is used in the development. The first approach adopted was to assume that the velocities of flow in the final solution would, for the purposes of calculating the design flows, be equal to a single uniform value of, say, 1.0 or 1.5 m/s. The actual design flows can then be calculated as a function of the total upstream equivalent impermeable catchment area and the time to concentration using the Rational method (see section 5.10.2).

With this simplification the design flow does not depend on the actual optimal set of manholes chosen or on the design of the pipe diameters or gradients. Hence a unique design flow can be specified for each possible manhole position in every run of the network. A computer program DPO was written to implement an ISDP model using this procedure.

Unfortunately comparison of flow velocities and computed flows for the resultant "optimal" design showed that large discrepancies resulted from such an approximate method. Velocities ranged from 0.5 to 3 times the original assumed value with resultant design flows out by up to 30%.

It was decided to use an iterative approach, with the new set of velocities from the "optimal" design being used to recalculate the design flows and the computer program DPO was altered to implement this. The velocities change abruptly at the manhole positions selected by the "optimal" design, but not at the other possible vacant manhole positions. This was recognised as a drawback which could lead to problems of convergence, but the method was nonetheless pursued to gain experience.

The iterations were continued until flow velocities converged to within a given tolerance. The first one or two iterations generally gave small changes in manhole positions and pipe diameters. Thereafter changes were generally limited to the pipe gradients. On one example the process failed to converge, with the solution hunting between two different manhole layouts. Generally, however, a fully

- 120 -

consistent and near optimal solution was obtained within about five iterations.

The method proved rather expensive in computer resources and could not be relied upon to converge properly. However, it was found that very rapid convergence could be achieved by first performing a minimum cover design by a conventional design procedure using all possible manhole positions, and using the resultant flows from this as the starting point of the iterations. The manhole positions and diameters resulting from the first iteration usually remained fixed during subsequent iterations with only the pipe gradient changing to accommodate changes in design flows. Moreover such a minimum cover design is also necessary to establish economical ranges for the pipe levels and diameters, hence this was the method adopted.

7.10 Experience and results of using preliminary program DPO

7.10.1 Introduction

DPO was written to develop and test ISDP and the approximate procedures for dealing with network dependent design flows.

Consequently there were frequent alterations and improvements to DPO during its working life with the program being finally superseded by MOD (see section 7.13) and DAPHOP (see section 7.16).

Hence only the general results, major limitations and conclusions will be presented here, as detailed results are somewhat meaningless in the light of subsequent improvements and alterations.

The Fortran coding for the final version of DPO is given in Appendix C.

7.10.2 The test networks

Several test networks were used in the development of DPO, the one shown in figure 7.6 (Network 2) being used extensively in the investigation of the sensitivity of the solution to the choice of parameters.

7.10.3 General results

The following general observations could be made about the preliminary runs of DPO.

a) Manhole positions and pipe diameters were relatively insensitive to the design flows used at each iteration, whereas pipe slopes varied considerably between iterations.

- 121 -



- 122 -

b) Manhole positions were quite sensitive to the choice of spacing of the discrete pipe depth state variable.

c) Optimal designs tended to lie within a 0.5 m zone below a level defined by a pipe at minimum possible cover.

d) Pipe diameters were always less than, and within two increments of, the diameter obtained from a minimum cover design.

e) Optimal costs were 5-15% lower than those of equivalent designs using the conventional minimum cover criterion.

f) Solutions obtained with discrete pipe depth increments less than about 0.15 m and adequate range of depths generally gave optimal manhole positions and pipe diameters on the first iteration.

7.10.4 Selection of Optimising Parameters

Thirteen runs were executed using DPO on network 2 (figure 7.6) for the purpose of establishing the sensitivity of the "optimal" solution to the choice of the two main optimising parameters. These parameters are

a) Spacing of discrete levels for pipe level state variable;

b) Spacing of possible intermediate manholes.

Checks were made on the results to ensure that there was a sufficient range of pipe levels and a sufficient number of discrete pipe diameters considered for the optimal solution to be within the bounds of the available values. For example if the results showed that the solution lay at or close to the lower bound of the pipe levels, the range of levels was increased without altering the spacing of levels and the optimal solution re-computed. If the solutions were found to be identical, it was assumed that the solution was then indeed optimal.

The results of these runs are shown in the graphs of figure 7.7, but should be treated with caution as they relate to a single network and are insufficient in number to establish any specific conclusions.

7.11 The variable manhole position model - MANVAR

7.11.1 Introduction

The results of the preceding section led to the formation of a new model which is both economic and robust. For convenience it is referred to as the MANVAR model, and is the basis of a fully commercial program, DAPHOP, which is described in section 7.16.

- 123 -



- 124 -

MANVAR essentially consists of four distinct stages. These are:

- a) producing a set of possible intermediate manhole positions;
- b) producing a minimum cover design and limits on the ISDP design;
- c) performing a coarse ISDP design;
- d) performing a final exact solution.

Stage (a) consists of taking the skeleton layout of the network and producing sets of possible intermediate manhole positions along all relevant runs. Ground levels and catchment areas are also produced for all the generated manhole positions.

Stage (b) consists of performing a minimum cover design on the network assuming a manhole at every possible location using the design method of one's choice (Rational, TRRL etc.) to obtain upper limits on the pipe levels and diameters and the design flow at every manhole location. Lower limits on level and diameter are also set at this stage.

Stage (c) consists of a single ISDP design using a coarse grid of discrete levels, the grid of discrete diameters and the sets of possible intermediate manholes. This gives a "coarse optimal" solution, for which the actual flows will differ somewhat from the true design flows, but which, from result (f) of section 7.10.3, will generally give the optimal values of manhole positions and pipe diameters for all pipes in the network.

Stage (d) consists of taking the new network of pipes of known diameter as defined by the optimal set of manhole positions and pipe diameters, and designing the pipe gradients using the chosen design method, thereby ensuring a fully consistent final design. This effectively truncates the iterative procedure of section 7.9.2 thereby making a far more efficient model with little or no penalty incurred.

7.11.3 Implementation of MANVAR

MANVAR was implemented as two computer programs, ASSEMB and MOD, linked by a data file (see figure 7.8). Input to MOD may be either from ASSEMB or direct from the user.



IMPLEMENTING THE MANVAR MODEL

FIGURE 7.8

7.12 Program ASSEMB

7.12.1 Introduction

As explained in section 7.7 an essential prerequisite of the ISDP process is the establishment of sets of possible intermediate manhole positions along all relevant runs.

This can be done by the user first deciding on the required spacing of the intermediate manholes and then calculating and inputting all such manhole positions. This was done manually for the earliest variable manhole test runs but is tedious and is a task best performed by the computer. Hence at an early stage in the development of MANVAR a subsidiary computer program called ASSEMB was written, part of the function of which was to take the skeleton layout input by a user and produce a full set of possible intermediate manhole positions together with their associated ground levels and catchment areas.

The second function was to define the upper limit on the pipe levels for the whole network, working from information on ground levels, obstructions, minimum pipe gradient and connectivity only. It is assumed that pipes at manholes are positioned such that the downstream soffit level is at or below the soffit level of the lowest upstream pipe. Hence this upper limit can be defined without reference to pipe diameters or flows, and hence in fact without the design method being defined.

The final function was to arrange data to a form convenient for the main program.

7.12.2 The program

A flow chart showing the essential features of ASSEMB is given in figure 7.9 and a full listing of the Fortran Program is given in Appendix D.

7.12.3 Input

The input to ASSEMB consists of a) the basic design parameters (e.g. minimum pipe gradient, minimum cover, minimum flow velocity, minimum and maximum manhole spacing). b) available pipe sizes.

- c) optimising parameters for main program.
- d) type, length and catchment width for each drainage run in the



FIGURE 7.9

network. (The "type" determines whether or not there is any catchment to be assigned along the length of the run, and whether intermediate manholes are to be placed along the run. For the purpose of assigning catchments to the intermediate manholes it is assumed, where relevant, that the catchment area is of a uniform width parallel to the run.)

- e) ground levels along each run.
- f) details of obstructions along each run.
- g) the connectivity of the network.

7.12.4 Output

The output from ASSEMB forms the complete input to the next part of MANVAR and consists of the following:

- a) optimising parameters
- b) pipe sizes
- c) basic design parameters
- d) number of runs in the network
- e) for each run: number of possible manholes
 - : positions along branch of each m/h
 - : cumulative catchment area for each m/h
 - : pipe soffit level for a min. cover design (= upper limit for DP design) for each m/h
 - : lower limit for DP design for each m/h
 - : range of feasible upstream connections for each m/h (governed by max. and min. m/h spacing)
 - : ground level data along run
 - : identification of any runs upstream

f) problem size (total number of possible manholes, total number of ground levels, probable maximum number of manholes in final design).

7.12.5 Use of ASSEMB

For ease of data preparation ASSEMB may be used either interactively or remotely. Data generated by ASSEMB is written onto a card-punch file for compatibility with manually created data, and to allow small modifications, (e.g. to the optimising parameters) by changing individual lines of the output without re-running ASSEMB. The card-punch file could be listed onto actual punch cards, but on the computer system used for this research it was more convenient to use card image files within the computer memory.

7.13 Program MOD

7.13.1 Introduction

The remainder of the MANVAR model is implemented by the computer program MOD.

Production of a minimum cover design could be performed manually or by an existing computer program (e.g. DAPHNE (ref. 56) or TRRL (ref. 6)). However MOD incorporates a minimum cover design procedure producing a minimum cover design for a network consisting of all possible intermediate manholes. The design method used is the Rational, with rainfall calculated by the modified Bilham formulae (ref. 5).

The heart of the program is the ISDP design of the network. This can only be performed using a computer due to the very large number of calculations involved (see 7.6.5). Computer storage and execution time become critical factors influencing the structure of the program and the choice of parameters for the optimisation.

The final part of MOD produces a fully consistent design for the network, based on the manhole positions and diameters chosen by the ISDP optimisation. This design essentially consists of finding the correct pipe gradients, the Rational method again being used.

7.13.2 The program

MOD consists of approximately 830 lines of standard Fortran, there being a main program and thirteen subroutines. A listing of the full program is given in Appendix E.

A flow chart showing the principal features of the program is given in figure 7.10.

To minimise storage requirements most data is stored in four large arrays which are dynamically addressed, thus preventing large redundant areas of storage, and reducing memory requirements to a modest size for a main-frame computer.

The program was written in National Computer Centre (NCC) Standard Fortran (ref. 57) with further limitations imposed by HECB standards (ref. 58). This was to enable the final commercial version to be machine independent and fully transferable to any large computer. Many of the subroutines used in MOD were used in the commercial version and elsewhere. This does however incur some penalties in



FLOW CHART FOR PROGRAM MOD

FIGURE 7.10

terms of the number of lines of coding and the execution time of the program. These are, however, felt to be minor compared to the benefits of interchangeability.

7.13.3 Input

Input data is either generated by ASSEMB or can be created manually. In either case it is on cards (either real cards or a card image file) and has the same format as the output from ASSEMB (see 7.12.4).

7.13.4 Output

÷

The output essentially consists of the final optimum design giving all final manhole places, pipe diameters, levels and gradients, together with flows, pipe capacities and details of cost.

Additional information is optionally available giving details of the initial minimum gradient design and details of the optimisation process but these were primarily for diagnosis in the event of failure during program development.

7.14 Results from using ASSEMB and MOD

7.14.1 Introduction

The computer runs using MOD may be divided into four groups:

- a) checking that results are consistent with DPO
- b) finding the effects of varying the optimising and design parameters
- c) checking performance on various networks
- d) preliminary investigations for the variable cross-drain problem

The results of a, b and c are presented and discussed below, and are tabulated in Table 7.1. Results from d will be presented and discussed in Chapter 8.

7.14.2 Checks on the consistency of MOD and DPO

Due to minor changes in the costing routines and in the method used to calculate rainfall, MOD could not be expected to be in full agreement with DPO.

Three examples using network 2 (fig. 7.6) were tested on MOD (see Table 7.1) and the results compared to those using DPO. A large measure of agreement was noted, with pipe diameters and manhole positions identical. There were small differences in pipe gradients

Network	ianhole Spacing SP (m)	Zone Depth RZ (m)	Number of Levels	Number of Diameters	hinimun Spacing (m)	'ilme of Antry te (mins)	Cost of Construction (£)	Execution Time (secs)
2	10	0.5	5	3	30	2	14064	19
2	30	0.5	5	3	30	2	14204	4
2	60	0.5	5	3	30	2	14329	1
3	120	0.5	5	3	30	2	108084	3
3	60	0.5	5.	3	3 0	2	107450	6
3	30	0.5	5	3	· 30	2	107323	19
3	10	0.5	5	3	30	2	106572	140
3	30	1.0	11	3	30	2	106984	75
3	30	1.0	11	3	30	2	106693	72
3	30	0.5/1.5	6	3	30	2	107114	26
3	30	0.5/1.5	8	3	30	2	106864	42
3	30	0.5/1.5	11	3	30	2	106643	75
3	30	0.6/1.2	7	3	30	2	1066 93	33
3	30	2.0	21	3	30	2	106220	243
3	30	3.0	31	3	30	2	106220	518
3	10	2.0	21	3	30	2	105860	1884
3	120	0.6	7	2	30	2	108084	4
3	60	0.6	7	2	30	2	107121	9
3	30	0.6	7	2	30	2	107021	30
3	10	0.6	7	2	30	2	106725	22 3
3	30	1.5	16	3	60	2	106220	108
3	20	1.0	11	3	6 0	2	106902	107
3	20	1.0	11	3	60	4	104067	107
3	20	1.0	11	3	60	6	102028	107
3	20	1.0	11	3	60	8	100103	106
3	20	1.0	11	3	60	10	98769	109

-

TESTS USING MOD

and costs, however, but these were thought acceptable. It was concluded that MOD was largely consistent with DPO.

7.14.3 The effect of varying the optimising parameters

The results of 23 runs using MOD on network 3 (figure 7.11) are tabulated in Table 7.1 Seventeen of these runs had identical design parameters, but varying optimising parameters. Execution times varied from 3 seconds to 1884 seconds with network construction costs ranging from £108100 to £105900 respectively, representing savings of from 2% to 4% over a likely minimum gradient solution (see fig. 7.12(b)) costing f110,100. A typical optimal solution is shown in fig. 7.12(a). Fig. 7.13 shows program execution time plotted against network costs on a logarithmic scale. The general trend shows that the optimal solution is approached as the time spent on computing increases. There must be an absolute value for the true optimal solution but this is unknown. If the intermediate manhole spacing was decreased to a very small distance and the spacing of levels greatly decreased whilst retaining a wide level zone this optimal solution would be approached. However, the execution time would be enormous and hence computing costs would be greatly in excess of any possible saving on construction cost.

On the other hand, very small execution times still show sizeable savings on construction costs, for negligible computing costs.

Obviously a balance must be achieved between computer costs and likely savings as explained in section 5.15.

Costs of running MOD on Liverpool University's I.C.L. 1906S computer work out at approximately £0.16 per second. Hence for a 100 second job the execution cost is £16.

For practical considerations (see section 5.15) it was felt that the ratio of (extra saving on construction costs)/(extra computing costs) should not fall below about 20 in a commercial program. This, applied on a diminishing returns basis, leads to a maximum execution time of about 150 secs for this network on the 1906S.

This, then, forms a restraint on the selection of the optimising parameters. The parameters that effect the performance of the optimisation and the program execution time are: a) spacing of possible intermediate manholes (SP) b) number of discrete pipe levels considered (M)

- 134 -


- 135 -



FIGURE 7.12

- 136 -

<u>note:</u> diameters in mm.



CONSTRUCTION COST vs EXECUTION TIME

FIGURE 7.13

- c) range of pipe levels considered (RZ)
- d) range of pipe diameters considered (J)

For the reasons described in section 7.10.3 (f) the spacing of discrete pipe levels was kept within the range 0.1 to 0.15, there being no apparent advantage in decreasing the spacing below this, and with increases of spacing likely to cause sub-optimal manhole positions and diameters. Hence M and RZ are linked by RZ \ddagger (M-1) \times 0.1. Generally the range of diameters considered was kept to 3, but reduced to 2 for four of the runs.

Hence in practice SP and RZ become the only two important parameters.

The effects of varying these are shown in figure 7.14. It can be seen that there appears to be a zone depth above which there is no further reduction in cost, this occurring at about 1.5 m. However, decreasing manhole spacing leads to increased savings with no such obvious limit.

The results are combined in fig. 7.15(a), the costs being given as approximate contours. The line joining points with execution times of 150 secs is also sketched in. If the execution time is restricted to this value, the correct choice of parameters for the most effective use of computer time is a zone depth of about 1.5 m with a manhole spacing of about 30 m.

Several runs were performed (Table 7.1) with RZ locally widening towards the end of long pipe branches especially where these branches ended well above the level of the main pipe in the minimum cover design. Such runs were found to be more efficient than those using constant values of RZ, using generally about half the computer time to achieve the same cost savings.

Four runs were performed using only 2 possible pipe diameters. These showed only small savings in computer time. The penalties involved in using only 2 diameters were not large in this example, but neither were the advantages. Hence it was felt unnecessary and rather unwise to adopt less than 3 diameters in general.

Most runs used a minimum manhole spacing of 30 m. The exceptions used a minimum spacing of 60 m and in the one case where comparison was possible this was rather more efficient, using less computer time than for an equivalent run with a 30 m minimum spacing and no cost penalty.



SENSITIVITY OF NETWORK COST TO OPTIMISING PARAMETERS: PROGRAM MOD

FIGURE 7.14



(a) NETWORK COSTS FOR COMBINATION OF OPTIMISING PARAMETERS





7.14.4 Varying the design parameters

Having established an optimal design program for drainage design it is relatively easy and informative to find the effects on the design and cost of construction of varying certain of the design parameters.

Such parameters, the values of which are at present selected on a rather arbitrary basis, could include minimum cover over the pipe, minimum slope, pipe roughness, maximum manhole spacing, storm return period and time of entry of runoff into the pipe system. Such a study was not the objective of this research, but five runs were performed on Network 3 (fig. 7.11) using varying times of entry (te) to the pipe system (see Rational Method, section 5.11.2) as it was felt that this may be greatly underestimated in current design practice.

At present te is generally taken to be 2 minutes. Fig. 7.15(b) shows the effect on the network cost of taking te equal to 2,4,6,8 and 10 minutes, resulting in reductions in network cost of up to 7.5%.

7.14.5 Tests on other networks

MOD was used on rather more complicated road drainage networks similar to Network 3 but with one or two additional cross-drains. These runs were primarily to investigate the possibilities of variable cross-drain optimisation and will not be discussed here except to say that in all cases MOD produced sensible results, with rather greater cost savings than for Network 3.

In addition one further design network was used as an example. This consisted of a single run as shown in figure 7.16. The plan length of pipe run between the two fixed position manholes at the ends was 840 m. The ground profile has three regions each at a different linear slope. Possible manhole positions were defined at 10 m spacing along the pipe run with a maximum permissible distance of 150 m between manholes. A 1 m zone was used for the levels with 11 discrete pipe levels at 0.1 m spacing. Three pipe diameters were used at each location.

The optimal design uses 6 intermediate manholes placed at the chainages shown. The cost of this optimum design is 12493 units and took about 120 secs execution time on the Liverpool 1906S computer.

- 141 -



DESIGN EXAMPLE FIGURE 7.16

- 142 -

In order to compare it with a traditional manual design a standard design was produced and costed by hand. The definition of such a standard design is, as in all variable plan problems, rather subjective. However, the following logical stages were taken: a) place manholes at both ground discontinuities; b) place manholes at equal spacings in each sloping portion subject to a maximum spacing of 150 m;

c) design pipes at minimum cover subject to a minimum gradient requirement.

Using the same cost function as for the optimal design the standard design cost was 14491 units, i.e. 16% more expensive than the computed optimal.

This example demonstrates that considerable savings can be made by making relatively small adjustments to manhole positions and gradients. Most of the saving in this example is effected by reducing the pipe diameter in the central region from 375 mm to 300 mm, and by reducing the final pipe in the run from 450 mm to 375 mm.

7.15 Conclusions from using MOD

Using MOD on the large and realistic road drainage example of Network 3 showed two important differences to the results obtained from using DPO on preliminary examples.

These were firstly that the savings likely to be achieved over a sensible minimum gradient design were substantially less than expected. They were approximately 3% to 4% as opposed to the 5% to 15% expected.

Secondly the optimal solution could only be obtained by taking a range of depths of about 1.5 m or more, as opposed to the expected range of about 0.6 m.

The first point could be explained by the long lengths of gently sloping ground profiles, typical of a road carriageway, and the small number of pipe intersections. In fact in examples involving additional cross-drains MOD produced rather larger cost savings. Hence it would seem likely that the most spectacular results are achieved with rapidly varying ground levels and complicated pipe networks.

The second point is again probably explained by the long lengths of the runs as opposed to the relatively short lengths in previous examples, or perhaps just to the overall larger size of the network. It may make sense to select the range of depth considered according to the maximum drainage path through the network, or have a variable range, the range at a manhole being a function of the length of the longest drainage path upstream of that manhole. This could be achieved by having a variable number of discrete levels, but the data handling routines would be rather more complicated than in MOD, and extra data would be required to define the number of levels at each point in the network. Nevertheless it remains a sensible approach for further investigation.

The choice of parameters to obtain the best solution for a reasonable outlay in computer resources is not obvious, and varies depending on the example. It seems reasonably clear that only 3 pipe diameters need be considered provided that the available diameters are in the standard 75 mm nominal increments, and that the original minimum gradient design produced does not result in artificially large diameters due to the use of a very small minimum gradient (less than, say 1 in 300).

Based on the information to date, a range of depths of at least 1.5 m is needed to ensure that the optimal design is not excluded. As 0.15 m is about the maximum spacing of discrete levels allowable this requires 11 discrete levels to ensure the solution is reasonably close to the optimal. This conclusion will need checking in the light of further experience in using MOD.

Having fixed the number of pipe diameters and suggested the range of levels and their spacing, it remains only to fix the spacing of intermediate manholes. The cost of the solution will continue to decrease as the manhole spacing decreases towards zero, whilst the computer costs involved rise rapidly. It is probably not worth while decreasing the spacing below 10 m, both on economic grounds and because of the desirability of keeping manholes at convenient chainages (see section 7.7), and spacings of 30 m or even 60 m may give reasonable answers with more acceptable computer execution times. Note that these figures relate to a maximum manhole spacing of 120 m, and are convenient fractions of 120 m. For another maximum manhole spacing other similar fractions of distance would be more appropriate.

The most important conclusion relating to the choice of parameters must however be that whatever computer program is developed for the MANVAR system, it must be flexible enough to incorporate changes in these parameters as experience is built up of their use.

It remains very clear that even though savings may not be as great as previously expected, large sums of money on construction costs can be saved by a very small outlay in computer time. This will always be worthwhile. Greater outlay on computing will save larger amounts on the construction costs. The extent of investment in computing time to obtain more savings in construction is then a matter of policy for those in charge of the design procedure, and may be controlled by careful selection of the optimising parameters.

7.16 A commercial program

7.16.1 Introduction

The funding that enabled the bulk of this research to take place was provided by the Department of Transport, Highway Engineering Computer Branch.

Their principal requirement was the production of a fully commercial optimal drainage design program for roads based on their existing DAPHNE highway drainage design program (ref. 56).

For reasons described in Chapter 8, it was decided that this program should be based on the MANVAR model and not on the CROSSVAR (variable cross drain) model described in that chapter.

For convenience the program will be referred to here as DAPHOP, although when released it will probably be as a user-selected option of a new version of DAPHNE.

7.16.2 The existing program DAPHNE

DAPHNE consists of approximately 8500 lines of Standard Fortran. Much of the coding is required for handling, interpreting and checking input data and outputting results and messages.

DAPHNE uses data which is already available in the form of computer files to define all the road geometry (alignment, crossfalls etc.), this information being available from running the BIPS suite of programs (ref. 59) for the design of highways. The DAPHNE user then defines the drainage network he requires, including all manhole positions, and the design parameters he wishes to use. DAPHNE then calculates all catchment areas, calculates design flows according to the Rational method and designs all pipe diameters with pipes at minimum possible cover.

7.16.3 The optimising version: DAPHOP

DAPHOP is structured on the MANVAR model, the details being shown in figure 7.17. Basically DAPHOP and the joint ASSEMB and MOD programs, are very similar except that DAPHOP uses the existing DAPHNE routines to establish a minimum cover design and to perform the final design of pipe gradients. The efficiency of some of the MOD routines and the data handling were improved before incorporation into DAPHOP, resulting in a generally more compact and efficient program than would otherwise have been possible.

DAPHOP is at present undergoing trials with the DOT before being released for general use.

7.17 Conclusions on the MANVAR model

Experience has shown that the MANVAR model is fully practicable as a storm drainage design program for networks in which there are branches along which unknown numbers of intermediate manholes are to be placed. This type of network is typical of highway drainage.

The extent to which the MANVAR produced design approaches the true optimal solution is dependent on the choice of parameters in the optimising routines. The choice of parameters determines the cost of running MANVAR and may be limited by the size of available computing memory.

Significant savings can always be made by adopting very coarse parameters (e.g. a zone depth of 0.6 m and a manhole spacing of 60 m) at minimal computing costs. Larger investment in computing will result in larger cost savings, there being a practical rather than a theoretical limitation on this.



STRUCTURE OF DAPHOP FIGURE 7.17

CHAPTER 8

THE VARIABLE CROSSDRAIN POSITION MODEL : CROSSVAR

- 8.1 Introduction
- 8.2 Defining the Problem
- 8.3 Methods of Approach
- 8.4 Fibonacci Search
 - 8.4.1 Trial using Fibonac1 search
 - 8.4.2 Results
 - 8.4.3 Conclusions
- 8.5 Polytope Search
 - 8.5.1 Introduction
 - 8.5.2 Results using a polytope search
 - 8.5.3 Conclusions
- 8.6 Dynamic Programming Approach
- 8.7 Variable Cross-drains and the Modified Serial System
- 8.8 Applying ISDP to the Variable Cross-drain Problem-
- 8.9 The Design of a Stage
- 8.10 Establishing the Ranges of Value for the State Variables
- 8.11 Design Flows
- 8.12 Cross-drain Sets with Networks Upstream
- 8.13 Cross-drain Sets Sharing a Common Base Cross-drain
- 8.14 A Practicable Model : CROSSVAR 8.14.1 Introduction 8.14.2 Structure of CROSSVAR
- 8.15 Program MODEX
- 8.16 Optimising Parameters for CROSSVAR
- 8.17 Program of Testing for CROSSVAR
- 8.18 Results Using the CROSSVAR Model
 - 8.18.1 Checking CROSSVAR with previous results
 - 8.18.2 Finding typical cross-drain spacings
 - 8.18.3 Stability of cross-drain positions to variation of parameters
 - 8.18.4 The effect of cross-drain resolution on the optimality of the solution
 - 8.18.5 Runs using other networks
- 8.19 Choice of Values of the Optimising Parameters
- 8.20 Conclusions on the Use of CROSSVAR - 148 -

Chapter 8. The Variable Cross-Drain Position Model - CROSSVAR

8.1 Introduction

In this chapter the second of the variable plan optimisation models for road drainage design is presented. This involves the determination of the number and position of cross-drains in a storm water drainage network for highways. As outlined in Section 6.3, this essentially completes the optimal design process for such networks.

8.2 Defining the Problem.

A typical network (Fig. 8.1) consists of carriageway drains (AD, BE, CF) connected to a base cross-drain (DEF) connected in turn to carrier drains (FG, GH).

However a number of additional cross-drains could be added (eg.IJK,LMN) thus diverting the flow from AD to AIJKF and ILMNF. Drains IL, LD will have zero flow at their upstream ends but will collect flow along their lengths.

Flow along BE is similarly diverted. The overall result is for the drains KN, NF to be substantially larger than CF was, and for IL, LD, JM, ME to be smaller. This may well be cheaper to construct than the original basic layout.

The problem is thus to find the number of cross-drains and their positions that will result in the network of minimum construction cost.

To make the model complete, it is necessary to find the number and positions of all intermediate manholes for each pipe run such as AI and the diameters and slopes of all pipes. This can be accomplished by incorporating the MANVAR model (see Chapter 7) into the present model.



(a) TYPICAL NETWORK



(b) POSITION OF SINGLE CROSS-DRAIN



FIGURE 8.1

8.3 Methods of approach.

Several different approaches were investigated in preliminary studies of the problem before D.P. was again selected as being the most likely contender. These preliminary studies used the MANVAR model as a step in a search procedure for the optimal solution and are described in the following sections.

8.4 Fibonacci Search.

If the problem is simplified greatly by assuming that only one additional cross-drain is required for the optimal solution, the well-known Fibonacci Search method (ref. 60) can be employed.

Defining the distance of the additional cross-drain from the base cross-drain as x (Fig. 8.1) the total network cost can be expressed as f(x) where f(x) can be evaluated for any feasible x by running the MANVAR model. It is then necessary to adopt a one-dimensional search technique to find the position of x that makes f(x) a minimum. It is necessary to make two assumptions, firstly, that there is a value of x that minimises f(x) within the range of x considered, and secondly that f(x) is unimodal within this range. It can then be shown (ref.61) that no grid search technique can be guaranteed to find the minimum in less function evaluations than the Fibonacci method.

Essentially the method consists of the following steps:

- i) consider a set of positions x covering the range of interest.
- ii) evaluate f(x) at a specified pair of points x_1 , x_2 .
- iii) as f(x) is unimodal, from the values of $f(x_1)$ and $f(x_2)$ determine whether the value of x that minimises f(x) is > x_1 or < x_2 . This narrows the range of x that need be considered.
- iv) with reduced range, and knowing one value of f(x) within this range (either $f(x_1)$ or $f(x_2)$) determine f(x) at a specified point x_3 and repeat the process.

- 151 -

The points x_1 , x_2 , x_3 etc. are determined by reference to the Fibonacci series of numbers 0, 1, 1, 2, 3, 5, 8, 13, 21,34 etc.

In this way the minimum value of f(x) can be found for a large number of possible positions of x with the minimum number of evaluations. e.g. for 88 positions of x, only 9 function evaluations need be made.

8.41 Trial Using Fibonacci Search.

An attempt was made to apply this technique to finding the optimum position of an additional cross-drain for Network 3 (fig. 7.11). 88 points were used, thus requiring a total of 9 evaluations. For this preliminary work, the MANVAR programs were used, with each evaluation requiring a new run using a manually altered set of input data.

Defining x as in Fig.8.1c, the problem can be stated as follows: Find the value of x that minimises the construction cost of the network, where 0 < x < 2000.

It was originally thought that x would be about half way along the 2000 m long parallel drainage runs. Hence the search was restricted to the region $560 \le x \le 1430$ Using a 10 m grid interval allows 88 possible positions for x, varying from point (1), x = 560 m to point (88), x = 1430 m. In general for point n, x = 550 + 10 n.

Assuming that the cost function is unimodal and has a minimum value within this range, the optimum value of x should then be obtained in 9 function evaluations, i.e. 9 runs of the MANVAR model, using networks defined by different positions of x. The choice of points at which to evaluate f(x) is shown in Fig. 8.2.



	Initial Value for Element								
Array	1	2	3	4	5	6	7	8	9
8	1	1	2	3	5	8	13	21	34
ь	-	2	3	5	8	13	21	34	55
c	-	1	2	3	5	8	13	21	34
d	-	2	3	5	8	13	21	34	55

FIBONACCI SEARCH OVER 88 INTERNAL POINTS

FIGURE 8.2

8.4.2 Results.

The results of applying Fibonacci search to the region $560 \le x \le 1430$ are shown in Table 8.1.

After just four evaluations it became clear by plotting a graph of the points (Fig.8.3) that the optimal solutions could well lie outside the range being investigated, thus invalidating the search technique.

Table 8.1 Fibonacci Search.

<u>Pt.</u>	<u></u>	Cost						
55	1100	£107297		Solution	in	range	1	to FF
34	890	£106561	J	Solucion				20 33
21	760	£106512		Solution	in	range	1	to 34
13	680	£106237		Solution	in	range	1	to 21

Consequently further values of x were used outside the range initially considered. These are tabulated in Table 8.2 in chronoligical order and plotted also on fig. 8.3.

Table 8.2

x	Cost
400	£105689
200	£105710
320	£105598
480	£105602
360	£105555

Two results are of immediate interest. Firstly the assumption of the likely position of the cross-drain was erroneous, and secondly the assumption of unimodal behaviour is also invalid, at least in close proximity to the optimal solution.

- 154 -



OPTIMAL POSITION OF A SINGLE CROSS-DRAIN

FIGURE 8.3

A comparison with two other solutions is also informative. Firstly if no additional cross-drain is used, the optimal solution using the same parameters as for the results above would cost £106 220 (£665 or 0.6%more expensive than the cheapest single cross-drain solution)

Secondly a manual, minimum cover design, with one additional cross-drain placed at x = 960 would cost £111 533 (£5978 or 5.7% more expensive). This represents a typical solution using current design practice.

8.4.3 Conclusions.

Had the Fibonacci search covered the whole of the range 0 < x < 2000it would probably have attained the optimal solution with a similar number of function evaluations as were actually employed using the graphical plot as a guide.

However as the function cannot be relied on to be unimodal, the technique may have foundered, and it is felt that it is therefore insuff sciently robust to be of general use in this application.

The optimal position of the cross-drain, being only 320 m upstream of the base cross-drain, suggests that several more cross-drains at similar spacings may be required for a truly optimal solution. Hence a simple univariable search is probably not appropriate.

8.5 Polytope Search.

8.5.1 Intrdouction.

As the Fibonacci method was limited to the possibility of a single cross-drain it was clearly of limited use, some more general method being desirable.

- 156 -

The nature of the objective function rules out almost all the recognised multivariable optimisation algorithms with the exception of the polytope search technique - sometimes known as the simplex method (ref. 60)

A polytope consists of a pattern of at least (n+1) points defining a non-zero volume in n-dimensional space. Hence if there are n variables in the problem, the first step is to evaluate the function at (n+1) appropriate points. The highest of these values is then discarded, and the function is evaluated at a new point, this being the reflection of the discarded point about the centroid of the remaining points. A new polytope is thus formed, the highest value of the function again being discarded and the process repeated.

Various rules can be applied to increase or decrease the size of the polytope and to deal with constraints on the position of the vertices.

8.5.2 Results using a Polytope search.

This technique was applied to finding the optimum positions of two cross-drains for Network 3 (fig. 8.4a). A simple two-dimensional simplex was used. As it was felt desirable to keep cross-drains to sensible chainages, (e.g. multiples of 100 m) a right-angled isosceles triangle was used (fig. 8.4b). The method was applied manually using MANVAR to evaluate the function at the chosen vertices.

Vertices (1) (2) and (3) (fig.8.4b) were evaluated initially. The results of all evaluations are given in Table 8.3. As point (2) gave the highes cost this vertex should have been reflected about the mid-point of (1)-(3) to give the new vertex. However this would have created a vertex at pt.(600,600), giving coincident positions for the 2 cross-drains. Hence the mearest position which preserved the shape of the polytope was chosen for the new vertex, this being point 4.

- 157 -









Evaluation of vertex (4) identified vertex (3) as having the current highest cost. This vertex was then reflected about the centre of (1)-(4)and the new vertex (5) was identified and evaluated.

Of the current vertices (1, 4 and 5), (1) was the most expensive. Hence a vertex at (6) was identified and evaluated.

Of the current vertices ((4) (5) and (6)), (5) was the most expensive. As the next vertex would have been point (400,400) giving coincident crossdrains, and there was no other new point available for a vertex whilst preserving the same size and shape of polytope, the logical next step would have been to decrease the polytope size and to continue the search. The search was, however, terminated here as it was felt that sufficient information had been gained about the method. At termination the cheapest vertex was (4) having cross-drains at 400 and 600 m from the base crossdrain.

Table 8.3	Polytop	e Search	
Vert	ex x1	<u>x</u> 2	Cost
1	4 00	800	106271
2	400	1000	106841
3	600	800	106367
4	400	600	105926
5	200	600	106249
6	200	400	105944

- 159 -

8.5.3 Conclusions.

The results confirmed one finding of the one dimensional search, namely that the optimal number of cross-drains was not obvious and could be quite large, due to the close spacing of the optimal cross-drains in the cases investigated.

Thus for a complete optimisation, polytope searches for 2, 3, 4 etc. variable cross-drain positions would have to be implemented. Although such a procedure is possible for up to about 6 variables using the Polytope method it would require a large number of function evaluations and be very inefficient.

Hence it was felt that although a Polytope search could be applied if the number of cross drains were known, the technique was not suited for the present case.

8.6 Dynamic Programming Approach.

It had become clear that any method which relied on the number of cross-drains being predetermined was of little practicable use, as results indicated that several cross-drains were normally economical, rather than zero, one or two.

The research done on the variable manhole position problem (Chapter 7) had shown that I.S.D.P. was capable of handling a similar situation and it was decided to investigate whether I.S.D.P. could be applied again.

8.7 Variable cross-drains and the modified serial system.

Section 7.6.2 introduces the concept of a modified serial system, essential for the implementation of I.S.D.P. Consider now a typical length of highway drainage consisting of a base cross-drain fixed in position into which run three roughly parallel carriageway drains. (Fig.8.5a)

It is possible to define a modified serial system in the following way.

- 160 -



Define a set of possible discrete cross-drain positions a, b, c, etc. along the length of the drainage network. (Fig. 8.5b)

Let each of these correspond to the downstream end of a stage in the modified serial system. The output from a stage is then the lower pipe level and larger pipe diameter of the pipes that meet at the downstream end of the cross-drain. The input to a stage is the output from one of the upstream stages. Figure 8.5cshows a typical stage and Figure 8.5d the modified serial system.

8.8 Applying I.S.D.P. to the variable cross-drain problem.

The I.S.D.P. can now be structured as follows.

For cross-drain (a) obtain minimum costs for a set of output states for the network consisting of the cross-drain at (a) plus the length of carriageway drains upstream of the cross-drain.

For cross-drain (b) there is either no cross-drain upstream, or a cross-drain at (a). Hence obtain minimum costs for a set of output states at (b) considering both of the following possible networks.

i) cross-drain (b) plus all carriageway drains upstream.

ii) cross-drain (b), plus carriageway drains from (a) to (b) plus the set of minimum costs corresponding to input states at (a).

In general a cross-drain stage may have any of the possible upstream cross-drains forming its input state, or none at all.

In this way the I.S.D.P. proceeds downstream to the base cross-drain, giving a set of minimum costs corresponding to a range of depths and diameters. The process may then be continued by conventional D.P. through to the downstream end of the network.

When the overall minimum network cost is identified the solution can be traced back up the network and the optimal plan identified.

- 162 -

8.9 The design of a stage

The design of a cross-drain stage consists of finding the least cost solutions for all possible networks upstream for a range of output states. As outlined in Section 8.8, this is accomplished by considering the nearest upstream cross-drain to be in each possible position, or there to be no upstream cross-drain. Considering just one position of the upstream cross-drain, Fig.8.5c, the problem can now be stated as follows:

Given a set of input costs corresponding to input states S(in) find the set of designs for the network between the cross-draing that minimises the costs corresponding to output states S(out).

This can be done conveniently using the MANVAR model, thus giving optimal intermediate manhole positions and optimal pipe slopes and diameters.

8.10 Establishing the ranges of value for the state variables

For any D.P. process the ranges of values of the state variables need to be defined, ie. in Fig.8.5c the limits on S(in) and S(out) need to be fixed.

The range of the pipe level could be fixed in relation to the ground level, but this is rather inefficient. It is far better to fix the range in relation to the maximum possible pipe level, which will generally coincide with the "minimum cover" design for a given upstream network.

Here, however, the network upstream is not predetermined, and hence there is no unique minimum gradient design from which to obtain an upper limit on pipe level.

Hence all possible upstream networks should strictly be considered in obtaining the upper limit on pipe level at a given cross-drain position.

- 163 -

If pipe diameter is also a state variable (i.e. if diameters are constrained not to decrease in a downstream direction) then the maximum pipe diameter may be obtained from the same "minimum cover " design procedure.

8.11 Design flows

So far in this chapter it has been assumed that design flows can readily be calculated for the individual components of the design stages and that flows at the main stages are independent of the networks upstream.

Taking the former point first, provided the inlet points for flows into all the carriageway pipes are defined (e.g. gullies) or are continuous along the pipe lengths (e.g. French drain) then for each subnetwork, design flows can readily be calculated as for the MANVAR model.

The latter point requires rather more attention. Consider the flow just downstream of F in two extreme cases (Fig. 8.6a and 8.6b)

Using the Rational method as an illustration, the design flow at F depends on the time to concentration t and the catchment area A. A will be equal for the two cases. However the time for case (a), t(a), will depend on the full flow velocity for pipes AD, DE, EF, being largely dependent on the velocity of flow in AD. For case (b), t(b), will depend largely on the velocity of flow in IF, which, being necessarily a larger pipe than AD, will usually have a significantly higher flow velocity.

As an example, if AD is of diameter 150 mm throughout its length, if IF is 225 mm throughout its length, if AD = 1000m and if IF = 900m and all pipes are at a gradient of 1 in 250, then t(a) = 1550 secs. and t(b) = 1260 secs. for typical pipes. For a 1 year storm, design flow Q(b) is then 18% higher than design flow Q(a).

8.12 Cross-drain sets with networks upstream

It is possible to have branches joining into the component drains of a cross-drain set. Such an arrangement is shown in Figure 8.7a. This

- 164 -







DESIGN FLOWS WITH VARIABLE NETWORK

FIGURE 8.6



(a) CROSS-DRAIN SET WITH BRANCHES JOINING



(b) CROSS-DRAIN SETS WITH A COMMON BASE

;

FIGURE 8.7

.

presents no theoretical difficulty, as the situation can be handled as follows:

- Identify all such branches, and identify the manholes (e.g. X, Y) at which they enter the cross-drain set.
- Define range of discrete output states for the branches at these manholes.
- Obtain by MANVAR the set of minimum costs for these discrete output states.

The main cross-drain set I.S.D.P. design may then proceed, incorporating the sets of branch costs.

8.13 Cross-drain sets sharing a common base cross-drain

Frequently two sets of parallel drains enter a common base drain, one from either side as in Figure 8.7b. This raises a theoretical difficulty with the proposed method, as it has so far been assumed that a cross-drain set can be designed in isolation from any other cross-drain set.

Imagine performing the I.S.D.P. process on set A. The final crossdrain position considered is the base cross-drain QP.

A range of states at P is considered, and the minimum costs of arriving at those states is obtained, considering only cross-drain set A, i.e. excluding the effect on the pipe level at Q of the drain entering Q from cross-drain set B.

These costs include the cost of drain QP.

Imagine now the I.S.D.P. process on cross-drain set B. A new set of minimum costs will be obtained for the states at P, again including the cost of the drain QP.

These two sets of costs at P can then be combined to form a single set of optimal upstream costs over the range of states at P. These will

- 167 -

however include drain QP twice. The D.P. will then proceed to the network outfall. The final network trace back will identify one state at P from which the optimal set of drains for A and B will be identified. However, base drain QP may well have two different designs for the two different cross-drain sets. Indeed the pipe levels at Q may well be quite incompatible.

Theoretically, then, it is necessary to consider the two cross-drain sets simultaneously in the I.S.D.P. process. Such a procedure would involve severe computational penalties even if a sound method could be evolved, and so was pursued no further.

In practice, therefore, cross-drain sets sharing a common base crossdrain are designed as if they had separate base cross-drains. It is in fact unlikely that the cost of the base cross-drain will exceed about 1% of the cost of the upstream drains in any normal network. Hence its effect on the positioning of the cross-drains is likely to be minimal. The problem of two differing trace-back solutions for the base cross-drain is overcome by using the procedure described in the following section which describes a practical model.

8.14 A practicable model - CROSSVAR

8.14.1 Introduction

A model based on the I.S.D.P. approach was developed for the variable cross-drain problem. The model, CROSSVAR, breaks the optimisation problem into three parts.

Firstly, cross-drain positions are determined. Next intermediate manhole positions and pipe diameters are found. Finally pipe slopes are determined. The last two stages are equivalent to the MANVAR model.

- 168 -

8.14.2 Structure of CROSSVAR

CROSSVAR is implemented by two programs, SORT and MODEX linked as shown in Fig.8.8. Both are generally run twice.

SORT accepts in card format input data describing the road geometry and network layout, and a set of design parameters. It outputs on magnetic tape a complete set of input data for MODEX.

A flow chart for SORT is given in Fig 8.9 and a program listing is given in Appendix F.

MODEX operates in one of two modes. If there are any cross-drain positions to determine, it will do so and output in card format the geometry corresponding to the resultant pipe network. This is then processed by SORT, and returned to MODEX which, because there are now no cross-drain positions to find, will operate in its second mode. In this mode it will perform the same function as program MOD in MANVAR (see Section 7.13) producing first a set of optimal manhole positions and pipe diameters and then a set of pipe gradients.

CROSSVAR was so structured for two main reasons. Firstly, as explained in Section 8.13, it is not possible to obtain a theoretically correct solution for the common case of two sets of cross-drains sharing a common base. Hence some approximation is necessary, the most satisfactory being to assume two base cross-drains independent of each other for the purpose of establishing cross-drain positions. Having established these, the optimal solution for the resulting network can then be obtained assuming a joint base cross-drain by re-running MODEX with no variable cross-drains.

The second reason is that of program efficiency. The time taken to produce a design with a single program run is an order of magnitude greater than that taken to run the program twice, once to establish cross-drain positions, and then to complete the design, whilst the results obtained are usually identical. The disadvantage is that the solution could be suboptimal in the latter case, if the cross-drain positions are sensitive to the choice of grid spacings for the manhole positions and pipe levels. (See Section 8.19)

- 169 -



IMPLEMENTING THE CROSSVAR MODEL

FIGURE 8.8


FLOW CHART FOR PROGRAM SORT

FIGURE 8.9

8.15 Program MODEX

Flow charts for program MODEX are given in Figures 8.10 and 8.11, and a program listing in Appendix G.

Essentially MODEX is a version of MOD, extended to include a variable cross-drain facility. The program takes each branch of the network sequentially from the upstream ends, and if the branch is not a member of a cross-drain set, performs an ISDP optimisation for a range of downstream states.

When a member of a cross-drain set is identified, control is switched to a subroutine XDSET. This subroutine controls the I.S.D.P. optimisation for the cross-drain set, identifying subnetworks between cross-drains, which are then optimised by calling SUBNET.

MODEX has an important refinement over MOD in that it was realised it was not essential to perform a minimum cover design at the start of the program. This was done in MOD to establish the upper limits on the state variables throughout the network. In MODEX, the minimum cover design is done for a branch at a time, just before that branch design is optimised. This overcomes the problem of the network not being defined initially.

MODEX consists of about 1900 lines of National Computer Centre Standard Fortren and is hence largely machine independent. Data is stored partly in two main arrays, these being dynamically addressed to minimise core storage requirements, and partly on two magnetic tape workfiles for bulk storage of data that is not being currently used by the program.



FLOW CHART FOR PROGRAM MODEX - Main Program

FIGURE 8.10



FLOW CHART FOR PROGRAM MODEX - Subroutine XDSET

FIGURE 8.11

- 174 -

8.16 Optimising Parameters for CROSSVAR

The CROSSVAR programs were written to cope with any reasonable combination of optimising parameters.

These parameters are as follows :

- 1) Cross-drain resolution, i.e. the spacing between possible cross-drain positions.
- 2) Manhole resolution, i.e. the spacing between possible manhole positions.
- 3) Number of pipe diameters considered at each manhole.
- 4) Width of the pipe level zone
- 5) Number of discrete levels within pipe level zone.

Rather than generating cross-drain positions at specific intervals of chainage along the road, it was thought better to specify the manholes that they would connect. Hence the spacing of possible cross-drain positions is in fact a multiple of the spacing of possible manhole positions.

The possible manhole positions are generated along each branch of the cross-drain set and numbered. Possible cross-drain positions are generated from the base cross-drain upstream and defined by the corresponding manhole numbers.

Using this system all possible manhole positions can be generated initially.

Parameter 1 is used only in the first run of SORT and MODEX. Parameters 2, 3, 4 and 5 are specified independently for the first and second runs of the program. Hence a coarse initial run can be used to establish cross-drain positions, followed by a finer process for establishing manhole positions and pipe diameters.

8.17 Program of Testing for CROSSVAR

A program of testing was devised for CROSSVAR, to check its validity and to choose suitable optimising parameters. The program was as follows:

- (1) Check compatibility with MANVAR
- (2) Find typical cross-drain spacing
- (3) Test whether I.S.D.P. can be truncated by considering only a certain number of possible upstream cross-drain positions
- (4) Test the stability of the solution (for cross-drain positions) to variations in the optimising parameters 2, 4 and 5
- (5) Find suitable values of optimising parameters 2 to 5
- (6) Find the effect of cross-drain resolution(parameter 1) on overall optimal cost
- (7) Choose suitable values for cross-drain resolution
- (8) Run using other networks

8.18 Results using the CROSSVAR model

8.18.1 Checking CROSSVAR with previous results

The first runs of the CROSSVAR model were to check that it was fully consistent with the MANVAR model. Hence these test runs were performed using CROSSVAR without its cross-drain optimising capability.

Three test runs were performed on two networks and compared to runs using MANVAR.

The first of these was on the network shown in Fig. 7.16 and described in Section 7.14.5. The results from the two models were identical.

The other two tests were on network 3 (Figure 7.11) using two sets of design parameters. The results differed very slightly from MANVAR. This was traced to a minor error in the MANVAR programs. However the results were substantially identical.

8.18.2 Finding Typical Cross-drain Spacings

The second set of runs were aimed at finding typical cross-drain spacings. Five runs were performed on network 4 (Fig.8.12a) using first of all a coarse spacing of possible cross-drain positions, and then gradually finer spacings.

The results of these runs are illustrated in Figure 8.13. For run 5, the spacing between cross-drains was limited to 250 m to reduce program execution time. The result is therefore somewhat artificial.

In general the runs show that optimal cross-drain spacings were between 150 m and 500 m for this network.

It would seem reasonable therefore to limit the maximum cross-drain spacing in order to truncate the I.S.D.P. process, thereby considerably reducing computation.

- 177 -



(a) NETWORK 4



(b) NETWORK 5

<u>note:</u> all dimensions in m.

FIGURE 8.12



CROSS-DRAIN POSITIONS FOR NETWORK 4 USING MODEX FIGURE 8.13

8.18.3 Stability of Cross-Drain Position to Variation of Parameters.

The third set of runs were to examine the stability of the optimal cross-drains to variations in three of the optimising parameters. All runs used Network 4 (Fig. 8.12)

a) Manhole Resolution

The first parameter investigated was the manhole resolution, which was allowed to vary from 25 m to 150 m (the maximum spacing permitted), whilst the cross-drain resolution was held at 150 m. Having established cross-drain positions, the second optimisation used a new set of optimising parameters which were identical for all runs.

Results for these runs are shown in Table 8.4

It can be seen that **alteration** of the manhole resolution from 25 m to 150 m produces just one change in the cross-drain positions. This change leads to a very slight (0.1%) increase in the cost of the final optimal network. Overall execution time for the program decreases by a factor of between 6 and 7.

In these circumstances it would seem reasonable to consider only a manhole resolution equal to the maximum specified spacing of manholes, assuming this to be not greatly in excess of 150 m.

b) Width of Pipe Level Zone

The second parameter investigated was the width of the pipe level zone in the initial stage of the model.

As a preliminary to this, one run was performed using a zero width of zone for this stage, i.e. a minimum cover design.

This produced a design which at £118 124 was about 1% more expensive than the optimum. The number of cross-drains generated was larger than

Run Number	Manhole Resolution (m)	200	Cross-drain location (m) (Base cross-drain at chainage 2000) 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0							Network Cost (£)	Run Time (secs)				
21 13 23	25 50 75		X X X X	X X X X		X	X X	X X X X		X X X X	X X X X			117143 117143 117267 117267	911 323 212 147
	Zone Depth (m)			•											
(23) 42 43 44	1.0 2.0 3.0 4.0		X X X X	X X X X		X X X X	X	x x x x x	x	X X X X	X X X	x		117267 117267 117267 117267 118058	147 148 149 146
	Number of Levels														
(44) 51 52 53	11 21 31 41		X X X X	X X X X		X X X X	X	X X X X	X	x x x x	X X X	X		118058 117267 117267 117267	146 ? 494 776

X = cross-drain position

SENSITIVITY OF CROSS-DRAIN POSITIONS TO OPTIMISING PARAMETERS

TABLE 8.4

that for any previous run, probably due to larger pipe sizes being generated by the low gradients.

As the manhole resolution was quite fine, (25 m), the run involved a moderate execution time of 232 secs. The idea of using a minimum gradient design to establish the cross-drain pattern was rejected as it was felt that better solutions could be obtained in less execution time by a suitable choice of parameters.

The main series of runs varied the width of zone from $1 \cdot 0$ m to $4 \cdot 0$ m, the top of the zone being at a minimum cover design level. The remaining optimising parameters were held constant, the values of manhole resolution being 150 m, cross-drain resolution being 150 m, number of levels being 11 and number of diameters being 3.

The results are shown in Table 8.4. These indicate that there is no advantage in having a wider zone whilst keeping the number of discrete levels the same. In fact, if the spacing between discrete levels becomes excessive, the solution may well become sub-optimal as in run 44.

c) Number of levels in zone

The last series of runs in this section investigated the effect of varying the number of levels within a zone of fixed width.

Four runs were performed using a standard zone width of 4.0 m, manhole and cross-drain resolution of 150 m and 3 pipe diameters. The number of levels were varied from 11 to 41, giving spacings of 0.4 m to 0.1 m between levels. The results are shown in Table 8.4.

Only the first section of CROSSVAR was implemented for these runs. This, however, was sufficient to show that the cross-drain positions remained stable for spacings up to 0.2 m.

8.18.4 The effect of cross-drain resolution on the optimality of the solution

The fourth set of runs were designed to find how the cost of the optimised network varied with the choice of cross-drain resolution.

Network 4 was again used, for cross-drain resolutions of from 100 m to 500 m. Manhole resolution and maximum manhole spacing were taken as 100 m. A zone width of 1 m with 11 discrete levels was adopted together with 3 possible pipe diameters.

The results are plotted on the graph of Figure 8.14.

As one would expect, the general trend is for the solution to decrease in cost as the spacing decreases. There is not however, a great increase in execution time and it would seem sensible to adopt a cross-drain resolution equal to maximum manhole spacing.

8.18.5 Runs using other Networks

Two runs were performed using other, more complicated, networks. These were primarily to test that the program was capable of handling such networks and to see whether the results it gave were sensible.

The networks used were Network 5, Figure 8.12b, and Network 3, Figure 7.11. The resulting designs are shown in Figure 8.15(a) and (b).

8.19 Choice of values for the Optimising Parameters

The choice of values for the optimising parameters is not obvious as it depends on a trade-off between computer resources and the degree of optimality of the design.

However, it seems from the examples used in this research that the first part of the program can be run with the following values of parameters and

- 183 -



SENSITIVITY OF NETWORK COST TO CROSS-DRAIN RESOLUTION

FIGURE 8.14



۰.

.

- 185 -

still, in most cases, achieve the optimal set of cross-drain positions for a given cross-drain resolution.

Manhole resolution = maximum manhole spacing Pipe level zone = 1.0 m Discrete pipe levels = 6 Discrete pipe sizes = 3

The choice of cross-drain resolution is less obvious. It would not seem sensible to decrease the spacing below the maximum manhole spacing. A value of cross-drain resolution equal to the maximum manhole spacing is thus suggested.

For the second part of the program, it is necessary to alter the manhole resolution to 25 m (or a convenient factor of the cross-drain resolution close to this value), and alter the number of pipe levels considered to 11. This will give a set of manhole positions and pipe diameters close to the optimal set, if not actually optimal.

8.20 Conclusions on the use of CROSSVAR

This chapter shows that an optimal drainage design model capable of dealing simultaneously with variable cross-drain positions, variable intermediate manhole positions, variable pipe diameters and gradients can be implemented using the I.S.D.P. approach.

Using a sensible choice of optimising parameters the execution times on a large computer are reasonable, the costs involved being only a small proportion of the likely saving on construction costs.

On the examples tested, the savings made by using the CROSSVAR model instead of the MANVAR model with fixed cross-drain positions are not large. As execution times for CROSSVAR are up to an order of magnitude larger than for MANVAR, it was decided not to proceed with a commercial version of CROSSVAR at this stage. It was felt that the extra program length, execution cost and documentation would have discouraged engineers from using it.

- 186 -

CHAPTER 9

A FURTHER VARIABLE PLAN OPTIMI SATION PROBLEM

9•1	Introduction
9.2	Connecting several sources of flow to a single main drain
9•3	A DP approach - MULTICON
9•4	Comments on the method

9.1 Introduction

In this chapter a further problem involving variable plan optimisation is examined and a method of solution is proposed. The proposed method has not been tested.

9.2 Connecting several sources of flow to a single main drain

A typical variable plan drainage problem might be posed as follows:

Given sources of known drainage flow at manholes A, B, C, D, E, F in Fig. 9.1a connect them in the least cost way to the outfall manhole 0, whilst satisfying all the usual drainage design constraints (see Section 2.3). Such problems may indeed be applicable to other forms of network, for instance water supply, roads and gas.

Wilson (ref. 25) attempted to form a model based on Geometric Programming, with the simplifying assumption that there was a straight main drain into which each of the others connected (eg AO in Fig.9.1b) However, he met with severe problems due to optimal solutions involving manholes coinciding and a large number of equality constraints.

9.3 <u>A DP approach - MULTICON</u>

It is necessary to make two simplifying assumptions. The first is that each manhole is connected to a single main drain. This excludes the possibility of a branch drain linking several manholes before connecting to a main drain. Secondly it is necessary to predefine the order in which the manholes are connected along the length of the main drain. In the example it is assumed they are connected in the order ABCDEF. Note that it is not necessary for the main drain to be straight.

For simplicity it is assumed that pipe diameters may increase down the network, although a restriction on pipe diameters can readily be incorporated.

Working from the upstream end of the network consider first manhole A. This can be considered as the start of the main drain. Consider now manhole B. A drain will run from manhole B to join the main drain at an unknown position, say B', or possibly the main drain could run through B. Consider a grid of points representing possible positions for this junction B'. This grid may include point B. (see Fig. 9.1c)



FIGURE 9.1

For each possible grid position B' calculate the minimum total cost of connections AB', BB', for each of a range of discrete pipe depths at B'. This could be done using the MANFIX model. Moving on to manhole C, consider a grid of possible junction points C' and obtain for each grid position and discrete depth the minimum total upstream cost. This consists of the cost of CC' plus the cost of B'C' plus the previously obtained cost of the network upstream of B', and may be obtained by using MANFIX on the subnetwork consisting of CC' and B'C' for every feasible position of junction B'.

In this way the design proceeds downstream and the minimum cost of the network can be found for a range of depths at the outfall manhole. Hence the overall minimum cost can be selected and the solution traced back up the network. A typical solution is shown in Fig. 9.1d. A flow chart for this procedure is shown in Fig. 9.2.

9.4 Comments on the method

The main disadvantage with the model is that the order of connection must be predetermined. For example, a solution in which C was connected into the main drain before B would not be considered. However it can allow B and C to be connected at the same point thus overcoming a problem that Wilson found (see Section 9.2) and indeed can allow the main drain to pass through the sources of flow.

If lengths between junctions, or lengths from source manholes to the main drain justify the inclusion of intermediate manholes, then MANVAR can be used to establish the minimum subnetwork costs.

A computer program has not yet been written to implement the method so no practical problems such as storage requirements or execution time can be discussed.

Applications in sewer system design could be considerable wherever some freedom of choice exists in the positioning of manholes. The discrete nature of the possible solutions may itself be valuable in accommodating problems in which manholes are limited to a small number of possible discrete positions. For example, this can arise if manholes are to be at street corners or in road verges.



FIGURE 9.2

CHAPTER 10

CONCLUSIONS AND AREAS FOR FURTHER STUDY

- 10.1 MANFIX
- 10.2 MANVAR
- 10.3 CROSSVAR
- 10.4 MULTICON
- 10.5 General Conclusions
- 10.6 Areas for Further Study

Chapter 10. Conclusions and Areas for Further Study

10.1 MANFIX

A study of previous work on the optimisation of fixed plan drainage networks provided a good basis for the development of a new and efficient Dynamic Programming model called MANFIX . This correctly handles the constraint that pipe diameters should not decrease in a downstream direction by the use of two state variables in the D.P. process. Although the number of elemental designs at each stage is thereby greatly enlarged, the actual computation time is not unduly increased due to each elemental design being greatly simplified. The final solution produced is fully consistent with the method chosen for determining design flows. This is accomplished by using the D.P. to establish the set of optimal pipe diameters, and then using these diameters in a final fully consistent design process to establish pipe gradients.

Results show that the set of optimal pipe diameters can be reliably found by the use of a very coarse D.P. grid, using an initial "minimum cover" design to establish a set of approximate flows and bounds on the D.P. process. Hence the process is comparable to DDDP in computer time and storage requirements with the possible advantage of simplicity.

A separate computer program was not developed for MANFIX, but the model was an essential foundation for MANVAR, the variable manhole position model.

10.2. MANVAR

The problem of optimising the number and position of manholes along the line of a non-branching drainage run required the introduction of a new type of D.P.

This was termed Indeterminate Stage Dynamic Programming (ISDP) as the number of stages in the final solution is not predetermined. The

- 193 -

concept of a set of discrete feasible positions for intermediate manholes provides the key to the problem, enabling ISDP to be used in an elegant procedure. The choice of a suitable set of optimising parameters leads to an efficient and fully practicable computer program. Whilst savings in construction cost over non-optimised schemes are not as great for road drainage networks as for other forms of drainage, it is clear that for a very modest computer running cost, large sums of money can still be saved. The model is flexible enough to allow a great deal of freedom in the choice of optimising parameters so that they can be altered at will to allow for varying computer costs.

Schemes designed using MANVAR produced solutions with levels close to the minimum possible cover but with generally smaller pipe sizes. This was accomplished by minor changes in pipe gradients and better positioning of manholes. The resultant designs are usually better from an engineering viewpoint in that less of the network is at minimum gradient and hence is less likely to suffer from siltation and blockage.

A fully commercial version of MANVAR for use in the design of new road drainage schemes (DAPHOP) was produced and is undergoing trials by the Department of Transport.

10.3. CROSSVAR

The ISDP process was used for the more complicated problem of finding the number and position of drains crossing the road to link parallel carriageway drains. A set of feasible cross-drain positions is first identified and a coarse ISDP design performed to establish the set of optimal cross-drains.

The resultant network is then optimised using the MANVAR procedure. In this way a near optimal design of cross-drains, manhole positions, pipe diameters and gradients can be achieved for a typical road drainage problem.

- 194 -

CROSSVAR is not fully optimal for design flows that are dependent on the network, but gives a solution which is probably very close to the optimum.

A practical computer program was written and tested successfully, but a fully commercial version was not produced.

Experience in the use of CROSSVAR showed that the optimal number of cross-drains in the typical road drainage program was not at all obvious, and that the overall network cost was not very sensitive to cross-drain positions.

10.4. MULTICON

MULTICON demonstrates the adaptability of the general D.P. approach to drainage network problems. A variable layout problem, completely different from those tackled by MANVAR and CROSSVAR, is solved by D.P. This is the case of a number of manholes connected by drains to a single main drain. The method involves defining a set of discrete possible positions for each junction manhole, and using manhole position, in effect, as a state variable. This idea developed from the use of ISDP in the MANVAR and CROSSVAR models, but as yet MULTICON has not been implemented.

10.5. General Conclusions

Due to the complexity of optimal layout problems for storm drainage systems no single "black box" algorithm is possible at present or likely to be developed in the foreseeable future.

However the general D.P. approach has been shown to be highly effective if correctly tailored to the individual type of network considered.

Road drainage is one such type of network and alternative network optimisation models have been developed and tested using ISDP. These

- 195 -

are both fully practicable and show worthwhile savings over non-optimised solutions, (see also Ref. 62).

To demonstrate the suitability of the general DP approach one further type of network was examined and a DP model formulated for its optimal solution.

10.6. Areas for Further Study

The possibilities for further work in the general area of storm water drainage optimisation are numerous. These possibilities include the development of optimisation models for other typical types of network. It is likely that any successful work in this area will involve some form of D.P. due to the serial nature of drainage systems.

The MULTICON model requires the writing and testing of a computer program to test its validity and efficiency. Problems could arise when using network dependent design flows and some approximate procedure may be required as in the MANFIX, MANVAR and CROSSVAR models. It may indeed be of greater use for foul sewerage networks, or as the basis of an optimisation model for other distribution systems (e.g. water supply aqueducts).

The concept of ISDP needs further exploration to see whether other engineering optimisation applications exist.

MANFIX, MANVAR and CROSSVAR have been implemented using the Rational design method for calculating design flows. The practical difficulties of using, say, the TRRL hydrograph method should be explored and the resulting designs compared with Rational designs.

The models developed have assumed no possibility of detaining flood waters by ponding in special tanks, ponds or oversized pipes. As this is likely to be the object of considerable attention in future years,

- 196 -

the incorporation of such storage items within an optimal design model should be investigated.

The existing models should be further tested to investigate their sensitivity to differing forms of cost function, as this is the factor of greatest uncertainty in any optimisation process. If the "optimal" design is very sensitive to the form of cost function then further work is required on establishing the most accurate cost functions possible. However if the designs are relatively insensitive to the form of the cost function then the optimisation models are valid without further research on costs.

MANFIX could be used as a tool to investigate the effect on the cost of a network of alterations to the design parameters. Such parameters are at present selected on the basis of judgement and experience with little knowledge of the cost penalties for being over-conservative.

A version of MANFIX for use on a mini-computer should be developed. This would enable the design of small housing estate and industrial drainage networks to be performed optimally in the smallest design office.

REFERENCES

(1) P.J.Colyer. "Storm drainage design methods- a literature review". Hydraulics Research Station, Wallingford, 1976. R.W.Pethick (2)Local Government "Economics of sewerage design", Report no. C218, Operational 1975. Research Unit (3) British Standards "British standard code of practice for sewerage", Institution CP 2005, 1968, London. (4) British Standards "British standards code of practice for building Institution drainage", CP 301, 1971, London. (5) Department of "A guide for engineers to the design of storm sewer systems", Road Note 35, 2nd. edition, HMSO, 1976. the Environment (6) L.H.Watkins, "Developments in urban hydrology in Great Britain", Road Research Laboratory, LN/885,1965. C.P.Young "Clay drain and sewer pipes including surface (7) British Standards water pipes and fittings", BS 65 and 540, Part 1, Institution 1971. London. (8) British Standards "As bestos-cement pipes, joints and fittings for Institution sewerage and drainage", BS 3656, 1973, London. (9) British Standards "Concrete unreinforced tubes and fittings with Ogee Institution joints for surface water drainage", BS 4101, 1967, London. (10) British Standards "Prestressed concrete pipes for drainage and Institution sewerage", BS 5178, 1975, London. (11) British Standards "Pitch impregnated fibre pipes and fittings for Institution below and above ground drainage", BS 2760, 1973, London. (12) British Standards "Unplasticized PVC underground drain pipe and Institution fittings", BS 4660, 1973. London. (13) British Standards "Unplasticized PVC pipe and fittings for gravity Institution sewers", BS 5481, 1977, London. (14) A.D.Haith " Vertical alignment of sewer and drainage systems by Dynamic Programming", PhD. thesis, M.I.T., Boston, 1976. (15) R.E.Bellman "Dynamic Programming", Princeton University Press, Princeton, New Jersey, 1957.

(16) J.J.Dajani,	"Optimal desitn of urban wastewater collection
	R.S.Gemmell,	networks", Proc. ASCE, SA6, 1972, pp853-67.
	E.K.Morlok	
(17	') J.S.Dajani,	"Capital cost minimisation of drainage networks",
	Y.Hasit	Proc. ASCE, EE2, 1974, pp 325-337.
(18) P.Yletyinen	"Optimal design of water supply and sewerage
		networks", Helsinki University of Technology
		Research Paper 48, 1973, Otaniemi, Finland.
(19) L.Von Dobschutz	"Mathematical models for the optimisation of
		pipe networks", Die Wasserwirtschaft, Vol.65,
		no.6, 1975, pp 160-164.
(20)) E.H.Iman,	"Damage and cost simulation in pumped storm sewers",
	J.A.McCorquodale,	Canadian Journal of Civil Engineering, Vol.6,
	J.K.Bewtra	1979, pp 129-138
(21)) P.F.Lemieux,	"Design of a stormwater sewer by nonlinear pr
	Y.Zech,	programming", Canadian Journal of Civil Engineering,
	R.Delarue	Vol 3, 1976, pp 83-89.
(22)	J.B.Rosen	"The gradient projection method for non-linear
		programming, part 1, linear constraints", Journal
		Soc. Ind.Appl. Math., Vol 8, 1960, pp 181-217.
(23)	R.K.Price	"Design of storm sewers for minimum construction
		cost", Proc. Int. Conf. on Urban Storm Drainage,
		University of Southampton, 1978. Pentech Press,
		pp 636-47
(24)	R.J.Duffin,	"Geometric programming- theory and application",
	£.L.Peterson,	Wiley, New York, 1967.
	C.Zener	
(25)	A.J.Wilson	"Optimization in computer aided building design",
1		PhD. thesis, University of Liverpool, 1973.
(26)	D.D.Meredith	"Dynamic programming with case study on planning
		and design of urban water facilities", Treatise
		on urban water systems, Colorado State University,
•		Fort Collins, 1971.
(27)	L.B.Merrit	"Computer based optimal design of sewer systems",
	R.H.Bogan	Proc. ASCE, EE1, 1973, pp35-53
(28)	S.Walsh	"Least cost method for sewer design", Proc ASC2,
	L.C.Brown	起3, 1973, pp333-45

1

(29) 3.Froise	"Least cost design of urban drainage networks",
	S.J.Burges	Proc. ASCE, WR1,1978, pp75-92
(30)) T.Liang	"Design of conduit system by dynamic programming",
		Proc. ASCE, HY3, 1971, pp383-393.
(31)	G.A.Walters	"Non-optimal dynamic programming algorithms in the
	A.B.Templeman	design of minimum cost drainage systems",
		Engineering Optimisation, Vol 4, 1 79
~ (32)	M.Heidari	"Discrete differential dynamic programming approach
	V.T.Chow	to water resources systems optimisation", Water
	P.V.Kokotovich	Resources Research, Vol 7, 1971, pp273-82.
	D.D.Meredith	
(33)	L.W.Mays	"Optimal layout and design of storm sewer systems",
		PhD. thesis, University of Illinois, 1976.
(34)	L.W.Mays	"Optimal cost design of branched sewer systems",
	B.C.Yen	Water Resources Research, Vol 11, 1975, pp 37-47.
(35)	B.C.Yen	"Designing storm sewers using the rational method",
	W.H.Tang	Water and Sewage Works, Oct. 1974, pp92-95, and
	L.W.Mays	Nov. 1974, pp84-85.
(36)	W.H.Tang	"Optimal risk based design of storm sewer networks",
	L.W.Mays	Proc. ASCE, EE3, 1975, pp 381-98.
	B.C.Yen	
(37)	L.W.Mays	"Optimal design of multilevel branching sewer
	H.G.Wenzel	systems", Water Resources Research, Vol 12, 1976,
		pp 913-917.
(38)	Y.Argaman	"Design of optimal sewerage systems", Proc. ASCE,
	U.Shamir	EE5, 1973, pp 703-716.
	E.Spivak	
(39)	P.Nopmongcol	"Multilevel incremental dynamic programming",
	A.J.Askew	Water Resources Research, Vol 12, 1976, pp 1291-7
(40)	V.T.Chow	"Computer time and memory requirements for DP and
	D.R.Maidment	DDDP in water resources systems analysis", Water
	G.W.Tauxe	Resources Research, Vol11, 1975, pp 621-28.
(41)	J.C.Liebman	"A heuristic aid for the design of sewer networks",
		Proc. ASCE, SA4, 1967, pp 81-90.
(42)	J.F.Barlow	"Cost optimisation of pipe sewerage systems",
		Proc. Inst. Civil Engineers, Pt 2, Vol 53, 1972,
		pp 57-64.

_

- 200 -

(43) I.H.Lowsley	"An implicit enumeration algorithm for optimal
	sewer layout", PhD. thesis, John Hopkins University,
	Baltimore, 1973.
(44) L.W.Mays	"Model for layout and design of sewer systems",
H.G.Wenzel	Proc. ASCE, Vol 102, WR2, 1976, pp 385-405.
J.S.Liebman	
(45) D.M.Farrar	"A procedure for calculating the cost of laying
	rigid sewer pipes", SR 333, Transport and Road
	Research Laboratory, Crowthorne, Berkshire, 1977.
(46) Davis, Belfield	"Architects and builders price book", 102nd edition,
and Everest	Spon Ltd., London, 1977.
(47) A.J.M.Harrison	"An analysis of sewer costs", Internal report,
	Hydraulics Research Station, Wallingford, 1974.
(48) Department of	"Notes for guidance, rural motorways, drainage",
the Environment	1973.
(49) R.Hooke	"Direct search solution of numerical and statistical
T.A.Jeeves	problems", Journal A.C.M., Vol 8, 1961, pp212-29.
(50) H.H.Rosenbrock	"An automatic method for finding the greatest
	or least value of a function", The Computer Journal,
	Vol 3, 1960, pp175-84.
(51) W.H.Swann	"Report on the development of a new direct
	search method of optimisation", ICI Ltd.,
	Note 64/3, 1964.
(52) W.C.Davidson	"Variable metric method for minimisation",
	Argonne Nat. Lab., 1959, ANL-5990 Rev.
(53) R.Fletcher	"A rapidly convergent descent method for
M.J.D.Powell	minimisation", Computer Journal, Vol 6, 1963, p 163.
(54) R.Fletcher	"Function minimisation by conjugate gradients",
C.M.Reeves	Computer Journal, Vol 7,1964, p149.
(55) P.E.Gill	"Numerical methods for constrained optimisation",
W.Murray (eds)	Academic Press, 1974.
(56) Department of	"Drainage analysis program for highway networks,
the Environment	DAPHNE", Highway Engineering Computer Laboratory,
	D.o.E., London.
(57) National	"Standard Fortran programming manual", National
Computing Centre	Computing Centre, 1972.
(58) Department of	"HECB programming instruction manual, vol 3,
the invironment	HECB subset of NCC Fortran standards", Highway
	Engineering Computer Branch, D.o.E., London.

- (59) Department of "British integrated program system for highway the Environment design", D.o.E., London.
- (60) L.C.W.Dixon "Nonlinear optimisation", Anglish Universities Press, London, 1972.
- (61) D.J.Wilde "Foundations of optimisation", Prentice-Hall,
 C.S.Beightler 1967.
- (62) A.B.Templeman "Optimal design of stormwater drainage networks
 G.A.Walters for roads", Proc. Inst. Civil Engineers, Pt 2, Vol 67; 1979.

APP-NDIX A

Cost Calculations Based On Spon's Architects and Builders Price Book
A1 General Cost of a pipe run between two manholes = length × cost per m run +
cost of upstream manhole.
Costs are adjusted where necessary to March 1977 prices.
A2 Cost per m run.
This consists of the following items:
(1) pipe supply
(2) excavation of trench by machine
(3) layering and compaction of backfill
(4) removal of surplus backfill
(5) support for trench excavation
(6) smoothing the trench bottom by hand
(7) supply and placing of bedding and haunching material
(8) laying and jointing of pipes
The costs of these items are taken as follows:
(1) <u>Pipe supply</u>
Manufacturers' quotes x 1.05 for wastage.

Diameter (mm)	150	225	300	375	450	525	600	675
Cost (£ per m)	1.86	4.03	5.80	8.35	12.55	15.22	19.83	25.21
Diameter (mm)	750	825	900					
Cost (£ per m)	30.30	34.90	42.60					

(2),(3),(4) Excavation, Compaction etc.

Machine and operator for 0.11 hours + 1.26 hours labour attendance on machine per cubic metre excavated = $\pounds 2.69 / m^3$

(5) Trench support

Trench depth (m)	Cost per m ² of trench wall
y <1.0	zero
1.0 ≤ y < 1.5	0.22 hours labour + 0.00165 m^3 timber = £0.46
1.5 ≤ y < 3.0	0.32 hours labour + $0.00335m^3$ timber = £0.73
3.0 ≤ y < 4.5	0.43 hours labour + 0.00335m ³ timber = \pounds 0.91

(6) Smoothing trench bottom

0.39 hours labour per
$$m^2 = \pm 0.56 / m^2$$

(7) Bedding and haunching

Cost of supply of bedding material + 2.6 hours labour per m^3 = £5.94 + £3.74 = £9.68 / m^3

(8) Laying of pipes

			and the second se					
Diameter (mm)	150	225	300	375	450	525	600	675
Labour (hrs/m)	0.25	0.32	0.40	0.46	0.53	0.60	0.67	0.75
Cost (£/m)	0.36	0.46	0.58	0.66	0.77	0.86	0.%	1.08
Diameter (mm)	750	825	900					
Labour (hrs/m)	0.82	0.89	0.95					
Cost (£/m)	1.18	1.28	1.37					

A3 Cost of a manhole

This consists of the following items:

(1) Excavation by machine

(2) Support of excavation walls

(3) Smoothing the bottom of the excavation by hand

(4) Placing in situ concrete base

(5) Supply and placing of precast concrete manhole rings

- (6) Placing concrete benching
- (7) Backfilling around manhole
- (8) Removal of surplus material
- (9) Supply and placing of concrete cover slab
- (10) Supply and placing of brickwork, access cover and frame
- (11.) Supply and fitting of step irons
- (12) Supply and placing of tapered ring sections (if required)

The costs of these items are taken as follows:

(1) Excavation

depth (m)	0 < y ≤ 1.5	1.5 < y 4 3.0	3.0 < y ≤ 4.5
$cost (\ell/m^3)$	5.38	7.34	9.28

(2) Wall support

depth	(m)	0 < y ≤ 1.5	1.5 < y š 3.0	3.0 < y ≤ 4.5
cost	(ϵ/m^3)	0.46	0.73	0.91

- (3) <u>Smoothing bottom of excavation</u> $\pounds 0.10/m^2$
- (4) <u>Placing concrete insitu base</u> £3.62/m²
- (5) Manhole rings

Manhole diameter (mm)	900	1050	1200	1500
Cost (£/m)	35.43	44.41	56.47	90.93

- (6) <u>Benching</u> £45.2/m³
- (7) <u>Backfilling</u> £1.64
- (8) <u>Removal of surplus</u> £1.87/m³
- (9) <u>Concrete cover slabs</u> Manhole diameter 900 1050 1200 1500 Cost (£) 23.11 29.26 39.47 63.24
- (10) <u>Access cover, frame and brickwork</u> £30.51 (lump sum)
- (11) <u>Step irons</u> £3.30 each
- (12) <u>Tapered ring sections</u>

Special s	ections	tapering	to 685	mm diamet	er
Diameter	900	1050	1200	1500]
Cost (£)	24.59	31.23	39.51	85.29	1

A4 Cost Functions

Using quantities taken from typical detail drawings (ref. 48), the cost functions quoted in section 4.3 have been developed. APPENDIA B

Cost Calculations Based on Farrar (ref. 45)

B1 General

Cost of a pipe run between two manholes = length x cost per m. run + cost of upstream manhole.

Four basic cost coefficients are defined, C1, C2, C3, C4 where cost of pipe supply = C1(0.025+D²) (D in metres) cost of a wheeled excavator = C2 £/hr cost of labour (general operative) =C3 £/hr cost of granular bedding material = C4 £/m³ Default values of C1, C2, C3, C4 are 40, 3.5, 1.6, and 3.0 respectively. All other costs are expressed as factors of these 4 basic rates.

An excavation factor F1 is defined for excavation in hard or difficult ground conditions. For normal conditions F1 = 1.0.

B2 Cost per m run.

Four types of drain are considered as show in the sketch below:



The operations involved are:

- (1) Excavation
- (2) Trench support
- (3) Pipe supply
- (4) Distribute and lay pipes
- (5) Place bedding material
- (6) Place backfill or free draining mater al
- (7) Compact backfill or free draining material
- (8) Supply granular bedding or free draining material
- (9) Remove surplus soil
The costs of these operations are as follows (all in ℓ/m run of drain) : (1) Excavation $Cost = b \times y \times K_1 \times F1$ where b= trench width, y = trench depth, and M = 0.09702 + 0.13003(2) Trench support Cost = 0y ≤ 1.5m Cost = 0.893 K 1.5 < y ≤ 3.0 Cost = 5.03 €. 3.0 < y (3) Pipe supply Drains type A and D Cost = $C1(0.025 + D^2)$ where D = pipe diameter (m) Drains type B and C Cost = $C1(0.05 + D^2 + D_2^2)$ where D_2 = upper pipe diameter. (4) Distribute and lay pipes Drains type A and D D < 0.3 Cost = 0.0455C2 + 0.201C3 $D \ge 0.3$ Cost = 0.21702 + 0.38903 Drains type B and C (assuming $D_0 \leq 0.3$) D < 0.3 Cost = 0.091C2 + 0.402C3 D ≥ 0.3 Cost = 0.263C2 + 0.590C3(5) Place bedding material Drains type A and D Cost = 0.0502 + 0.32103 Drains type B and C Cost = 0.102 + 0.64203(6) <u>Place backfill or free draining material</u> $Cost = K_2 \times V_2$ where $V_2 = volume$ of backfill or free draining material excluding bedding per m run, and $\kappa_2 = 0.078102 + 0.10403$ (7) Compact backfill or free draining material Cost $\aleph_3 \times \nu_2$ where $\aleph_3 = 0.05702 + 0.22203$ (8) Supply granular bedding or free draining material Cost = C4 x v_3 where v_3 = total volume of granular or free draining material per m run. (9) <u>Remove surplus soil from site</u> Cost = $\mathbf{x}_{1} \times \mathbf{v}_{4} \times 0.912$ where \mathbf{v}_{4} = volume of spoil

```
B3 Cost of a manhole
```

```
This consists of :
```

- (1) Excavation
- (2) Supports for excavation
- (3) Place concrete base
- (4) Place rings
- (5) Benching
- (6) Place concrete road slab
- (7) Backfill
- (8) Remove spoil
- (9) Supply concrete
- (10) Supply rings
- (11) Supply slab
- (12) Supply fittings
- (13) Place brickwork and fittings
- The costs of these operations are as follows (all in \pounds)
- (1) Excavation
 - Cost = K F1 x Volex where Volex = volume of excavation
- (2) Supports
 - $y \leq 1.5$ Cost = 0 $1.5 < y \leq 3.0$ Cost = length x 1.79 K, where length = length3.0 < yCost = length x 10.1 K of a side of square hole
- (3) <u>Place concrete base</u> Cost = 0.10502 + 2.2203
- (4) <u>Place manhole ring and joint</u> Cost = 0.643C2 + C3 (per ring)
- (5) <u>Place benching</u> Cost = 0.105C2 + 6.67C3
- (6) <u>Place concrete road slab</u> Cost = 0.157C2 + 0.667C3
- (7) Place backfill and compact Cost = (0.135C2 + 0.326C3) x Volback where Volback = volume of backfill.
- (8) <u>Remove surplus fill</u> Cost = 0.912 x x Volspoil where Volspoil = volume of spoil
 (9) <u>Supply concrete for benching and base slab</u>
 - Cost = 3.6704 x Volconc where Volconc = volume of concrete

- (10) Supply precast concrete manhole rings $Cost = 0.391 \times C1 \times (diam)^2$ per m where diam = manhole diameter (m)
- (11) Supply road slab $Cost = C1(1.04 \times diam - 0.68)$
- (12) Supply fittings
 ie. frame, cover, step irons, bricks
 Cost = C1
- (13) <u>Place brickwork and fittings</u> Cost = 0.10502 + 5.3303

B4 Costing Routine

Using the above unit rates the cost per m run and manhole costs are calculated by the subroutine COSTIT, which then gives the total cost of the pipe run.

COSTIT identifies the pipe type, calculates volumes for the mean pipe depth along a run and hence calculates the costs. APPENDIX C

PROGRAM DPO

	0007	CLARENT ECITION: 26/5/7H ALTERATIONS TO GROUND CALCS, COSTS, ABRAY SIZES
	8030	COMPON/NP/D(20), SMIN, SMAX, DMIN, NPAX, SPMIN, SPMAK, PFND, JEND, T, ND, WK
	0009	COMMUN/UNERE/IN(50,6),LN1,LN2,LN3,LP1,LP2,LP3,LP3
	0130	DIMENSION NIP(8000), PIP(8000), NIT(4000), PIT(4000)
	0011	LOGICAL OK
	512	CSPECISY MAXIMUM ARRAY SIZES
	0613	TMANAAUDO
	0014	
	0014	
	0013	
	0010	
	0017	CREAD DESIGN PARAPPIERS
	0018	CALL DATA1
	0619	C+READ SYSTEM GEOPETRY AND SYDRE IN PERMANENT ANKAYS
	0520	CALL DATAZ(%\$P,P\$P,%\$T,P\$T,1MAX,JMAX,KMAX,LMAX)
	0021	LOBNIP(2)
	0622	NOR5 18(1)
•	0023	CHANNEL THITTAL FLOW VALUES TO ZERO
	0624	
	0,64	
	0523	A THE TOLE AV ASSUMING & S W/S VEL IN PIPES FOR TIME TO GONG.
	0020	
	0647	
	0048	00 5.1 1 1 1
	6656	5 017(1)=1,5
•	0630	6 OK=, TRUF,
	0031	w COSTS=0
-	0035	5JUL 41-41
	0033	CALL FLOWS(NIP,PIP,KWAX,LWAX,PIY,JWAX,T,UK)
	0634	1F(OK) 60 TC 30
•	0635	CPROCEED WITH DESIGN
	0034	60 20 I=1.LC
	0413	CALL FOMM (MTP, PIP, WIT, PIT, KWAX, LMAX, IMAX, IMAX, INCOSTS)
	0037	ALL MARIN SUTA, ATA, MIT, BIT, MAX, JMAX, KMAX, LMAX, 13
	0038	CALL NEW CHIPPEPPERPERPENDENCE MAX, MAX, MAX, TAIJ)
•	0630	GALL THALL CRITICATION POLICY AND
	0040	20 CONTINUE
	0641	CALL TRACE (WIT, DIT, NIP, PID, IMAX, JMAX, CMAX, CMAX)
۰.	- 0642	GU TU 6
	0643	30 CALL PRINT(NIP,PIP,KPAX,LMAX)
	0744	STOP
	0645	END
1		
1		MANA JANATU 436. LANG ANARAS
•	R4C (Ab. 964	MENTY PENNIN JEON HAVE DIAMA
. 1		
1		
•	0646	SUBROUTINE DATA1 Subroutine data1
:	0C46 0C47	SUBROUTINE DATA1 Cummon/dp/d(20),Smin,Smak,Dmin,Dmax,Spnin,Spmak,meur,JEnd,T,nd,Rk
	0046 0047 0048	SUBROUTINE DATA1 COMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAK,MEY7,JEND,7,4D,%K Common/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAK,MEY7,JEND,7,4D,%K CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES
:	0C46 0C47 0C48 0C49	SUBROUTINE DATA1 COMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPMIN,SPMAK,MEY1,JEND,T,ND,RK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND
•	0C46 0C47 0C48 0C49 0C30	SUBROUTINE DATA1 COMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAK,MEY7,JEND,T,4D,9K Common/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAK,MEY7,JEND,T,4D,9K CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,5002) MEND,JEND WAITE (6,2003) MEND,JEND WAITE (6,2003) MEND,JEND
:	0C46 0C47 0C48 0C49 0C50 0C51	SUBROUTINE DATA1 COMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAX,MEY7,JEND,T,VD,VK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (3,1002) MEND,JEND WAITE (6,2005) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETERS
	0C46 0C47 0C48 0C49 0C50 0C51 0C52	SUBROUTINE DATA1 COMMON/OP/D(20);SMIN,SMAX;DMIN,DMAX;SPNIN,SPMAX;MEN1,JEND;T,ND;RK CRéad in Numrer of Vertical Zones and Pipe Choices Read (5,1002) mend;Jend Write (6,2005) mend;Jend CRéad in Numrer of Pipe Sizes Available and Their Dinmetfres Read (5,1000) NP;(D(13,1=1,NP)
•	0C46 0C47 0C48 0C49 0C50 0C51 0C52 0C53	SUBROUTINE DATA1 CUMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPMIN,SPMAK,MENJ,JEND,T,ND,RK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND WHITE (6,2005) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETFRS READ (5,1000) WF,(D(1),1=1,NP) WRITE (6,2000) (D(1),1=1,NP)
1	0C46 0C47 0C49 0C30 0C31 0C32 0C53	SUBROUTINE DATA1 CUMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAK,MEY7,JEND,T,ND,RK CUMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAK,MEY7,JEND,T,ND,RK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND GREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETERS READ (5,1000) NB,(D(1),I=1,NP) WRITE (6,2000) (D(1),I=1,NP) WRITE (6,2000) (D(1),I=1,NP) CREAD IN PIPE ROLONNESS IN MM.
	0C44 0C47 0C48 0C50 0C51 0C52 0C53 0C55	SUBROUTINE DATA1 COMMON/OP/D(20);SMIN,SMAX;DMIN,DMAX;SPNIN,SPMAX,MEUY;JEND;T;ND;RK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND;JEND UNITE (6,2005) MEND;JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIN DINMETFRS READ (5,1000) NP;(D(1);101;NP) URITE (6,2000) (D(1);101;NP) CREAD IN PIPE ROLEMNESS IN MM. READ (5,1001) RK
:	0C46 0C47 0C40 0C50 0C51 0C52 0C53 0C54 0C55	SUBROUTINE DATA1 CGMMON/DP/D(20).SMIN,SM4K,DMIN,DM4X,SPNIN,SPMAK,MEY1,JEND,7,4D,9K CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND WAITE (6,2005) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETFRS READ (5,1000) NP,(D(1),1=1,NP) WRITE (6,2000) (D(1),1=1,NP) CREAD IN PIPE ROLENNESS IN MM. READ (5,1001) RK WRITE(6,2006) RK
:	0C46 0C47 0C49 0C50 0C51 0C53 0C53 0C54 0C55 0C54	SUBROUTINE DATA1 CUMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPMIN,SPMAK,MENJ,JEND,T,ND,RK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND WHITE (4,2005) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETFRS READ (5,1003) (D(I),I=1,NP) WRITE (6,2000) (D(I),I=1,NP) CREAD IN PIPE POLEMNESS IN MM. READ (5,1001) RK WRITE(6,2006) RK READ (5,1001) RK
	0C44 0C47 0C49 0C30 0C31 0C32 0C33 0C53 0C53 0C53	SUBROUTINE DATA1 COMMON/DP/D(20);SMIN,SMAX;DMIN,DMAX;SPNIN,SPMAX,MEUY;JEND,T,ND,RK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND UNITE (6,2003) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIN DINMETERS READ (5,1003) NB,(D(1),1=1,NP) URITE (6,2003) (D(1),1=1,NP) CREAD IN PIPE ROLENNESS IN MM. READ (5,1001) RV MAITE(6,2004) RV MAITE(6,2004) RV MAITE(6,2004) RV MAITE(6,2004) RV CREAD IN TIME DF ENTRY
	0C44 0C47 0C49 0C50 0C51 0C53 0C53 0C54 0C55 0C55 0C55 0C57 0C58	SUBROUTINE DATA1 COMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAK,MEY1,JEND,T,4D,9K CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETFRS READ (5,1003) NP,(D(13,1=1,NP) WRITE (6,2003) (D(1),1=1,NP) WRITE (6,2003) (D(1),1=1,NP) CREAD IN PIPE ROLEMMENS IN MM. READ (5,10013 RK MRITE(6,2006) RK READ (5,10013 TIME
	0C46 0C47 0C49 0C50 0C51 0C53 0C53 0C53 0C55 0C55 0C55 0C55 0C55	SUBROUTINE DATA1 COMMON/DP/D(20).SMIN,SMAX,DMIN,DMAX,SPMIN,SPMAK,MEY1,JEND,7,4D,9K CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND WAITE (6,2005) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETFRS READ (5,1000) NP,(D(1),1=1,NP) WRITE (6,2003) (D(1),1=1,NP) CREAD IN PIPE ROLENNESS IN MM. READ (5,1001) RK WAITE(6,2006) RK READ (5,1001) RK WAITE (6,2006) RK READ (5,1001) TIME WRITE (6,2006) TIME
:	0C46 0C47 0C49 0C30 0C31 0C33 0C34 0C33 0C34 0C35 0C36 0C38 0C38 0C38	SUBROUTINE DATA1 CUMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPMIN,SPMAK,MENJ,JEND,T,ND,RK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND WHITE (4,2005) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETFRS READ (5,1003) MP,(D(1),1=1,NP) WRITE (6,2000) (D(1),1=1,NP) CREAD IN PIPE POLENNESS IN MM. READ (5,1001) RK WRITE(6,2006) RK READ (5,1001) TIME WRITE (6,2006) TIME THTURFR0.0
:	0044 0047 0049 0050 0051 0052 0053 0053 0053 0053 0053 0053 0053	SUBROUTINE DATA1 CIMMON/OP/D[2[0];SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAK,MEY1,JEND,T,ND,RK CREAD IN NUMBER OF VERTICAL ZONES AND PIDE CHOICES READ (5,1002) MEND,JEND CREAD IN NUMBER OF PIDE SIZES AVAILABLE AND THEIN DINMETFRS READ (5,1003) NB,(D(1),1=1,NP) WRITE (6,2003) (D(1),1=1,NP) CREAD IN PIPE ROLEMNESS IN MM. READ (5,1001) RK MRITE(6,2006) RK READ (5,1001) TIME READ (5,1001) TIME WRITE (6,2006) TIME T=TIME=60.0 CREAD IN NIM AND MAX PIPE SLOPES
	0C46 0C47 0C49 0C50 0C51 0C53 0C53 0C53 0C55 0C55 0C55 0C55 0C55	SUBROUTINE DATA1 CGMMON/DP/0(20).SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAK,MEY1,JEND,7,4D,9K CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND WAITE (6,2005) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETFRS READ (5,1000) NP,(D(1),1=1,NP) WRITE (6,2000) (D(1),1=1,NP) WRITE (6,2000) (D(1),1=1,NP) CREAD IN PIPE ROLENNESS IN MM. READ (5,1001) RK WAITE(6,2006) RK READ (5,1001) RK WAITE (6,2006) RK READ (5,1001) TIME WRITE (6,2006) TIME TOTIME 60,0 CREAD IN MIN AND MAX SIPP SLOPES CREAD IN MIN AND MAX SIPP SLOPES
:	0C46 0C47 0C49 0C30 0C31 0C33 0C33 0C35 0C35 0C36 0C37 0C38 0C37 0C38 0C39 0C41 0C63	SUBROUTINE DATA1 CUMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPMIN,SPMAK,MEY1,JEND,7,ND,RK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND WAITE (4,2005) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETFRS READ (5,1000) NF,(D(1),1=1,NP) WRITE (4,2000) (D(1),1=1,NP) CREAD IN PIPE ROLENNESS IN MM. READ (5,1001) RK WRITE(6,2006) RK READ (5,1001) TIME WRITE (6,2006) TIME THE (6,2006) TIME THE (6,2006) TIME READ (5,1001) SMIN,SMAK READ (5,1001) SMIN,SMAK
	0044 0047 0049 0050 0051 0052 0053 0053 0053 0053 0053 0053 0053	SUBROUTINE DATA1 CIMMON/OP/D[2]20],SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAK,MEY1,JEND,T,ND,RK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND URITE (6,2003) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIN DINMETFRS READ (5,1003) NP,(D(1),1=1,NP) URITE (6,2003) (D(1),1=1,NP) URITE (6,2003) (D(1),1=1,NP) CREAD IN PIPE ROLEMNESS IN NM. READ (5,1001) RK MEITE(6,2004) RK READ (5,1001) TIME URITE (6,2004) TIME TETIMETAD. CREAD IN MIN AND MAX PIPE SLOPES READ (5,1001) SMIN,SMAX WEITE (6,2001) SMIN,SMAX
	0C46 0C47 0C49 0C50 0C51 0C53 0C53 0C55 0C55 0C55 0C55 0C55 0C55	SUBROUTINE DATA1 CGMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAK,MEY1,JEND,7,4D,9K CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND UNITE (4,2005) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETFRS READ (5,1000) NP,(D(I),I=1,NP) UNITE (4,2000) (D(I),I=1,NP) UNITE (4,2000) (D(I),I=1,NP) UNITE (4,2000) (D(I),I=1,NP) CREAD IN PIPE ROLEMMENS IN MM. READ (5,1001) RU MAITE (4,2006) RU READ (5,1001) RU MAITE (4,2006) RU READ (5,1001) TIME UNITE (6,2006) TIME T=TIME460,0 CREAD IN MIN AND MAX PIPE SLOPES READ (5,1001) SMIN,SMAX UNITE (6,2001) SMIN,SMAX CREAD IN MIN AND MAX DEPTM OF COVER
	0046 0047 0049 0050 0051 0053 0053 0053 0053 0055 0055	SUBROUTINE DATA1 CUMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPMIN,SPMAK,MEY1,JEND,7,4D,4K CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND WAITE (6,2005) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETFRS READ (5,1000) NP,(D(1),1=1,NP) WRITE (6,2000) (D(1),1=1,NP) WRITE (6,2000) (D(1),1=1,NP) CREAD IN PIPE ROLENNESS IN MM. READ (5,1001) RK WRITE(6,2004) RL READ (5,1001) TIME WRITE (6,2004) TIME THIME460,0 CREAD IN MIN AND MAX PIPE SLOPES READ (5,1001) SMIN,SMAX CREAD IN MIN AND MAX DEPTN OF COVEP NEAD (5,1001) DMIN,DMAX
	0044 0047 0049 0030 0031 0032 0031 0032 0034 0053 00534 00535 0054 00537 00536 0054 0064 0064 0066 0067	SUBROUTINE DATA1 CIMMON/DP/D[20];SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAK,MEUT,JEND,T,ND,RK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND UNITE (6,2003) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIN DINMETFRS READ (5,1003) MP,(D(1),101,NP) UNITE (6,2000) (D(1),101,NP) CREAD IN PIPE FOLGHNESS IN MM. READ (5,1001) RK MAITE (6,2006) RK READ (5,1001) TIME UNITE (6,2006) TIME TOTIMETE(6,2006) TIME TOTIMETE(6,2006) TIME CREAD IN MIN AND MAX PIPE SLOPES READ (5,1001) SMIN,SMAX CREAD IN MIN AND MAX PIPE SLOPES READ (5,1001) SMIN,SMAX CREAD IN MIN AND MAX PIPE SLOPES READ (5,1001) SMIN,SMAX UNITE (6,2002) DMIN,DMAX UNITE (6,2002) DMIN,DMAX
	0044 0047 0049 0051 0051 0053 0053 0053 0053 0055 0055	SUBROUTINE DATA1 COMMON/OP/D(200),SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAK,MEY1,JEND,T,ND,RK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND UNITE (6,2005) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETFRS READ (5,1003) NP,(D(13,1=1,NP) UNITE (6,2000) (D(1),1=1,NP) UNITE (6,2000) (D(1),1=1,NP) CREAD IN PIPE ROLEMMENS IN MM. READ (5,1001) RK MAITE(6,2006) RK READ (5,1001) TIME TTIME OF ENTRY READ (5,1001) TIME TTIME 00.0 CREAD IN MIN AND MAX SIPE SLOPES READ (5,1001) SMIN,SMAX CREAD IN MIN AND MAX DEPTN OF COVEP MEAD (5,1001) SMIN,SMAX UNITE (6,2002) DMIN,DMAX UNITE (6,2002) DMIN,DMAX UNITE (6,2002) DMIN,DMAX UNITE (6,2003) DMIN,DMAX UNITE (6,2003) DMIN,DMAX UNITE (6,2003) DMIN,DMAX UNITE (6,2003) DMIN,DMAX CREAD IN MIN AND MAX MANHOLE SPACING
	0C46 0C47 0C49 0C50 0C51 0C53 0C53 0C53 0C55 0C55 0C55 0C55 0C55	SUBROUTINE DATA1 CIMMON/DP/D(20).SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAK,MEN1,JEND,7,ND,RK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (3,1002) MEND,JEND UNITE (0,2003) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIN DIAMETFRS READ (3,1000) MB,(D(1),I=1,NP) UNITE (0,2000) (D(1),I=1,NP) CREAD IN PIPE ROCEMNESS IN MM. READ (3,1001) RX UNITE(0,2004) RX READ (3,1001) RX UNITE (0,2004) RX READ (3,1001) TIME UNITE (0,2004) TIME TUTINE=060.0 CREAD IN MIN AND MAX BIPP SLOPES READ (3,1001) SFIN,SMAX CREAD IN MIN AND MAX DEPTN OF COVEP NEAD (3,1001) DIN,DMAX UNITE (0,2002) NFIN,DMAX UNITE (0,2002) NFIN,DMAX CREAD IN MIN AND MAX PANHOLE SPACING READ (3,1001) SPIN,SPMAX
	0C44 0C49 0C49 0C30 0C31 0C32 0C33 0C34 0C35 0C36 0C37 0C36 0C63 0C63 0C63 0C63 0C63 0C64 0C64 0C65	SUBROUTINE DATA1 CIMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAX,MENJ,JEND,7,ND,RK CREAD IN NUMBER OF VERTICAL ZOMES AND PIPE CHOICES READ (5,1002) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILARLE AND THEIR DINMETFRS READ (5,1003) NG,(D(1),1=1,NP) URITE (6,2000) (D(1),1=1,NP) CREAD IN PIPE ROLEMMERS IN MM. READ (5,1001) RK URITE(6,2006) RK READ (5,1001) RK URITE (6,2006) TIME THIME460.0 CREAD IN MIN AND MAX BIPE SLOPES READ (5,1001) STIN,SMAX URITE (6,2001) SHIN,SMAX CREAD IN MIN AND MAX DIPF NOF COVEP MEAD (5,1001) SPIN,SMAX URITE (6,2002) DMIN,DMAX CREAD IN MIN AND MAX PANNOLE SPACING READ (5,1001) SPIN,SMAX URITE (5,2003) SPIN,SMAX
	0044 00647 00648 00051 0051 0053 00053 00053 00057 00057 00057 00057 00064 0064 0064 00668 00667 00668 00667 00668	SUBROUTINE DATA1 CQHHOM/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPMIN,SPMAK,MEY3,JEND,T,ND,RK CREAD IN NUMBER OF VERTICAL ZOMES AND PIPE CHOICES READ (5,1002) MEND,JEND GREAD IN NUMBER OF PIPE SIZES AVAILANLE AND THEIR DINMETFRES READ (5,1003) NO.(D(1),I=1,NP) URITE (6,2003) NO.(D(1),I=1,NP) URITE (6,2003) NO.(D(1),I=1,NP) CREAD IN PIPE ROLOMNEDS IN MM. READ (5,1003) RK WRITE (6,2004) RK READ (5,1003) RK WRITE (6,2004) RK WRITE (6,2004) TIME WRITE (6,2004) TIME WRITE (6,2003) TIME CREAD IN MIN AND MAX DIPP SLOPES READ (5,1003) SMIN,SMAX CREAD IN MIN AND MAX DEPTH OF COVER MEAD (5,1003) DMIN,DMAX CREAD IN MIN AND MAX DEPTH OF COVER MEAD (5,1003) SMIN,SMAX URITE (6,2003) SPHIN,SPMAX WRITE (6,2003) SPHIN,SPMAX WRITE (6,2003) SPHIN,SPMAX WRITE (6,2003) SPHIN,SPMAX WRITE (6,2003) SPHIN,SPMAX WRITE (6,2003) SPHIN,SPMAX
	0C46 0C47 0C49 0C50 0C51 0C53 0C53 0C55 0C55 0C55 0C55 0C55 0C55	SUBROUTINE DATA1 CIMMON/DP/D(20),SWIN,SMAX.OMIN,DMAX,SPMIN,SPMA4,MEVI,JEND,T,MD,MK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (5,1002) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETFRES READ (5,1003) MB,(D(1),101,NP) WRITE (6,2000) (D(1),101,NP) CREAD IN PIPE ROLGAMESS IN MM. READ (5,1001) RK WRITE(6,2006) RK REAR(1000,0 CREAD IN MIN AND MAX BIPE SLOPES READ (5,1001) SWIN,SMAX WRITE (6,2004) TIME TETIME460,0 CREAD IN MIN AND MAX BIPE SLOPES READ (5,1001) SWIN,SMAX URITE (6,2002) DWIN,DMAX WRITE (6,2003) NHN,DMAX WRITE (6,2003) SWIN,SMAX CREAD IN MIN AND MAX BIPE SLOPES READ (5,1001) SWIN,SMAX URITE (6,2003) TIME TETIME460,0 CREAD IN MIN AND MAX BIPE SLOPES READ (5,1001) SWIN,SMAX URITE (6,2003) TIME HEAD (5,1001) SWIN,SMAX CREAD IN MIN AND MAX BIPE SLOPES READ (1,001) SWIN,SMAX CREAD IN MIN AND MAX BIPE SLOPES READ (1,001) SWIN,SMAX URITE (6,2003) SWIN,SMAX CREAD IN MIN AND MAX BIPE SLOPES READ (1,001) SWIN,SMAX CREAD IN MIN AND MAX BIPE SLOPES READ (1,001) SWIN,SMAX URITE (6,2003) SWIN,SMAX CREAD IN MIN AND MAX BIPE SLOPES READ (1,001) SWIN,SMAX CREAD IN DIAGNOSTICS LEVEL READ (1,002) MD
	0C46 0C47 0C49 0C50 0C51 0C53 0C53 0C53 0C55 0C55 0C55 0C55 0C55	SUBROUTINE DATA1 COMMON/DP/D(20),SMIN,SMAX,DMIN,DMAX,SPMIN,SPMAX,MEY3,JEND,T,ND,RK CREAD IN NUMBER OF VERTICAL ZOMES AND PIPE CHOICES READ (5,1002) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETFOS READ (5,1000) MR,(D(1),I=1,NP) URITE (6,2000) (D(1),I=1,NP) CREAD IN PIPE POLONNESS IN MM. READ (5,1001) NK MRITE(6,2006) RK READ (5,1001) NK MRITE(6,2006) RK READ (5,1001) SKIN,SMAX CREAD IN MIN AND MAX SIPP SLOPES READ (5,1001) SWIN,SMAX CREAD IN MIN AND MAX OFPTH OF COVER MRITE (6,2003) RWIN,SMAX CREAD IN MIN AND PAX OFPTH OF COVER MRAD (5,1001) SWIN,SMAX CREAD IN MIN AND MAX OFPTH OF COVER MRAD (5,1001) SWIN,SMAX CREAD IN MIN AND PAX WANDLE SPACING READ (5,1001) SWIN,SMAX CREAD IN MIN AND PAX WANDLE SPACING READ (5,1001) SWIN,SMAX CREAD IN DIAGNOSTICS LEVEL READ (5,1002) MD
	0C44 0C47 0C48 0C49 0C31 0C32 0C34 0C33 0C34 0C53 0C34 0C57 0C58 0C63 0C63 0C64 0C64 0C64 0C66 0C66 0C66 0C66 0C66	SUBROUTINE DATA1 COMMON/DP/D(20).SMIN,SMAX,DMIN,DMAX,SPMIN,SPMAX,MEYD,JEND,T,ND,RK CREAD IN NUMBER OF VERTICAL ZOMES AND PIPE CHOICES READ (D)(02) MENDJEND WHITE (4,2003) MENDJEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIN DINMETERS READ (D)(00) NG(1),101,NP) WHITE (4,2000) (D(1),101,NP) CREAD IN PIPE BOLGHNESS IN NM. READ (3,1001) RK MATE(4,2004) RK READ (3,1001) TIME WHITE (4,2004) TIME THIME400,0 CREAD IN NIN AND MAX BIPE SLOPES READ (3,1001) SMIN,SMAX WHITE (4,2001) SMIN,SMAX WHITE (4,2001) SMIN,SMAX CREAD IN NIN AND MAX DIPE SLOPES READ (3,1001) SMIN,SMAX WHITE (4,2003) DMIN,DMAX CREAD IN NIN AND MAX DIPE SLOPES READ (3,1001) SMIN,SMAX CREAD IN NIN AND MAX DIPE SLOPES READ (3,1001) SMIN,SMAX CREAD IN NIN AND MAX MANDLE SPACING READ (3,1001) SMIN,SMAX CREAD IN NIN AND MAX MANDLE SPACING READ (3,1001) SPIN,SPMAX WHITE (4,2003) NFIN,SPMAX WHITE (4,2003) NFIN,SPMAX CREAD IN DIA AND FAX MANDLE SPACING READ (3,1002) MD RETURM ADD (5,1002) MD RETURM
	0C46 0C47 0C49 0C59 0C53 0C53 0C53 0C55 0C55 0C55 0C55 0C55	SUBROUTINE DATA1 CIMMOM/DP/D(20).SMIN,SMAX,DMIN,DMAX,SPMIN,SDMAK,MEVIN,JEND,T.ND.PK CREAD IN NUMBER OF VERTICAL ZOMES AND PIPE CHOICES READ C3,10023 MEND,JEND WHITE (6,2003) MENDJEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DIIMETFES READ C3,1003 ME, (D(1), 1=1,NP) WHITE (6,2000) NP, (D(1), 1=1,NP) WHITE (6,2000) RECOMMENS IN MM. READ C3,10013 RE WHITE (6,2006) RE READ C3,10013 RE WHITE (6,2006) RE READ C3,10013 SHN,SMAX CREAD IN TIME DF ENTRY READ C3,10013 SHN,SMAX CREAD IN TIME OF ENTRY READ C3,10013 SHN,SMAX CREAD IN MIN AND MAX SIPE SLOPES READ C3,10013 SHN,SMAX CREAD IN MIN AND MAX OF PTN OF COVER MEAD C3,10013 SHN,SMAX CREAD IN MIN AND MAX OF PTN OF COVER MEAD C3,10013 SHN,SMAX CREAD IN MIN AND MAX CAN WHITE (6,2003) NPIN,SMAX CREAD IN MIN AND MAX CREAD IN DIAGNOSTICS LEVEL READ C3,10023 NB RETLAN 1000 FORMAT (16,2(/1078,33))
	0C46 0C47 0C49 0C50 0C51 0C53 0C53 0C55 0C55 0C55 0C55 0C55 0C55	SUBROUTINE DATA1 CIMMOM/DP/D(20)SHIN,SMAX,DMIN,DMAX,SPMIN,SDMAX,HEVI,JEND,T,VD,VK CREAD IN NUMBER OF VERTICAL ZOMES AND PIPE CHOICES READ (5,1002) MENDJEND UNITE (4,2005) MENDJEND CREAD IN NUMBER OF PIPE SIZES AVAILARLE AND THEIR DINMETFRS READ (5,1000) M. (D(1),IR1,NP) URITE (4,2000) (D(1),IR1,NP) CREAD IN PIPE ROLENTRY READ (3,1001) RK URITE (6,2006) RK RCARC(J1000,D CREAD IN TIPE DE ENTRY READ (5,1001) SMIN,SMAX URITE (6,2004) TIME TTIMER00,D CREAD IN MIN AND MAX SIPP SLOPES READ (5,1001) SMIN,SMAA URITE (6,2001) SMIN,SMAA URITE (6,2001) SMIN,SMAA URITE (6,2001) DMIN,DMAX CREAD IN MIN AND MAX SIPP SLOPES READ (5,1001) SMIN,SMAA URITE (6,2003) TIME TTIMER00,D CREAD IN MIN AND MAX SIPP SLOPES READ (5,1001) SMIN,SMAA URITE (6,2003) TIME TETIMER00,D CREAD IN MIN AND MAX SIPP SLOPES READ (5,1001) SMIN,SMAA URITE (6,2003) TIME READ (5,1001) SMIN,SMAA CREAD IN DIAGNOSTICS LEVEL READ (5,1002) ND RETLAW 1000 FORMAT (16,22/1078,33) 1001 FORMAT (264,3)
-	0C44 0C49 0C49 0C31 0C32 0C33 0C34 0C35 0C36 0C37 0C36 0C63 0C63 0C63 0C64 0C65 0C66 0C66 0C66 0C66 0C66 0C66 0C66	SUBROUTINE DATA1 CGMMON/DP/0(20)SMIN,SMAX,DMIN,DMAX,SPMIN,SPMAK,MEUN,JEND,T.ND, %K CREAD IN NUMBER OF VENTICAL ZOMES AND PIPE CHOICES READ (3,1002) MEND,JEND WHITE (4,2003) MEND,JEND CREAD IN NUMBER OF DIGE SIZES AVAILARLE AND THEIR DINMETFES READ (3,1003) N#,(D(1),1=1,NP) WHITE (4,2000) ND(1),1=1,NP) CREAD IN PIPE POLENNESS IN MM. READ (3,1001) NW WHITE (4,2006) NW READ (3,1001) TIPE WHITE (4,2006) NUME TOTIME*60.0 CREAD IN TIME OF ENTRY READ (3,1001) SUIN,SMAX WHITE (4,2005) SUIN,SMAX CREAD IN MIN AND MAX SIPP SLOPES READ (3,1001) SUIN,SMAX WHITE (4,2002) DDIN,DMAX WHITE (4,2003) SUIN,SMAX CREAD IN MIN AND MAX MANDLE SPACING READ (3,1001) SPIN,SMAX CREAD IN NIN AND MAX MANDLE SPACING READ (3,1001) SPIN,SMAX CREAD IN NIN AND MAX MANDLE SPACING READ (3,1001) SPIN,SMAX CREAD IN DIA MD MAX MANDLE SPACING READ (3,1002) NDIN,SMAX CREAD IN DIA MD MAX MANDLE SPACING READ (3,2002) NDIN,SMAX CREAD IN DIA MD MAX MANDLE SPACING READ (3,2002) NDIN,SMAX CREAD IN DIA AND MAX MANDLE SPACING READ (3,1002) SPIN,SMAX CREAD IN DIA AND MAX MANDLE SPACING READ (3,2002) ND READ (3,2002) ND READ (3,2002) ND READ (3,2002) ND READ (3,2002) ND READ (3,1007) SPIN,SMAX CREAD IN DIAGNOSTICS LEVEL READ (3,1007) MENTAND 1000 FORMAT (226,3) 1001 FORMAT (226,3) 1002 FORMAT (226,3)
	0C44 0C47 0C49 0C53 0C53 0C53 0C55 0C55 0C55 0C55 0C55	SUBROUTINE DATA1 COMMON/DP/0(200)SMIN,SMAX,DMIN,DMAX,SPNIN,SPNAA,MEY1,JEND,T.40,%K CREAD IN NUMBER OF VERTICAL ZOMES AND PIPE CHOICES READ (3,1002) MEND,JEND UHITE (6,2003) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETFES READ (3,1001) MP,(D(1),Ie1,NP) URITE (6,2000) (0(1),Ie1,NP) CREAD IN PIPE ROCENESS IN MM. READ (3,1001) RK WEITE (6,2006) RK READ (3,1001) TIME TOTIME 06,2006) RK READ (3,1001) TIME TOTIME 06,2006) TIME TOTIME (6,2006) TIME TOTIME (6,2006) TIME CREAD IN NIN AND MAX SIPE SLOPES READ (3,1001) SPIN,SMAX URITE (6,2001) SPIN,SMAX URITE (6,2002) DHN,DMAX URITE (6,2003) SPIN,SMAX CREAD IN NIN AND MAX MANOLE SPACING READ (3,1001) SPIN,SMAX URITE (6,2003) SPIN,SMAX CREAD IN NIN AND MAX MANOLE SPACING READ (3,1001) SPIN,SMAX URITE (6,2002) ND READ (3,1002) ND READ (
	0C46 0C47 0C49 0C50 0C51 0C53 0C53 0C55 0C55 0C55 0C55 0C55 0C55	SUBROUTINE DATA1 COMPON/DP/D(20)SMIN, SMAX, DMIN, DMAX, SPNIN, SPMAA, MEUN, JEND, T, ND, NK COMPON/DP/D(20)SMIN, SMAX, DMIN, DMAX, SPNIN, SPMAA, MEUN, JEND, T, ND, NK CREAD IN NUMBER OF VERTICAL ZOMES AND PIPE CHOICES READ (5,1002) MEND, JEND WHITE (4,2003) MEND, JEND CREAD IN NIMBER OF PIPE SIZES AVAILABLE AND THEIN DINMETFRES READ (5,1000) MA, (D(1), 141, NP) CREAD IN PIPE POLEMNESS IN MN. READ (5,1001) RT MEITE (4,2006) RK READ (5,1001) TIPE WRITE (4,2006) TIPE TETINE400,0 CREAD IN TIPE OF ENTRY READ (5,1001) SMIN, SMAA WRITE (4,2003) SMIN, SMAA CREAD IN MIN AND PAX SIPP SLOPES READ (5,1001) SMIN, SMAA CREAD IN MIN AND PAX OFFYN OF COVER MEAO (5,1001) SMIN, SMAA WRITE (4,2002) NPIN, DMAX WRITE (4,2003) SPIN, SPAN CREAD IN DIA MED FAX PANNOLE SPACING READ (5,1001) SPIN, SMAX CREAD IN DIA MD FAX PANNOLE SPACING READ (5,1002) MD READ (5,1002) MD READ (5,1002) MD READ (5,1003) SMIN, SMAX CREAD IN DIA AND PAX PANNOLE SPACING READ (5,1002) MD READ (5,1003) SPIN, SPMAX CREAD IN DIAGNOSTICS LEVEL READ (5,1002) MD RETLEM 1000 FORMAT (10,22/1058,33) 1002 FORMAT (214) 2000 FORMAT (214) 2001 FORMAT (3X, 23MIN AND MAX PIPE SLOPES, 12X, 256,3) 2001 FORMAT (3X, 23MIN AND MAX PIPE SLOPES, 12X, 256,3)
	0C44 0C49 0C49 0C52 0C53 0C53 0C55 0C55 0C55 0C55 0C55 0C55	SUBROUTINE DATA1 COMMOR/DP/D (20),SMIN,SMAX,DMIN,DMAX,SPNIN,SPMAA,MENN,JEND,T,ND,TK CREAD IN NUMBER OF VERTICAL ZOMES AND PIPE CHOICES READ (3,1002) MEND,JEND WHITE (4,2005) MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILARLE AND THEIR DINMETFRES READ (3,1003 ND,(D(1),101,ND) CREAD IN PIPE POLENNESS IN MM. READ (3,1001) RK MEITE (4,2004) RX MEITE (4,2004) RX MEITE (4,2004) RX MEITE (4,2004) RX MEITE (4,2004) TIME TTIME OF DENTRY READ (3,1001) TIME MEITE (4,2004) TIME TTIME 0,1001) SMIN,SMAX GREAD IN MIN AND MAX DIPP SLOPES READ (3,1001) SMIN,SMAX MEITE (4,2003) DNIN,DMAX MEITE (4,2003) DNIN,DMAX MEITE (4,2004) TIME TTIME 460,0 CREAD IN MIN AND MAX DIPP SLOPES READ (3,1001) SMIN,SMAX CREAD IN MIN AND MAX DIPP SLOPES READ (3,1002) ND RETURN 1000 FORMAT (226,3) 1001 FORMAT (226,3) 1001 FORMAT (256,3) 1001 FORMAT (3X,23MIN AND MAX DIPP SLOPES,12X,278,3) 2000 FORMAT (3X,23MIN AND MAX COVER,10X,278,3) 2001 FORMAT (3X,23MIN AND MAX COVER,10X,278,3) 2002 FORMAT (3X,23MIN AND MAX COVER,10X,278,3) 2004 FORMAT (3X,23MIN AND MAX COVER,10X,278,3) 2005 FORMAT (3X,23MIN AND MAX COVER,10X,278,3) 2004 FORMAT (3X,23MIN AND MAX COVER,10X,278,3) 2005 FORMAT (3X,74MIN AND MAX COVER,10X,278,3) 2006 FORMAT (3X,74MIN AND MAX COVER,10X,278,3) 2007 FORMAT (3
	0044 00647 00648 00031 0032 00031 00034 00034 00037 00034 00037 00036 0004 00063 00064 00067 00067 00073 00073 00073 00073 00073 00073	SUBROUTINE DATA1 C(H=0A/DF/D(20),SHIN,SMAX,DMIN,DMAX,SPNIN,SDMAA,MENN,JEND,T,ND,NK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (3,1002) MEND,JEND WITE (4,2003) MEND,JEND WITE (4,2003) MEND,JEND WITE (4,2003) (0(1),I=1,NP) WITE (4,2003) (0(1),I=1,NP) CREAD IN PIPE FOLGHNESS IN MN. READ (3,1001) NK WEITE (4,2004) RK READ (3,1001) TIPE TTIME 40,2004) RK READ (3,1001) TIPE TTIME 40,0 CREAD IN TIPE DF ENTRY READ (3,1001) SHIN,SMAX CREAD IN TIPE DF ENTRY READ (3,1001) SHIN,SMAX CREAD IN TIPE DF ENTRY READ (3,1001) SHIN,SMAX CREAD IN MIN AND MAX SIPP SLOPES READ (3,1001) DMIN,DMAX WITE (4,2002) DMIN,SMAX CREAD IN MIN AND PAX OFFN OF COVEP MEAD (3,1001) SHIN,SMAX CREAD IN TIME NAX MOLE SPACING READ (3,1001) SPHIN,SMAX CREAD IN DIA(C) SPHIN,SMAX CREAD IN TIME AND PAX MANDLE SPACING READ (3,1002) ND RETURN 1000 FORMAT (14,2(/10F8,3)) 1001 FORMAT (2F8,3) 1002 FORMAT (2F8,3) 2005 FORMAT (5X,123MHA AND MAX PIPE SLOPES,12X,2F8,3) 2005 FORMAT (5X,134MHA AND MAX COVFE,1AX,2F8,3) 2005 FORMAT (5X,134MHA AND MAX COVFE,1AX,2F8,3) 2005 FORMAT (5X,134MHA AND MAX COVFE,1AX,2F8,3) 2005 FORMAT (5X,134MHA AND MAX MAX MAX SPERS,100,2F8,3) 2005 FORMAT (5X,134MHA AND MAX MAX MAX P278,3)
	0C46 0C47 0C49 0C53 0C53 0C53 0C55 0C55 0C55 0C55 0C55	SUBROWTINE DATA1 COMPON/OP/D(20), SMIN, SMAX, DMIN, DMAX, SPMIN, SPMAA, MENN, JEND, 7, ND, NK CREAD IN NUMBER OF VERTICAL ZONES AND PIPE CHOICES READ (3,1002) MEND, JEND WHITE (4,2003) MEND, JEND WHITE (4,2003) MEND, JEND CREAD IN NUMBER OF PIPE SIJES AVAILABLE AND THEIR DINMETFRES READ (3,1001) NK WHITE (4,2004) REAL READ (3,1001) RK WHITE (4,2004) REAL READ (3,1001) RK WHITE (4,2004) TIME THIME (4,2004) TIME THIME (4,2004) TIME WHITE (4,2004) TIME THIME (4,2004) TIME THIME (4,2004) TIME THIME (4,2004) TIME THIME (4,2004) TIME THIME (4,2004) TIME CREAD IN TIME OF ENTRY READ (3,1001) SMIN, SMAX WHITE (4,2003) TIME THIME (4,2003) TIME THIME (4,2003) TIME READ (3,1001) SMIN, SMAX WHITE (4,2003) TIME, SMAX WHITE (4,2003) SMIN, SMAX WHITE (4,2004) SMIN, SMAX WHITE (4,2003) SMIN, SMAX WHITE (4,2003) SMIN, SMAX WHITE (4,2004) SMIN, SMAX WHITE (4,2004) SMIN, SMAX WHITE (4,2004) SMIN, SMAX WHITE (4,2003) SMIN, SMAX WHITE (4,2004) SMIN,
	0C44 0C49 0C49 0C52 0C53 0C53 0C55 0C55 0C55 0C55 0C55 0C55	SUBROUTINE DATA1 CIMMON/DP/B(20).SMIN,SMAX,DMIN,DMAX,SPMIN,SPMAA,MEY1,JEND,T,ND,TK CREAD IN NUMBER OF VERTICAL ZOMES AND PIPE CHOICES READ (3,1002) MEND,JEND CBAD IN NUMBER OF DIPE SIZES AVAILABLE AND THEIR DINMETFRES READ (3,1003) ME,O(1).I=1,NP) MRITE (4,2004) MR,O(1).I=1,NP) CREAD IN PIPE ROLENKESS IN MM. READ (3,1001) RK WRITE(4,2006) RK WRITE(4,2006) RK WRITE(4,2006) TIME THIEFOD O CREAD IN TIME DF ENTRY READ (3,1001) SPIN,SMAX WRITE (4,2006) TIME THIEFOD.O CREAD IN TIME OF ENTRY READ (3,1001) SPIN,SMAX WRITE (4,2001) SPIN,SMAX WRITE (4,2001) SPIN,SMAX WRITE (4,2001) DFIN,DMAX CREAD IN NIN AND MAX DIPE SLOPES READ (3,1001) SPIN,SMAX WRITE (4,2002) DNIN,DMAX CREAD IN NIN AND MAX DIPE SLOPES READ (3,1001) SPIN,SMAX WRITE (4,2003) NPHIN,SMAX CREAD IN NIN AND MAX MINE SPACING READ (3,1002) DNIN,DMAX CREAD IN DIANOSTICS LEVEL READ (3,1002) NDIN,SPNAX WRITE (4,2003) NPHIN,SMAX CREAD IN DIANOSTICS LEVEL READ (3,1002) NDIN,SPNAX MRITE (4,2003) NPHIN,SMAX CREAD IN DIANOSTICS LEVEL READ (3,1002) NDIN,SPNAX MRITE (4,2003) NPHIN,SMAX CREAD IN DIANOSTICS LEVEL READ (3,1002, NDI RETLAW 1000 FORMAT (216,2) 2001 FORMAT (3X,23MHIN AND MAX PIPE SLOPES,122,288,3) 2002 FORMAT (3X,23MHIN AND MAX PIPE SLOPES,124,288,3) 2003 FORMAT (3X,23MHIN AND MAX MINEPACING,288,3) 2004 FORMAT (3X,73MHIN AND MAX MINEPACING,288,3) 2005 FORMAT (3X,73MHIN AND MAX MINEPACING,288,3) 2006 FORMAT (3X,73MHIN AND MAX MINEPACING,288,3) 2006 FORMAT (3X,73MHIN AND MAX MINEPACING,288,3) 2006 FORMAT (3X,73MHIN AND MAX MINEPACING,288,3) 2007 FORMAT (3X,73MHIN AND MAX MINEPACING,288,3) 2006 FORMAT (3X,73MHINE OF ENTRY,56,45,45 HINE 2006 FORMAT (3X,73MHINE OF ENTRY,56,45,45 HINE 2007 FORMAT (3X,73MHINE OF ENTRY,5
	0044 00647 00648 00031 00032 00034 00033 00034 00034 00034 00034 00034 00034 00034 00034 00034 0004 0004 0004 00077 000000	SUBROUTINE BATA1 CUMPON/DP/0(20).SMIN,SMAX,DMIN,DMAX,SPMIN,SPMAA,MEV1,JEND,T,ND,TK CREAD IN NUMBER OF VERTICAL ZOMES AND PIPE CHOICES READ (3,1002) MEND,JEND CREAD IN NUMBER OF DIPE SIZES AVAILABLE AND THEIR DIAMETERS READ (3,1003) WE,OC(1).FR,NP) WRITE (6,2000) (O(1).JUT,NP) CREAD IN PIPE ROLEMMESS IN NM. READ (3,1003) RC MRITE (6,2006) RC READ (3,1003) RC MRITE (6,2006) THE WRITE (6,2006) THE TTIME40.0 CREAD IN TIME DI ENTRY READ (3,1003) SHIN,SMAX WRITE (6,2001) SHIN,SMAX WRITE (6,2001) SHIN,SMAX WRITE (6,2001) SHIN,SMAX WRITE (6,2002) DHIN,DMAX WRITE (6,2002) DHIN,DMAX WRITE (6,2003) SHIN,SMAX CREAD IN MIN AND PAX SIPP SLOPES READ (3,1003) SHIN,SMAX WRITE (6,2003) SHIN,SMAX WRITE (6,2003) SHIN,SMAX CREAD IN DIAM AND PAX SIPP SLOPES READ (3,1003) SHIN,SMAX CREAD IN MIN AND PAX SIPP SLOPES READ (3,1003) SHIN,SMAX WRITE (6,2003) SHIN,SMAX CREAD IN DIAM AND PAX SIPP SLOPES READ (3,1003) SHIN,SMAX WRITE (4,2002) DHIN,DMAX WRITE (4,2002) SHIN,SMAX CREAD IN DIAMO PAX SIPP SLOPES READ (3,1003) SHIN,SMAX CREAD IN DIAMOVER SIGNAN CREAD IN DIAMOVER SIGNAN C
	0C44 0C47 0C49 0C53 0C53 0C53 0C53 0C55 0C55 0C55 0C55	SUBROUTINE DATA1 CJMMON/DP/0(20).SMIN,SMAX,DMIN,DMAX,SPMIN,SPMAA,MEV1,JEND,T,ND,TK CREAD IN NUMBER OF VERTICAL ZOMES AND PIPE CHOICES READ (3,1002) MENDJEND CREAD IN NUMBER OF DIPE SIZES AVAILABLE AND THEIR DINMETFES READ (3,1002) MENDJEND CREAD IN NUMBER OF DIPE SIZES AVAILABLE AND THEIR DINMETFES READ (3,1001) ME,OC(1).IET,NP) WHITE (4,2000) ME,OC(1).IET,NP) CREAD IN TIPE BCLEMMEDS IN MN. READ (3,1001) RT WHITE (4,2006) RU READ (3,1001) RT WHITE (4,2006) TIPE WHITE (4,2006) TIPE WHITE (4,2006) TIPE WHITE (4,2006) SPIN,SMAX CREAD IN TIPE DE ENTRY READ (3,1001) SPIN,SMAX WHITE (4,2006) SPIN,SMAX CREAD IN MIN AND PAX DIPP SLOPES READ (3,1001) SPIN,SMAX CREAD IN MIN AND PAX DIPP SLOPES READ (3,1001) SPIN,SMAX CREAD IN MIN AND PAX DIPP SLOPES READ (3,1001) SPIN,SMAX CREAD IN MIN AND PAX PIPP SLOPES READ (3,1001) SPIN,SMAX CREAD IN MIN AND PAX PANDLE BPACING READ (3,1001) SPIN,SMAX CREAD IN MIN AND PAX PANDLE BPACING READ (3,1001) SPIN,SMAX CREAD IN DIA AND PAX PANDLE BPACING READ (3,1002) ND READ (3,1002) ND REA
	0C44 0C49 0C49 0C52 0C53 0C53 0C55 0C55 0C55 0C55 0C55 0C55	SUBROUTINE DATA1 CIMPON/DF/D(20),SMIN,SMAX,DMIN,DMAX,SPMIN,SDMA4,MEY1,JEND,T,ND,TE CIMPON/DF/D(20),SMIN,SMAX,DMIN,DMAX,SPMIN,SDMA4,MEY1,JEND,T,ND,TE CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DIIMETFRS READ (5,1002) MEND,JEND WRITE (6,2000) (D(1),IR1,MP) WRITE (6,2000) RE READ (5,1001) RT MREAD (5,1001) RT MREAD (5,1001) THE TRIME (6,2000) THE TRIME (6,2000) THE TRIME (6,2001) SPIN,SMAX WRITE (6,2001) SPIN,SMAX WRITE (6,2001) SPIN,SMAX CREAD IN TIME OF ENTRY READ (5,1001) SPIN,SMAX WRITE (6,2001) SPIN,SMAX WRITE (6,2001) SPIN,SMAX WRITE (6,2001) SPIN,SMAX CREAD IN MIN AND MAX DIPF SLOPES READ (5,1001) SPIN,SMAX WRITE (6,2001) SPIN,SMAX WRITE (6,2001) SPIN,SMAX WRITE (6,2001) SPIN,SMAX WRITE (6,2003) NDN,SMAX WRITE (6,2003) NDN,SMAX CREAD IN MIN AND MAX DIPF SLOPES READ (5,1001) SPIN,SMAX WRITE (6,2003) NDN,SMAX CREAD IN MIN AND MAX DIPF SLOPES READ (5,1001) SPIN,SMAX WRITE (6,2003) NDN,SMAX C
	0044 00647 00648 00031 00032 00034 00034 00034 00034 00034 0004 000	SUBBOUTINE DATA1 CJAMOW/DP/D/DJ/DJ/DJ/DJ/DJ/DJ/DJ/DJ/DJ/DJ/DJ/DJ/D
	0044 00647 00648 00630 00631 00633 00633 00634 00637 00636 00664 00663 00664 00663 00664 00663 00664 00667 00677 00676 00677 00678 00677 00683 00683 00683	SUBBOUTINE DATA1 CIMPON/DF/D(20)SMIN,SMAX,DMIN,DMAX,SPNIN,SPMA4,MEY1,JEND,T,ND,TK CIMPON/DF/D(20)MEND,JEND WATTE (6,2003)MEND,JEND WATTE (6,2003)MEND,JEND CREAD IN NUMBER OF PIPE SIZES AVAILABLE AND THEIR DINMETRES READ (5,1000) WA,(D(1),IE1,MEN) WATTE (6,2000) READ,JEND CREAD IN TIPE OCENHESS IN MN. READ (3,1001) RT WATTE (6,2000) TIPE TETIMEEGO,O CREAD IN TIPE OF ENTRY READ (3,1001) STIL MATTE (6,2001) TIPE WATTE (6,2001) TIPE WATTE (6,2001) STIL MATTE (6,2001) STIL CREAD IN MIN AND MAX SIDE SLOPES READ (3,1001) STIL,SMA4 WATTE (6,2002) MENANULE SPACING READ (3,1001) STIL,SMA4 WATTE (6,2002) MENANULE SPACING READ (3,1001) STIL,SMA4 WATTE (6,2002) MENANULE CREAD IN MIN AND MAX DEPTH OF COVER MEAD (3,1001) STIL,SMA4 WATTE (6,2002) MENANULE CREAD IN MIN AND MAX DEPTH OF COVER MEAD (3,1001) SPIN,SMA4 WATTE (6,2002) MENANULE CREAD IN MIN AND MAX DEPTH OF COVER MEAD (3,1001) SPIN,SMA4 WATTE (6,2002) MENANULE CREAD IN MIN AND MAX DEPTH OF COVER MEAD (3,1001) SPIN,SMA4 WATTE (6,2002) MENANULE CREAD IN DIAKONAX CREAD IN DIAKONAX C

•

.

.

0683	SUBBOUTINE DATA2 (NIP,PIP,NIT,PIT,144X,JMAX,KMAX,LMAX)
0086	CSUBROUTINE CALLS GEON AND SORTS OUT ADDRESSES FOR PER'S ARRAYS
0687	CONMON/DP/D(20), SMIN, SMAX, DMIN, MMAX, SPHIN, SPMAK, MEND, JEND, T, ND, RE
0048	CONMON/UNERE/IN(50,4),LN1,LN2,LN3,LP1,LP2,LP3
00.00	DIMENSION NIT(IMAX), DIT(JMAX), NID(KMAX), DID(LMAX)
0000	CALL BEOM (NIP,PIP,KWAX,LMAX)
0791	108410(2)
0692	CDESING ADDRESSES FOR ARRAYS WIP AND PIP
1930	NTOT=0
4910	
0095	HTQT=0
A930	D0 10 1=1,L0
0097	1M(1,1)#2+2+NT07+L707+3+1
8930	[N(],4)=1+4+NT07+2+MT0T
0099	J=14(1,1)
0100	NTOT=NTOT+LIP(J)
0101	1 N (1 , 2) = 3+2+NTOT+LTOT+3+1
0102	IN(I, 3)=IN(I, 2)+1
0103	J=IN(I,T)
0104	LT07=LT07+N1P(J)
0105	J=IN(I,Z)
0104	MT0T#MT0T+N1P(J)
0107	to continue
0108	LN1=2+2+NTQ7+L707+3+(L0+1)
0109	LN2=LN3+MEND+JE+D+N3P(6)
0110	LNS=LNZ+MEND+JEND+LO+1
0111	LP109+6+NT07+2+HT07
0112	LP2=LP1+MEND+JE+D+3
0113	6P3#6P2+N1P2+3+1
0114	WAITE (6007,00)
0115	MHILE (0'5000) ((IM(I'Y)')=1'+7''I#1'+7')
0116	WRITE (6,2001) LN1,LN2,LN3,LP1,LP2,LP3
0112	RETURN
0118	1000 FORMAT (149/10X,36HADDRESSES STARED IN ARRAY IN(LO,6))
0119	2000 FORMAT (6110)
0520	2007 608447 (140/3x,64L47=,14,5%,64L42=,14,5%,64L43=,16,5%,64L43=,16,
0161	15%,6%68%,16,5%,6%683%,163
0122	END

. . .

;

277; NAME DATAZ -----LENGTH

•

1

.

- 212 -

SUBPOUTINE GEOM (NIB,PID,KMAX,LMAX) DIMENSION NIP(KMAX),PIP(LMAX) ----PROBLEM SIZE READ (5,1000) NIP(2) 0123 0124 ć-0126 ---- SET COUNTERS 1510 ¢ -K=4 0128 1.0 C----READ IN DATA FOR EACH BRANCH 0130 0131 01 3 2 01 3 3 00 40 Jet.LO 0134 K=K+1 READ (3,1000) NH 0135 0134 0137 NIP(K)=NN NIP(K)=NN C----RAD IN CHAINAGES, AREAS, TOP AND BOTTOM ZONE LEVELS DO 10 M=1,4 L=L+1 0138 0139 0140 NuL+NN+1 READ(5,2000) (PIP(1),1+L,N) 0141 LUN 10 CONTINUE 0143 C----READ IN FIRST AND LAST MANHOLES DO 20 No1 / 2 Kakoi 0145 0146 0147 N=K+NN=1 READ (5,1000) (NIP(1),1=K,N) 0148 0149 0150 0151 Kul 20 CONTINUE C-----READ IN GROUND LEVEL DATA 0152 K=K+1 0153 0154 0155 0156 READ (5,1000) NG DO 30 M#1,2 1-1+1 NaL+N&+1 READ (5,2070) (PIP(I),I=L,N) 0157 0158 0139 Len 30 CONTINUE C----READ IN CONNECTIONS UPSTREAM 0141 0142 READ (3,1000) NR NIP(4)=NR 0165 0165 0166 0167 IF (NR.EQ.0) \$5 70 40 K=K+1 N=K+VR+1 0168 0169 0170 READ (S, 1000) (NIP(1)/I=K,N) KUN 60 CONTINUE 40 CINTINUF READ (3,1003) NIP(1),NIP(3),NIP(4) C----PAINT OUT DATA AS STORED WRITE (4,5000) WRITE (4,3000) (NIP(1),I=1,K) WRITE (4,4003) (PIP(1),I=1,K) RETURN 1000 FORMAT (10[3] 2000 FORMAT (10F1,3) 3000 FORMAT (10F10,5) 5000 FORMAT (10F10,5) 0171 0172 0173 0175 0174 0177 0179 0180 0182 END

ENC OF SERMFNY, LENGTH 257, NAME SEOM

.

1

1

6183	SUBROUTINE FLOWE(NID, PID, KMAX, LMAX, PIT, JMAX, T, JK)
0184	COMMON/WHE4E/IN(50,4),LN1,LN2,LN3,LP1,L#2,LP3
0185	DIMENSION HIP(KWAX),PIP(LWAX),PIT(JWAX)
0124	LUGICAL OK
0187	Casaaderine netuan praido
0188	RP#1,0
0189	FO ##15#(5)
0190	44=0
0191	CFUR EACH RUN
0192	BO 20 1=1.LC
0193	TUPUT
0194	CIDENTIFY NUMBER OF U/S BRANCHES
0195	K=1+(1,3)
OTRA	N=N1P(K)
0197	1F (N.EG.O) 60 TG 5
0198	50 3 Julin
0199	Ka14(1,3)+J
0200	KaJMAX-NJP (K)
0201	WHITE(0,1001) K.PIT(K)
0202	IF (PIT(K),GT,TUP) TUP=P1T(K)
0203	3 CONTINUE
0204	5 WRITE (6,1001) 1,70P
0205	Ku[N(];+7)
0206	NuN tPCK3
0207	DI\$T\$=0.0
0208	
0209	CIDENAILA DISAUCE VIGAR MA LADA FARA FURSIONE WA CONTINU
0210	Kaik(1,4)+J-1
0211	D121=#1#(K)=D1271
0212	DISTIBUTE
0213	CCALCULATE TIME
0214	NA=NA+1
0215	TINE=025T/#2T(N#3+TU#
0214	TUPetime
0217	CIDENTIFY AREA
0218	L=1N(1,1)
0219	x=I+C1+C3+419(C3+J=1
0220	ANEADDID(K)
0221	CCALCULATE FLOW FROM WILHAMANDELAND HAIAPHEE POHNOCA
2220	CALL RAIN (40,TIME/60,0,RI)
0243	K=L92-1+NA
0224	FLOWMAREA+N1/3,4E6
0252	
0559	WRITE (A, TODD) J, NA, DIST, SIT(NA), THEY RECEIPTION
0227	B1F(K)=FLOW
0350	10 CONTINUE
U267	
0230	P[T](4)=F7=8
1212	97178 19379997 1757575757 20. 60978948
0333	ev continue
0233	421U4N 9ANA FARMAY JAAY 315,388 8,843 8,3844 43
0235	1999 - FURTET STURISIZIJET STUTESZISTES - 2057 1897 - 40 - 40 - 40 - 40 - 40 - 40 - 40 - 4
0234	1991
	•

ENE OF SEGMENT, LENGTH 278, NAME FLOWS

.

٠

•

.

.

!

• ••

.

i

ł

•

- 214 -

SUBPOUTINE CONS (LIP, PIP, NIT, PIT, KMAX, LMAX, IMAX, JMAX, L, HCOSTS) COMMON/OP/O(20), SMIN, SMAX, DMIN, DMAX, SPMIN, SPMAX, MEND, JEHD, T, ND, RK COMMON/UMERE/IN(50, 4), LH1, LH2, LH3, LH1, LH2, LP3 0237 0238 0339 DIMENSION NIP(KMAX), PIP(LMAX), NIT(IMAX), PIT(JMAX) 0240 MJEHEND+JEND 0241 C----- HOW MANY U/S P1PES7 K41H(1,3) 0243 NaNIP(K) 0244 NAMIP(C) IF (N,GT,O) GO TC 30 C----NO U/S PIPESI SPT COST OF ARRIVAL AT 1ST MANHOLE IN RUN TO ZERO DO 10 J=1.MJ 10 PIT(J)=0.0 IF (I.EQ.1) 60 TO 70 C----STORE D/S COSTS FOR LAST RUN IN NEXT SECTION OF D/S COSTS ARRAY K=N(I=1.13) 0245 0247 0248 0249 0251 NENDENIF(K) 0232 1233 J1=(NEHD=1)+HJ+1 0234 J2=+E+D+MJ NCOSTS=+COSTS+1 0522 0256 0257 0258 0259 K=LP1+(NCD\$75+1)+MJ=1 K1=K+1 N= 20 J=J1+J2 20 010(6)=017(3) 0260 WHITE (4,1001) 41.K WRITE (4,1001) 41.K WRITE (4,1002) (010(1K),1K=K1,K) GO TO PO C---ONE U/S PIPEI SET U/S COSTS =0/S COSTS FOR LAST RUN 0241 0245 0263 30 K=IN(1=1,1) N&ND=N{P(K) J1=(NEND=1)+HJ+1 0203 0266 0267 JS#NEND+NJ K=0 D0 40 j=j1,j2 K=K+1 0269 0270 0271 0272 40 PIT(K)=PIT(J) IF (N.E0.1) GO TC 70 C----TWO OR THREE U/S PIPESI U/S COSTS=SUM OF D/S COSTS 0273 0276 0275 Heh-1 0274 D0 60 L=1.M J=LP1.MJ+(NC0\$T\$=1)=1 0278 00 50 K#1+#J Peter. 0280 #17(K)##17(K)+#1#(J) 1950 2920 0284 0285 0286 0287 RETURN 1001 FORMAT (5x, 3PHD/S CO3TS, LAST RUN, STORED IN PEP FRON, 16, TH T7, 14) 1002 FORMAT (10F12, 5) 1005 FORMAT (5x, 17HU/S CO3TS FOR RUN, 16, 19H (NO, OF U/S PIPES#, 14, 14)) 0288 0290 END END OF SEGMENT, LENGTH 317, NAME COMB 0291 0292 SUBROUTINE PRINT (NIP, PIP, XMAX, LMAX) DIMENSION NEPCKMAX3, FIPCLMAX3 RETURN 0293 END

END OF SEGMENT, LENGTH 32, NAME PRINT

•

ł

1

	SUBROUTINE WARUN (NIP,PIP,VIT,PIT,IMAX,JMAX,KMAX,LMAX,WAV)
0104	DIMENSION NIT(IMAX), DIT(JHAX), NIP(KMAX), DIP(LMAX), LEV(20),
1111	1111(20),LL2(20),015T(20),448A(20)
A298	COMPONIUNEREIIN(SOIA), LNI, LNZ, LNS, LPI, LPZ, LPS
0390	COMMUNIND/DELD(20), SMIN, SMAX, DMIN, DMAX, SPMIN, SPMAX, MEND, JEND, T, ND, HK
0300	LUGICAL NEWY
0301	144114,420,43
2302	145m14(H4UH,2)
0305	EN3=EN6NRUN,33
0304	1 # # 0 1 # (# 4 U # 5) # 1
0305	LENDANIP(INZ)
0304	
0307	CDEFINE D/B STAIR
0308	
0304	EU HUHTT DARAMETERS DEPENDENT ON "N"
0310	hFLNn, TRUE.
0311	
0313	KLANUINI+H
0314	KLBNWKLAN+NEND
0315	XM=PIP(«XN)
0318	KZTOBEING+Z+NENR+N
0317	K2807 = K2707 + VEN0
0318	2709-919 (K2709)
0319	
0320	KQN=0 TE Juping 20 43 20 10
0341	
2520	00 5 11-1,NB
4560	Kuth(11,1)
0345	5 KON=KON+HIP(K)=1
0326	10 KANAKANATAS
0327	J=0
0358	30 jæj+1
0349	MeQ
0330	
0331	HJN6((44))JJE/(000000)
0336	PIT(NJN) = 4949999.7
0333	
0335	
0334	44 NURNUS
0337	- <u>Kinnojyé</u> ten
0338	XNN=P3P{XXNN}
0339	KZTESNG+Z+NEND+NN
0340	279919(427)
0341	KZBEKZTANEND
0342	
0343	IF C.RUT.NEENJ OU IV IJU Componentie taveramentave genuide isvels
0343	
0344	KGXL=1%6+6+VEND+LEND+L
0347	15 (PIP(KGXL),GT,PIP(KXNN)+0,01) 40 70 60
0348	50 CONTINUE
0349	40 LIEL -
0331	KYALTENAL Kali - Javi - Irna
0152	· Jaci
0333	440.0
0334	LEVELOI
0335	IF (L1.EQ.LEND) GO TC 140
0336	IF (PI#(KGXL1+1),GT,XN+0,01) 60 10 160
033/	L3=L1+1 Do 20 (0) 1 (0)
0339	KGXLATNÍAÍANSNOALSNOAI
0160	16 (PIJ(KGHL). GT. HA-0. 01) 40 TO AD
0361	70 CONTINUE
0342	80 L24L-1
0343	KGZLZ#IN6+6+hEND+LZ
0364	KGXLZ=KGZLZ+LFNA
0305	CDEFINE AREA OF LONG SECTION ABOVE STRAIGHT LINE
0344	A = + P J P (K G X L 1 + 1 3 + (P J P (K G Z L 1 3 + P] P (K G Z L 2 + 1 3) + 0 , 5
0367	00 85 LL+L7+L2
0368	KGKL
0369	KUREENVEEEVETV 45. AAAptelekatii\+eetP(katii+t\=Dtb/katii+t\+a
03/0	σμ μμτριπικοπορισισισισισισισισισισισισισισισισισισισ
0371	CIS GROUND LEVEL, CONCAVE, CONVEX OR VARIABLE
0171	65L0PE+(PIP(KG2L2+1)+PIP(KG2L1+1))/(XH+XHN)
0374	00 130 L=L1/L2
0375	KGZLOINGOGANENDOL
0376	KGXL=KGZL+LEND
0377	PSLOPE4(P1P(KG2L)=P1P(KG2L1=1))/(P1P(KGXL)=XNN)
0374	1 (PSLOPE, LT, 63LOPE=0,00003) GO TO 90
03/9	F (#\$LAPE, LT, WILLPE#0, A0007) 6A TO 130
0369	60 TO (110/100/130//LEVEL 00 60 TO (120.110.100)/LEVEL
0382	90 90 17 5'505'905'905'905'55'55 800 1255196
0383	60 TO 160
0184	110 LEVELOS
0385	GO TO 130
0386	120 LEVELOD
0367	130 CONTINUE

. ...

- 216 -

C----DEFINE GROUND CONDITIONS AND DISTANCE BETWEEN MANMOLES 140 NTHUNN-NIP (KLAN)+1 LEV (NTH)=LEVEL LL1 (NTH)=L1 LL2 (NTH)=L2 0392 0157 (N7H)=XN=XNH AREACTIVIEA WAITE (6,2002) NANAANTHALEVELALIALZADIST(NTH)AA C----DEPINE UPSTREAM STATE (U/S PIPE DIAMETER AND CRIJEN LEVEL) ISO NYMENN-NIP(KLAN)+I 11=0 160 11=11+1 MMm 0 170 MM8M4+1 MMJJNN&(NN+1)+JEhD+MEND+(JJ+1)+MEND+MM MMJJNN&(NN-1)+JENDOMEND+(JJ-1)+MEND+MM C-----CHECK FEASIBILITY OF SOLUTION C-----(U/S STATE FEASIBLE7) IF (DIT(MMJJNN), GT,999999,0) GO TO 250 C-----(PIPE SLOPE -ITMIN RESTAINTS7) ZUS#27-FLQAT(MM-1)+(ZT-ZB)/FLQAT(MEND-1) SLOPE=(ZUS+2DS)/DIST(NTH) IF (SLOPE,LT,SMIN=0.00001) GO TO 260 IF (SLOPE,LT,SMIN=0.0001) GO TO 250 C-----(PIPE CAPACITY SLFFICIENT? COLEMMOK-WHITE FORMULA) ROSOF(SLOPE+013) C----(PIRE CAPACITY SUPPLIEURIF LOLEMANDL-MAILE FORGELF SQ#SQRT(SLORE+D(J)) GFULL=+6,927+D(J)+D(J)+SQ+ALQ610(RK/3,7/D(J)+0,6671E-6/49/D(J)) IF (GFULL,LT,FIP(KGM)) 60 TO 250 C-----(DEPTH OF COVER RESTRAINTS VIOLATED?) LEVEL=LEV(NTH) L1=LL1(NTH) L20L2(474) 60 TO (240,180,220,180),LEVEL 180 DO 200 L=L1,L2 KOXLEIN4+6+NEND+LEND+L KGZL=KGXL=LEND [F (ZUS=SLOPE+(PIP(KGXL)+XNN),GT,PIP(KGZL)+DMIN+0,U1) GN TO 250 210 IF (LEVEL.EG.2) GO TO 240 220 DO 230 L=L1,L2 KGXL=1N4+4+NEND+LEND+L KGXL=1N4+4+NEND+LEND+L KGZL=KGXL=LEND KGZL=KGXL=LEND IF (ZUS=SLOPE+(PIP(KGXL)=XNN),LT,PIP(KGZL)=DMAK) GO TO 260 230 CONTINUE C----SOLUTION IS FEASIBLE SO COST AND COMPARE WITH PREVIOUS CHEAPEST 240 KG2L1=144+4+NEND+L1) KGZLIBINGSGENENDSI KGZLIBINGSGEIT-LI-LZ CALL CORTIT (J,AREA(NTH),DID(KGZLIGI)-ZUS,DID(KGZLZGI)-ZUS, IDIST(NTH),C) C=C+DIT(MHJUHN) IF(C,GT,DIT(MJN)=0,J01) GO YO 250 PIT(MJN)=C NIT(MJN)=(NH=1)+JEND=MEND=(JJ=1)=MEND=M4 NIT(MJN)=(NH=1)+JEND=MEND=(JJ=1)=MEND=M4 OF (MIN)=(NH=1)+JEND=MEND=(JJ=1)=MEND=M4 WRITE (6, 2005) N.J.M.NN.JJ.MM. 417("JN).C 1# (ND.E9.2) IP (ND, E4, 6) NT (EC. 200 C----MOVE ON TO NEXT U/S STATE 250 IF (MM, LT, MEND) 60 TO 170 260 IF (JJ, LT, J) 60 TO 160 IF (NN, LT, VIP(6(DN)) 60 TO 64 C----MOVE 04 TO NEXT D/S STATE NEMM, FALSE, 14 (M) 14 MEND 40 TO 40 NEWN#, FALSE, IF (M, LT, MEND) 60 T0 40 IF(J, LT, JEND) 60 T0 30 IF (N, LT, NEND) 60 T0 20 Return 2002 Format (1N0, 15H0ROUND FROM M/H, 16.64T0 M/H, 514, 2(F7, 5)) 2005 Format (10%, 3H0/S, 315, 10%, 34U/S, 613, F15, 3) END END

END OF SEGMENT, LENGTH 942, NAME NARUN

.

•

.

į

ł

SURROUTINE COSTIT (J, AREA, DUS, DDS, DIST, COST) 0454 DEPTHE(NUS+NOS)/2, D+AREA/DIST GO TO (10,20,30,40,50,60,70) ,J 10 COST=DIST+(2,8+4,1+DEPTH)+30, D+70, 0+DUS 0455 0454 0457 0458 20 COSTEDIST+(5,7+4,1+DEPTH)+50,0+70,0+DUS 0439 RETURN 0460 0461 JO COST +DIST + (8.9+4.1+DEPTH)+30.0+75.0+DUS 0462 RETURN 40 CUST+DIST+(12,3+4,4+BEPTH)+30,0+80,0+8US RETURN 0444 COST#DIST+(15,9+4,7+0EPTH)+30;0+85,1+0US 50 0465 ETURN 0444 40 COSTADIST+ (10.7+5.0+0EPTH)+30.0+90.0+0US 4467 RETURN 0468 0469 70 CUSTODIST+(23,7+5,3+BEPTH)+30,0+95,0+84 RETURN 0470 0471 EuD END OF SEGMENT, LENGTH 136, NAME COSTIT SUBROUTINE TRAIL (NIT, PIT, NIP, PIP, IMAX, JMAX, KMAX, LMAX, MRUN, 1) 04/2 DIMENSION NITCIMAX), DITCIMAX), NIDCEMAX), DIDCEMAX) COMMON/DD/DC203, SMIN, SMAX, DMIN, DMAX, SPMIN, SPMAX, MEND, JEMD, T, MD, RK 0473 COMMON/UNERE/IN(50,4).LN1.LN2.LN3.LP1.LP2.L#5 0475 INT=IN(VRUN,1) NEND=NIP(IN1) 0478 0477 KELN2-1+HEND+JEND+(NEUN-1) 0478 NJNEND=(NEND=1)+JEND+MEND 0479 0480 N=NEND+MEND+JEND DO 60 J=1, JEN0 DO 60 M=1, MEND 0481 0482 KuK+1 C-----NIPENIPERS) IS 187 U/S REFERENCE NUMBER IN TRACE RACK FROM 1/5 4840 C----STATE (J,M) 0485 NUN+1 0484 C-----NET (N) BREFERENCE BACK ACROSS M/H FROM D/S STATE(J.4) 0487 MJNENDEMJNEND+1 C----PIT (MJNEND)*COST OF SOLUTION TO 0/S STATE(J,M) C----ASSUME NO CHANGE OF REFERENCE ACROSS THE M/M 0488 0490 NIT(N) #MJNEND 0491 C----IF CUST OF ARRIVAL AT A HIGHER LEVEL IS CHEAPER ABOPT THIS COST C----AND ALTER REFERENCE IF (N.LE.1) 60 TO 10 IF (PIT(HJNEND), LT, PIT(HJNEND-1)) 60 TO 10 0492 0493 0494 0495 0496 0497 NITENSENITENESS C----IF COST OF ARRIVAL WITH A SMALLER DIAMETER IS CHEADER, ANOPT THIS 0498 C----COST AND ALTER REFERENCE 10 16 (J.LE.1) 60 TC 20 1-MJ4END-MEND 0449 0500 0101 15 (PIT(MJ4E40).LT.PIT(I)) 60 70 20 PIT(MJ4E40)=PIT(I) 5060 0903 0904 ISN-NEND 0505 NETCHINNETCES 20 CONTINUE C--+1ST U/S REF. VO. IN TRACE BACK FROM D/S STATE(J,M) IS REF. ACRISS M/H 0504 0508 11=11+1 30 MAJAAAANIP(LJ) NIP(KJ)=NIT(N) IS(PIT(MJNEND),GT,000000,03 NIB(IJ)=0 C----ESTABLISH THE NBST OF THE TRACE BACK NIP(KJ)=NIT(N) IJ-IJ-T 0909 0511 0312 0513 11=11+1 0514 IF(MAJANA, LT, MENDOJEND) GO TO 40 NIP(IJ)=NIT(MAJANA) 0516 GO TO SO COMPANY AND END POINTS IN ARRAY WIP FOR TRACE MACK FROM DIS 0512 0518 0519 C---- STATE(J,H) 0520 0521 40 JJ=IJ=1 IF (ND,EQ.0) 60 TO 40 IK#NIP(#) 0522 0523 WRITE(0,2003) J.M. PIT(MJNEND) 0525 00 50 [1=14,1] 10H=(NIP(II)=1)/(MEND+JEND)+1 0326 100=(NIP(II)-1+(104-1)+MEND+JENN)/MEND+1 0527 102=41=(11)=(10++1)+MEND+JEND=(100=1)+MEND 0528 50 WAITE (4,2006) 100,100,102 60 CONTINUE 0530 2003 FORMAT (1H0,10X,3HJ= ,13,5X,3HME ,13,10X,19HM/H U/S 5 15X,5HCOSTE,F12,3) 2004 FURMAT (32X,318) 0531 LEVFLA 0533 0534 END 0535 END OF SEGMENT, LENGTH 384, NAME TRAIL

والمادة ومساهم المحمور والمراسون

SUBBOUTINE TRACE (NIT, DIT, NIP, DIP, IMAX, JMAX, KMAX, LMAX) 0536 DIMENSION NIT(IMAX), DIT(JMAX), NID(KMAX), PID(LMAX), JDS(17), MAS(17) COMMON/DD/ D(20), SMIN, SMAX, DMIN, DMAX, SDMIN, SPMAX, MEND, JEND, T, NA, RK COMMON/UMERE/ IN(50,4), LN1, LN2, LN3, LP1, LP2, LP3 0537 0538 0539 C----TOTAL NUMBER OF BRANCHES AND MANHOLES 0540 LOSNIP(2) 0541 NO=NIP(1) 0542 K-IN(LO,1) NEND-NIP(K) 0544 0545 C-----START AND END ELEMENTS FOR FINAL COSTS IN ARRAY FIT 0544 MJOMENDOJEND 0348 11=(NEND=1)+FJ+1 ISSNENDONJ 0549 C-----WHICH IS CHEAPEST ICOST=U COST=99999.8 0550 0551 0552 00 10 101112 17 (017(1),68,COST) 60 70 10 COSTOPIT(1) 0333 0555 0556 0557 ICOST=I 10 CONTINUE TO CONTINUE C----NO FEASIBLE SOLUTION? IF (ICOST,EG.O) STOP C-----IDENTIFY THE DOBASTREAM STATE I=ICOST-I1+1 J=(I-1)/MEND+1 N=I-(J=1)+MEND 0358 0559 0341 0562 0563 0544 NHUNELO 0545 L=0 LLEO 0566 C----- IDENTIFY START AND END ELEMENTS KI, K2 IN ARRAY HIP FOR TRACE BACK 0367 0568 C-----UP BRANCH GIVEN RUN NUMBER .J AND M 15 KmlN2-1+(NRUN+1)+JEND+MEND+(J-1)+MEND+M K1=N1P(K) K2=N1P(K+1)+1 03/0 0371 K=IN(NRUN,1) NENDENIP(K) 0572 0574 N3=0 DO 16 NN#1, NRUN 0575 KalN(NN:1) 16 N34N34N1P(K)=1 0576 C----TRACE BACK ALONG BRANCH NRUN FROM (J,M) DO 20 44K1,K2 0378 0579 0580 1=1+1 NITCLS=NRUN 0582 NA= (N] P (K) = 1) / (FEND+JEND) + 1 0383 L=L+1 0584 NITCLIANA 0383 J4=(N]P(K)=1={NA=1}+#ENN+JEND)/#END+1 0586 L=L+1 NET(L)#JA MA=N1P(K)={NA=1}+MEND+JEND+(JA=1)+MEND L=L+1 0587 0588 9529 NIT(L) WA G----IDENTIFY SOFFIT LEVEL 0590 0391 U592 U593 KK1=IN(48UN,6)+2+4E40-1+44 KK2=KK1+HE40 ZUS=PIP(KK2)+(PIP(KK1)-PIP(KK2))+FLGAT("END=MA)/FLGAT(MEND=1) C----IDENTIFY DIAMETER AND M/H NUMBEP DIAMUS=D(JA) 0996 0395 0396 0597 NHUSBNA C----IDENTIFY CHAINAGE C-----IDENTIFY CNAINAGE KKI=IN(NRUN,6)-1+N4 XUS=PIP(KK1) IF (K.EQ,K1) GO TO 31 C----CALCULATE PIPE SLOPE SLOPE=(7US=205)/(XDS=XUS) C----CALCULATE VELOCITY FROM COLEBROOK=WHITE FORMULA SQASQRT(SLOPE+DIAM) 0399 0600 0401 0402 0603 0404 0405 VEL=-8,818+89+4L0810(8K/3,7/01A4+0,6471E=6/89/01AM) 0404 C----STORE VELOCITY NISHS-NENDOMHUSO 0407 0608 N2=N1=#HUS+##D5=1 0409 0410 00 30 NEN1,42 30 PIT(N)=VEL 0611 C----MOVE ON TO VEXT MANNGLE S1 ZDS=ZUS XDS=KUS 9612 0613 DIAMODIANUS 0415 0414 MHDSOMHUS 20 CONTINUE 0617

:

0418	CNUW AT UPSTREAM END GF RUN
0419	CHOW MANY UBSTREAM RUNST
0440	K=14(NRU4,3)
0441	K=N1P(K)+1
5540	60 TO (70,60,50,40),K
0423	40 LL=LL+1
0424	JOS(LL)=JA
0425	NOS(LL)=MA
0424	SO LLELLET
0427	JOS(LL)=JA
0448	MOS(LL)=NA
0449	60 HAUN-RUN-T
0430	
0631	
0632	60 TO 15
0633	70 IF (HRUN,E9,1) 6C 70 80
0634	
0635	J=JD\$(LL)
0434	M=MDS(LL)
9637	LL#LL#1
0638	60 70 15
0639	80 WRITE (4,1003) CCST
0640	MULLE CH'43043 CHILLETSSTELLER
0641	WAITE (6,1006)
0642	WHITE (A,1002) (PIT(H))HUT FYARACH ARRAYS
(16 4 J	Connell Diagnosticas on stania, Dot stands music
0644	$1 \neq (ND, EQ, U) = NETUNN$
0643	
0646	Weite Charge Charge Charge States and Anna Anna Anna Anna Anna Anna Anna
0647	adite (0,5003) (bit()))aditation
08+8	WAITE (6,5003) (PIP(1))1=(,6=A)
0649	RETURN
0630	1001 FORMAY (10X, SHAUN, 13/3X, SHAVA/15/5X/844/4
0621	1002 FORMAY (10F12.3)
0632	1003 FORMAY EINDYION, SINTHACE WALK OF CHEMPEST SOLUTION FATTHE
0653	
0634	1016 FORMAY (ING//SX, CANNEW FULL FLOW FIPE VECUCITES/)
0633	3005 EQHMAT (5018)
0636	SOJ FURNAT (10E12,5)
0637	END

ENC OF SEGMENT, LENGTH 610, NAME TOACE

.

.

.

ł

٠

. . .

.

1

.

.

.

.

.

.

0658 0659 C-----Y#RETURN PERIOD(Y#S),T#TIHE(MINKS),RIWINTENSITY (#M/HR) 0660 RI#60,/T+((Y#T#702,20)++,28169+2,56) 0661 16 (RI=3), 10,10,40 0662 10 IF(#I) 20,30,30 0663 20 RI=0,0 0664 30 RETURN 0665 40 NI=0 0665 C----ITERATION LOOP FCR HC(LAND FORMULA 0665 0674 0674 0674 0674 0674 0675 0675 0675 07 STOP 0677 00 TO SO 0677 00 TO SO 0677 00 TO SO 0677 00 TO SO 0677 07 STOP 0677 0677 07 STOP 0677 07 STOP 0677 07 STOP 0677 07 STOP 07 STOP 0677 07 STOP 07 STOP

.

•

ENE OF SEGMENT, LENGTH 94, NAME RAIN

•.

• 、

0678 FINISH

- 220 -

. •

APPENDIX D

PROGRAM ASSEMB

.

:

/ L-----FEADS DATA 14 SIMPLIFIED FORM AND OUTPUTS TO CARD PUNCH FILE FUR & C PROGRAMS DPO AND HOD. SUITABLE FOR INTERACTIVE USE + FIMENSION D(20),GL(300),GX(300),DIST(300),KA(300),KB(300),X(300), 0 12(300),GROUND(300),OBLEV(20),OBDIST(20),NBR(10),AREADS(30), 1 27TUPDS(30),ZTUP(300),ZAUT(300),AREA(SUU) - FOO GOODAL - FOO FOO FOO - FOO - FOO - FOO FOO - FOO 10 11 12 15 14
 Ib
 PFAD (S,2001) SIIN.

 15
 IF (SHIN, LE,E) SMIN=0.004

 17
 SMAX=0.1

 18
 URITE (G.1002)

 19
 READ (S,2001) VLLHIN

 20
 IF (VELIIN, LE,E) VELMIN=0.7

 21
 URITE (G.1003)

 22
 READ (S,2001) VLLHIN

 23
 IF (VELIIN, LE,E) VELMAX=6.D

 24
 IWITE (G.1003)

 25
 READ (S,2001) DHIN

 26
 IF (OMIN, LE,E) OMIN=1.0

 27
 URITE (G.1005)

 28
 READ (S,2001) DHIN

 26
 IF (OMIN, LE,E) DMIN=1.0

 27
 URITE (G.1005)

 28
 READ (S,2001) DHIN

 29
 IF (DMAX, LE,E) DMAX=4.0

 30
 URITE (G.1006)

 31
 PEAD (S,2001) TIME

 32
 IF (TIME, LE,E) TIME=2.0

 33
 IF (RK, LE,E) MMAN=4.0

 34
 PEAD (S,2001) SPMIN

 35
 IF (RK, LE,E) TIME=2.0

 36
 URITE (G.1008)

 37
 READ (S,2001) SPMAX

 38
 IF (SPMAX, LE,E) SPMIN=3 1.5 16 50 10 CONTINUE 60 TO 30 51 52 20 119=7

 53
 f(1)=,150

 54
 f(2)=,225

 55
 D(3)=,300
 / ...

 56
 D(4)=,375

 57
 r(5)=,450

 58
 D(6)=,525

 59
 D(7)=,600

 60
 C====READ IN PROGRAM CONTROL VALUES

 61
 JO 'FRITE (0,1012)

 62
 READ (5,2001) D2

 63
 IF (D2,LE,E) D2=0.5

 64
 'FRITE (0,1013)

 65
 FEAD (5,2002) HEND

 66
 IF (MEND,EQ,0) HEND

 t(1)=,150 53

 64
 4'RITE (G,1013)

 65
 PEAD (S,2002) HEND

 66
 1f (HEND,EQ,0) HEND=S

 67
 URITE (G,1014)

 68
 PEAD (S,2002) JEND

 69
 1f (JEND,EQ,0) JEND=4

 70
 URITE (G,1015)

 71
 PEAD (S,2002) JEND

 72
 URITE (G,1016)

 73
 PEAD (S,2001) SPMM

 74
 If (SPMN,LE,E) SPHN=10.0

 75
 CP=P==-READ IN ND, OF DRANCHES IN DESIGN PROBLEM

 76
 URITE (G,1017)

 77
 PEAD (S,2002) LU

 78
 C===-URITE OUT DATA TO FURHATTED FILE

 79
 URITE (1,3002) (D(1),I=1,NP)

 80
 URITE (1,3002) KK

 81
 URITE (1,3002) SHIN,SMAX

 84
 URITE (1,3002) SHIN,SMAX

 85
 URITE (1,3002) SHIN,SMAX

 84
 URITE (1,3003) LD

 85
 URITE (1,3003) LD

 86
 DISTOTEU,O

 87
 FWT0T=0

 89
 FWT0T=0

 #9 KNT0T=0 90 KLTOT=0
 VI
 CF-F--PEAD IN DATA FOR EACH BRANCH

 92
 NO 290 [=1, [0]

 93
 URITE (G, 1018) I

 94
 FEAD (S, 2003) ITYPE, Y, DU

 95
 CF-F---FREAD IN GROUND LEVEL DATA
 96 97 98 J=0 40 J=J+1 +'RITE (0,1019) PEAD (5,2001) GL(J) URITE (6,1020) PEAD (5,2001) GX(J) SF(GX(J),LT,Y=0.1) 99 100 101 GO TO 40 104

۰.

١

ς.

٠

.

、 ·

١

* +

• •

.

- 222 -

TUS CHARTE POSITIONS OF HANHOLES TUS IF (ITYPE.E4.D) GO TO 70 TUS 60 DIST(T)=0.0 PIST(2)=V FN=2 100 107 RN=2 108 GG TO 90 109 G====15 LENGTH LESS THAN TWICE HINIM M SPACE467 110 70 15 (Y,LT,2,0+59).1N=0,13 GD TO 60 111 DIST(1)=0,0 AIST(2)=AHAX7(SPMIN,SPMH) FA=3+1NT((Y+DIST(2)=SPMIN+0,13/SPHH) 142 113 TO 80 K=5,NN DJST(K)=DJST(K=1)+8PMH 114 115 IF (K.EQ.NN) DIST(K)=Y BD CONTINUE 116 117 BD CONTINUE 118 CP-P-FENERATE PERMISSIBLE MANHOLE CONNECTIONS #A(1)=1 119 120 KB(1)=1 NO 340 H=2,NH NO 300 L=1,H If (Dist(H)+Dist(L),Gt,SPMAX+0.1) 60 to 300 121 122 123 TA(H)=L CD TO 310 3LD CONTINUE 310 DO 320 L=1,H IF (DIST(H)=DIST(L),LT,SPMIN=0,1) 60 TO 330 124 120 127 128 129 320 CONTINUE 330 KB(H)=L=1 340 CONTINUE 130 131 132 200 -CALCULATE GROUND LEVELS 40 GROUND(1)=GL(1) 133 TECHNALT.3) 50 TO 130 Ktenna to 120 K=2,KK to 100 H=1,J IF (6x(1),6T,01ST(K)+0,U1) GO TO 110 FOUND 154 155 137 158 139 100 CONTINUE 110 FROUND(K)=GL(M=1)+(GL(H)=GL(M=1))/(GX(M)=6X(M=1))+ 141 1(DIST(K)=6X(M=1)) 142 420 FONTINUE 145 130 REOUND(NN)=GL(J) 144 KK=1 KNui 146 KL=0 147 140 KL-KL+1 TF (DIST(KM)-GX(KN),GT,U,01) GO TO 165 X(KL)=DIST(KM) Z(KL)=GROUND(KM) IF (KM,EQ,NN) GU TO 166 JF (DIST(KM)=GX(KN),GT,=0,01) KN=K4+1 KM=KM+1 148 149 150 151 152 155 CD TO 140 145 X(RL)=GX(KN) X(KL)=GL(KN) 154 155 156 157 Fh=KN+1 GD TO 140 ----FEAD IN DETAILS OF GESTRUCTIONS 446 NOB=D 450 NOB=NDB+1 158 159 Tr--160 161 POB=NDB=T WRITE (6,1021) PEAD (5,2001) ORLEV(NOB) IF (OBLEV(NOB).LE.=998,9) &0 TO 169 URITE (6,1020) READ (5,2001) OBDIST(NOG) GO TO 150 POD=NOD=1 162 163 164 165 156 167 168 160 008=N08=1 169 COMPONENT IN UPSTREAM CONNECTIONS 170 171 170 NUB=NUB+1 URITE (6,1022) READ (5,2002) NBR(NUB) IF (NBR(NUB),EQ.0) GO TU 180 GD TO 170 180 NUB-NUB-1 172 1/3 176 176 1/2 CP--P-IDENTIFY TOP OF ZONE AT UPSTREAH END UP RUN 1/8 ZTOPUS=GRUUND(1)=DMIN 1/9 AREAUS=0.0 AREAUSED, O IF (NUB, EQ.U) GO TO 200 DO 190 JJ=1, NUB K=NBR(JJ) APEAUS=AREAUS+AREADS(K) 180 181 182 183 IF (2TOPDS(K). LT. 2TUPUS) ZTOPUS=ZTOPDS(K) 144 190 CONTINUE 185

.

.

ς.

.

.

•

.

ĸ

.

ς.

.

.

•

•

ς.

.

ς.

in the second se	
186	CPCEFINE TOP OF ZONE ALONG THE BRANCH
187	200 ZIN=ZTOPUS
188	Zn#=9999999,9
189	ZGL=9994499,9
190	CO 270 JJP1, NN
191	ZOUT=GROUND(JJ)=DMIN
192	ZTOP(JJ)=AMIN1(ZIN,ZOB,ZGL,ZOUT)
193	JF (JJ,EQ,NN) GO TO 270
194	214=270P(JJ)-S41N+(D197(JJ+7)+D197(JJ))
195	1F(NOB,E9.0) GO TO 220
196	PO 210 K=1,NOB
197	IF (DBDIST(K)_ST_DIST(JJ).AND.OBDIST(K).LT.DIST(JJ+1)) GU TU 290
198	210 CONTINUE
199	220 205-999999.9
200	60 TO 240
201	230 208=08LEV(K)-6H3H+(DIST(JJ+1)=08DIST(K))
202	240 DO 250 K=1,J
205	IF (GX(K), GT, DIST(JJ), AND, GX(K), LT, DIST(JJ+1)) 60 TO 260
204	250 CONTINUE
205	26L=999999,9
206	60 TO 270
207	Z60 ZGL=GL(K)=SHIN+(DIST(JJ+1)=GX(K))
208	270 CONTINUE
209	ZTOPDS(I)=ZTOP(NN)
210	IF ((ZTUPUS-ZTOPDS(I))/Y.LT.SMAX-0,00001) 60 TO 275
211	ZTOPUS#ZTUPD\${}}+(\$11AX#0,00002)+Y
212	60 YO 200
213	275 DO 280 JJ=9,NN
214	ZBOT{JJ}=ZTOP{JJ}=DZ
215	ZBD AREA(JJ)=AREAUS+DIST(JJ)=DW
216	AREADS(\$)=AREA(NN)
217	MNTOTENNTOTENNET
218	<07#XL707+XL#1
219	D18707=D18707=V
220	CP-P+PWRITE DATA TO FILE
221	WRITE (1,3003) HN
222	WRITE (1.3005) (DIST(K) K=1, NN)
223	VRITE (1,3006) (AREA(K),K=1,HN)
224	WRITE (1,3002) (270P(K)/K=1/NN)
225	WRITE (1,3002) (2807(K),K=1,NN)
226	VRITE (1,3003) (KA(K),K=1,NN)
227	VRITE (1,3003) (KB(K)/K=1/NN)
228	VRITE (1.3003) KL
279	VHITE (1.3002) (7(K).KH1.KL)
230	UNITE (1.3005) (X(K).K=1.KL)
231	
2.52	
233	UPITE (1.5003) (NRB(K),KD1,NUB)
234	
235	CrewerPROBLEM S12r
236	NOMHAVELO+INT(DISTOT/50.0)+1
237	
238	KLTOT=KLTUT+1
239	WRITE (1,3003) NNTOT,KLTOT,NOMWAY
240	URITE (6.1023)
241	STOP
242	1001 FORMAT (10X,36H++++FOR DEFAULT VALUE INPUT ZERO+++/5X+
243	447HHINI(UM GRADIENTE)
244	1002 FURMAT (5X,17HMINIMUM VELOCITY#)
245	1003 FORMAT (5X,17HMAXIMUM VELOCITY=)
246	1004 FORMAT (5X,14HMINIMUM COVER=)
261	1005 FURNAT (SX,14HNAXINUM COVER=)
248	1006 FURHAT (SX,20HTIME OF ENTRY(MINS)=)
249	1007 FURHAT (\$X,15HPIPE ROUGHNESS=)
250	TCOB FORHAT (5X,24HMINIMUM HANHOLE \$PACING=)
251	1009 FORMAT (SX,24HMAXIMUM MANHOLE SPACING#)
252	1010 FORMAT (5X,61NFUR LIBRARY PIPE SIZES ENTER ZEROLOTHERWISE ENTER LO
255	1.0F PIPES)
254	TOTT FORMAT (SX+14HPIPE SIZE(MH)=)
255	1012 FURHAT (5x,14HDEPTH OF ZONE=)
256	1013 FORMAT (5X,17HNUMBER OF LEVELS=)
257	1016 FORMAT (SX,16HNUMBER OF PIPES=)
658	1015 FORMAT (5X,18HDIAGNOSTICS LEVEL=)
259	1016 FORMAT (5X,20HSPACING OF POSSIBLE MANHOLESE)
260	1017 FORHAT (5X/19HNUMBER OF BRANCHESB)
261	1018 FORMAT (5%,10HBRANCH NO.,16,5%,43HBNTER TYPE(U UN 1)/LENGTH AND DH
262	TAINED WIDTH
263	1019 FORMAT (5%,18HENTER GROUND LEVEL)
264	1020 FORMAT (5X, JOHENTER DISTANCE FROM UPSTREAM MANHOLE)
602	1721 FORMAT (SX,42HENTER OUSTRUCTION LEVEL(=949 TO TERMINATE))
266	1022 FORMAT (5X,48HENTER UPSTREAM BRANCH NUMBER (ZERO TO TERMINATE))
267	1023 FORMAT (5X,10(1H+),18HEXECUTION FINISHED,10(1H+))
668	2001 FORMAT (F0.0)
209	2002 FURNAT (10)
210	2003 FORMAT(10,2F0.0)
271	3001 FORHAT (216)
272	3042 FURHAT (1058,3)
275	3003 FORHAT (1615)
274	3004 FURHAT (1048.0)
275	5005 FORHAT (1088.1)
276	END

APPENDIX E

PROGRAM MOD

1 CURPENT EPITION124.6.76, LARGE SIZE VEASION 2 CONMON/AP/D(20), GMIN(20), GMAX(20), DMIN, NMAX, MEND, JEND, T, ND, RK, RP 3 COMMON/UMERF/IN(50,4), LN1, LN2, LN3, LP1, LP2, LP3 4 DIMENTION NIP(20000), PIP(10000), NIT(13000), PIT(13000) S LOGICAL DE • C-----SPECIFY MAXEMUM AFRAY SIZES IMAX=13000 7 JHAX#13000 KHAY=20100 8 10 11 C----REAR DERIGN PARAMETERS 42 CALL DATA1 43 C----REAR SYRTEM GEOMETRY AND STORE IN PERMANENT ARRAYS CALL DATA2(NIT,PIT,NIP,PIP,IMAX,JMAX,KMAX,LMAX) LOBHIP(2) •4 NOWNIP(1) 16 17 C----SET INITIAL FLOW VALUES TO ZERO 40 1 PIP(1)=0.0 NC0475=0 1J=1N1=1 20 21 22 C----PROBUCE & MINIMUM GRADIFNT DESIGN 23 CALL MGRAD (NIP,PIP,KMAX,LMAX) 24 C----PRODUCE OPTIMUM DESIGN RASED ON MINIMUM GRADIENT FLOWS DO PO JEN,LO CALL CONB (NIT,PIT,NIP,PIP,KMAX,LMAX,IMAX,JMAX,I,HC(STS) CALL CONB (NIT,PIT,NIP,PIP,IMAX,JMAX,KMAX,LMAX,I, CALL NBRUN (NIT,PIT,NIP,PIP,IMAX,JMAX,KMAX,LMAX,I) CALL TRAIL (NIT,PIT,NIP,PIP,IMAX,JMAX,KMAX,LMAX,I) 25 26 27 28 28 CALL THAIL CHIT, DIT, NIP, PIP, IMAX, JMAX, LMAX, L

 SUBEOUTINE DATA*

 10
 COMMON/DF/C(2075GMIN(20),GMAX(20),DMIN,DMAX,MEND,JEND,T,ND,

 11
 C-----REAR IN NUMBER OF VERTICAL ZONES AND PIPE CHUICES

 12
 REAR (S,1002) HEND,JEND

 13
 REAR (S,2003) HEND,JEND

 14
 REAR (S,2003) HEND,JEND

 15
 WEITP (G,2003) HEND,JEND

 16
 C-----REAR IN NUMBER OF PIPE SIZES AVAILABLE AND THEIB DIAMETERS

 16
 REAR (S,1001) NP,(D(1),I=1,MP)

 17
 C-----REAR IN NUMBER OF PIPE SIZES AVAILABLE AND THEIB DIAMETERS

 18
 REAR (S,1001) NP,(D(1),I=1,MP)

 19
 WEITP (G,2004) NP,(D(1),I=1,MP)

 10
 METTE (G,2004) RESE IN MM,

 13
 C------REAR IN BIPE ROUGHNESS IN MM,

 14
 REAR (S,1001) THE

 15
 WEITPE(G,2004) RE

 16
 REAR (S,1001) TIME

 17
 C------REAR IN WIN AND MAX BIPE SLOPES

 18
 REAN (S,1001) SUIN,SMAX

 19
 C------FIX MIN AND MAX GRADJENTS FOR EACH PIPE SIZE

 10
 GMIN,SMAX

 11
 MIN AND MAX GRADJENTS FOR EACH PIPE SIZE

 10
 GMIN AND MAX BEACH OF COUST

</tabul> ٩5 SUBPOUTINE DATAS CONMON/ NOT DECOT GHIN (20) , GMAX (20) , D'IN, THAX, MENT, JEUD, T. NN, RK, RD 16 7 2

 56
 GMIH(1)=SMIH

 57
 10
 GMAY(1)=SMAX

 58
 C====RFAN IN MIN AND MAX DEPTH OF COVER

 59
 PFAN (5,1001) DHIN, DMAX

 60
 WEIYE (6,2002) DHIN, DMAX

 61
 WEIYE (6,2003) DHIN, DMAX

 62
 REAN (5,1001) SPMIN, SPMAX

 63
 WEIYE (6,2003) SPMIN, SPMAX

 64
 C====RFAN IN DIAGNOSTICS LEVEL

 65
 REAN (5,1002) ND

 64
 RETURN

 RETURN 66 1000 FORMAT (16,2(/1058,3)) 1001 FORMAT (258,3) 1002 FORMAT (258,3) 2000 FORMAT (216) 2000 FORMAT (31,16HPIPE DIAMETERS,219,1058,3) 47 £ 8 49 70 2000 FORWAT (\$X,16HPIPE DIAMETERS,21%,10F8,3) 2001 FORMAT (\$X,23H'IN AND MAX PIPE SLOVES,12X,2F8,3) 2002 FORMAT (\$X,71HPIN AND MAX COVER,14X,2F8,3) 2003 FORMAT (\$X,73HIIN AND MAX MAN SPACING,2F8,3) 2006 FORMAT (\$X,73HIIN AND MAX MAN SPACING,2F8,3) 2005 FORMAT (\$X,73HIIN AND MAX MAN SPACING,2F8,3) 2006 FORMAT (\$X,75HIN AND MAX MAN SPACING,2F8,3) 2006 FORMAT (\$X,75HIN AND MAX MAN SPACING,2F8,3) 2006 FORMAT (\$X,75HPIPE ROUGHNESSM,F4,3,6H MM,) 71 72 73 75 77 END . .

....

	GINGONTATUS DATAD (NETOPETORESPECTORESPAK)
70	Commesting out the CALLS GEOM AND SORTS OUT ADDRESSES FOR PERM, ARRAYS
- 10	COMMONT NOT D(20) + SHIN(20) + SHAK(20) + DHIN THAR + HEND JEND T + NP+ * ++*
	COMMON/ UNFRE/IN (50,4), LN1, LN2, LN3, LP1, LP2, LP3
	NINENSION NIT(IMAX), PIT(JMA4), NIP(K"AX)+PIP(LMAX)
	CALL GEON (HIP, PIP, KMAX, LMAX)
	10=+10 (2)
	C
	NTQT=0
87	
	NT0T=0 ·
	00 10 1=1.Ln
	[4{2,1}#2+2+4T07+LT07+3+1
71	14(7,6)=1+6++707+2+4707
· ? 2	J # 2 K (Z + 9)
~ 3	NT0T#NT0T+NIP(J) -
94	14(1,2)=3+2+HIOT+LTOT+3+1
- 5	IN(1,3)=IN(1,2)+1
76	J=1+(1, 1)
27	LTOTALTOTANIP(J)
78	Jain (195)
79	HT0784T074N1PCJ3
1-0	18 CON44NUE
121	LN1=2+2+NT07+LT07+J+(L0+1)
1.15	rusarujonindelēkienik(4)
1.13	FA3#FNS+NEND+1END+1E
174	LP1=1+4+NT07+2+HT07
1.12	LDS=LD1+HFND+JEND+S
1.26	LP3uLP2+N1P(1)+1
1.77	IF (ND, FO, US RETURN
178	M#14E (4'1000)
1 19	MAIAE (V'SODD) ((IN(I'Y)'TEL'Y')'TEL'Y'TEL
110	waive (6,2001) LN1,LN2,LN3,LP1,LP2,LP3
444	RETIEN
115	TOND FORMAT (140/102,34HADDRESSES STORED IN ARRAY INCLUSE)
113	2000 FURMAT (4110)
446	2001 FORMAT (140/5X,44L418,14,5X,64L428,14,74,44L434,14,54,64L4)
115	15x,4HLPP=,14,5X,4HLP3=,14)
1 4	END

•

•

SUBPOUTINE GEUM (N19, PIP, KMAX, LMAX) ECHMON/R0/D(20), SMIN(20), SMAX(20), DHIN, DMAX, MEND, JEHD, T, ND, RK, RD 117 4*8 140 DIMENSION NIP(KPAX), DIMENSION NIP(KPAX), DIMENSION NIP(KPAX), DIMENSION NIP(KPAX), DIMENSION, 123 K=4 124 1=0 176 LOWIP(7) 177 C----REAR IN DATA FOR EACH BRANGN 128 DO 40 J=1+LO K=K+1 READ (5,1000) NN 129 130 131 NIP(KJANN 132 C----REAN IN CHAINAGESIAREASITUP AND BOTTUP ZONE LEVELS DD 10 M=1,4 Lul41 Nul4NN#1 #E48(5,2000) (#1#((),1=L,4) 133 134 135 136 137 10 CONTINUE 138 140 CO-PO-BEAN IN STRET AND LAST MANHOLES 140 DO 20 Mai;2 141 K=K+1 942 943 966 -----READ (5,1000) (NIP(1),144,N) 149 NIP(K)=NG Dn 10 H=1.2 151 152 153 L=L+1 NUL+NG=1 READ (5,2000) (PIP(1),1=L+N) 154 LON 155 30 CONTINUE 186 Componertan in Connections Upstream 157 158 149 K=K+1 READ (5,1000) NR NIP(K)=NN IF (NN,EO.O) GO TO 40 K=K+1 N=K+NN=1 160 949 142 163 READ (5,1000) (NIP(1),1=4,8)

 163
 WEAR (3,1000) (NIP(1),1=K,W)

 144
 CONTINUE

 145
 40 CONTINUE

 144
 RE49 (3,1000) NIP(1),NIP(3),NIP(4)

 147
 C----PBINT OUT DATA AS \$TORED

 148
 IF (ND,E0.0) RETUR(

 149
 WBITE (4,3000) (NIP(1),1=1,K)

 170
 WBITE (6,4000) (DIS(1),1=1,K)

 171
 WBITE (6,4000) (DIS(1),1=1,L)

 172
 RETURN

 144 171 MELT 172 RETURN 173 1000 FORMAT (1415) 174 2000 FORMAT (10FA,3) 175 3000 FORMAT (2015) 176 4000 FORMAT (10F10"3) 177 5000 FORMAT (10F10"3) 177 5000 FORMAT (1M0/10X/45HDATA STORED IN ARRAYS NIP AND PIP AS FOLLOWS1) 178 END

SIDEOUTINE CUMB (LIT, PIT, NIP, PIP, KIAA, LMAY, IMAY, IMAY, I, COSTS) COMPON/DE/DE20), GMIN(20), GMAR(20), D'IN, DMAX, MEND, JEHD, T, ND, RA, RE COMMON/WHERE/IN(50, 6), LN1, LN2, LN3, LP1, LP2, LP3 179 1-0 DIMENSION NIPCKMAX), PIPCLMAX), NITCIMAX), PITCJMAX) 182 123 MJEMENDAJENO 146 Gauna HOM MANY U/8 PEPEST KK#1411.53 N#41P(K#3 1 .5 176 NEWTP(KW) 977 C----SET COST OF ARRIVAL AT 1ST MANHOLE IN RUN TO ZERO 178 DO 10 J=1+MJ 179 18 0170J=0 170 IF (I[0]]) GO TO 27 171 C----STORE D/S COSTS FUR LAST PUN IN NEXT SECTION DF D/S COSTS ARRAY 172 KEIN(1-1,1) 173 ENDERLING 126 NENNUNSP(K) J1#(NFND=1)+MJ+1 903 904 975 . . J2=NEND=MJ NCOCTS=NCOSTS+1 K=LP1+(NCOSTS=1}=MJ=1 106 107 178 K1##+1 100 51, PLEL 04 00 200 K#K+1 200 K4K4 201 20 PIP(K)=PIT(J) 202 WRITE (4,1001) K1,K 203 WRITE (4,1002) (PIP(IK),IK=K1,K) 204 21 IF (N,NE,0) GO TO 25 203 C=====DEFINE INFEASIBLE PIPE ZONES AT UPSTREAM ENDS OF METWORK 206 IF (JEND.EQ.1) GO TO 70 207 J2#J4ND=1 276 K#0 0 22 J#1, J2 D0 22 Me1, MEND 278 244 210 211 55 bil(4)=00000.0 2"5 GN TO 70 2"4 C-----OVE, THO OR THREE U/S PIPES 2*3 C----FIND D/4 TOP OF ZUNE 2*6 25 K=IN(1,1) 2*7 NENDDS=NIP(K) L=IN(I,4)+2+NEN0DS 27D9=PI=(L) 218 2.9 220 C-----FIND D/S MAX DIAMETER 271 JI=1-1 222 K=LN3 273 DO 30 L=1.11 271 272 273 276 275 LL#IN(L,1) 30 K#K+NIP(LL) 226 NDDSaN1+(*) 227 DO 40 LV01,N 278 C----FIND NO OF U/S RUN 229 K=KK+H=LK+1 230 NETUSELS 231 CHARTERS AND BOTTOM OF 2048 K#[V(NR,1) NENNUSEN[P(K) 272 214 215 L=1+(NR,4)+3+NE47U3-1 ZTUR=P1+(L) 236 LUL+NFNDUS 277 ZBUREPIP(L) 278 C----FIND U/S HAX DIAHETER 219 K=L=3 00 60 L=1,N= 240 LL=IN(L,1) 40 K=K+NIP(LL) 241 242 243 NDURENTP(K+1) 244 Comparing Correspondence Betheen U/B and D/B states DO 41 M=2, MEND 2115=27US=FLAT(H=1)=(2TUS=2RUS)/FLOAT(MEND=1) 15 (2US, LT, 2TDS=0,001) 60 TO 42 245 246 247 248 41 CONTINUE HEMEND+1 250 42 #1=#=2 DO 43 J=9, JEND NUS=NDUS=JEND+J TF (NUS_E9, NDDS=JEND+1) 60 TO 44 251

272 NUSENDUS-JEND+J 273 IF (NUS.EQ.VDDS-JEND 274 43 CONTINUF 275 J-JEND 276 44 J1#J=1 277 K=0 278 LREFELN2+MJ+(NR=1)=1

.....

. .

...

.

 257
 K=0

 258
 LREF#LN2+MJ+(NR=1)=

 249
 D0.50
 J=1.JFND

 240
 JUSamtRAC(JEN0, J=J)
 J

 241
 D0.50
 H=1, HEND

 242
 K=441
 Z

 243
 MUSamtND (MEND, H=1)

```
2% C=====DEFINF FORT OF U/S STATF
2% L=L01+Ni+(NCOSTS-T)+(JU%=1)+MEND+MU%=1
2% C=====ADD COST AT STAFT OF 0/% RUN
2% D====ADD COST AT STAFT OF 0/% RUN
2% D====ADD COST AT STAFT OF 0/% RUN
2% C=====IDENTIFY TREEE BACK TOF 0/% RUN
2% C=====IDENTIFY TREEE BACK TEST REWEE ACROSS MANNOLE FROM STATE (JUS,MU%3),
2% C=====ADD ASSIGN THIS REFFRENCE TO STATE (J,M)
2% C=====ADD ASSIGN THIS REFFRENCE TO STATE (J,M)
2% C=====ADD ASSIGN THIS REFFRENCE TO STATE (J,M)
2% NIP(LNET)=NIP(L)
2% NIP(LNET)=NIP(L)
2% ADD ASSIGN THICLS
2% FOUNTE (A,1002) (PIT(C),K=1,MJ)
2% AETIGN
2% AETIGN
2% OF SENTAT (IN0/JSX,37HD/S CCSTS,L4ST RUN,STORED IN PIP FROM,I6,3M TO,
1% IIA)
2% IIAD FORMAT (IN0/JSX,37HD/S COSTS FOR PUN,16,19M (NO, OF U/S PIPES=,16,
1% END
```

۰,

.

;

SUBROUTINE NERUE (NTT, PIT, NIP, PIP, 144X, JMAX, KHAX, LHAX, NOUS) SUBHULTINF NBRUP (NIT/PIT/NIP/PIP/L'AA/JMAA/KUAX/LMAA/NUU/J DIMENSION NIT(IMAX), PIT(JMAX), NIP(K'AX), PIP(LMAX), EFV(20), 1LL1(20), LL2(2U), JIST(20), ARE4(2D) COMMON/PMERF/IN(SO,6), LN1, LN2, LN3, LP1, LP2, LP3 COMMON/PMERF/IN(SO,6), LN1, LN2, LN3, LP1, LP2, LP3 COMMON/PMERF/IN(SO,6), LN1, LN2, LN3, LP1, LP2, LP3 215 27e 2.7 289 INSTERL NEWN 200 271 znż INZaINENAIN,2) 273 274 275 tadetacantiny 33 [N4=14(+=UN, 43=1 LENN=41=11N23 276 NENNUNIFILMIS 277 C----DEFINE 9/S STATE N#1 20 N#N+1 208 200 310 C--P-DEFINE PARAMETERS DEPENDENT ON "N" NEWNE, TRUE. 311 312 ********** 374 KLANDKLAN+NEND 305 377 KZENTEKZTAPONEND 2708#FI#(K2TOP) 2807#FI#(K28UT) 318 . 319 KONEO 3.4 0 311 NBON QUN-1 3.2 10 KIW42KOAPTASK)-1 2 KAMACANPHIS(K)-1 2 Liat'HU 2 Liat'HU 315 314 315 316 . KONAKONALP2AN-2 317 318 j∎0 319 320 321 30 J=J+1 J=J=Q=NT#(KJMG)=JENC+J M=0 322 40 MaN+4 MJNG((N-1)=JEND="END+(J=1)="END+") PIT(MJN) =000000,0
 JCW
 PITTMIN3
 B000000,0

 375
 IF
 (JPIPE,LT,1) GO TU 270

 376
 Enseztop-sinat(H=1)+(ZTCP-ZBOT)/siJAT(HEND=1)

 377
 C====0EFINE

 378
 NN=NIP(VLAN)=1

 379
 L4
 370 370 371 372 373 44 NNEVN+1 KXNNETN4+NN ١ ZT#PIP(CZT) KZ##KZT+NEND 334 339 347 341 IF (PIPCKAKL)"GT. #I#(#XLA)+0.01) GD TD 60 60 [1#[KGX[1#KAKL KGZL1#KGXL1#LFND 342 343 344 345 12=11=1 316 A=0.0 LEVPL=1 IF (L1,F0,LFND) 60 TU 140 IF TDTP(KGXL1+1),GT,XN+0,013 60 TO 140 347 349 L301707 D0 70 L=L3,LEND KgXL01N44404END4LEND4L IF (PIP(KgXL),GT,XN=0,01) 60 70 80 70 Continue 351 352 353 356 355 A0 L7=L=1 K62L2=1N4+4+NEND+L2 316
 336
 K#CL2=K#CZL2=LEND

 377
 KGXL2=K#ZL2=LEND

 378
 C====DEFINE AFFA OF LONG SECTION ABOVE STRAIGHT LINE

 359
 A===1P(vdXL1=1)+(PIP(K6ZL1)=PIP(K6ZL2+1))+0,5

 360
 DU A5

 371
 KGZLL=IN4+4=NEN1+LL
 359 340 KGXLL#K4ZLL#LEND 85 A#A#PJP(KGXLL)+(PJP(KGZLL+1)-PIP(KGZLL+1))+0.5 142 343 344 A#A=#1#(#GXL2+1)+(#1#(#G2L1-1)=#1#(#G2L2))+0.5

.

...

```
115 C----IS ADAUND LEVEL, CUNCAVE, CONVEX OR VARIABLE

356 GELOPE#(PIP(KG2L2+1)+PIP(KG2L1+1))/(XH+XNN)
356
                   00 130 1011.12
348
                    KGZL#IN4+4+HEND+L
                  &GAL #KGPL+LFND
#SLMPF#CPIP(KG2L)=PIP(KG2L1=1))/(PIP(KGXL)=XNN)
349
            1# 194LAPE, LT GELOPE-0, COOD1) 6A TO 90

1# (#8LAPE, LT GELOPE-0, COOD1) 6A TO 90

1# (#8LAPE, LT GELOPE-0, TOOO1) 5A TO 130

6A TO (110, 100, 130), LEVFL

00 60 TO (120, 140, 100), LEVEL
371
372
1 7 1
374
           100 LEVFL=4
60 70 140
375
376
377
378
           110 LEVEL=3
                   60 10 130
           120 LEVEL=2
130 CONTINUE
379
310
341 C----DEFINE GROUND CONVITIONS AND DISTANCE OFTWEEN MANHOLES
           160 NTHENNENSPEKLAND+1
312
313
                   LEVENTHIELEVEL
                   LL1 (NTH)=L1
LL2 (NTH)=L2
315
316
                   DISTENTHINKHTANH
317 AREA (NTH)AA
317 AREA (NTH)AA
318 IF (ND; GT.O.) WRITE (G,2002) N,NN,NTH,LEVEL,L1,L2,DIST(NTH)AA
319 C----DEFINE UDITPEAN STATE (U/S AIPE DIAMETER AND CROWN LEVEL)
310 150 NTHENNEN(EKLAN)+1
311
           JJ#A
160 JJ=JJ+1
NM#A
372
373
374
         170. HManuel
375 MMJJNN&{NN=1}=J#ND=HEND=(JJ=1]=MEND=MM
376 C====CHECK #F451HILITY D# SOLUTION
377 C-----(U/S STATE FEASIBLE?)
378 JF (PIT(MMJJNN), GT, 99
378 IF (PIT(MMJJNN), GT, 999999, 0) GO TO 250

370 C----(PIPE SLOPE WITHIN #557#41#157)

470 20582705LOAT(MM-1)+(27-7H)/FLOAT(ME40-1)
411 SLOPE=(7US-TOS)/DIST(NTH)

612 IF (SLOPE,LT,GHIA(JPJPE)=0,30001) G1 T0 260

413 IF (SLOPE,GT,GMAX(JPJPE)=0,30001) G1 T0 250

414 G=====(PIPE CAPACTIY SUFFICIE4T7)
                 CALL VELOC (SLOPE, DIJPIPE), *K, VFL, AFULL)
415
616 IF (GFULL LT, PIP(KQN)) GO TO 250
617 C----CHEPTH OF COVER RESTRAINTS VIOLATED7)
618 LEVFLELPV(NTH)
                                                                                              · /
419
                   L1=L1(NTH)
           L2=L12(474)
G7 T0 (240,180,220,180),LEVFL
180 D7 200 L=L1,L2
KGXL=IN6+6+UEND+LEND+L
610
411
412
4*3
414
4*5
                   ##21##6#1-1#ND
1# f2U$=$10#E+(#I#(#6#1)=#N#);6T.#1#(#6£1)=8#I#40.01) 60 70 250
416
           200 CONTINUE
           210 1# (LEVEL.En.2) 60 TO 240
270 DN 730 Lul1.L2
KGKLWIN4+6+5END+LEN7+L
418
419
470
671
479 KARLDINASGARENASENASE
470 KARLDINASGARENASENASE
571 IF KUSSELOSE+(PIPKKKRL)=XNNJ%LT,PIPKKGRL)=MMAX) GO TO P60
472 230 Continue
473 C=====SOLUTION 15 FEASIBLE SO COST AND COMPARE WITH PREVIOUS CHEAPEST
424
           260 KGZL1=1×6+4+NEND+L1
                   KG2L2=KG2L3=L3=L2
CALL CUGTIT (JPIPE,AREA(NTH),PIP(KG7L1=1)=ZUS,PIP(KG2L2+1)=ZDS,
475
                 10157(NTH),C)
C#C+PIT(MMJJNN)
478
479
                    IF(P.GT. #IT(MUN)=0,001) GO TO 250
410
                   PITEMJNSEC
                   NJ 7 (HJN9 = (NN=9 3=JEND=HEND=LJJ=9 3=HEND=HH
477
474
           260 KJJHG=KJHG=N+NN

        446
        260
        RJJMGHELING-NONN

        475
        JJPIPENIP(KJJMG)-JENDJJ

        476
        1F
        CJJPIPE, LTJPIPE, AND, JJ, LT, JEND)
        GO
        TO
        160

        477
        1F
        (NN, LT, NIP(KLHN))
        GO
        TO
        44

        478
        C-----MOVP
        ON
        TO
        HEXT
        D/S
        STATE

           NEWNE FALSE.
270 IF (M.LT.MEND) GD TO 40
IF(J.LT.JEND) GD TO 30
IF (N.LT.YEND) GD TO 30
479
460
441
442
41.3
                    2002 FORMAT (10X,15HGRCUND FPOM M/H,16,6HTO M/H,514,2(FP.3))
646
665
                   END.
```

1

- 232 -

	SUBROUTINE COSTIT (J, AREA, BUS, BOS, DIST, COST)
	BEATHELPHEADED/2. GAAREA/DIST
	60 TO (10.20.30.40.50.60.70) .J
40	Ensyan147+12.8+6.4+0EPT#3+50,0+PG,0+DUS
	RETHEN
20	CAST=0147+15.7+4.1+06P74)+30.0+70.0+DUS
	#FTURN
30	COST #D[47+ (8,9+4,1+NEPTH)+30,0+75,0+DUS
	RETURN
40	COSTAD147+(12"5+4.4+DEPTH)+30"0+80.0+0U\$
	RETURN
50	COSTED141+(15,9+4.7+DEPTH)+30,0+85.0+DUS
	RETURN
60	COST=DIST+(19;7+5,0+DEPTH)+30;0+90,7+DUS
	RETURN
70	COST + DI + T+ (23"7+5, 3+ DEPTH)+30, 0+95, 0+DUS
•	RETURN
	END
	41) 20 30 40 50 60 70

۰.

.

١

\$UBPOUTINE THALL (NIT, PIT, NIP, PIP, 1"AX, JHAX, KMAX, LMAX, NBUH, []) DIMFHSION NIT(I"AX), PIT(JMAH), NIP(KMAX), PIP(LMAX) COMMON/AP/0(20), GHIK(20), GMAX(20), D"IK, DMAX, MEND, JEND, T, NP, RK, RP COMMON/APF/IN(50,6), LNI, LN2, LN3, LNI, LP2, LP3 Thistmentain 444 456 1.1.01.0(.0.0.1,1) wfw.m.1.p(1.1) K.m.1.2.1.6MEN.7.0.JEN.D.+(.4.8.0.4-1) K.m.1.2.1.6MEN.7.0.JEN.D.+(.4.8.0.4-1) 648 449 470 471 MINFNO=(NEN1-1)+JEND+MEND 472 473 474 N=N=ND+D+D+JE+D 00 40 J=1.JEND 00 40 M=1.HEYD 475 KWK44 676 C----NIP(NIP(KI)) IS 1ST U/S REFERENCE NUMBER IN TRACE BACK FROM D/S 677 C----STATE (J.M) 678 NEN41 677 C----STATE (J,M) 678 NENA 679 C----STRINGOEFEVENCE HACK ACROSS P/M FROM D/S STATE(J,M) 640 MANDADE/JNEND 641 C----FIT(MJNEND)ECUST CE SOLLTION TO D/S STATE(J,M) 642 C----ASSIME NO CHANGE CE REFERENCE ACROSS THE M/M 643 NIT(M)EUDEND 644 C----IF COST OF ARRIVAL AT A HIGHER LEVEL IS CHEAPER ADOPT THIS COST 645 C----AND ALTER REFERENCE 646 IF (M,LE,1) 60 TO 10 647 IF (MJNEND)ELT, BIT(MJNEND-1)) 60 TO 10 648 PIT(MJNEND)EIT(MJNEND-1) 649 ALTERNOPIT(MJNEND-1) 649 ALTERNOPIT(MJNEND-1) 649 ALTERNOPIT(MJNEND-1) 649 ALTERNOPIT(MJNEND-1) 640 ALTERNOPIT(MJNEND)EIT(MJNEND-1) 640 ALTERNOPIT(MJNEND-1) 640 ALTERNOPIT(MJNEND)EIT(MJNEND-1) 640 ALTERNOPIT(MJNEND)EIT(MJNEND-1) 640 ALTERNOPIT(MJNEND)EIT(MJNEND-1) 640 ALTERNOPIT(MJNEND)EIT(MJNEND-1) 640 ALTERNOPIT(MJNEND)EIT(MJNEND-1) 640 ALTERNOPIT(MJNEND)EIT(MJNEND-1) 640 ALTERNOPIT(MJNEND)EIT(MJNEND)EIT(MJNEND-1) 640 ALTERNOPIT(MJNEND)EIT(ATT MITCHINETENETS 470 CHMHHETENETS OF ARRIVAL WITH A SMALLER DIAMETER IS CHEAPER, ANOPT THIS 471 CHHHETENETS ANN ALTER BEFERENCE 472 10 IF (J;LE,1) GO TO 20 473 IEMJNEMNHEND 444 15 181TEMUNEND3.LT.#1TE133 60 70 20 P1T(MJNFND)=P1T(1) 476 ISN-WFAR 477 N1T(N)=N1T(1) 478 20 CONTINUE 479 C---SRT U/S NEF, NO. IN TRACE PACK FROM D/S STATE(J,M) IS REF. ACROSS M/H 500 11=11+1 511 NTPERSHTJ. 512 513 5-2 NID(IJ)=NIT(N) 5-3 IF(PIT(=JNEXD),GT,999999,03 NID(IJ)=0 5-4 E====ESTABLIRH THE REST OF THE TRACE BACK 575 575 577 578 579 30 MAJANARNIP(TJ) 11#TJ+1 1f(MAJANA,LT,MEND+JFND) 60 TO 60 NEPEEJSONITEMAJANAS 60 TO 30 40 LJ=1J=1 \$10 5*1 5*2 5*3 60 CONTINUE RETURN £ N D .

SUBBOUTINE TRACE CUIT, DIT, NIP, FIP, IMAX, JHAX, KHAN, LHAX, LD 5°6 5°5 SHBBOUTINE TRACE ("IT, DIT, NIT, FID, I"AA, JPAA, KPAT, LPAK, L) DIMENSION NIT(IPAK), DIT(JMAK), NID(KMAK), DID(LMAK), JDS(10), MOS(10) COMMON/PD/ D(20), (MIN(20), GMAK(20), PMIN, DMAK, MEND, JEND, T, ND, BK, RP COMMON/UMERIA/ IN(50,4), LN1, LN2, LN3, LP1, LP2, LP3 516 517 S'B C----TOTAL NUMBER OF BEANCHES AND MANHOLES 5-0 (0447847) 570 NOWYD(7) 521 Caarannimee of Manholes in Last Pun KHIN(10,17 572 523 NENNANIP(4) 574 C-----STANT AND END ELEMENTS FOR FINAL COSTS IN ARRAY PIT 525 NUMENDAJEND 525 576 [1=(NEND=1)+HJ+1 527 12=NEND+HJ \$ 20 ICOT=0 CCST#000000.8 D0 10 I=11,12 IF (PIT(I).6E,COST7 G0 T0 10 COST#PIT(I) ICOT=I CONT=1 570 571 572 573 574
 516
 517
 10 CONTINUE

 518
 C----NO REASIBLE SOLUTIONE

 519
 IF (ICOST, EG, D) STUP

 518
 C-----IDPNTIFY THE DOWNSTREAM STATE

 519
 IFIPOST-I1+1

 14/1-13/MEND+1
 \$ 61 MetalJ=1)+HEND 542 NEUMOLO 545 1.00 LL=0 565 C----- IDENTIEV START AND END ELIMENTS IN ARRAY NID FOR TRACE RACK 546 C-----UD RRANCH GIVEN RUN NUMBER, J AND M 547 15 KHLN2-1+(NRUN-1)+JEND+MEN(+(J-1)+MEND+M KUNTPERS-1 548 549 C----TRAFE BACK ALONG BRANCH NEUN FROM (J.H) \$50 20 K+K+1 551 552 553 LeLes NITELSEVEN NA= (N1P(x)=1)/(1-EhD=JEND)+1 556 1+1+1 NITILIANA 556 JAEENIBCK)-1-(NA-1)-FENDAJEND)/MEND+1 337 337 338 339 LaLa1 NTTEL)=JA
 PARMIP(K)=(hAn1)+PEND+JEhr

 540
 L=L+1

 541
 NIT(L)=MA

 542
 IP(NA:GT:1) GU TO 20

 543
 C====NOW AT UPSTPEAH FHD UF RUH

 546
 C====NOW MANY UPSTREAM HUNS?

 545
 K#IMENRUA.33
 MAENIP(K)=(HA=1)+FEND+JEND+(JAP1)+HEND 546 -----Gn Tn (70,60,50,40),« 40 LL=LL+1 Jrsflly=JA MDSflly=MA 548 549 570 571 572 573 50 LL=LL+1 JD\$(LL)=JA MD\$(LL)=MA 574 575 60 NRUNENRUN=1 J=J& M=MA 576 577 60 TO 15 70 15 (NEUN (FU, 1) CO TO 80 578 579 NOUNANRIN-1 JEJPS(LL) Menng(LL) 590 5.81 LI=LL=1 G0 T0 14 R0 WPITE (4,1003) (057 WPITE (4,1001) (N37(11),11=1,L) 542 513 594 595 596 517 RETURN 1001 FORMAT (10X, SHRUN, 17, 5X, 3HM/H, 13, 5X, 8HU/S DIAH, 13, 5X, SHLEVEL, 13) 1003 FORMAT (SH1///104, 31HTRACE BACK OF CHEAPEST SOLUTION, SX, SHCOST=, 598 1#12,3) 399 590 END

571	\$UBPOUTINF RAIN {VoteR1}	·
5-2	! CFXERPTURY PEPIUD(YRS),YOTINE(PINS),RTWINTENSITY \$PH/WD3	
513	R1=A0"/Y++((Y+T+2U2,76)++,28169+2,54)	
574	16 (81-33.) 10,10,40	
575	10 1F(#1) 20+30+30	
576	20 81=0.0	
377	30 BETHEN	
578	40 wjen	
5-9	CTTEPATION LOUP FUP HULLAND FOPHULA	
610	50 F#ALOG(15240,0/V/PI/T+(PI+T/1524,0+0,1)++3,55)+7,0+0,0346+RI	
611	DF==1(0/#1+3,55/(#1+152,4/T)+(0314	
6^2	X=F/DF	
613	#1##T=X	
674	IF (ARS(X)=0,01) 10,10,40-	
615	60 1\$ (N1+10) 80,80,70	
516	CEPROR	
6 7	70 STOP	
678	40 NIMNI41	
519	GO TO 50	
610	END	

· ,

•

.

. .

611 SI BONITINE MURAN (NID, PIF, FMAX, LMAX) 62 C----BRONNICES A MINIFUL GRADIENT INESIGNI 63 COMMON/ND/ N(20), GHIN(20), GMAX(20), DMIN, DMAX, HEND, JEND, T, NO, RK, PP 64 C(MMON/UNERE/ IF(50, 4), LN1, LN2, LN3, LD1, LD2, LD3 65 DIMENSION NIP(KTAX), PIP(LMAX), TNS(20), MD(50) 6"6 C----SET INITIAL VAL"ES 6"7 RD#1.0 618 1-5414A 619 IR .LN 3=1 671 FURNIb(5) 621 IF (ND, WE, 0) WRITE (6, 1001) 672 C-----EAR EACH BRANCH IN TURN J=IN(1,1) 625 626 NNEWSP(J) 677 C-----INENTIFY TIME TO UPSTREAM END OF BRANCH, AND MAX, UPSTREAM DIAM, J=1+(1,3) 428 \$29 NRRENIP(J) 610 105+1 671 672 673 NPIPE=1 IF (NAR, EQ. 0) GO TO 20 DO 10 4K=1+18R 674 Juj+1 J=3-4 K=NTP(J) IF (TDS(K),GT,TUS) TUS=TUS(K) IF (MD(K);GT,N#IPF) \PIPE=MD(K) 675 676 617 10 CONTINUE 618 439 C----- IDENTIFY PUSITIONS IN PIP FOR CHAINAGE, AREA, LEVEL FOR FIRST M/H 59'15N101' 02 641 661 13-12+44 442 2115=#1#(33) 663 466 D15TUS=1.0 GLG NID(IB)=NDIPE GLG NID(IB)=NDIPE GLT C----FOD FACH HANNULE POSITIUN DOWN THE BRANCH GLB DO 45 KD2.NH 649 650 651 J1=J1+1 J2=J2+1 J3=J5+1 452 015705=P1P(J1) 653 654 205=PEP(J3) DISTODICTOS-DISTUS 615 APEA=PIP(J2) 676 SLOPE=(7US-2US)/UIST 677 C----CALCULATE VELUCITY OF FLUW AND PIPE CAPACITY 678 30 CALL VELOC (SLOPE, D(NPIPE), RK, VFL, CAP) 659 C-----FALCULATE PAINFALL AND FLOW TIMPOTURODIST/VEL CALL RAIN (RP,TIME/60,0,81) FLOWARFAOPI/3,466 14 (FLOW.LT,CAP) GQ TO 40 NPIPEONPIPEO1 660 641 642 643 644 645 GO TO 30 646 C----STORE FLOW AND PIPE SIZE NUMBER 647 60 14-14-1 16=18+1 P1P(1A)=FLOU 548 449 670 NIP(IN)=NP]PE IF (ND, NE U) UNITE (6,1002) I.K.D(NPIPE), ISLOPE, DIST, AREA, VEL, TIMF, RI, CAP, FLOU š71 672 473 TISSTIME 21:5+205 474 675 DISTUSEDISTOS 45 CONTINUE TOSEIJATIME HOCTIENEIPE 676 477 678 379 50 CONTINUE RETURN 1001 FURMAT (1H1///,5X,91HBRANCH D/S M/H DIAM SLOPE LENGTH 1APEA VELOCITY TIME RAINFALL CAPACITY FLOW) 4002 FURMAT (7X,I3,5X,I3,6X,F8,3,F8,4,F8,1,F8,0,F8,1,F8,3,2F10,6) 610 691 642 693 END 6=4 495 SUBPOUTINE VELUC (SLUPE,DIAM,RK,V,4) 676 C----Calculates Flows From Colebrook-44ITE Formula 677 Sobrort(Slope+Diam) V--P. A1A+SQ+ALUG10(4K/3,7/DIAM+0,6471E+6/S9/DIAH) 6*A 6-9 Q=V+DIAM+DIAH+9,7854

......

.

670 RFTIAN 471 €ND

SUBHOUTINE LEVELS (NIT, PIT, NIP, PIP, 1MAX, JMAX, KMAX, LMAK, LMAK, LMAK, COMMON/DP/D(20), GHIN(20), GHAX(20), DHIN, DMAX, MEND, JEUD, T, ND, RK, HD COMMON/UHERF/IN(50, 4), LN1, LN2, LN3, LP1, LP4, LP3 DIMPNSION NIT(IMAX), PIT(JMAX), NIP(KMAX), PIP(LMAX), TOS(50), 27(50) UHITE (4, 1002) COSTED, A LORA SUBROUTINE LEVELS ENIT, PIT, NIP, PIP, IMAX, JHAX, KHAX, EMAK, EMAK, EMAK, EMAK, EMAK, EMAK, EMAK, EMAK, EMAK, 672 673 676 575 576 577 578 10=0 479 LJ=1N3+1 10 L0=10+1 700 KOIN(LO.1) THE NEWPENJERS NOT NEWPENJERS AND MENCE USS TIME OF FLOW AND LEVEL 776 KHINCLO,3) NARHNIPCK) TUS=T DIS+U5=0_0 7-6 11#10(10.4)+4+NFHD 610+#PTP(11) 718 739 710 2HSUGLUS-PHIN 712 7*3 7*4 NGONIP(L) 17 (NRR, EG. C) GR TO 60 DR TO SHI NRR 7.5 #=x+1 716 LautPLKY TUSHAMARY (TUS, TOS(1)) 7*8 7*9 2113#3#1+1412US,201135 30 CONTINUE 720 C----DEFINE FIRE DIANETER, D/S M/H, CATCHHENT AREA 721 772 723 40 LHITHLNIT-4 N=NTT(L+1T=2) J=NTT(L+1T=1) 726 725 736 111=11+" JPIPE=MAXO(NIP(LJ1)=JEND+J.1) DIAMADEJPIPES 778 778 779 770 731 L#IN(L0,6)+1+N D15705=01P(L) DISTANISTOSANISTUS LELONEND APEAupzacis 72 C----- FIND REQUIPED GRADIENT 73 CALL GRADE(JPIPF,DI9T,AREA,TUS,SLOPE,G,GMAX,VEL) 773 CALL GRIDE(JPIPF/DIT/) 774 C----FIND D/S GOUND LEVEL 775 L20(1+NG 776 D/ SD J=17/L2 716 717 718 719 K=1+46 IF (PIP(K), 67, DISTDS=0, 1) 60 70 60 50 CONTINUE 40 GLD4=P1P(1) 741 15+1 741 LC#7 742 20\$#QLD4=tM1% 743 C=====14 HIN \$LOPE SOLUTION FRABIALE7 744 IF f2HS=SLOPE=DIST, GT, 20\$+0,001) GU TO 65 205=205=5LCPE+DIST GO TO 70 45 SLOPE=(ZUS=ZDS)/DIST IF (SLUPF,LT,GHAK(JPIPE)+0,0001) GO TU 68 ZUS=205=GHAY(JPIPE)=DIST 765 746 747 749 750 751 752 SLOPE=GNAX (JPIPE) AB CALL VELOC(GLOPE,DIAM,RK,VEL,GMAX) TIMBA(THS+DIST/VEL)/60,0 CALL RAIN (99,TIME,RI) GRAFEA+01/3,6E6 ----CHECK GOGUND COVER EN ROUTE 753 754 755 8--70 13+11+1 756 757 14=12=1 GARPA=0.0 IF (13.97"14) GO TO 90 DZNAX=0.0 799 760 DO PO INLAILA KHIANG 741 AD DZMAKWAHAX1(ZUSWSLUDE+(PIP(K)+DISTUS)+DIP(1)+DHIN,DZMAX) 743 744 2US#ZIIS=DZHAX 745 ZOSHZOSHDZNAX

```
746 C-----CALPULATE AREA OF LONG SECTION ABOVE STALIGHT LINE
747
                #1+1+NA
7/8
                H2#12+NS
                GARBAN-NIP(+1)+(PIP(L3)+PIP(L2))+0.5
749
                00 100 tel5.14
771
                K#1+NG
         100 GAREA=GAREA=DIA(2)+(DIB(1+1)=DIB(1=1))+7,5
GAREA=GAPFA=DIB(+2)+(DIB(11)=DIB((4))+0,5
90 CALL COTTIT (JDIDE,GAREA,GLUS=205,GLDS=205,7157,C)
772
773
774
775
776
777
               WRITE (A.1001) LOIDIST, DIAMISLOPE, 208, GLUS, GLDS, AREA, GAREA, 9,
19444, VEL, C.COST
TUSATUSADIST/VEL
778
779
770
71
                2115#ZNS
GLUR#GLNS
                D157US=015705
773
775
716
                11012
                LJ=LJ1
TD$(L0)=T#$
7×6
7*7
7*7
7*9
7*9
                20(10)=705
                IF (LO,LY,HIP(2)) 60 TO 10
AFTHEN
       1001 FORMAT (140,16,59,3,57,1,57,4,458,3,58,0,58,3,297,4,58,1,2510,1)
1002 FORMAT (1417/3X,114HBRANCH LENGTH NIAM SLAPE U/S SL H/S SL H/S
1 GL D/B GL AREA GROUND A, FLOW CAPACITY VEL. COST 5
703
772
773
              ZUH)
               END
```

```
775 SUBBOITTINE GRADE (JPIPE, DIST, AREA, THS, SLOPE, G, GFULL, V)
776 C----CALPULATES REGUIRED SLOPE OF A PIPE ACCORDING TO RATIONAL METHOD
777 COMMON/DP/D(20), GMIN(20), GMAX(20), DMIN, DMAX, MEND, JEND, T, ND, RK, RP
778 LOGICAL NPLUS, MINUS
778
                    K=0
                    ROU
NPLUGG,FALSF,
MINUSS,FALSF,
SLOPEGGUIN(JPIPE)
SLUMENGIN(JPIPE)
DIAMOD(JPIPE)
5 CALL VE(CC(SLOPE,DIAM,RK,V,OFULL)
TIMFOTUQ-DIGT/V
CALL RAIN(HP,TIME/60,0,RI)
QRAPEA=DI/3,666
IF (Q.LT.OFULL,AND,K,EQ,D) #ETUAM
SLOPE=0,02045+Q=0/DIAM++5+(ALOGIO(RK/3,7/DIAM+4,1365/(Q/DIAM/
810
811
                  11;1418=4)=+0,89))++(=2)
                    K=K+1
8•2
8•3
                    1F (K"LF.7) 60 TO 5
              10 K=K+1
314
315
                    CALL VELOC(SLOPE+DIAM+RK+V+OFULL)
                    TIMFETUS+DIST/V
CALL RAIN (7P,TIME/60,0,R1)
GEAPEA+01/3,666
516
517
             8.8
820
871
872
573
                   NPLIISE, TRUE,
MINUSE, FALSE,
874
875
              GO TO TA STURN STURN
876
                    SLOPE=SLOPE+.949
                    HINHS=, TRUE,
828
828
829
                    NPLUS=, FALSE,
60 TO 10
830
                    END
```

APPENDIX F

PROGRAM SORT

.

:

	POSTOPROGRAM REATS DATA FUR MODEN IN SIMPLIFIED FORM AND OUTPUTS TO
9	C PASNETIC TAPE FILF
10	,
11	17(300),0400 (01) (01) (01) (01) (01) (01) (01) (0
19	REVIND 7
14	END. NODDOT
15	
16	ALAD CONTRACTOR TO SEC
14	G (VEFTHE DEFAULT VALUES)
19	n11111, 3
21	NEL117400.7
22	VEL'IAX=G.n
23	() () () () () () () () () () () () () (
25	657Ae7#7, 075
26	nu 510 1#1,15
27	SID DITADSTART+FLOATIJ #DELTA
29	WEITE (6,2001) PMIN, RK, VELUIN, VELUAX, SMIN, (0(1), 1=1, 15)
30	URITE (?) [MIN, RK, VELIIIN, VELHAX, SMIN, (D(1), 1+1, 15)
31	RU TO 560 820 /8 /00105 CT"CN STOP
34	ny SSO 1a1.//PTYPE
54	BEAD 45, 1001) DI SH, RK, VELHIN, VELMAX, SMIN
35	exerc/1000.r
30	בר, רשו הצר טון 51.0 Allen G
58	(=). (=)
59	569 30341
60 14	READ (7,1977) F(J)
42	1F (D(J).GT.E.AND.J.LT.15) GU TU 54 ^A
43	UNITE (6.2001) PHINARKAVELIIINAVELHAXASHINA (P(11) +11=1+15)
••	
4) 4 A	0 200 CUNTINUL 18 (V27V2F, 2000) 60 70 590
47	500 NU 570 1=1,20
48	570 n(1)=0.3
50	NO SEO INVERSE O
51	URITE (6,2001) (D(11),11=1,20)
52	550 WITE (7) (D(11), 11=1,20)
71 54) 390 CUNTINUE C
	READ (4.10(1) T.RP
56	1F (T.LE.E) YHZ.0
57	7 75 (80, 18, 19, 1987,0) Thán (18, 19, 1987,0)
59	C TETD IA HILLIA. " ID WYXIMIN WYANOFE BACING
5Ú	BEAD (4-1073) SENSII, SUNAK
21 A2	16 (CDHAX.LE.F) SPHAX#150.0
63	URITE (6,2701) T, RP, SPHIN, SPMAX
A.6	, WRITE (7) T,RP,SPMIN,SPMAX Theremore an in orthisation beregoriance parameters
	READ (4.1901) DZIRESHH
67	READ (4.4002) I'ENDIJENDIKXD
68	; 1F (DZ.LE.E) FZWC.5 te (mestim le s) pertimesa.0
70	17 (KXD_LE.C) KyD=4
71	IF (MEND, LE, C) 1 ENDE
72	TP (JETD.LT.C) JETOME CBEAD IN DIACHOSTICS LEVEL
75	DEAD (6,1907) 40
75	Commenterd IN ANILER CE BEANCHES IN BURIC PAULI ANARTEN
76	9 8240 (5,5902) LC 19178 (4,2002) ISND,JEND,NO,LO
7 #	URITE (7) IFND.JEND.ND.LD
79	UNTTE (6.2001) [2
40	urite (7) CZ

•

.

.

•

86 ____3#0 40 __=j+1 #7
 #EAD
 (5,1071)
 GL(J)

 #EAD
 (5,1071)
 GX(J)

 15(GX(J).LT.Y-0.D(1))
 GO T(

 C-----GE/IERATE
 PCSITIONS
 OF MANHOLES
 RB 89 GO TO 40 90 91 15 (ITYPE.LT. 10.DR. ITYPE.GT. 100) 60 TO 70 60 5157(1)=0.0 92 93 NIST(2)=V NN=2 94

 05
 NH=2

 96
 R0 TO 90

 97
 C====15 LENGTH LESS THAN TWICE HINI"UM \$PACING?

 98
 70 IF (V.LT.2.0+SP) Jh=0.4.0R.V.LT.RESHN+0.13 GO TO 60

 99
 DIST(1)=0.0

 101
 HN=3.5INT((Y=DIS=SPIIN+0.13/RESHH)

 102
 DIST(1)=V

 103
 BIST(UN-1)=V=015

 104
 F (UN.EQ.3) GO TO 81

 105
 DIST(UN-1)=V

 104
 F (UN.EQ.3) GO TO 81

 105
 DIST(UN-1)=V

 104
 F (UN.EQ.3) GO TO 81

 105
 AU.EQ.30 GO TO 81

 95 104 102 105 104 195 106 DIST(L)=DIST(L+1)-RESHM 108 80 BUNTINUE 109 C-----GENERATE PERHISSIBLE HANHOLE CONNECTIONS (+++REPUNDANT++++) 31 KA(1)=1 110 EBS13=7 AU 340 Mm2+1.N BU 300 Lm1+1 4f (DIST(U)=DIST(L)+GT+SPNAX+0+1) 60 TO 300 EA(U)=L 111 x8(1)=1 112 113 114 115 GU TO 310 116 117 300 CUNTINUE 310 h0 320 La11 1F (0157(H)-DIST(L)-LT-SPHIN-0-1) 60 10 330 118 114 320 FUNTINUE 120 330 x0(")=L-1 340 CONTINUE 121 162 G-----CALCULATE GROUP'S LEVELS 90 GROUND(1)_GL(1) 125 125 1F(NN"LT.3) GU TO 130 IF(IN)LT.3) GU TO 130
EK=[Mm]
NU 120 K=2,KK
NU 100 1=1,J
IF (GX(").GT.UIST(K)+0.001) GO TO 130
100 CULTINUE
110 GROUND(K)=GL(H=1)+(GL(H)=GL(H=1))/(GX(")=GX(H=1))+
1(UIST(K)=GX(H=1))
120 CULTINUE
130 GROUND(NU)=GL(J)
==1 126 127 129 1 51 1 52 1 3 3 134 135 gH#1 137 #Lan 140 KL=KL+1 1 58 TF (DIST(K")-CX(K"),GT_0,001) GO TO 145 W(KL)=DIST(KH) 1 39 FEKL)=GROUPD(KP) 141 1F (KH.EG.KL) 50 TO 146 1F (DIST(KT)=6X(K+),6T,=0.007) KH=KN+1 142 145 KIIEKH+1 145 60 TO 140 165 X(KL)=GX(K!') 144 7(KL)=GL(K%) #1=KN+1 147 148 149 GU TO 140 150 C----READ IN UPSTREAN CONNECTIONS 151 152 154 154 146 NUB=0 170 NUB=NUA+1 BEAD (5,1002) HER(NUB) TF (NBR(NUD),FQ,D) GO TO 180 GU TO 170 155
 133
 GU TU TUT

 156
 130 HUBBHURH

 157
 C====FETINE

 157
 C====FETINE

 154
 NU TOT JJ=ZINH

 154
 NU TOT JJ=ZINH

 159
 190

 159
 190
 140 AREA(1)=0, 141 C----DEFINE DEFSETS 142 DU 200 JJ=1,44 145 200 OFF(JJ)=0FFSET

~

-

.

+-

÷

.

.

.

-

.

.

~

		A ARTICLE AND
	9 6 4	CBEFINE ARDILUTE COALMANN
•	145	A0 210 JJ#1/15
•	166	210 A35X(JJ)#5TC#+D1#+D1\$T(JJ)
	447	JAITE (6.2002) STYPE NULINNAKI
		UNITE (T) ITYDE-UUDANN.#1
	165	
	169	AD. ISAN C. 1.5 Marine
	170	WHITE (6,20017 (OFF(K),K#1,NN)
	171	urfte (7) {OFF(K)+K=9+1%}
÷-	. 7 .	10175 (6.2004) (ACSX(K), KE4. NN)
	1.6	
	173	UNITE TTY TAISACHICKET, WAS
	176	WRITE (6,2001) (DIST(K),K=1,NN)
•	175	UNITE 175 (DIST(K)/K=1/NN)
	474	UNITE (A.2001) (AREA(K), KM1, NA)
	1/0	
	111	
	178	15 (408, 20, 0) Gr to 020
	179	URITE (6,2002) (HIR(K),K=1,HUB)
	180	WRITE (7) (NER(K)+K=1+NUB)
	181	625 CONTINUE
		UNITE (6.2004) (2(K) KM4.KL)
		10178 (7) (7(4), KR4, KL)
-	103	
	106	WEITE (9/2001) (X(K)/K=1/KL)
	185	URITE (7) (X(K))KHT,KL)
	106	290 CONTINUE
+	167	CBEAD IN HUILER OF CROSS DRAIN SETS
		SEAD (5.1007) LYDSET
		10195 (4.3062) LVD4/9
•	107	METTE FRENCH LUNSEL
	990	WHITE (7) (XDSET
	171	14 2NAD2E1-E0-01 CO TO 420
	192	C FUR EACH SET
•	491	NO 440 TOTAL XESET
		Commenter IN REPAILEL DRANCHER
		Canada ta subdett beverte
•	143	
	170	618 JeJ.4
	• • •	
	197	#EAD (5,1002) KER(J)
	197	96AD (5,1002) KER(J) 15 (KBR(J), FE, 0) 60 70 400
•	197 198 199	BEAD (5,1002) Krr(J) TF (KBR(J), FE, 0) GQ TO 400
•	197 198 199 200	DEAD (5,1002) KER(J) IF (KBR(J), HE', 0) GO TO 400 C
•	197 198 199 200 201	BEAD (5,1002) KIR(J) IF (KBR(J), HE, 0) GO TO 400 IF (KBR(J), HE, 0) GO TO 400 C
•	197 198 199 200 201	BEAD (5,1002) KIR(J) IF (KBR(J), HE, 0) GO YO 400 IF (KBR(J), HE, 0) GO YO 400 CGRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J
	197 198 199 200 201 202	DEAD (5,1002) KER(J) IF (KBR(J), EE, 0) GO TO 400 IF (KBR(J), EE, 0) GO TO 400 CGRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J URITE (7) J
• • •	197 198 199 200 201 202 202	DEAD (5,1002) KIR(J) IF (KBR(J), HE, 0) GQ TO 400 CQRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J UPITE (7) J CFOR EACH BRANCH IN THIS SET
*	197 198 199 200 201 202 205 205	•EAD (5,1002) KIR(J) IF (KBR(J), HE, 0) GO TO 400 IF (KBR(J), HE, 0) GO TO 400 CGRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J URITE (7) J CFOR EACH BRANCH IN THIS SET DÚ 410 KH1/J
•	197 198 199 200 201 202 203 204 204	θΕΑΠ (5,1002) KrR(J) IF (KBR(J), HE, 0) GQ TO 400 CGRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J URITE (7) J CFOR EACH BRANCH IN THIS SET nú 410 Kel, J KIRAKBR(K)
*	197 198 198 200 201 202 203 204 205	•EAD (5,1002) KIR(J) IF (KBR(J), HE, 0) GQ TO 400 IF (KBR(J), HE, 0) GQ TO 400 CQRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J UPITE (7) J CFOR EACH BRANCH IN THIS SET NÚ 410 KE1, J KURAEKBR(K) UPITHINGERDAN
•	197 198 199 201 202 203 204 205 204	•EAD (5,1002) KIR(J) IF (KBR(J), HE, 0) GO TO 400 IF (KBR(J), HE, 0) GO TO 400 CGRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J URITE (7) J CFOR EACH DRANCH IN THIS SET nú 410 KEI,J KURAEKBR(K) MENUMIN (KRRA)
* *	197 198 199 200 201 202 203 204 205 204 205 204 205	•EAD (S,1002) KrR(J) IF (KBR(J), FE, 0) GQ TO 400 J=1 C====-GRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (G,2002) J UPITE (7) C====-FOR EACH BRANCH IN THIS SET nú 410 Km1,J KBR(K) M=4UMINH(KBRA) LA=14AXOC(1)=13/KXD,13
	197 198 199 200 201 202 205 205 205 205 205 205 205 205 205	•EAD (S, 1002) KrR(J) IF (KBR(J), HE, 0) GQ TO 400 IF (KBR(J), HE, 0) GQ TO 400 CGRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J UPITE (7) J CFOR EACH BRANCH IN THIS SET NÚ 410 K=1, J KURA=KBR(K) M=MUMIN(KRRA) LA=MAXOC(M=1)/KXD, 1) L=LA
•	197 198 199 200 201 202 205 205 205 205 205 205 205 205 205	•EAD (S,1002) KrR(J) IF (KBR(J), LE, 0) GO TO 400 IF (KBR(J), LE, 0) GO TO 400 IF (KBR(J), LE, 0) GO TO 400 IF (RITE (6,2002) J URITE (6,2002) J URITE (73) J CFOR EACH URANCH IN THIS SET nú 410 K=1, J KURA=KBR(K3) M=4UH1H (KRRA) LA=11AX0((1)=1)/(KXD,1)) L=A UNCL)=11
	197 198 199 200 200 200 200 200 200 200 200 200 2	•EAD (5,1002) KrR(J) IF (KBR(J), HE, 0) GQ TO 400 J=J ¹ C=====GRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J URITE (7) C=====FOR EACH BRANCH IN THIS SET NÚ 410 KH1,J KURAHKBR(K) M=UUNIH (KNRA) LAHAKOK ((I=1)/KXD,1) L=LA 'H(L)=11 IF (L;EQ,1) GU TO 406
	197 198 199 201 202 205 205 205 205 205 205 205 205 205	•EAD (S, 1002) Kr R(J) IF (KBR(J), HE, 0) GQ TO 400 JEJ1 C=====GRITE OUT NUMBER OF BRANCHES IN THIS RET URITE (6,2002) J UPITE (7) J C=====FOR EACH BRANCH IN THIS SET NÚ 410 K=1,J KURA=KBR(K) M=YUHNH(KNRA) LA=HAXO((H=1)/KXD,1) L=LA NH(L)=H IF (L]EQ,1) GC TO 406 405 I=L=1
	197 198 199 201 201 205 205 205 205 205 205 205 205 205 205	•EAD (S, 1002) KIR(J) IF (KBR(J), LE, 0) GO TO 400 IE, 1 URITE (6, 2002) J URITE (7) J CBR EACH URANCH IN THIS SET NÚ 410 KE1; J KURAEKBR(K) M=4UHH(KRRA) LAHAXO((H-1)/KXD, 1) L=LA TH(L)=H IF (L, EQ, 1) GO TO 406 405 L=L-1 MELEKD
•	197 198 199 201 202 204 205 205 205 205 205 205 205 205 205 205	•EAD (S, 1072) Kr R(J) IF (KBR(J), HE, 7) GQ TO 400 J=J=1 URITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J UPITE (7) J CFOR EACH BRANCH IN THIS SET NÚ 410 K=1,J KURAMKBR(K) M=UUNIH(KRRA) LA=NAXO((N=1)/KXD,1) L=LA NH(L)=N If (L,EQ,1) GC TO 406 405 L=L-1 M=N=N
	197 198 202 202 202 205 205 205 205 205 205 205	•EAD (S, 1002) KrR(J) IF (KBR(J), HE, 0) GQ TO 400 JEJE C====-QRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J UPITE (73 J C====COR EACH BRANCH IN THIS SET NÚ 410 KE1, J KURAEKBR(K) MENUMIN (KNRA) LABIAXO ((I=1)/KXD, 1) L=LA NM(L)=11 TF (L]EQ,1) GU TO 406 405 L=L-1 Mell=KD MM(L)=11
	197 198 198 201 202 203 204 204 204 204 204 204 204 204 204 204	#EAD (S,1002) KrR(J) IF (K&R(J), FE, 0) GQ TO 400 J=1 C=====GRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (G,2002) J URITE (7) J C=====FOR EACH BRANCH IN THIS SET nú 410 Kal,J KRAMEBR(K) M=4UHIH (KRRA) LABIAXOC(II=1)/KXD,1) L=LA rif(L)=II TF (L,EQ,1) GC TO 406 405 L=L=1 M=(L)=II rf (L,EQ,1) GC TO 405
	197 198 199 200 200 200 200 200 200 200 200 200 2	•EAD (S,1002) KrR(J) IF (KBR(J), HE, 0) GQ TO 400 J=J=1 URITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J UBITE (7) J CFOR EACH BRANCH IN THIS SET NO 410 K=1,J CURA=KBR(K) M=YUHHH (KRRA) LASHAKBR(K) M=YUHHH (KRRA) LASHAKOC(H=1)/KXD,1) L=LA NH(L)=H IF (L,EQ,1) GC TO 406 405 L=L-1 M=H-KXD MH(L)=H IF (L,GT,1) GC TO 405 406 IF (K,NE,1) GC TO 407
	197 198 201 202 205 205 205 205 205 205 205 205 205	• EAD (S, 1002) Kr R(J) IF (KBR(J), HE, 0) GQ TO 400 JEJT C=====QRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6, 2002) J UPITE (7) J C=====FOR EACH BRANCH IN THIS SET NÚ 410 KE1, J KURAEKBR(K) M=YUMIN (KRRA) LATIAXO ((I=1)/KXD, 1) L=LA NM(L)=II IF (L;EQ,1) GU TO 406 405 L=L=1 MeII=KXD MM(L)=II IF (L;GT,1) GU TO 405 406 IF (K;NE,1) GG TO 407 URITE (6, 2072) LA
	197 198 198 201 202 203 203 203 203 203 203 203 203 203	<pre>EAD (S,10^2) KrR(J) IF (K&R(J), FE, 0) GQ TO 400 J=J¹ C=====GRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J UBJTE (7) C=====FOR EACH BRANCH IN THIS SET nú 410 Kal, J KURA=KBR(K) M=UUNIH (KNRA) LA=HAXO((H=1)/KXD,1) L=LA 'H(L)=H TF (L:Eq.1) GC TO 406 405 L=L=1 M=H=KXD M=(L)=H TF (L:GT.1) GC TO 405 406 IF (K:NE.1) GC TO 407 JRITE (6,2072) LA HITE (C2) LA</pre>
	197 198 198 201 201 201 201 201 201 201 201 201 201	<pre>BEAD (S,1002) KTR(J) IF (KBR(J), HE, 0) GQ TO 400 J=J=1 URITE (G,2002) J URITE (G,2002) J URITE (7) J CFOR EACH BRANCH IN THIS SET NU 410 K=1,J CURA=KBR(K) M=YUHHH (KRRA) LASHAXO((H=1)/KXD,1) L=LA 'HH(L)=H IF (L,Eq.1) GC TO 406 405 L=L=1 M=H=KXD M=(L=H) IF (L,GT,1) GC TO 405 406 IF (K,ME.1) GC TO 407 URITE (G,2002) LA UNITE (C) LA UNI</pre>
	97 198 202 203 203 203 203 203 203 203 203 203	•EAD (S,1002) KrR(J) IF (KBR(J), HE, 0) GQ TO 400 IF (KBR(J), HE, 0) GQ TO 400 URITE (0,2002) J URITE (73 J CFOR EACH BRANCH IN THIS SET nú 410 K=1, J KURAKBR(K) M=YUMIN (KRRA) LA=14AX0C(II=1)/KXD,1) L=LA THE (C, Eq.1) GC TO 406 405 L=L-1 M=11-KXD M=12 M=12 URITE (6,2002) LA URITE (6,2002) KBRA
	97 198 198 201 202 203 203 203 203 203 203 203 203 203	<pre>EAD (S,10^2) KTR(J) IF (K&R(J), HE, 0) GQ TO 400 J=J¹ C=====GRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J URITE (7) J C=====FOR EACH BRANCH IN THIS SET nú 410 Kal, J KURA=KBR(K) M=UUNIH (KNRA) LA=HAXO((H=1)/KXD,1) L=LA 'H(L)=H TF (L:Eq.1) GC TO 406 405 L=L=1 M=H=KXD M=(L)=H TF (L:GT.1) GC TO 405 406 IF (K:NE.1) GC TO 407 JRITE (6,2072) LA HITE (72 LA URITE (72 KBRA URITE (72 KBRA URITE (72 KBRA) URITE (72 KBRA) URITE (72 KBRA) URITE (72 KBRA) URITE (72 KBRA) URITE (72 KBRA) URITE (72 KBRA)</pre>
	97 198 198 201 201 201 201 201 201 201 201 201 201	<pre>UN = EAD (S,10^2) KrR(J) IF (KBR(J), HE, 0) GQ TO 400 J=J=1 C=====GRITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J URITE (7) J C=====FOR EACH BRANCH IN THIS SET NO 410 K=1,J CURA=KBR(K) M=1UHIH (KRRA) LA=NAXO((N=1)/KXD,1) L=LA N=U=KXD M=</pre>
	97 198 198 201 201 201 201 201 201 201 201 201 201	<pre>UNU = EAD (S,10^2) KrR(J) IF (K&R(J), FE, 0) GQ TO 400 J=J=1 C=====EQR EACH UNUER OF BRANCHES IN THIS SET URITE (G,2002) J UBJTE (T) J C=====FOR EACH URANCH IN THIS SET nú 410 K=1,J KURA=KBR(K) M=4UHHH(KRRA) LA=14AXO((1=1)/KXD,1) L=LA "M(L)=11 IF (L,EQ,1) GC TO 406 405 L=L=1 M=1=KXD M=1=KXD M=1=KXD M=1=KXD M=1E (T, 1) GC TO 405 406 IF (K,NE,1) GC TO 405 406 IF (K,NE,1) GC TO 407 URITE (G,2002) KBRA URITE (G,2002) KBRA URITE (T) KBHA URITE (T) (HI(G),N=1,LA) URITE (T) (HI(G),N=1,LA) URITE (T) (HI(G),N=1,LA)</pre>
	97 198 201 201 201 201 201 201 201 201 201 201	<pre>UN =EAD (S,10^2) KrR(J) IF (KBR(J), HE, 0) GQ TO 400 J=J=1 C=====WITE OUT NUMBER OF BRANCHES IN THIS SET URITE (G,2002) J UBITE (7) J C=====FOR EACH BRANCH IN THIS SET NO 410 K=1,J KURA=KBR(K) M=UUNIN (KNRA) LA=NAXO((N=1)/KXD,1) L=LA NH(L)=N TF (L,EQ,1) GC TO 406 405 L=L=1 M=N=KXD M=(L)=N M=(L)=</pre>
	9789 1988 2012 2012 2015 2015 2015 2015 2015 2015	<pre>EAD (S,10^2) KrR(J) IF (KBR(J), HE, 0) GQ TO 400 J=J=1 C=====GRITE OUT NUMBER OF BRANCHES IN THIS RET URITE (6,2002) J URITE (7) J C=====FOR EACH BRANCH IN THIS SET NO 410 K=1,J CURAEKBR(K) M= NUHHH (KRRA) LASTAXO((II=1)/KXD,1) LELA NH(L)=II IF (L,Eq.1) GC TO 406 405 L=L=1 M=II=KXD M=II=KXD M=II=KXD M=II=KXD M=II=C73 LA URITE (7) KBRA URITE (6,2002) KBRA URITE (7) KBRA URITE (7) KBRA URITE (7) (MII(I],N=1,LA) URITE (7) KOM 410 CUNTINUE 420 KTOP 410 CUNTINUE 420 KTOP 410 CUNTINUE 420 KTOP 410 CUNTINUE 420 KTOP 410 CUNTINUE 420 KTOP 410 CUNTINUE 420 KTOP MENTINE 420 KTOP MENTINE MEN</pre>
	9789 1988 2012 2012 2015 2015 2015 2015 2015 2015	<pre>UN = EAD (S,10^2) KrR(J) IF (K&R(J), FE, 0) GQ TO 400 J=J=1 C=====KR EACH UNALER OF BRANCHES IN THIS SET URITE (G,2002) J URITE (7) J C====FOR EACH URANCH IN THIS SET nú 410 Kal, J KURA=KBR(K) M=UUNIH (KRRA) LA=HAXO((H=1)/KXD,1) L=LA 'H(L)=H TF (L'EQ,1) GC TO 406 405 L=L=1 M=H=KXD M=(L)=H TF (L'GT,1) GC TO 406 405 L=L=1 M=H=KXD M=(L)=H TF (L'GT,1) GC TO 405 406 IF (K'NE,13 GG TO 407 JRITE (G,2072) LA URITE (73 KBRA URITE (73 KBRA URITE (73 KBRA URITE (73 KBRA URITE (73 (HIC(13,N=1,LA)) URITE (73</pre>
	978 1988 2012 2012 2015 2015 2015 2015 2015 2015	<pre>UNU = EAD (S,10^2) Kr R(J) IF (KBR(J), HE, 0) GQ TO 400 J=J=1 URITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J UPITE (7) J C=====FOR EACH URANCH IN THIS SET nû 410 Km1,J KURAMKBR(K) M=UUNIH (KRRA) LA=HAXO((H=T)/KXD,1) L=LA HM(L)=H IF (L,Eq.1) GC TO 406 405 L=L=1 M=H=KXD MM(L)=H IF (L,Eq.1) GC TO 406 405 L=L=1 M=H=KXD MM(L)=H IF (C,Eq.1) GC TO 407 JRITE (6,2072) LA URITE (7) LA 407 UHITE (6,2002) KBRA URITE (7) KBHA URITE (7) (HH(L),N=1,LA) URITE (7) (HH(L),K=1,LA) URITE (7) (HH(L),K=1,LA) (HIAT (F0,H) 1001 FURIAT (F0,H</pre>
	9789 1988 2012 2015 2015 2015 2015 2015 2015 2015	<pre>EAD (S,10^2) K(R(J) IF (KBR(J), HE, 0) GQ TO 400 J=J=1 C=====GRITE OUT NUMBER OF BRANCHES IN THIS RET URITE (6,2002) J URITE (7) J C=====FOR EACH DRANCH IN THIS SET NO 410 K=1,J CURAEKBR(K) M=YUHHH (KRRA) LASTAXOC(II=1)/KXD,1) LELA 'HH(L)=II IF (L,EQ,1) GC TO 406 405 L=L=1 M=II=KXD MM(L)=II IF (L,GT,1) GC TO 405 405 L=L=1 M=II=KXD MM(L)=II IF (C,EQ,1) GC TO 407 JRITE (6,2072) LA URITE (7) LA URITE (7) KBRA URITE (7) KBRA URITE (7) (MI(G),N=1,LA) URITE (7) (MI(G),N=1,LA) URITE (7) (MI(G),N=1,LA) URITE (7) (MI(C),N=1,LA) URITE (7) (MI(C),N=1,LA) URITE (7) (MI(C),N=1,LA) URITE (7) (MI(C),N=1,LA) URITE (7) (MI(C),N=1,LA) URITE (7) (MI(C),N=1,LA) URITE (7) (MI(C),N=1,LA) 410 CUNTINUE 420 STOP 1004 FURHAT (10) 1072 FURMAT (10) 1072 FURMA</pre>
	9789 1998 2011 2012 2015 2015 2015 2015 2015 2015	<pre>UNU = EAD (S,10^2) KrR(J) IF (KBR(J), HE, 0) GQ TO 400 J=J=1 C=====KR EAU NUMBER OF BRANCHES IN THIS SET URITE (G,2002) J URITE (7) J C=====FOR EACH DRANCH IN THIS SET nú 410 Kal, J KURA=KBR(K) M=UUNIH (KNRA) LA=HAX0((H=1)/KXD,1) L=LA 'H(L)=H TF (L'EQ,1) GC TO 406 405 L=L=1 M=H=KXD M=(L)=H TF (L'GT,1) GC TO 406 405 L=L=1 M=H=KXD M=(L)=H TF (L'GT,1) GC TO 405 406 IF (K'NE,13 GG TO 407 JRITE (G,2072) LA URITE (73 KBRA URITE (73 KBRA URITE (73 KBRA URITE (73 KBRA URITE (73 (HI:(13,N=1,LA)) URITE (73 (HI:(13,N=1,LA))) URITE (73 (HI:(13,N=1,LA)) URITE (73 (HI:(13,N=1,LA))) URITE (73 (HI:(13,N=1,LA)))) URITE (73 (HI:(13,N=1,LA))) URITE (73 (HI:(13,N=1,LA)))) URITE (73 (HI:(13,N=1,LA)))) URITE (73 (HI:(13,N=1,LA)))) URITE (73 (HI:(13,N=1,LA))))) URITE (73 (HI:(13,N=1,LA)))))))) URITE (73 (HI:(13,N=1,LA))))))))))))))))))))))))))))))))))))</pre>
	978 1978 2012 2014 2015 2015 2015 2015 2015 2015 2015 2015	<pre>UNU =EAD (S,10^2) K(R(J) IF (KBR(J), HE, 0) GQ TO 400 J=J=1 URITE OUT NUMBER OF BRANCHES IN THIS SET URITE (6,2002) J UFTE (7) J C=====FOR EACH URANCH IN THIS SET nû 410 Km1,J KURAMKBR(K) M=UUNIH (KNRA) LA=HAXO((H=T)/KXD,1) L=LA H=H=KXD M=(L)=H TF (L;Eq.1) GC TO 406 405 L=L=1 M=H=KXD M=(L)=H TF (C;Eq.1) GC TO 406 405 L=L=1 M=H=KXD M=(L)=H TF (C;Eq.1) GC TO 407 JRITE (6,2072) LA URITE (7) LA 407 UHITE (C6,2002) KBRA URITE (7) KBHA URITE (7) (HH:(L),N=1,LA) URITE (7) (HH:(L),K=1,LA) URITE (7) (HH:(L),K=1,LA) CUNTINUE 420 STOP 1001 FURHAT (1012)</pre>
	9989 19989 2013 2014 2014 2014 2014 2014 2014 2014 2014	<pre>UN = EAD (S,10^2) K(R(J) IF (KBR(J), HE, 0) GQ TO 400 J=J=1 URITE (0,2002) J URITE (6,2002) J URITE (7) J C=====FOR EACH DRANCH IN THIS SET NO 410 K=1,J CURA=KBR(K) M=4UHHH (KRRA) LA=HAXO((H=1)/KXD,1) L=LA 'HM(L)=H IF (L,Eq,1) GC TO 406 405 L=L=1 M=H=KXD M=(L=K) M=(L=K</pre>
	97897897899999999999999999999999999999	<pre>UNU =EAD (S,10^2) KrR(J) IF (KBR(J), HE, 0) GQ TO 400 J=J=1 C=====WITE OUT NUMBER OF BRANCHES IN THIS SET URITE (G,2002) J URITE (7) J C=====FOR EACH DRANCH IN THIS SET nú 410 Kal, J KURA=KBR(K) M=UUNIH (KNRA) LA=HAX0((H=1)/KXD,1) L=LA 'H(L)=H TF (L:Eq.1) GC TO 406 405 L=L=1 M=H=KXD M=(L)=H TF (L:GT.1) GC TO 406 405 L=L=1 M=H=KXD M=(L)=H TF (C,2072) LA URITE (7) LA 407 URITE (6,2072) LA URITE (7) KBRA URITE (7) KBRA URITE (7) (HI:(H),N=1,LA) URITE (7) (HI:(H),N=1,LA) URITE (7) (O, H) 1004 FURHAT (10112) END CHIEN CHIEN CONTINUE CONTAT (10112) END CHIEN CONTAT (10112) END CONTAT (10112) CONTAT (10112) CONT</pre>

. .

.

.

.

:

-

- 242 -

•
:

.

.

PROGRAM MODEX

APPENDIX G

•

.

.

	- and the second s
0009	CURRENT EDITIONIZO.03.74
0010	DIMENSION RE(0000),GA(12500)
0011	COMMON/DATA/NSET, NORAN, NO, NPIPE, NXDSET, NITEM, KITEM, NK
7012	COMMON/PARAM/PDA(9,20), MEND, JEND, MJ, ND, T. RP, ZO, OMAX, DMIN, DELTA
	CANDENSING BE/TD 1900.111, TOX (10.12), TDA. IDL. 10T. 10K. TOG. IDD. 10J.
0013	
0014	
0016	CoesesBECIES HYILMA WARAA SISKe
0016	NKE=8500
0017	NGAR12500
0010	ALL
0010	
0019	00 1 1914MAE
0020	1 KE(1)=0
0021	DE 2 ISLANGA
0022	
0023	DW 3 1=1.50
0024	DØ 3 Jel,11
0025	3 ID(1,J)=0
0026	DØ 4 1=1.10
0001	
0027	
0054	a IDA(1,J)=0
0029	NITEMBO
0030	KITEMa0
0031	NETAD
0012	
0010	ALL AND THE ALTA PARM MACHETTE TARE FILE
0033	Condergroup in Data Franchic tare rice
0034	CALL DATANT (REJGA, WREINGA)
0032	Commagranches have been sequentially prochedidesign cach in lurn,
0034	C EXCEPT FOR COMPONENTS OF X=DRAIN SETS WHICH ARE DESIGNED
0037	C BY CALLING SUBRPUTINE XDBET
0038	
0030	
0039	
0040	COOPOOLS THE BRANCH THE START OF A CHOSS DWAIN SELL
0041	IF (ID(NB,1),GT,100) GO TO 40
0042	CawageDESIGN THIS BRANCH TO A HINIHUM GRADIENT
0043	TE (NO 17 A) ER TA 18
0044	ANTIE (01001) ((10(11))10=1011)01=1450)
0046	WRITE (6,1002) (GA(I),I=1,NKE)
0044	WRITE (6,1001) IDA.IDL.IDT.IDK.IDG.IDD.IDJ.IDP.IDR
0047	AN FALL MEDAD (ME.GA.NKE.NGA)
	jo bet notar (neigera dere
0040	Ceessals Bint Orather Ceets
0049	NN#(IDG+Z+ID(NB,J))
0050	CALL COMS (KE.GA.NKE.NGA.NN.GA(IDG),KE(IDK))
0051	CPRODUCT OFTIMAL OFSIGNS FOR THIS BRANCH FOR RANGE OF D/S STATES
0089	PALL DEDING (ME.CA. NME. NCA.)
0032	
0123	COOLOOPHODUCE TRACE BACK UP BRANCH FOR RANGE OF D/S STATES
0054	CALL TRAIL («E,GA,NKE,NGA)
0055	20 IF (NB.LT.NBRAN) GR TO 10
0054	SO CALL TRACE (KE,GA,NKE,NGA,NGG,NKK)
0057	TE INVORET CT. O. PALL BUTBUT INF. CA. NVE. NGA
0058	
0050	
0059	
COCC.	CB RANCH is s tart p f a x=drain set: design this set
0061	40 NSETENSET+1
0062	CALL XDSET (KE.GA.NKE.NGA)
0041	TE (ND.) T. 4) CB TE 50
0044	AT LOGE 24 ADDE ANDE AND AN AND AND ADDE ADDE ADDE
	WITE (031001) HETSHERANGERTPESHEDETSHITEH
0009	WHITE (6,1001) HEND, JEND, HJ, ND
0044	WRITE (6,1002) ((PDA(I,J),Jm1,20),Im1,9)
0067	WRITE (6+1002) TARPAZD.DHAX.DHIN.DELTA
0068	WRITE (6.1001) IDA. IDL. 10K. 10G. 100. 10J. 10P. 10R. 10H. 10A
00AB	MD17F (A.1001) ((10(1,1),101.11).Tel.MBAN)
	HATOR IN INCLUSION SUBJECTED STATES AND
00/0	WRITE (0,1001) ((17A(1,J),JE1,12),IB1,WADBET)
0071	WHITE (8,1001) (RE(1),101,NKE)
0072	WRITE (6,1002) (GA(1),Is1,NGA)
0073	ConnoulPoate the current branch number
	ED NEWLASTOX (MEFT.6)=1
0075	
D076	1001 FBRMAT (2016)
0077	1002 FBRHAT (10F12.3)
0078	END
30.0	

•

END OF SEGMENT, LENGTH 477, NAME HODEX

:

•

.

:

•

SUBRBUTINE XDSET (KE,GA,NKE,NGA) C====THIS SUBRBUTINE PRADUCES A SET OF APTIMAL DESIGNS FAR A RANGE PF C STATE VARIANLES AT D/S END AF A X=DRAIN SET DIMENSIAN KE(NKE),GA(NGA) COMMON/DATA/NSET,NRAN,NO,NFIPE,NXDSET,NITEM,KITEM,NK COMMON/PARAM/PDA(9,20),MEND,JEND,MJ,ND,T,RP,ZD,DMAX,DMIN,DELTA COMMON/PARAM/PDA(9,20),MEND,JEND,MJ,ND,T,RP,ZD,DMAX,DMIN,DELTA COMMON/PARAM/PDA(9,20),MEND,JEND,MJ,ND,T,RP,ZD,DMAX,DMIN,DELTA COMMON/PARAM/PDA(9,20),MEND,JEND,MJ,ND,T,RP,ZD,DMAX,DMIN,DELTA COMMON/PARAM/PDA(9,20),MEND,JEND,MJ,ND,T,RP,ZD,DMAX,DMIN,DELTA 1 IDP, IDR, IDH, IDB IF (ND, GT, O) WRITE (6, 2000) NSET E-----IDENTIPY NUMBER OF X-DRAIN POSITIONS IN CURRENT X-DRAIN BET WENDETDIGHT ANDER JANAN AND GA RESUIRED FOR STONING TRACE AND COST DATA AT X-DRAIN POSITIONS LASTKERIDK-NYENDAMJOL £ LASTGADIDG+NXEND+(MJ+47-1 TF (LASTRE. GT. NKE-HJ) CALL HEBAGE (6.0) IF (LASTRA. GE. IDP) CALL HEBAGE (7.0) SET COSTS ARTIFICIALLY HIGH Ċ DE 1 INIDE,LASTEA 1 GA(I)#999999,9 C#####DEFINE START BF RUN PARAMETERS AND COSTS FOR EACH MEMBER OF SET KENB J=IDX(NSET,6) C====(for Each Branch In the Set) D0 10 I=1,J C====(IDENTIFY BRANCH NUMBER AND PIPE TYPE) C----(IDENTIFY BRANCH NUMBER AND FIFE TYPE) NB=IDX(NSET,I) NFIFE=HBD(ID(NB,1),10) IF (I.EG.1) NFIFEI=NFIFE C-----(DEFINE START OF RUN VALUES) L=ID(NB,10) CALL UPVAL (KE.GA,NKE.NGA.TUB,AREAUS,ZUB,DUS,GA(L)) C----(STARE START OF RUN VALUES) L=IDR+(I=1)=(HJ+4) GA(1)=TUB GA(L)=TUS GATL+1 JEAREAUS GA(L+2)=2US GA(L+3)=DUS C=====C@MBINE AND STBRE UPSTREAM CRSTS CALL SIZED (PDA, NPIPE, JI, DUS) CALL COMB (KE, GA, NKE, NGA, L+4, ZUS, JI) 10 CRNTINUE NBEK C----CANSIDER EACH X-DRAIN POSITION IN TURN STARTING AT U/S END NXBO 20 NX=NX+1 COMMOCENSIDER FIVE NEAREST U/S X-DRAINS, (CAN START WITH NO U/S X-DRAIN) NUX=MAXO(NX=6,=1) NCR035=0 40 NUXENUX+1 NCR8SSUNCROSS+1 NCR833DNCR05341 C----IDENTIFY UPSTREAM VALUES AND COSTS FOR CURRENT U/S X/D IF (NUX,EG.O) GO TO 50 K=IOG-(NUX-[]+(HJ+4)-1 L=IDR=1 LL=HJ+4 DE 45 KKEL-LL KaK+1 L=L+1 45 GAILJEGAIKJ Co---WRITE CONTENTS OF RE, GA, COMMONS "WHERE" AND "DATA" TO MAG TAPE 50 CALL HRITHT (RE, GA, LASTRE, LASTGA) CALL SUBNET (KE, GA, NKE, NGA, NUX) CALL SUBNET (KE, GA, NKE, NGA, NUX) C====RESTORE CONTENTS OF KE, GA, COMMONS "WHERE" AND "DATA" CALL READHT (KE, GA, LASTKE, LASTGA) C====STORE VALUES OF TIME, AREA, LEVEL AND DIAMETER AT X/D POINT NX KEIDG+(NX+1)+(MJ+4) IF (NCR038,EQ.1) GA(K+1)+GA(IDR+1) IF (NCR038,EQ.1) GA(K)=0 GA(K)=(GA(K)+FL0AT(NCR038+1)+GA(IDR))/FL0AT(NCR038) IF (NCR033.GT.1) GE TE 54 2810=-999999.9 DØLD=0.0 GØ TØ 55 54 20LD=64(K+2) DALDEGA(K+3) 55 ZLASTEGA(IDR+2) DLASTEGA(IDR+3) ZNEW=AMAX1 (ZOLD, ZLAST) DNEWHAMAX1 (DOLD, DLAST) GA(K+2)=ZNEH GA (K+3) BONEN IF (NCR033.GT.1) 00 TO 57 HOLDEHEND G8 TH 58 57 HBLD#IFIX((ZNEW=Z8LD+DELTA=D,D1)/DELTA) 56 HLAST#IFIX((ZNEW=Z4AST+DELTA=D,D1)/DELTA) CALL SIZED (PDA, NPIPEI, NALD, DALD) Call Sized (PDA, NPIPEI, NLAST, DLAST) NNEWEMAXO(NOLD, NLAST) JOLDENNEN-NOLD JLASTONNEW-NLAST

0084 0087 0068

0009

0090

0092

0093

0094

0096

0097

0100

0114

0115

0119 0120

0121

0122

0124

0125

0126

0128

0129

0135

0136

0130 0130 0140

0141 0142 0143

0153

0154

0156

0157

0160

0161 0162 0163

0164

0165

0166 0167 0168

0169 0170 0171

0172

.

. •

1

• •

t Ey

,

.

.

01.74	THE ALTER ALD AND LAST CASTS TO NEW REFERENCE GRID AND SELFCT CHEAPEST
0174	
0176	JI=MIHO(J+JØLD, JEND) ·
0177	J2#HIND(J+JLAST, JEND)
0178	D0 60 I=1,#END
0179	Mame ND+1 = I
0180]KE[NK+(NX+])#NJ+(J=])#NEND#N=1
0161	ICH IDC0(NK=1)=(NU+4)+(J=1)=NENU++>3
0195	
0103	
0100	IF (H1.LT.1) GB TB 601
0146	IG0LD=IDG+(NX=1)+(MJ+4)+(J1=1)+MEND+M1+3
0147	1x81De30x+(\\=134MJ+(J1+1)+MEND+M1=1
0188	C#LD=G4(1G#LD)
0189	KØLN=KË(IKËLD)
0190	601 CLASTAGGGGGG, G
0191	
0194	TULASTANUE-MJA(J2-1)AMENDAMA
0194	CLASTEGA (IGLAST)
0195	KLASTEKE(IKLAST)
0196	602 KE(1K)=0
0197	GA(IG)#9898989, 4
0198	IF (CLAST, LT, CFLD) GG TH GUS
0199	
0201	GAIGJECELD
0202	GA TA 60
0203	603 KE(IK)=KLAST
0204	GA(IG)+CLAST
0203	eg Centiniz Pul-tokajuv-11.001
n207	
0208	1K2=1K1+MJ=1
0209	1G2=1G1+HJ=1
0210	K1=K+3
0211	IF (ND,LT,1) G0 T6 65
5120	WRITE (0,2701) NX,161,162
0214	while to serve a two tips is the server as the server and the server as
0215	WRITE $(6,2004)$ [K1.]K2
0216	WRITE (6,2005) (KE(I),IWIK1,IK2)
0217	CIS THERE ANOTHER POSSIBLE POSITION FOR U/S X+DRAINT
0218	65 IF (NUX.LT.NX-1) GA TO 40
0219	CIS THERE ANATHER POSSIBLE POSITION FOR X-DRAINT
0220	IF (NI,LT, NIEND) GH TU ZU
0229	LEINITIAL VALUES DE LIMESANENSLEVELSDAMETEN AND COULS
0223	K=10G+(NXEND=1)+(MJ+4)+3
0224	L=IDT+NB=1
0552	GA(L)=GA(K=3)
0220	L=IDA+NB=1 0.40
0228	
0229	
0230	L=IDD+NB=1
0231	GA(L)=GA(K)
0232	08 70 I=1,43
0233	
0234	70 64(1)=64(4)
0236	IF (ND.LT.1) G0 T0 75
0237	WRITE (6,2006)
0238	WRITE (6,2004) IOK,IK2
0528	WRITE (6,2003) (KE(1),IEIDK,IKZ)
0240	TO DE DING INALE DATA FOR AND BET IN MADE THE FILLS
n747	1F (KITEN-ER-D) 68 78 98
0243	08 80 I=1, 417EH
0244	80 READ (8)
0245	90 DB 100 IEIDK, IK2
0246	100 WRITE (B) WE(1) WRITE (B) KE(1)B23.GA(TDD43)
0248	KITEMAKITEMAIK2=IDK+2
0249	RETURN
0250	2000 FORMAT (1H0/////SX,13HENTERED XDSET/DX,22HCROSSDRAIN SET NUMBERA,
0251	170) Deal chort flug av Jaugettna, sei 117760 deun te perse deltn. 16/101.
0252	SUCI FRAMAS (INUIDA, JONESIJAL BELUIJON VEHN IN LAUGE VARTALESTINA)
0254	2002 FARMAT (1X,F11,J,9F12.J)
0255	2003 FORMAT (5x, 5HTIME=, F12, 3, 6H AREA=, F12, 3, 11H MAX, LEVEL=, F12, 3,
0256	110H MAX, DIANA, F12, 3)
0257	2004 FORMAT (5%,41HREFERENCE NUMBERS STURED IN ARRAV KE FROM,16,
0238	ien to fio) 2008 Fridmat (11.15.1916)
0260	2005 FBRMAT (1H0/5%,27HDESIGN OF %/D SET COMPLETED)
0261	END

. . .

END OF SEGNENT, LENGTH ODD, NAME XDSET

SUBREUTINE WRITHT(KE,GA,NKE,NGA) DIMENSION KE(NKE),GA(NEA) COMMON/DATA/N(5),NITEM,KIFEM,NK COMMON/DERE/J(2331) COMMON/PEREMFFDA(9,203,MEND_JEND,NJ,ND,T,RF,ZD,DMAX,DMIN,DELTA JF(ND_&T_D) HRITE(6,2000) REWIND 4 JF(KITEM,ED.O) 50 TO 5 DB 3 I=:,KITEM 3 READ (6) 5 WRITE(0) (KE(I),I=1,NKE) WRITE(0) (KE(I),I=1,NKE) WRITE(0) (KII),I=1,S) WRITE(0) (N(I),I=1,S) WRITE(0) KITEM,KITEM,NK WRITE(0) KITEM,KITEM,NK WRITE(0) CJ(J,I=2,2331) IF (NITEM,EQ.O) RETURN REWIND 9 DB 10 I=1,NITEM READ (9) K,R,S 10 WRITE(0) K,R,S 2000 FORMAT (SX,14MENTERED WRITMT) RETURN FND 2920 0263 0264 0265 0266 0267 0269 0270 0271 0272 0273 0274 0276 0277 0278 0279 0280 0241 0283 RETURN D284 END END SP SEGMENT, LENGTH 17D, NAME WRITHT \$UBRBUTINE READHT (KE,GA,NKE,NGA) DIMENSIGN KE(NKE),GA(NGA) CBMHBN/DATA/N(5),NITEM,KITEM,NK CGMHBN/DATA/N(5),NITEM,KITEM,NK CGMHBN/DARAM/PDA(9,20),MEND,JEND,MJ,ND,T,RP,ZD,DMAX,DMIN,DFLTA IF (ND.GT,O) WRITE (6,2000) REWIND 8 IF (KITEM,E0,0) G0 T0 5 DG 3 I=1,KITEM 3 READ (8) B READ (8) (KE(I),I=1,NKE) READ (8) (KE(I),I=1,NKE) READ (8) (KE(I),I=1,S) READ (8) (KIC(I),I=1,S) READ (8) (I(I),I=1,S) READ (8) (I(I),I=1,S) IF (NITEM,KITEM,NK READ (8) (J(I),I=1,23) IF (NITEM,EQ,O) RETURN REWIND 9 DG 10 I=1,NITEM READ (8) K,R,S 10 WRITE (9) K,R,S 2000 F&RHAT (SX,14HENTERED READHT) RETURN FHD 0285 0287 0288 0290 0291 0292 0293 0294 0295 0296 0298 0299 0300 0301 0302 0303 0304 0305 RETURN 0307 END

END OF SEGMENT, LENGTH

.

.

1 : 3)

: 4

.

ł

Ŧ.

.

.

..

.

.

;

ł

1

· • • •

-

170. NAME READMT

.

I.

SUBRAUTINE DATANT (KF, GA. NKE, NGA) Componies Routine Reads the Design Parameters and Basic Data fund a Componies Tape File and Defines the Contents of Common "WHERE" DIMENSION KE(NKE), GA(NGA) COMMEN/DATA/MOET, NORAN, NO, NO IPE, NKOSET, NI TEM, KI TEM, NK COMMEN/PARAM/PDA (0,20), HENO, JENO, NJ, NO, T, RP, ZO, DMAX, DMIN, DEL TA COMMEN/HHERE/ID(200,11), IDX(10,12), IDA, IDL, IDT, IDK, IDG, IND, IDJ, 110P, IDR, ID4, ID8 NGEO NK=0 CONTREAD IN FIFE TABLE DATA DB 1 IC1.9 1 READ (7) (PDA(1,J),J=1.20) Contrered in time of Entry(Secs), Return Period(Years), Min and Max spacing READ (7) T, RP, DMIN, DMAX Contrered in State Variable Parameters, DIAG Level and ND, OF BRANCHES PEAD (7) MENT, TEND, MERAM C---- READ IN PIPE TABLE DATA CONVERTING AND IN STATE VARIABLE PARAMETERS, DIAG LEVEL AND NO. SP BRANCHES READ (7) MEND, JEND, ND, NBRAN IF (ND, GT, 0) MRITE(6,2000) IF (NBRAN, GT,200) CALL MEBAGE(15, NBRAN) MJAMENDJEND CONVERTING VALUE SP DEPTH INCREMENT DELTA DELTA 099099, S IF (MEND.GT.1) DELTA 20/FLBAT(MEND=1) CONVERT SALUE SP DEPTH INCREMENT DELTA DE 50 MBG1, NBRAN CUNNER OF GROUND LEVELS READ (7) (10(NB, 1), 141, 4) READ (7) (ID(NB,1),I=1.4) ---READ IN OFFSETS AND ABSOLUTE CHAINAGES NNWID(NB,3) 2 IF (NGA,LT,NG+4+NN+2+ID(NB,4)) CALL MESAGE (3,0) ID(NB,6)=NG+1 NG1=ID(NB,6) ME2=NG+1 NG2=NG1+NN=1 READ (7) (GA(N), Nung1, Ng2) 10(NB, 7)=NG+1+NN NGIEID(N8,7) NG2=NG1+NN=1 READ (7) (GA(N), NENG1, NG2) NG=NG2 --READ IN CHAINAGES AND INCREMENTS OF AREA ID (NB,8)=NG+1 NG1=ID(NB,8) NG2=NG1+NV=1 READ (7) (GA(N),N=NG1,NG2) ID(NB,9)=NG+1+NN NG1=ID(NB,9) NG2=NG1+NN=1 READ (7) (64(N); NeNG1; NG2) NG=NG2 NGUNG2 C----READ IN U/S BRANCH NUMBERS IF (ID(N8,2),E0,0) G0 T0 30 ID(N8,5) ank() NK1=ID(N8,5) . NK2sNK1+ID(N8,2)=1 IF (NKE,LT,NK2) CALL CALL HEBAGE (4,0) READ (7) (KE(N), NENK1, NK2) NKENK2 C---- READ IN GROUND LEVEL DATA 30 ID(N8.10)=NG+1 NG1=ID(N8,10) NG2=NG1+1D(N8,4)=1 READ (7) (GA(N), N=NG1, NG2) ID(NB, 11)=NG+1+ID(NB, 4) NG1=ID(NA,11) NG2=NG1+ID(N8,4)=1 READ (7) (GA(N), NENG1, NG2) NG#NG2 SO CONTINUE

••-

0308

0309 0310 0311

0312 0313 0314

0315

0317

0337

0343

0344

0346

0347

0348

0349

0350 0351 0352

0357

0358

0366

0348

0369

0370

0371

0372

0373

0374

0375

0376

0377

0378

.

3

1

1.1

1.1

1

.

!

.

.

.

- 248 -

0379	Connered IN NUMBER OF MORAIN BETS
0380	READ (7) WEDBET
0361	IF (NXDSET) 90,90,60
0382	Consumption EACH BETT
0383	60 ng 80 Ng1,NXDSET
0384	Connaread in Number of Branches in this set
0385	READ (7) IDX(N;6)
0386	NUMBEIDX(N,6)
0387	CPHEREAD IN NUMBER OF X-DRAINS
0388	READ (7) IDX(N,7)
0389	NUMXOBIDX(N,7)
0390	IF (NKE, LY, NK+NUNXD+NUHB) CALL HEBAGE (4,0)
0391	COMMENDER EACH BRANCHI
0392	De 70 Hel, NUHB
0393	CREAD IN BRANCH NUMBER
0394	READ (7) IDX(NAM)
0395	
0396	COMMENTAL IN THE NUMBERS OF THE MANS WHICH WAVE ADDRESS COMMENTAL
0397	
0390	NRZENRIANUTIONI Read and supported to the second second
0399	NEAD (1) (NE(NI) NI-NNI) NKE)
0400	
0401	
2405	
0403	AD TOVENKAT
0404	
0405	
0406	
0407	IDDE IDLANGWAN
0408	IOJEIOOANEMAM
0409	
0410	IDKENGASKJUJ 108-103-8-4 1-20
0411	
0412	IP (IDG, GT, IDF) CALL MESAGE (SID)
0413	
0414	
0413	100 ANIIS (BITOR) I'(LOV(I'))'ANIA
0416	WRITE (8,1003)
0417	
0410	110 MATE (0)1003 10(boy(1)3)1m01401
0420	1272 TZU HDTTE 14.1004) TTME.88.0MTN RMAY.MEND.JEND.70
0421	ALIUM Muit (01004) lintintintinuutinentier.
0422	DODA FRANT IST. LANENTEOFR DATANTI
0423	1011 FREMAT (501, 12WEEF) TREADY/SOX, 12(1H+)/10X, 4HTYPE, 15X, 5HCHVER,
0424	IIIX OMDRICHNERS 154 AMMIN. VEL 124 AMMAX. VEL 11X. OMMIN. GRAD./
0425	230X.3H(M).18X.4H(M) .18X.8H(M/8).15X.5H(M/8))
0426	1002 FRRMAT (AX, TA, F20, 3, F20, 4, 2F20, 3, F20, 4)
0427	1003 FORMAT (1HD. 10X. 4HTYPE, SOX. GHOTAMETERS)
0428	1004 FURHAT (IND/IDX. 19HTIME BF ENTRY (MINS) F11. 3.5%-
0429	118HARTURN PERIBO(VRS), F12, 3/10X, 22HMIN, MANHALE SPACING(M), F8, 3, 5%,
3430	222HHAX, HANHOLE SPACING(H), FB. 3/10%, IG. 15H VERTICAL ZANES, 10%, IG.
9431	311H PIPE 20NES, 14HZUNE DEPTH(H) ,FG.3/1H1)
0432	1005 FBRMAT (4x,I6,1577,3)
0433	END

•

	·
0434	SUBROUTINE SETUP (KE,GA,NKE,NGA,NK,NUX)
0435	DIMENSION RE(NRE), GA(NGA), CMAIN(100), AMEA(100), AMEA
D436	1030FF(3),030L(3),104(11) CRMMBN/DATA/NGET,NBBAN,NB,NBTOF,NYASFT,NTTFM,KTTEM,NK
0437	COMMENT CALAFORT (0, 20) MEND JENDAMI, NO. T. RP. 20, DMAX, DMIN, DFLTA
0438	CAMMEN/WHERE/ID(200,11), IDX(10,12), IDA, IDL, IDT, IDK, ING, IND, IDJ,
0439	I DR. IDR. IDR.
0441	IF (ND.6T.0) WRITE (6,2000)
0442	1F (ND.671) WRITE (6.1001) NUX, NX
0443	NITEMOD
0444	REWIND 9
0445	Commander INE WE, BY PARALLEL LINES IN THIS X/D BET
D448	NRARAS IDK (WEET, 6)
0447	16A#0
0448	IKEAU
0449	Conceptur lack urainers ling
0430	CONTRACTOR AND NO. OF DRAINAGE LINE
0451	NUMEIDX(NSET,I)
0453	CFIND OLD NUMBERS OF 1ST AND LAST MANHOLES
0454	20 IF (NUX,EQ.0) GB TA 30
0455	J=1Dx (NSE7, 147)+NUX=1
0456	HHS BKE (J)
0457	GE TU 40
Q430	30 MILLI 40 ILTOV/NETT.14714NV-1
0459	
0441	CIDENTIFY OLD CHAINAGES OF FIRST AND LAST MANHOLES
0462	L=ID(NUM,8)+MH1=1
0463	CHINGA(L)
0464	LeiD(NUM,8)+MHZ=1
0485	CH2864(L)
0480	Coossidentify Enginades and Areas for Each min
U487	NG 80 [SHU1.MU2
0470	L = I O (NUM, 83 + J= 1
0471	CHAIN(K)=GA(L)=CH1
0472	L=ID(NUM,9)+J+1
0473	AREA(K)=GA(L)
.0474	50 CONTINUE
0475	CIDENTIFY GROUND LEVEL DATA
0476	NGLEID(NUM,4)
0477	DE 60 JEI,NGL
0478	$\begin{bmatrix} 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 $
0480	
0481	
0482	DE 60 K=J,NGL
0483	
0484	L1=ID(NUM,10)+K+1
0485	
0450	XGL (L)=GA (L2)=CM1 701 /1 >=CA (L1)
0488	
0489	AC CONTINUE
0490	Community UPSTREAM BRANCH
0491	90 NU8=0
0492	IF (NUX,EQ_Q,BR,I,EQ,1) NUB=1
0493	
0496	
0495	
0497	IDN(2)=NUB
0498	JON(J)=N
0499	10N(4)=L
0500	IDN(5)=0
0501	IF (NUR,EQ.0) GB TO 95
0502	IKE#IKE+1
0503	IDN(3)BIKE
0504	RE(1RE)ENUPB
0505	TUN(T)=IUNTI TUN(T)=IUNTI
0507	
0508	1DN(9)=1DN(8)+N
0509	IDN(10)=1DN(9)+N
0910	10N(11)=10N(10)+L
0511	Coursestare G.L. AND AFFSET FOR D/S M/H
0512	DSGL(I)=ZGL(L)
0513	FF1D(MAH 0)+WH5+1
0514	DSWFF(T)=GA(LL)
0010	COOPERSTURE THE DATA FUR THIS BRANCH IN RE,GA AND COMMON WHENE
	COODOCSIGHT CHAINERS AND AREAS!
0518	K80 1040104444
0519	08 100 Jel-N
0520	IGAOIGAOI
0521	IGR#IGA+N
0522	KeK+1
0523	GA(IGA)#CHAIN(K)
0524	GA(1GB)#AREA(K)
0525	1DD CUNTINUE

وهادهانا يورده يويوا الداديان IGA=IGA+N 0527 K=0 0528 00 110 J=1.L 0529 IGABIGA+1 0530 IG8#IGA+L 0531 0532 KaK+1 GA(IGA)=ZGL(K) 0533 GA(IG8)=XGL(K) 110 CONTINUE 0534 0535 IIC CONTINCE IGABIGA+L Cenemo SPECIFY ID'S DS 120 Jate11 ID(I+J)BIDN(J) 120 CONTINUE 0536 0538 0539 0540 £ 0541 0542 1JB11+WPARA=2 0843 KENPARA+1 0544 DØ 130 I=11,1J K=K=1 0545 0540 ID(1,1)=11 ID(1,2)=2 0847 ID(I,2)=2 IF (I,EG,II) ID(I,2)=1 ID(I,3)=2 ID(I,5)=IKE+1 DØ 125 JJ=6,11 JK=1+2+(JJ=6) 125 ID(I,J)=IGA-JK p====DEFINE CMAINAGES AND AREAS GA(IGA+5)=0.0 GA(IGA+6)=ABS(DB0FF(K)=DS0FF(K=1)) GA(IGA+6)=ABS(DB0FF(K)=DS0FF(K=1)) 0548 0549 0550 0551 0552 0553 0554 0555 0556 C. 0557 0558 0559 GA(1GA+7)=0.0 GA([GA+8]=0.0 GA([GA+9]=056L(K) 0560 0561 GA([GA+10]=DSGL(K=1) 0562 BA(IGA+11)=0.0 GA(IGA+12)=GA(IGA+6) . 0563 0564 IGAEIGA+12 C----DEFINE U/S BRANCHES KE([KE+1]=K 0565 0566 0567 IF(1.60,11) 68 78 128 KE(1KE+2)=1=1 0568 0569 0570 128 IKE#IKE+ID(1,2) 0571 130 CONTINUE C----CREATE UPSTREAM BRANCH DATA FOR DUNNY PIPE DIS OF X/D SET 0572 0573 I=IJ+1 ID(1,2)=2 ID(1,5)=1KE+1 0574 0575 0576 KE(1KE+1)=1 0577 KELIKE+218IJ IKE=IKE+2 0578 NBRANET+1 0579 0580 NBRA=1+1 IF (NUX,EG.O) NORA=I+NPARA NB=0 0581 0582 0583 IDT#IGA+1 IDA=IDT+NBRA IDL=IDA+NBRA 1584 0585 IDD=IDL+NBRA 0586 0587 IDK#IKE+1 TOJEIDD+NBRA 0588 IDG#IDJ+HJ+NBRA 0589 C----CREATE DUNMY PIPE VALUES UPSTREAM OF X/D BET AS REQUIRED 0590 0591 Ast. NP=1 0592 L=IDR+1 0593 IF (NUX.NE.D) GE TE 140 LEIDP+1 0594 0595 NPENPARA 0594 140 00 160 II=1,** f=I+1 0597 0596 K=IDT+I=1 0599 GA(K)=GA(L+1) 0600 K=IDA+I=1 0401 GA(K)=GA(L+2) K=IDL+I=1 0602 0603 GA(K)=GA(L+3) K=IDD+I=1 GA(K)=GA(L+4) 0405 0606 0607 LELA K=IDJ=1+HJ=(I=1) D0 159 JI=1,HJ K=K+1 0608 0609 0610 0611 L=L+1 159 GA(K)=GA(L) JJ#JJ+1 NUMEIDX(NSET,JJ) 0613 0614 ID(I,1)#ID(NUH,1) 0616 160 CONTINUE RETURN 1001 FORMAT (1H0,5%,20HUPSTREAM CROSSDRAINE,I6, 124H DOWNSTREAM CROSSDRAINE,I6) 2000 FORMAT (1H0/////S%,13HENTERED SETUP) END 0617 0619 0620 0621 END OF SEGMENT, LENGTH 609, NAME SETUP - 251 -

. .

0022	SUBRAUTINE SUBNET (KE,GA,NKE,NGA,NUX)
0023	L BVER A RANGE OF D/S STATES
0625	DIMENSION RECORE), GAINGA)
0626	COMMEN/DATA/ NOET, NORAN, NO, NPIPE, NXDJET, NITET, TITET, TH
0627	COMMON/PARAM/PDA(4)2034 MENDS JENDS 45, NDA 101, 107, 108, 106, 100, 103, 103, 103, 103, 103, 103, 103
0628	110P.10R.10H.108
0630	IF (ND, GT, 0) WRITE (6,2000)
0631	Constantine wave been sequentially arderedidestan each in than
0632	OF 100 NBEI, NERAN
0633	CHARACTERIEN REANCH TE A MINIMUM GRADIENT
0634	CALL MONED (NEJON,MEJMON)
0636	NN=1DG+2+ID(NB,3)
0637	CALL CONB (KE, GA, NKE, NGA, NN, GA(IDG), KE(IDK))
0638	Communications for this branch for range of D/S states
639	CALL NBRUN (RE,GA,NRE,NGA) Denner Thefe Back up Branch for Dange of D/8 States
	TF (NB, FG.) CALL TRAIL (KE, GA, NKE, NGA)
642	100 CONTINUE
643	NB=NBRAN+1
644	NPIPE=HBD(ID(I,1),10)
645	Kain(NB,10)+10(1,4)=1 call - UBVAD (MF.CA NMF. NGA. TUB.A.T.D.GA(K))
847	GA(IDR) #TUS
648	GA(IDR+1)=A
649	GA (SOR+2)=Z
650	GA(tDR+J)=0
651	CALL SIZED (FDA, NPIFE, I, D)
632 463	CALL COMS (REJUGA,NREJNGA,10844,231) Compositive May isvei and May othe Nimber For Branch 1
454	ZNAWGA(IDL)
655	DHAGGA (IDD)
656	CALL SIZED (PDA, NPIPE, JHAX, DHA)
657	C++++FOR EACH D/S STATE, TRACE BACK TO OBTAIN REF TO STATE U/S OF SUBNET
658	C+RFAD TRAIL DATA FREM MAG TAPE
1639	REWIND 9
	NATION I
862	NKSNKS1
663	190 READ (9) KE(NK), REAL, REAL
664	REWIND 9
665	NK1=NKE+MJ
606	NGIEIDHAJ Funda Farm N/R Ryaya
166 8	
0669	J1=I~JEND+J
670	DDS#PDA(NPIPE,J1)
671	06 200 H=1, HEND
3072	NK18*K1+1 Kf2ut1}=0
0874	
0675	IF (GA(NG1),GT,999994,0) GA T# 200
0676	ZDS#Z=DELTA#FL#AT(H=1)
0677	CJDENTIFY U/S H AND J
0678	CALL BRINGE (GA,NGA,ZD3,DD3,ZMAX,JMAX,MP1FF,100,71,01) TE JUD OT 15 HDTE (A 3000) TOS.DDS,ZMAX,JMAX,MP1FF,1DJ,H1,J1,H,J
00/7	$IF = \{0, 0, 0, 1\} + MEND + M = 1$
0681	KE(NX1)MNUX4MJ+KE(X)
0682	200 CONTINUE
0683	IF (ND_LT_2) RETURN
0684	NKIBNKEPFJ41 Udive (4.101) MV(.NKF
0003	#FILE (#)10021 MM1/ME
0487	RETURN
0688	2000 FORMAT (SX, LAHENTERED SUBNET)
0689	1001 FORMAT (1HO/DX,47HREFERENCES ACROSS SUBNETWORK STORED IN KE FRAM ,
0690	116:4M TØ (10) 1000 FORMAT (17,18,1976)
	INNE ERWART (1919/19/0)
0691	3000 FRRMAT (1X,3F8,3,716)

,

•

0594	BUBRAUTINE AAIN (4.T.AI)
0695	Gannay VERETURN PERIAD(VRB), THTINE(HINS), RIHINTENBITY (HHAHR)
0696	R1=60,/T+((Y+T+202,26)++,28159-2,54)
0697	1F (RI=33.) 10,10,40
0678	10 #I=#HAX1 (#1-0-0)
0699	RETURN
0700	40 NI=0
0701	CITERATION LOOP FOR HOLLAND FORMULA
0702	50 NIENI+1
0703	FEALPG(15240,0/Y/RI/T+(RI+T/1524,0+0,1)++3,55)+1,0+0,0314+RI
0704	DF==1,0/RI+3,55/(RI+152,4/T)+,0314
0705	X=F/DF
0706	RI=RI=X
0707	IF (ABS(X)=0,01) 10,10,60
0708	60 IF (NI.LT.10) GO TO 50
0709	CALL HERAGE(17,0)
0710	END

.

END OF SEGMENT, LENGTH 98, NAME RAIN

 0711
 SUBREUTINE VELAC (SLBPE, DIAM.RK, V, 0)

 0712
 C====CALCULATES FLBHS PRBM COLEBROOK=WHITE FORMULA

 0713
 SG#SORT(SLGPE=DIAM)

 0714
 V==6.818+SG+ALGGIO(RK/3,7/DIAM+0,6471E=6/SG/DIAM)

 0715
 G=V+DIAM+0,7054

 0716
 RETURN

 0717
 END

.

:

.

END OF SEGMENT, LENGTH 53, NAME VELOC

.

.

1718	SUBREUTINE HERAD (KE, GA, HKE, NGA)
0719	Correcting SUBRBUTINE DESIGNS AN INDIVIDUAL BRANCH TO HIMING CHARTER
0720	DIMENSION KE(NKE),GA(HGA)
0721	COMMENDATA/NSET, MARAN, NA, NA JEFE, MAUSET, MITES, MILLS
0722	C_{MMBN} / M_{ME} C_{T} / T_{T} / $T_$
0723	
0726	TF (ND.GT.O) WRITE (6.2000)
0726	CoDEFINE PIPE TYPE
0727	NFIFE=M8D(10(NB,1),10)
0728	CHARGENTIFY GROUND LEVEL AT UPSTREAM MANHPLE
0729	N=ID(NB,10)
0730	
0731	Conseptate Stati P NON VELCES
0732	CALL RAIN (RP.TUS/60.0.RI)
0734	CHIDENTIFY NUMBER OF MANHOLES IN THIS BRANCH
0735	NMM=10(N8,3)
0736	ИАТЕЛЬКОМИНО (ИНИОТ) СИЛ
0737	IF (AKE, LT, AVAL) CALL MERAGE (B, UVAL)
0738	NATEIDEVNHA (MIAS)
0739	IF (MUR,LI,MVAL) CALL MERADE (MYMMAL) Metato:
0740	
0741	
0743	CA(NG1)=ZUB
0744	GA (NG2) WAREAUS +RI/3.6E6
0748	JJOKE(NK1)
0746	NeID(NB,8)
0747	MUSEID(NB,11)
0748	LUSEID(NB, 10)
0749	Kaid(NG,9)
0730	
0783	
0753	CHORENER FACH MANHOLE POSITION
0754	D9 50 1.02, NMM
0755	NSN 61
0756	NGI aNGI+1
0757	
0758	nn i Erre i e i Phùchaith i e i
0759	
0761	ZDS=ZUS-DIST+GMIN
0762	Ma MUS
0763	LaLus
0764	30 M##+1
0765	
0700	
0768	ZDS#AMIN1 (Z=C8VER=(CHDS=X)+CMIN_ZDR)
0769	1F (CHQ8,GT,X+0,001) GR TO 3D
0770	SLOPE=(ZUS=ZDS)/DIST
0771	KaK+1
0772	TAREADS=AREAUS+GA(K)
0773	CCALCULATE REDUINED DIAMETER
0774	35 DISPORTIPE, JJ)
0775	17 (JJ,67,67,67,97,77,97,197,97,97,97,97,77,77,77,77,77,77,77,77,7
0777	1P (V-PDA(NPIPE,3)) 36,36,36
0778	36 1F (V-PDA(NPIPE,4)) 42,42,43
0779	42 TFETUS+DIST/V
0780	CALL RAIN (RP,TF/60.0,RI)
0781	FLOWEARCADS+RI/3.6F6
0782	IF (FL8+,LE,0) G0 T0 40
0783	37 JJ#JJ+1
0784	
0783	JJ VEFDA(MFIFC,J)
0789	TFUTUSFUS FALL DATM (DD.78/40.0.01)
W* W*	PMPP

•

•

....

0788	FLBHMAREADS=RI/3.6E6
0789	Q=0,7854+DI+V
0790	17 (Qofler) 37,39,39
0791	39 CALL FINDG (DI,V,PDA(NPIPE,2),8LOPE)
0792	ZD\$=ZU8=DI\$T+SL&PE
0793	Mamus
0794	LELUS
0795	41 MeM41
0796	X=G4(M)
0797	1=1+1
0798	Z=GA(L)
0799	2D\$=AHIN1 (Z=C8VER= (CHD3=X)+8L4PE, ZD\$)
0800	IF (CHD3.GT.X+0.001) GB TB 41
0801	
2005	43 VS-DA(NFIPE,4)
0803	
0604	
0000	
0808	AA FALL STNDG (DI.V. PDA(NPTPF. 2). 31 8973
0809	71184F 48703+8L8FE +0757+70
0810	15 (ZUSNEW, GT, ZUS) GO TO 40
0811	IF (NHH.GT.2) CALL HERAGE (19.1)
0812	GA (NG1=1) BZUSNEW
0813	IF (ND.GT.D) WRITE (6.1001) TUBNEW
0814	CSTURE DAWNSTREAM ZANE LEVEL FLOW AND PIPE SIZE
0815	40 GA(NG1)=ZDS
0816	GA (NG2) #FLOW
0817	KE (NK1)BJJ
0818	CREDEPINE UPSTREAM VALUES
0819	CHUS=CHOS
0820	ZUS=ZDS
0821	AREAUSBAREADS
0822	TUSETF
0823	MU\$zM
0824	
0825	50 CANTINUE
0426	CSTORE END BF BRANCH VALUES
0827	
0020	
0027	
0030	98 (7) = A (2) 9 M = Tol (A) 4 = (
0812	
0811	
0834	
0835	IDZ#IDG+2+N4H=1
0836	IDY=IDK+NMH=1
0837	IF (ND, LT, S) RETURN
0838	WRITE (6,1002) IDT, 102, 10K, 10Y
0839	WRITE (6,1001) (GA(1),1=107,102)
0840	WRITE (6,1002) (KE(I),I#IDK,IDY)
0841	RETURN
0642	1001 FØRMAT (1X,F11,J,9F12,3)
0843	1002 FORMAT (11,13,1916)
0844	2000 FURMAT (3141JAENIERED MGMAU)
0843	

.

-

. .

:

0546	SUBROUTINE UPVAL (RE, GA, NKE, NGA, TUS, AREANS, EUS, DUS, GLUS)
0847	COMMENTARY AF RUN VALUES OF TIME AREA LEVEL AND DIAMETEN
0848	C FOR BRANCH NB
0849	DIMENSION KE(NKE),GA(NGA)
0850	COMMON/DATA/NSET, NORAN, NO, NDIPE, NXDSET, NITEH, KITEM, NK
0851	COMMON/PARAM/PDA(9,20), MEND, JEND, MJ, NO, T, RP, ZD, DMAX, DWIN, DELTA
0852	C8HHRN/WHERE/ID(200,11),IDX(10,12),IOA,IDL,IDT,IOK,IDG,IDD,IDJ,
0853	1 IDF, IDF, IDF, IDF
0854	IF (ND.GT.O) WRITE (6,2000)
0855	
0436	AREAU8#0.0
0857	_ 215=999999,9
0850	
0859	CIDENTIFY NUMBER OF UPSTREAM BRANCHES
0860	5 NU6=ID(N6,2)
0861	IF (NUA,EQ.Q) BE TO 20
0962	N=I0(N4,5)-1
0063	Commerter Each Upstream Branch
0864	06 10 Te1, NU6
0865	
0866	CJDENTIFY BRANCH NUMBER AND UPSTREAM VALUES
0867	
0866	MEIDT+NBRA-1
0869	TUŞBAMAX1 (TUŞ,GA(M))
0870	MEID4+NBR4=1
0871	AREAUS=AREAUS+GA(H)
0872	MEIDL+NBRA-1
0873	ZU5=4MIN1(ZU8,GA(M))
0874	MaIDD+NBR4=1
0875	DUSBAMAX1 (OUS,GA(M))
0876	10 CONTINUE
0877	20 ZUS#AMIN1(ZUS;GLUS=POA(NPIPE;I))
0878	GA(IDG)=ZUS
0879	CALL SIZED (PDA, NPIPE, KE(IOK), DUS)
0880	IF (ND.LT.1) RETURN.
0881	ŴRITE (Ĝ.2001) NO,NUB,TUS,AREAUS,ZUS,DUS
0882	RETURN
0883	2000 FURHAT (5x,13HENTERED UPVAL)
0884	2001 FØRHAT (5%,6HBRANCH,I6,I8,23HBRANCHEB_UPSTREAH,TIME=,F12,3,
0445	15HAREAU,F12,3,6HLEVELu,F12,3,5HDIAM=,F12,3)
0886	END

.....

-

END OF SEGMENT, LENGTH 212, NAME UPVAL

•

•

.

:

.

0887	SUBRAUTINE, COMA (KE,GA,NKE,NGA,NN,ZTDS,NDDS)
0886	DIMENSTON KE (NKE), GA (NGA)
0569	COMMON/DATA/NSET, NORAN, NO, NOTPE, NXOSET, NITEM, KITEM, NK
0890	COMMON/PARAM/PDA(9,20), MEND, JEND, HJ, NO, T, RP, 20, NAX, DAIN, DELTA
0891	COMMON/WHERE/ID(200,11),IDX(10,12),IDA,IDLe IDT+ID4+IPG+ID1+ID4+ID4+ID4+ID4+ID4+ID4+ID4+ID4+ID4+ID4
069.2	110P, 10R, 10H, 10H
0893	14 (ND.67.1) HEITE (4-2000)
0894	CoosesET COST OF ARRIVAL AT FIRST MANNULE IN NUM TO CONS
0895	104 jū žeta
0896	1010/WRL
0897	
0848	NUMBIC(NT/2)
0844	tr (JFND_FQ_1) 60 TO 100
0000	CONTROL OFFINE INFEASIBLE PIPE ZONES AT THIS UPSTREAM END OF THE NETWORK
0902	J2eJENDe1
0903	K=NN=1
0904	D# 20 J=1,J2
0905	D8 20 ME1, MEND
0906	
0907	20 GA(K)=999999,9
0908	GU TW 100
0909	JU KEJU(NB/J/4) Composition (NC) (NC) (NC) (NC) (NC) (NC) (NC) (NC)
0910	
0912	C(FIND U/S PIPE NUMBER. TYPE AND MAX. DIAM. NUMBER)
0913	Kak+1
0914	NR#KE(K)
0915	NPIPUSened(IDINR,1),10)
0914	LaloD+NR+1
0917	CALL SIZED (PD4, HPIPUS, JI, GA(L))
0918	L=IDL+NR=1
0919	ZTUS=GA(L)
0920	DG 80 J=1, JE40
0951	JJENDOS-JEND+J
0922	[F (J] 62.0) GB TO 40
0953	
0929	45 TT 60
0943	
0927	
0928	
1926	De 60 JUS#1.JEND
0930	JJUS=JI-JEND+JUS
0931	IF (JJUS, LT, 6) G8 T8 60
0932	DUS=PDA(NPIPUS+JJUS)
. 0933	IF (DU3,GT,DD3+0,D01) 49 18 63
0934	00 55 MUSE1, HEND
0935	ZUSHZTUSHPLHAT(MUSH))+DELTA
0936	
0437	
0950	55 CONTINUE
0940	60 CANTINUE
0941	65 LENN-1+(J-1)+MEND+M
0942	GA{L)WAMIN1{GA(L)+CBSTA,994994,9}
0943	70 CONTINUE
0944	60 CONTINUE
- 0945	GU CENTINIE
0946	LOG IF (NO.LI.I) WEIDEN
0947	HUTTE (A. 1001) NE. WIE. NU. 1
0949	WRITE (6.1002) (GA(1).Imnn.L)
0950	RETURN
0951	110 LENN-1+(J-1)=MEND
0952	DE 12D KEWES, MEND
0953	L=L+1
0954	120 GA(L)=4999999,9
0955	GU TE 80
0956	1001 FORMAT (1HD,5%,20HU/S COSTS FOR BRANCH,16,19H (NG, AF U/S PIPES=,
0957	
0935	1002 FURTAI (LAFLISJOFLES)
0040	SOOD LAMMAL (21) THEN FRED CAMP)
4799	E40

ц., н. **н.** ч

END OF SEGMENT, LENGTH 407. NAME COMO

,

•

```
SUBROUTINE NARUN (KE.GA.NKE.NGA)
0961
                                 SUBRAUTINE NAKUN (KEJGA, MEJAK)
DIMENSION KE(NKE),GA(NGA)
COMMON/DATA/NSET,NBRAN,NB,NPIPE,NXDSET,NITEM,KITEM,NK
COMMON/PARAM/PDA(9,20),MEND,JEND,MJ,ND,T,RP,ZD,DMAX,DMIN,DFLTA
COMMON/MHERE/ID(200,11),IOX(10,12),IDA,INL,IDT,IDK,IDG,IDD,IDJ,
0962
0963
0964
n965
                               LOP. IDR, IDR, IDR
LOGICAL NEWN
IF (ND.GT. 0) - WAITE (4.2000)
LENDEID (ND.4)
0966
0967
0968
0970
                                  NENDEID(NB,3)
                                  IDH#10G+2+NEND
0971
                                  ID8=IDK+NEND
0972
0973
0974
0975
                                  ID9= TOH+NEND+45
                                  IDH=IDS+(NEND=1)+HJ
                                  KC8ST=TDH+HJ=1
0976
                                  KREF #108-1
0978
                                  N=1
                        C----DEFINE D/S STATE (MANHOLE N)
0979
                             10 N=N+1
                        C---- (DEFINE PARAHETERS DEPENDENT ON "N")
0980
                        C----- (DEFINE PARAMETERS
NEWNS, TRUE,
C----- (RELATIVE CHAINAGE)
KEID (NR, 8)+N=1
XDS=GA(K)
0981
0982
0983
0984
                            ---- (TRP OF ZONE)
Keing+N=1
0986
                                  ZTDS#GA(K)
0987
                        C---- (FLOW)
0988
0989
                                  KEIDG+N=1+NEND
                                  QNEGA (K)
0000
0990
0991
0992
0993
0994
                             ---- (HAX, PIPE DIAN, NUMBER)
                                 Katok+Nel
                                  NDNUKE (K)
                                  J=0
0995
                        C----DEFINE D/8 STATE (DIAMETER J)
                            20 JEJ+1
JPIPEENDN-JEND+J
0996
0997
0998
                                  Ha0
                        C----DEFINE D/8 STATE (LEVEL H)
0999
                        30 MBM+1
C=====0/S STATE (M,J,N) IS NOW DEFINEDISET COST OF ARRIVAL AT (M,J,N)
C ARTIFICIALLY HIGH
1000
1001
1002
                                  KCOST=KCBST+1
1003
                                  KREF=KREF+1
GA(KC8ST)=999999.9
1004
1005
                                  KE (KREF)=0
1008
                        C----DØES JPIPE CØRRESPOND TØ A REAL PIPE?
IF (JPIPE,LT.6) GØ TØ 300
ZDS#ZTD#=FLØAT(H=1)+DELTA
1007
1008
1009
                                  NNBO
1011
                       C----DEFINE UPSTREAM STATE (MANMOLE)
NREFED
1013
                             40 NNENN+1
                              K=ID(NR,0)+NN=1
1014
                       XUSUGA(K)
C+----IS DISTANCE BETWEEN MANNALES PERMISIBLE?
IF (XDS=XUS,GT,DMAX+0,001,AND,NEND,GT,2) GØ TØ 40
IF (XDS=XUS,LT,DMIN=0,001,AND,NEND,GT,2) GØ TØ 290
NREF=NREF+1
inis
1016
1017
1018
1020
                                  KEIDG+NN=1
1021
                                  IF (NKE,LT,IDH=1+J=NREF)
IF (NGA,LT,IDG=1+2=NREF)
NK=IDH+3+(NREF=1)
                                                                              CALL MESAGE (10,0)
CALL MESAGE (11,0)
1023
                                                                                    . .
1024
                                  NG=IDQ+2+(NREF=1)
                       ZTUS=GA(K)
IF (-NGT.NEWN) GO TO 150
C====DEFINE PARAMETERS DEPENDENT ON COMBINATION OF N AND NN
C=====(DEFINE INTERNEDIATE GROUND LEVELD)
1025
1024
1027
1028
                                 DØ 50 L=2,LEND
K=ID(NB,11)+L=1
IF (GA(K),GT,XUS+D,001)
1029
1030
                                                                           64 TB 60
1031 1032
                             50 CONTINUE
1033
                             60 L1=L
                                  KGXLISK
1034
                                 KGZLISK-LEND
L2=L1=1
1035
1037
                                  A=0,0
1038
                                  LEVEL+1
                                  IF (L1.E0.LEND) GU TO 140
IF (GA(KGXL1+1),GT,XDS+0,001) GO TO 140
1039
1041
                                  13=11+1
                                 DØ 70 L=L3,LEND
K=ID(NR,11)+L=1
IF (GA(K),GT,XDS=0,001) GØ 70 80
1042
1043
1044
1045
                             TO CONTINUE
                             80 L2=L=1
KGXL2=ID(NR+11)+L2=1
1044
1047
1040
                                 KGZLZ#KGXLZ=LEND
```

and a second second

	CONTRACTOR AREA OF LANG SECTION ADDRES STRAIGHT LINE FROM NN TH N
1.000	
1050	¥84@\${K@\$C141}={@\${K@\$C1}=@\${K@\$C\$+f}}=0\$3
1051	DØ 85 LL#L1/L2
1052	#C#1 + #ID (NB.11)+LL#1
1023	
1054	85 AEA+GA(KGXLL)+(GA(KGZLL+L)+GA(KGZLL+L)+GA))+00,0
1055	
1000	TA CORIND LEVEL CANCAVE CANVEX OR VARIABLE
1034	
1057	Q278554 [Q7 (K07744])_AA (K07714 [)]\ (FN94 YN9]
1058	DØ 130 L=L1,L2
1080	KCNI NTD (NA. 11)+L#1
1034	
1000	
1001	PSLBPE#(GA(462L)=GA(462L)=1))/(G+(462L)=x08)
1062	IF (PSLOPE,LT.GSLOPF=0.00001) GO TO 90
1043	TF (PSLBPF.LT.GSLGPE+0.00001) GC TC 130
1003	CA TE (110,100,130) (EVE)
Inde	
1065	90 GU TH (120,140,100) CEVEL
1066	100 LEVEL#4
1067	69 10 140
1040	110 IFVELAS
1000	
1009	
1070	120 LEVEL#Z
1071	130 CONTINUE
1077	COMPACT AND STARF GROUND CONDITIONS AND DISTANCE BETWEEN HANNPLES
10/3	the realization that a state of the state of
1074	140 KEINKIELEAFT
1075	4E{4Ka}}e
1074	KF (1)K + 2) BL 2
1010	
1077	GA (NG) X D S X D S
1078	GA(NG+1)=A
1079	TF IND.CT.t3 WRTTF (6.2009) NN.N.NREF.LEVEL.LI.L2.GA(NG).A
1000	COMPANY CONTRACTOR AND ANALY CONTRACTOR
1000	Change in Californ State (Die Alle State Californie)
1091	120 DISTEGAING)
1082	AREA=GA(NG+1)
1081	
1004	6.3 #RE [(K + 3.)
1085	L2=XE (4X +2)
1084	
1007	100 23-33-1
1088	KEIDK+WN+1
1089	JJP1PFEKE(K)=JEND+JJ
1000	
1000	
1001	CAAAAOFAINE NASIKKAM BIVIE (CMAMN PEAFF JUL)
1092	MMEO
1093	17D MMBHM+1
1004	TO BE THIS IS A FUNDED AND TAKE IN A AMENDAMM
1046	
1000	CONSECUTION LEASEDILITY HE SELUTION
1080	Conner(U/S STATE FEASIBLET)
1097	IF (GA(ICBST).GT.999999.0) GE TE 250
1098	Conser (PIPE SI BPE WITHIN RESTRAINTET)
1000	
1077	
1100	2Cabfa(102-102)\0121
1101	IF (SLAPE.LT.POA(NPIPE.5)=0.00001) GO TO ZOD
1102	C(PIPE CAPACITY SUFFICIENTY)
1103	PALL WELST ISLEDE, MALINETHE, THTHEY, MALINETEF, PI, VEL, OFULLI
1104	17 (GPULL-LI-GH) 69 19 200
1105	C+«(VELUCITV ACCEPTABLEY)
1106	1F (VEL.LT.PDA(NPIPE.3)=0.001) GA TA 260
1109	TE (VEL GT. PDA(NPTPE, A)AO OOR) GA TA 280
1107	ar (tegetgrowtingtegetgrowg) weite but
1100	Chanad (Debin of Code (Singing Alegaical)
1100	CM 10 (540,140,240,140)/FEAEP
1110	180 DØ 200 L=L1.L2
	KGXI BTDING. 113al et
1115	
1113	IF (ZUS=SLUPC+(GA(KGXL)=XUB).GT.GA(KGZL)=PDA(NPIPC+I)+D.OO)) GH
1114	170 250
	200 CANTINUE
1110	PROPAGATION TO FFASTALE A PORT AND PROPAGE STTM DESUTATE PARADERY
1110	Provide Aline for the second and the stateme with a stateme fully and the second second fully and the second
1117	240 KGZLIBIDUNNIUJ+L1+1
1118	KGZL2#ID{N##107+L2#1
1118	CALI CASTIT (JPIPE=5.AREA.GA(KGZL1=1)=ZUS.GA(KGZL2=1)=ZDS.DIST.C)
IICU	Latenal (total)
1121	IF(C,GT,GA(RC03T)=0.001) GD Y0 200
1122	GA (KC@ST)=C
1121	KE (KREFTE(NNUL)AMJA(JJO1)AMPNDAMM
1163	TELENCY JELEVIE JELEVIE JELEVIE VILLEN UN TELEN VAR VIR TAR TIR.
1144	It (MARIET were continued with an interaction of the continued of the cont
1125	lpda(npipe,jpipe),pda(npipe,jjpipe),kref,kcøst,ke(kref),ga(kcøst)
1124	CHOVF ON TO NEXT U/S STATE
1127	250 1F (HH.I T. MEND) 68 78 170
	AGA TE LINELIE AND TO TO TO TO TO TANK AN AN AND AND AND AND AND AND AND AND A
1140	CO IP (JFIFCHIGUFFFCHUGJGELIGJEND) 68 IN 180
1120	IF (NN.LT.N-13 G0 T0 40

...

.

•

۰

1130	CHUVE ON TO NEXT DIS STATE
1131	290 NEWND.FALSE.
1432	300 1F (M.LT.HEND) GR TØ 30
1135	TF (J.LT.JEND) GR TO 20
	TE (N.LT. NEHD) GO TO 10
4434	FREE DOWNSTREAM COSTS FOR THIS BRANCH
1170	TeTOHA(NENDel]eNJe1
1130	detole(NBel)#HJ#1
113/	
1138	
1139	Jog Jan Kat Mi
1140	DE JIO KEIJHA
1141	
1142	J=J+1
1143	GA(J)=GA(I)
1144	310 CONTINUE
1145	IF (ND,EG,O) HETUHN
1146	WRITE (6,2005) NB
1147	WRITE (6,2006) (GA(J),JeJA,J8)
1148	I=I0T+N8=1
1149	J=ID4+N8+1
1150	
1151	
1152	WRITE (6,2007) NB,GA(I),GA(J),GA(K),BA(L)
1153	RETURN
1154	2000 FORMAT (5x,13HENTERED NORUN)
.1155 .	2002 FRRHAT (LOX, 15HGROUND FROM M/H, 14,6HTO M/H, 514,2(PO.3)/5X,
1150	1108HN J H NN JJ HH XDS XUS ZDS ZUG
1157	2 555 DUS KREF KCOST BEF COST)
1156	2004 FORMAT (1x,615,6F9.3,316,F12.3)
1159	2005 FORMAT (1HO.SK.20HOFS COSTS FOR WANCH.15)
1140	2006 FORMAT (1X-F11-3-9F12-3)
1141	2007 FARMAT (1HD-5%,7HARANCH -16,26H DOWNSTREAM VALUES: TIME=,F12.3,
1142	17H A@FAN. F12. 3. AN LEVEL N. F12. 3. 7H OLANG. F12. 3//)
1149	

-

END OF SEGMENT, LENGTH 1083, NAME NORUN

• · ·

.

 1164
 SUGRAUTINE COSTIT IJ, AREA, DUS, DDS, D1ST, COST)

 1165
 DEPTHE (DUS+DDS)/2, 0+AREA/DIST

 1166
 GR TE (10,20,30,40,30,60,70), J

 1167
 10 COST=DIST=(2,0+4,1+DEPTH)+30,0+70,0+0US

 1168
 RETURN

 1169
 20 COST=DIST=(2,0+4,1+DEPTH)+30,0+70,0+0US

 1169
 20 COST=DIST=(5,7+4,1+DEPTH)+30,0+70,0+0US

 1170
 RETURN

 1171
 30 COST=DIST=(12,3+4,4+DEPTH)+30,0+75,0+0US

 1172
 RETURN

 1173
 40 COST=DIST=(12,3+4,4+DEPTH)+30,0+80,0+0US

 1174
 RETURN

 1175
 50 COST=DIST=(15,9+4,7+0EPTH)+30,0+85,0+0US

 1176
 RETURN

 1177
 60 COST=DIST=(19,7+6,0+DEPTH)+30,0+85,0+0US

 1176
 RETURN

 1177
 60 COST=DIST=(19,7+6,0+DEPTH)+30,0+90,0+0US

 1176
 RETURN

 1177
 60 COST=DIST=(19,7+6,0+DEPTH)+30,0+90,0+0US

 1177
 70 COST=DIST=(19,7+5,3+DEPTH)+30,0+90,0+0US

 1180
 RETURN

 1181
 END

END OF SEGMENT, LENGTH 139, NAME COSTIT

.

:

SUBRAUTINE TRAIL (KE,GA,NKF,NGA) C-----TRACES BACK UP A BRANCH FOR EACH D/S STATE AND STORES THE THACE BN C MAGNETIC TAPE FILE DIMENSION KE(NKE),GA(NGA),KTEMP(200),ZTEMP(200),DTEMP(200) COMMON/DATA/NSET,NARAN,NB,NPIPE,NXDSET,NITEM,KITEM,NK COMMON/DATA/NSET,NARAN,NB,NPIPE,NXDSET,NITEM,KITEM,NK COMMON/DATA/NSET,NARAN,NB,JEND,MJ,ND,T,RP,ZD,OMAX,DMIN,DELTA COMMON/WHERE/ID(200,11),IDX(10,12),IOA,IDL,IDT,IDK,IDG,IDD,IDJ, 1182 1183 1184 1185 1188 1191 1192 1193 NN=0 DØ 50 I=1,MJ C====IDENTIFY TRACE BACK REPERENCE FROM END MANHOLE 1194 1195 1196 KeK+1 HJNEKE (K) COMMENTE REFERENCE TO TAPE 10 IF (MKDSET.GT.O.AND.MJN.GT.MJ) GO TO 20 COMMENTER IF (MJN.NE.O) GO TO 30 ZHO.O DOC 0 1197 1196 1200 1505 1203 1204 1205 1206 LUIDK+N+1 1207 JMAX=KE(L) JPIPE=JMAX=JEND+J 1208 1209 1210 DUPDA(NPIPE, JPIPE) 1211 LaIDG+N=1 1515 Z=GA(L)=DELTA+FLØAT(M=1) 40 WRITE (9) MJN, 2, D NNENN+1 1513 1213 1214 1215 1216 1217 1218 1219 IF (MJN.LE.MJ) GB TB 50 C==== IDENTIFY NEXT TRACE BACK REFERENCE 20 Laidb=mj+mjn=1 MJN=KE(L) MJNEKE(L) GB TØ 10 S0 CONTINUE C====CHECK GN CONTENTS OF MAG, TAPE IF (M0,LT,2) GO TO 40 DØ 60 I=1,NN 60 RACKGPACE 9 DØ 70 I=1,NN READ (0) KTEMP(I),ITEMP(I),DTEMP(I) 70 CONTINUE WRITE (6,1000) (KTEMP(I),I=1,NN) WRITE (6,1001) (ZTEMP(I),I=1,NN) WRITE (6,1001) (DTEMP(I),I=1,NN) WRITE (6,1001) (DTEMP(I),I=1,NN) WRITE (6,1001) (DTEMP(I),I=1,NN) WRITE (6,1001) (DTEMP(I),I=1,NN) RETURN 1220 1221 1222 1223 1224 1225 1226 1228 1230 1231 RETURN 2000 FORMAT (5x,13MENTERED TRAIL) 1000 FORMAT (1x,15,1916) 1001 FORMAT (1x,F11,3,9F12,3) 1232 1522 1234 1235 FND 1234 END OF SEGMENT, LENGTH 284, NAME TRAIL

· ·

	AUROAUTANE BRADER (CA. NCA. 708.008.7MAY.JNAY.MP[#E2.MN.#].J])
1237	
7526	DIRENSION CALINGAL OCT WEND TEND AT NO C RD. TH. THAT. DELTA
1524	CEMMEN/PARAM/POR(9,20), HEND, JEND, HO, HO, HO, HO, HO, HANDER, A MANAMELS
1240	Commentings the values of MI,JI FAR AN UPSTREAM PARE AT A PARTIC PIPE
1241	CCARRESPONDING TO LEVELS AND DIAMETERS ZOB, ODS WE OUTGOING THE
1242	COMPASSION THAT U/S LEVEL .GE, ZDS, AND U/S DIAM .LE. DDS AND U/S CHST IS "IN
1243	IF (ND_GT_1) WRITE (6+2000)
1244	H1=0
1946	.11mn
	C4571164000000. 0
1440	
1247	ve svav tenna t
1240	
1249	IF (JA.LT.O) GW TO CO
1250	DUSEPDA(NPIPEZ, JA)
1251	IF (NUS.GT.005+0.001) GW TW 30
1252	DE 10 HW1, MEND
1253	ZUS#ZMAX#DELTA#FLØAT(M#1)
1254	IF (ZUS.LT.203=0.001) GB TO 20
1255	KENN+(JOI)+MEND+MOI
1256	C887=GA(K)
1257	1F (COST.GE.COSTUS=0.001) 60 70 10
1258	COSTUSECOST
1258	
1240	.11.8.7
1261	
1242	SA PANTANIP
1941	
1964	an trunt its lighter this .
1200	CADA LANDEL (949TANCAICHEA OKTARE)
1544	E ND

مادم المحمدين

END BP SEGMENT, LENGTH 165. NAME BRIDGE

 1266
 SUBROUTINE SIZED (PDA, NPIPE, J, D)

 1267
 C====FINDS THE PIPE NUMBER J CORRESPONDING TO BR GREATER THAN DIAM D

 1268
 DIMENSIGN PDA(0,20)

 1269
 DB 10 Ja6,20

 1270
 IF (PDA(NPIPE, J)_GE, D=0,0D1) RETURN

 1271
 10 CONTINUE

 1272
 CALL MESAGE (16, NPIPE)

 1273
 END

END OF SEGMENT, LENGTH 55. NAME SIZED

•

.

SUBRAUTINE TRACE (ME.GA, NKF, NGA, IG, IK) 1274 DIMENSION KE (NKE), GA (NGA) DIMENSION RELAREJ,GALNGAJ LUGICAL LX COMMON/DATA/NSET,NORAN,NO,NPIPE,NXDSET,NITEM,KITEM,NK COMMON/PARAM/POA(9,20),MEND,JENO,MJ,ND,T,RP,ZD,DMAX,DMIN,NFLTA COMMON/MMERE/ID(200,11),IDX(10,13),IDA,IDL,IDT,**SOK-IOG-I**DD,JDJ, 1276 1270 1270 110P, 10R, 10H, 10B 1F (ND. GT. D) WRITE (8,2000) IF (LX) IGENGA IF (LX) IGENGA 1282 1243 1284 1245 1286 NKONKENI E----IDENTIFY CHEAPEST END STATE 1599 1=10J+(NB=1)+HJ=1 1289 KREF=0 CØST=999999,9 1290 1291 1292 1293 DØ 20 Jai, MJ Islei IF (GA(I), GT, CUST-0,001) GO TO 20 COST-GA(I) 1294 1295 KREFAJ 20 CONTINUE 1294 1297 HRITE (6,2003) COST, KREF IF (KREP, NE, 0) 60 TO 30 WRITE (6,2004) 1298 1300 \$78P BTUP C----IDENTIFY BUTFALL LEVEL AND DIAMETER BD J=(KREF-1)/MEND+1 MakREF-[J-1)*MEND [sidd+MgRaN=] 1305 1305 1304 1305 1306 DHAGGA (1) 1307 1=10L+NBRAN=1 ZHAX+GA(1) 1308 ZEZMAX-DELTA+FLOAT(M-1) 1309 CALL SIZED (PDA, NPIPE, JJ, DHA) 1310 1311 JJ=JJ=JEND+J D=PDA(NPIPE, JJ) 1312 WRITE (6,2009) Z,D C----BET DOWNSTREAM LEVELS AND DIAMETERS TO -DODDDDD.D 1313 1314 1315 IGJalog IG4#IDG+2+NBRAN+1 1317 D# 40 1=163,164 1318 1319 1320 40 GA(1)==999999,9 C====READ TRAIL DATA FOR BRANCH FROM M.T.FILE INTO ARRAYS GA AND KE 41 IGI=IG4=1 -. INTELGENI IF (LX) IKIEIDK-1 NMEO LITEMEO 4 NMENMOJ 5 BACKSPACE 9 701-10-0 1321 1322 1324 BACKSPACE 9 IGI=IGI+2 IF (LX) IKI=IKI+1 IF (IGI.GT.NGA) CALL MESAGE (1,0) IF (LX.AND.IMI.GT.NKE) CALL MESAGE (2,0) MEAD (9) J.GA(IGI).GA(IGI+1) LITEMELITEMAL LITEMELITEMAL 1326 1320 1329 1323 1225 1221 IF (LK) KE(IK1)=J RACKSPACE 9 IF (J=MJ) 10,10,5 10 IF (MJ=NM) 11,11, 1334 11,11,4 11 IG2#IG1+1 1336 1337 161=164+1 IF (ND,LT,2) GB TB 50 WRITE (6,2002) IGI,IG2 WRITE (6,2001) (GA(I),IGIGI,IG2) 50 IF (LX) GR TB 140 Coto-IDENTIFY UPSTREAM END BF BRANCH LEVEL AND DIAMETER 1338 1339 1340 1342 IGL=IG1+2+(MJ=KREP) 1343 ZUB=GA(IGL) 1344 DUS=GA(IGL+1) 1345 52 WRITE (6,2005) NB,ZUS,DUS C----SET D/S LEVEL AND DIAMETER PBR UPSTREAM PIPES C----IDENTIFY UPSTREAM PIPES 1346 1347 1349 51 N=10(N8,2) IF (N.EQ.0) GB TB 60 I=10(NB.5)=1 DØ 55 J=1,N I=1+1 1350 1351 1352 1353 1354 K=KE(1) L1=IG3+K=1 L2=IG3+K=1+NBRAN 1355 1357 GA(L1)=ZUS GA(L2)=DUS SS CENTINUE 1350 1359 60 IF (ND. GT.1) 70 N8*N8+1 WRITE (6,2001) (64(1),1-163,164) 1360 1361 IF (NB.LE.O) RETURN I=IG3+NB=1 1362 1363

1364 1345 1 ------1367 1368 71 CONTINUE 1367 68 TO FO CommoFIND D/S LEVEL AND DIAMETER 1370 1371 1372 75 ZDSaGA(1) 1373 IEI+NSRAN DOSGA(I) NPIPESMOD(ID(NB,1),10) CHANNERFIND CORRESPONDENCE FROM O'S PIPE TO CURRENT PIPE 1374 1375 1376 1377 I=IDL+NB=1 THAYSGA(I) 1378 I=100+NR+1 ISIDD+NR+1 OMAWGA(I) CALL SIZED (PDA,NPIPE,J,DMA) WASIDJ+NJ+(NB+1) CALL BRIDGE (GA,NGA,ZDS,DDS,ZMAX,J,NPIPE,NN,M1,J1) KREFS(J)+1)+MEND+M1 COMMON STHEN THAN THE MAIN MEMBER OF A CRESSDRAIN SET? IF (ID(NB,1),LT,IO1) GO TO 41 NXENDSIDX(NSET.7) IF (ID(AMXEND+MJ,GE,NK) CALL MESABE (12,7) COMMON GEN IN X/O DET TASEE DATA INTO ME, AND MAX LEVEL AND DISM U/S PF SET AD REWIND A 1380 1381 1382 1383 1385 1386 1388 1349 1390 AG REWIND &

 WITEM #KITEM=NXEND+MJ=1

 IF (KITEM, EG, 0) GØ TØ 100

 DØ 90 I=1,KITEM

 #8 READ (0)

 100 J=NXEND+MJ

 1391 1392 1393 1395 1396 IsIDK+1 08 110 K=1,J 19141 110 READ (8) XE(I) READ (8) ZHAN,DHA 1397 1396 1399 1400 NKEVK-1 1401 1402 KE (NK) BNXEND 1403 IF (ND.GT.C) WRITE (6.2006) NSET, NXEND, KREF C+---IDENTIFY UPSTREAM REFERENCE 1404 1405 I=IDK+(NXEND=1)=MJ+KREF=1 1406 120 KREFake(1) 1407 NXDa (KREF+1)/NJ 1408 KREFEKREFONJONXD IF (ND.GT.O) WRITE (6,2006) NSET, NXD, KREF KNMH=IOX(NSET, 8)+NXD=1 1410 1411 NHHEKE (KNHH) I=IDX(NSET,1) KABSEID(I,7)+NMH+1 IF (NKD,GT.0) WRITE (6,2008) GA(KABS) 1412 1414 1415 NKENK-1 1416 KE(NK)=NXD IF (NXD.EQ.0) GO TO 130 I=IOK+(NXD-1)+MJ+KREF+1 1417 1418 GO TO 120 C----IDENTIFY LEVEL AND DIAMETER AT UPSTREAM END PF CROSSDRAIN SET 130 J=(KREF-1)/MEND+1 1419 1420 1421 HEKREF=(J=1)+MEND ZUSHZMAX=DELTA+FLBAT(N=1) 1423 CALL SIZED (PDA, NPIPE, J2, DNA) JJ=J1-JEND+J 1424 1425 DUS=POA(NPIPE, JJ) 1426 1427 1428 NSET=HSET=1 GO TO 52 C----FIND END MANHOLE NUMBER LEVEL AND DIAMETER 1429 140 JECKREF-1)/HEND+1 1430 MaKREF-MEND+(J=1) 1431 NEID(NB,3) 1433 I=IOL+NB=1 ZENDEGA(I)=DELTA=FLBAT(H=1) 1434 INIDD+NB=1 1435 TALL SIZED (PDA, NPIPE, JI, GA(I)) 1436 JJ=J1=JEND+J 1437 DENDEPDA(NPIPE:JJ) 1438 WRITE(6,2007) N, J, H, DEND, ZEND 1439 GA(IG)=DEND 1441 KE(IK)=N 1442 IG=IG=1 1443 JK=1K+1

1435 NATKISI 1446 NJEI 1447 D0 150 IEI,LITEM 1448 NJENJEI 1449 IF (NJ.(EQ.KREF) GB TØ 160 1440 IF (NJ.(EQ.KREF) GB TØ 150 1441 NJENJEI 1442 ISO CENTINUE 1453 150 CENTINUE 1453 160 NCEIGZ22411 1454 JONEKENADE(M=1)+MJE11/MENDE1 1455 JEKE(NAD=(M=1)+MJE11/MENDE1 1456 NEKE(NAD=(M=1)+MJE11/MENDE1 1457 ZUSEGA(NC:1) 1458 DUSEGA(NC:1) 1459 DUSEGA(NC:1) 1451 JF (IZ,GT.IG) 1452 IF (IX,GT.IK) 1453 JEC.TICN 1454 JF (IX,GT.IK) 1455 JEC.TICN 1466 IKEIK=1 1467 JEC.TICN 1468 ICCTCL 1469 JEC.TICN 1464 JEC.TICN 1465 ICCTCL 1466 ICCTCL 1467 JEC.TICN 1468 ICCTCL <	1444	CFIND STARTS OF TRACE BACK SEQUENCES FOR D/S STATE (H, J)
1446 NJEI 1447 DB 150 [F1,L[TFM 1448 IF (MJ,EG,MEF) GE TE 160 1440 IF (MJ,EG,MEF) GE TE 160 1440 IF (MJ,EG,MEF) GE TE 160 1440 IF (MJ,EG,MEF) GE TE 160 1441 MJENJAL 1452 150 CENTINUE 1453 160 NC=162-2±11 1454 170 NETE (MJ)=1/M3,1/MEND41 1455 JACKE(MA)=(M-1)+MJENJAMEND4[J=1] 1456 MERT(MA)=(M-1)+MJENJAMEND4[J=1] 1457 JUSGA(MC) 1458 DUSGA(MC) 1459 DUSGA(MC) 1451 JE (E (MA)=(M-1)+MJENJAMENDALEVELSI STERE M/MS AND DIAMS. 1458 DUSGA(MC) 1459 DUSGA(MC) 1450 METE (6,2007) NJJ,M.DUS,ZUS 1451 IF (E (E,G) CE (C C C C C C C C C C C C C C C C C C	1445	NAWIKI41
DB ISO TELLITEM 1446 IP ILLITEM 1446 IP (MJ,CG,MEF) GB TB 180 1450 IP (MZ(MA),GT,MJ) GB TB 150 1451 150 CBNTINUE 1452 150 CBNTINUE 1453 160 MG(MZ)=2=2=101 1454 170 MG(MZ)=4=3=4:1/MEND=(J=1) 1455 JACKE(MA)=(M-1)+MJ=MEND=(J=1) 1456 IND (MA)=(M-1)+MJ=MEND=(J=1) 1457 JUSAGA(MC) OUSAGA(MC) 0USAGA(MC) OUSAGA(MC) OUSAGA(MC) 1458 IP (IG2,GT,IG) CALL MESAGE (13,0) 1459 IP (IG2,GT,IG) CALL MESAGE (14,0) 1461 IG2=G=1 IA 1462 IF (IK,GT,IK) CALL MESAGE (14,0) 1463 IG2=G=1 IA 1464 IG2=G=1 IA 1465 IC=1 IA 1466 IG2=G=1 1467 IP (IN/////IX,ISHENTERED TRACE) 1471 2000 FORMAT (IX//IX/IX,ISHENTERED TRACE) <td>1446</td> <td>f a t</td>	1446	f a t
1449 N_ASNA-1 1449 IF (NJ,EG,KEF) GB TØ 100 1450 IF (NZ(NA),GT,HJ) GB TØ 190 1451 IDD CENTINUE 1452 IDD CENTINUE 1453 100 NEUTG2-2=011 1454 170 NEUTE(NA)-1)+MJ=1/MEND01 1455 Je(ME(NA)=(N=1)+MJ=H/MEND0(J=1) 1456 170 NEUTE(NA)-1)+MJ=HJ=1/MEND0(J=1) 1457 JUSSGA(NC) 00JSGA(NC+1) 00JSGA(NC+1) 1458 OUSSGA(NC+1) 1459 Converting it and it a	1447	DØ 150 I=1.LITEM
IF (N.J.EG.KREF) GE TE 180 1450 IF (K.E.M.J.).GE TE 190 1451 ISD CBNTINUE (M.J.M.J.). 1451 ISD CBNTINUE (M.J.EG.Z.241) 1453 ISD CBNTINUE (M.J.EG.Z.241) 1453 JSC CBNTINUE (M.J.EG.Z.241) 1453 JSC CBNTINUE (J.S.C.241) 1454 IYD WELKE(MAJS-(M.J.)+MJ-11/MENDO1 1455 JSC CMAJSCH (M.J.)+MJ-11/MENDO1 1456 DUSSGA(MCC) 1457 JSSGA(MCC) 1458 DUSSGA(MCC) 1459 DUSSGA(MCC) 1450 METE (S.Z.007) N.J.M.DUS,ZUS 1451 IF (IK.1.GT.IG) 1452 IF (IK.1.GT.IG) 1454 G.Z.(IG.) 1455 IK.(IG.IG.IG) 1466 IK.I.K.I.K.I.K.I. 1467 IF (N.J.E.I.) ED TB 32 1468 IK.I.K.I. 1464 GA.I.G.POUS 1465 IK.I.K.I.I.S.I.TARIL DATAL 1466 IK.I.K.I.I.S.I.TARIL DATAL 1470 2000 FORMAT	1448	NABNA-1
IF (RE(NA), GT, MJ) 60 T0 150 1450 NJMJ+1 1451 150 CBNTINUE 1452 150 CBNTINUE 1453 160 NCCTG2-241+1 1454 170 NEKE(NA)=(J/MJ+1/MEND+1 1455 Ja(KE(NA)=(N-1)+MJ+1/MEND+1 1456 JUSSGA(NC-1) 1457 ZUSSGA(NC-1) 1458 OUSSGA(NC-1) 1459 ZUSSGA(NC-1) 1450 DUSSGA(NC-1) 1451 ZUSSGA(NC-1) 1452 DUSSGA(NC-1) 1453 OUSSGA(NC-1) 1454 DUSSGA(NC-1) 1455 JUSSGA(NC-1) 1456 DUSSGA(NC-1) 1457 ZUSSGA(NC-1) 1458 OUSSGA(NC-1) 1459 IF (IS, GT, IS) 1450 IF (IS, GT, IS) 1451 IF (IS, GT, IS) 1452 IF (N, LE, I) ES TB 52 1454 MABNA=1 1465 ICS (IS - 0) 1471 2000 FBRMAT (IS, FINTAL DATAL DATAL LEVELS AND DIAMETERS STARED IN GA PROM.	1449	IF (NJ.EG.KREF) GØ TØ 180
identifie identifie 1432 150 CBNTINUE 1433 150 WETCTAJ-17/M341 1434 170 WETCTAJ-17/M341 1435 Jacket (Maj-(Maj)amj=1/MEND+1 1436 170 WETCTAJ-17/M341 1437 Justack(Mc) 1438 Justack(Mc) 1439 Cusack(Mc) 1439 Cusack(Mc) 1439 Cusack(Mc) 1449 Justack(Mc) 1450 Cusack(Mc) 1451 Cusack(Mc) 1452 JF (IX,GT,IK) 1461 IF (IX,GT,IK) 1462 JF (IX,GT,IK) 1463 Mc(IG)=OUS 1464 Ga(IG)=OUS 1465 ICe(IG)=I 1466 ICe(IG)=I 1468 ICe(IG)=I 1469 McENC=2 1471 2000 FBRMAT (IX,SINTRAL LEVEL FREED TRACE) 1472 2001 FBRMAT (IX,SINTRAL LEVEL SAND DIAMETERB STORED IN GA PROM. 1471 2000 FBRMAT (IX,SINTRAL LEVEL SAND DIAMETERB STORED IN GA PROM. 1472 2001 FBRMAT	1450	IF (KE(NA).GT.HJ) 69 70 190
1452 150 CÖNTINUE 1453 160 NC=IG2=2+1:1 1454 170 NGTC (NA)=17/M3+1 1455 Ja(KC(NA)=(N=1)+MJ=MEND+1 1456 US=GA(NC) 1457 ZUS=GA(NC) 1458 OUS=GA(NC) 1459 Conservative But M/H POSITIONS DIAMS AND LEVELS: STORE H/MS AND DIAMS. 1450 WRITE BUT M/H POSITIONS DIAMS AND LEVELS: STORE H/MS AND DIAMS. 1451 IP (IG2,GT.IG) CALL MESAGE (13,0) 1452 IF (IX:GT.IK) CALL MESAGE (14,0) 1453 KE(IK)=N CALL MESAGE (14,0) 1454 IF (IX:GT.IK) CALL MESAGE (14,0) 1455 IX:IC=1 IS 1466 IC=IC=1 IS 1467 IF (N,LE,1) DS 70 52 IS 1468 IC=IC=1 IS 1469 IC=IC=1 IS 1460 IC=IC=1 IS 1461 IC=IC=1 IS 1462 IF (N,LE,1) IS 9712,3) IS 1471 2000 FORMAT (INO/////X,IS,ISHUBATERD TRACE) IS 1472 2001 FORMAT (INO////IX,SHENTERED TRACE) IS	1461	4 J + 1
1433 160 NG NC#16222111 1454 170 NG NC#10+10+10+10+10+10+10+10+10+10+10+10+10+1	1452	13D CONTINUE
1434 170 Nitit (Na)=17/M341 1435 Ja(ME(NA)=(M=1)/MJ=11/MEND+1) 1436 Number (Na)=(M=1)=MD=1(J=1) 1437 ZUSAGA(NC+1) 1438 DUS#GA(NC+1) 1439 ZUSAGA(NC+1) 1439 ZUSAGA(NC+1) 1439 DUS#GA(NC+1) 1439 ZUSAGA(NC+1) 1430 NRTE (0,2007) N.J.M.DUS,ZUS 1440 NRTE (0,2007) N.J.M.DUS,ZUS 1441 IF (122,GT.GC) 1440 IF (122,GT.GC) 1441 GA(TG)=DUS 1442 IF (122,GT.GC) 1443 KE(TN)=N 1444 IGTG=GT.GC) 1445 IK=[X=1] 1446 IGTG=GT.GC) 1447 IGT (S=1) 1448 IGT (G=1) 1449 IF (N.LE.1) 1444 IGT (G=1) 1447 IGT (G=1) 1448 IGT (G=1) 1449 IGT (G=1) 1447 IGT (G=1) 1447 ZOOD FORMAT (1X,F11.3) 1448 NABNA=1 1447 <th>1453</th> <th>160 NC#162=2+1+1</th>	1453	160 NC#162=2+1+1
Je(KE(WA)=(Wa))=MJ=(1/MEND=(J=1) 1435 1436 1437 1438 1439 1439 1439 1439 1439 1439 1439 1439 1439 1439 1439 1439 1439 1439 1439 1450 1451 1451 1451 1451 1451 1451 1461 1471 1460 1471 1460 1471 1460 1471 1470 1470 1470 1470 1470 1471 1471 1470 1471 1470 1471 1470 1471 1472 1472 1474 1475 1475 1475 1475 1475 1475 1475 1476 1477 1476 1477 1478 1478 1478 1479 1474 1474 1475 14	1454	170 NECKE(NA)-17/M3+1
1436 Hexte(Ha)=(H=1)=HEND*(J=1) 1437 ZUSEGA(MC) 1438 OUSEGA(MC+1) 1439 ZUSEGA(MC+1) 1430 OUSEGA(MC+1) 1431 ZUSEGA(MC+1) 1432 WRITE (0,2007) N.J.M.DUS,ZUS 1460 IF (IX.GT.IG) 1461 IF (IX.GT.IG) 1462 IF (IX.GT.IG) 1463 KE(IX)=N 1464 GA(IG)=DUS 1465 IK=[X=1] 1466 IK=[X=1] 1466 IK=[X=1] 1466 IK=[X=1] 1466 IK=[X=1] 1466 IK=[X=1] 1466 IK=[X=1] 1467 IF (N.LE.1) GB YB B2 1468 IK=[X=1] 1469 NENAC=2 1470 GB YB 170 2000 F GBMAT (IX.SINTRAIL DATAC LEVELS AND DIAMETERB STORED IN GA PROM. 1471 2000 F GBMAT (IX.SINTRAIL DATAC LEVELS AND DIAMETERB STORED IN GA PROM. 1472 2003 F GBMAT (IX.SINTRAIL DATAC LEVELS AND DIAMETERB STORED IN GA PROM. 1473 2004 F GBMAT (IX.SINTRAIL DATAC LEVELS AND DIAMETERB STORED IN GA PROM. <	1455	Ja (KE (NA) w (No1) of Jot 17 MENDO1
1457 JUSEGA(NC) 1458 OUSEGA(NC+1) 1459 C====wRTTE BUT M/H POSITIBNS DIAMS AND LEVELS: BTERE H/HS AND DIAMS, 1460 NRITE (6,2007) N,J,M,DUS,ZUS 1461 IF (122,GT.IG) CALL MESAGE (13,0) 1462 IF (1K1,GT.IK) CALL MESAGE (14,0) 1463 KE(IX)=N CALL MESAGE (14,0) 1464 GA(IG)=DUS IK 1465 IK=[X=1] GB TE 1466 IG=IG=1 IG=IG=1 1466 IG=IG=1 IG=IG=1 1468 NC=NC=2 IG 1470 GB TE IT IG=IG=1 1471 2000 FERMAT (1X,F11,3,9F12,3) IG=IG=1 1473 2001 FERMAT (1MO////1X,13HENTERED TRACE) IG=IG=1 1471 2000 FERMAT (1MO////1X,13HENTERED TRACE) IG=IG=1 1472 2001 FERMAT (1MO////1X,13HENTERED TRACE) IG=IG=1 1473 2002 FERMAT (1MO////1X,13HENTERED TRACE) IG=IG=1 1474 2003 FERMAT (1MO//IA,16,3S,15HURAEST SELUTIAN CBSTS,F12,3,5X,10HEND STATEs, IIG=1 1475 2004 FERMAT (1X,20HNE FEASIBLE SELUTIAN) IG=FE,F5,3,5X, 1476	1456	Make (NA)= (Na1)=MJ=MEND+(J=1)
1456 DUB SGA (NC+1) 1456 C====WRITE BUT M/M PBBJITBNS DIAMB AND LEVELS: BTORE M/MS AND DIAMS, 1460 WRITE (6,2007) N.J.M.DUS,ZUS 1461 IF (IG2,GT.IG) CALL MESAGE (13,0) 1462 IF (IK1,GT.IK) CALL MESAGE (14,0) 1463 KE(IK3=N CALL MESAGE (14,0) 1464 GA(IG3=DUS IK=IK=1 1466 IG=IG=1 IK=IK=1 1466 IG=GE=1 IK=IK=1 1466 IG=TG=1 IK=IK=1 1466 IG=TG=1 IK=IK=1 1466 IG=TG=1 IK=IK=1 1467 IF (N.LE_1) DB TB B2 NABNA=1 1468 IK=NC=2 IK=NC=2 1470 G0 T0 FORMAT (IX.FI1.3.9F12.3) IG=IG=1 1471 2000 FORMAT (IX.SIMTRAIL DATAL LEVELS AND DIAMETERS STORED IN GA PROM. 1472 2001 FORMAT (IX.SIMTRAIL DATAL LEVELS AND DIAMETERS STORED IN GA PROM. 1471 2002 FORMAT (IX.SIMTRAIL DATAL LEVELS AND DIAMETERS STORED IN GA PROM. 1472 2003 FORMAT (IX.SIMTRAIL DATAL LEVELS AND DIAMETERS STORED IN GA PROM. 1471 2004 FORMAT (IX.SIMTRAIL DATAL LEVELS AND DIAMETERS STORED IN GA PROM. 1472 <th>1457</th> <th>ZUSAGA(NC)</th>	1457	ZUSAGA(NC)
1430 Commendiate But Min Positions Diams and levels: Stere Mins and Diams, MRITE (6,2007) N,J,M,DUS,ZUS 1460 IF (162,67.163) CALL MESAGE (13,0) 1461 IF (181,67.1K) CALL MESAGE (14,0) 1463 KC(1K)=N CALL MESAGE (14,0) 1464 GA(16)=DUS I 1465 IKE(IK)=N CALL MESAGE (14,0) 1466 ICE[G=1 I 1467 IF (N,LE.1) DB TB 52 MARNA=1 1468 NCENC=2 I 1471 2000 FORMAT (1x,51NTRAIL DATAG LEVELS AND DIAMETERS STORED IN GA PROM, IIII IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	1458	DUS #GA (NC+1)
1460 WRITE (6,2007) W, J, M, DUS, ZUS 1461 IF (IG2, GT, IG) CALL MESAGE (13,0) 1461 IF (IK1, GT, IK3) CALL MESAGE (14,0) 1462 IF (IK1, GT, IK3) CALL MESAGE (14,0) 1463 GA(IG)=DUS IK=IK-1 1464 IC=IG=I IK=IK-1 1465 IF (N, LE.1) ES TB 52 1466 IC=IG=I 1467 IF (N, LE.1) ES TB 52 1468 IC=IG=I 1469 NEWNA=I 1469 NEWNA=I 1469 NEWNA=I 1469 NEWNA=I 1470 GO TORMAT (1x,FII, 3, OF12, 3) 1471 2000 FORMAT (1x,SIMTRAIL DATAL LEVELS AND DIAMETERS STORED IN GA FROM, 1471 2002 FORMAT (1x,SIMTRAIL DATAL LEVELS AND DIAMETERS STORED IN GA FROM, 1473 2003 FORMAT (1x,SIMTRAIL DATAL LEVELS AND DIAMETERS STORED IN GA FROM, 1474 ITO.44M TØ, IS 1475 2003 FORMAT (1x,OHNO FEASIBLE SOLUTION) 1476 IS 1477 2004 FORMAT (1x,OHNO FEASIBLE SOLUTION) 1478 2005 FORMAT (1x,OHNO FEASIBLE SOLUTION) 1479 1004 FORMAT (1x,OHNO FEASI	1456	COMPANY AND DIAMS DIAMS AND LEVELSE STORE M/MS AND DIAMS.
1461 IP (IG2,GT,IG) CALL MESAGE (13,0) 1462 IF (IX1,GT,IK) CALL MESAGE (14,0) 1463 NE (IX)=N CALL MESAGE (14,0) 1464 GAIGD=008 IX=IX=1 1466 IG=IG=1 IX=IX=1 1466 IG=IG=1 IX=IX=1 1466 IG=IG=1 IX=IX=1 1468 IX=IX=1 DS 70 52 1469 NC=NC=2 IX=1X=1 1470 G0 70 170 G0 70 170 1471 2000 FORMAT (1X,FI1.3,9F12,3) IX=IX=1 1472 2001 FORMAT (1X,SINTRAIL DATAG LEVELS AND DIAMETERS STORED IN GA FROM, IIO,4M FØ,169 1473 2002 FORMAT (1X,SINTRAIL DATAG LEVELS AND DIAMETERS STORED IN GA FROM, IIO,4M FØ,169 1474 110,4M FØ,164 1475 2002 FORMAT (1X,20HNG FEASIBLE SOLUTION) 1476 1160 1477 2004 FORMAT (1X,20HNG FEASIBLE SOLUTION) 1478 2005 FORMAT (1X,10HERNERER,16,3X,15HUPSTREAH LEVEL=,79,3,5X, I0HUPSTREAM DIAMETER,76,3) 1480 2007 FORMAT (1X,14HCR3SDRAIN SET,IG,3X,17HCR0SSDRAIN NUHBER,IS, 1481 13X,5HSTATE:10 1482 200F FORMAT (1X,20HNEN EROSSDRAIN AT CHAINAGE,F16,3)	1440	HRITE (6,2007) N.J.H.DUS,ZUS
1462 IF (IKI,GT,IK) CALL MESAGE (14,0) 1463 KE(IK)=N 1464 GA(IG)=DUS 1465 IK=IK=1 1466 IG=IG=I 1466 IF (N,LE,1) ED TD B2 1466 IG=IG=I 1468 IF (N,LE,1) ED TD B2 1468 NE=NC=2 1470 G0 TD ITO 1471 2000 FORMAT (IX,FI1.3,0F12,J) 1472 2001 FORMAT (IX,FI1.3,0F12,J) 1473 2002 FORMAT (IX,SINTRALL DATAL LEVELS AND DIAMETERS STORED IN G4 PROM, 1473 2002 FORMAT (IX,SINTRALL DATAL LEVELS AND DIAMETERS STORED IN G4 PROM, 1474 110,4H TO ,IS) 1475 2002 FORMAT (IX,SINTRAL DATAL LEVELS AND DIAMETERS STORED IN G4 PROM, 1474 110,4H TO ,IS) 1475 2005 FORMAT (IX,20HNO FEASIBLE SOLUTION) 1476 116) 1477 2004 FORMAT (IX,0HORANCH,IG,SX,ISHUPSTREAM LEVEL=,F9,J,SX, 1478 2005 FORMAT (IX,0HORANCH,IG,SX,ISHUPSTREAM LEVEL=,F9,J,SX, 1479 116HUPSTREAM DIAMETER=,F6,S) 1470 2005 FORMAT (IX,14HCROSSDRAIN SET,IS,SX,IFMCROSSDRAIN NUMBER,IS, 1480 2007 FORMAT (IX,14HCROSSDRAIN SET,IS,SX,	1461	IP (IG2, GT, IG) CALL MESAGE (13,0)
1463 KE(IK)SN 1464 GA(IG)=DUS 1465 IK=IK=1 1466 IG=IG=1 1466 IG=IG=1 1466 IF (N,LE,1) GB TB 52 1468 IF (N,LE,1) GB TB 52 1469 NC=NC=2 1470 GB TØ 170 1471 2000 FGRMAT (1x,F11,J.9F12,J) 1472 2001 FGRMAT (1x,F11,J.9F12,J) 1473 2002 FGRMAT (1x,SINTRAIL DATA4 1474 IT0,4M TØ ,F03 1475 2003 FGRMAT (1x,23HCHEAPEST SOLUTION CBSTS,F12,J,5X,10HEND STATE*, 1476 II63 1477 2004 FURMAT (1X,20HNØ FEASIBLE SØLUTION) 1478 2005 FGRMAT (1X,0HBRANCH,I6,5X,15HUPSTREAM LEVEL=,F9,J,5X, 1479 II6HUPSTREAM DIAHETER*,F6,33 1479 1004 FURMAT (1X,14HCRSSDRAIN SET,IS,3X,17HCRØSSDRAIN NUMBER,IS, 1480 2005 FGRMAT (1X,14HCRSSDRAIN SET,IS,3X,17HCRØSSDRAIN NUMBER,IS, 1481 13X,5HSTATE.IG) 1482 2007 FGRMAT (1X,7HHANHØLE,IG,9H OIAM,NR.,IG,10H LEVEL NØ.,IG, 1483 I9M DIAMETER*,F6.3,6M LEVEL,F9.3) 1484 2009 FGRMAT (1X,20HNE CRESSDRAIN AT CHAINAGE,F16.3) 1483	1462	IF (IK1.GT.IK) CALL MESAGE (14,0)
1464 GA(IG)=DUS 1466 IK=IK=1 1466 IG=IG=1 1466 IG=IG=1 1467 IF (N,LE.1) GB TB 32 1468 NABNA=1 1469 NC=NC=2 1470 GB TB 170 1471 2000 FBRMAT (1M0////1x,13HENTERED TRACE) 1472 2001 FBRMAT (1x,F11.3,9F12,3) 1473 2002 FBRMAT (1x,SINTRAIL DATA: LEVELS AND DIAMETERS STORED IN GA FROM, 1474 ITB, 4M TB, IEB SOLUTION CBSTS,F12,3,5X,10HEND STATES, 1475 2003 FBRMAT (1x,20HNB FEASIBLE SOLUTION) 160 1477 2004 FBRMAT (1x,20HNB FEASIBLE SOLUTION) 1477 1478 2005 FBRMAT (1x,0HBRANCH,IG,5X,15HUPSTREAM LEVEL=,F9,3,5X, 10HEND STATES, 1479 116HUPSTREAM DIAMETERU,F6,31 13X,5HSTATE,IG) 13X,5HSTATE,IG) 1480 2007 FBRMAT (1X,7HMANHBLE,IG,9H DIAM,NR,1G,10H LEVEL NB,,IG, 1482 2007 FBRMAT (1X,7HMANHBLE,IG,9H DIAM,NR,1G,10H LEVEL NB,,IG, 1481 2007 FBRMAT (1X,2GHNEM CRESSDRAIN AT CHAINAGE,F16,3) 19H DIAMETER,F6,3,6H LEVEL,F9,3) 1483 2008 FBRMAT (1X,2GHNEM CRESSDRAIN AT CHAINAGE,F16,3) 19H DIAMETERU,F3,5H LEVELS,P12,3,19H BUTFALL DIAMETERU,F12,3) 1484 2008 FBRM	1463	KE (IK)=N
1466 IX=IX=1 1466 IG=IG=1 1466 IF (N,LE,1) GB TB 32 1468 IF (N,LE,1) GB TB 32 1468 NARNA=1 1469 NC=NC=2 1470 G0 T0 I70 1471 2000 F0RMAT (1M0////1X,13HENTERED TRACE) 1472 2001 F0RMAT (1X,F11.3.00F12,3) 1473 2002 F0RMAT (1X,SIMTRAIL DATAL LEVELS AND DIAMETERS STORED IN GA FROM, 1473 2003 F0RMAT (1X,SIMTRAIL DATAL LEVELS AND DIAMETERS STORED IN GA FROM, 1474 110,444 F0,163 1475 2003 F0RMAT (1X,SIMTRAIL DATAL LEVELS AND DIAMETERS STORED IN GA FROM, 1474 110,444 F0,163 1475 2003 F0RMAT (1X,20HN0 FEASIBLE S0LUTION) 1476 1163 1477 2005 F0RMAT (1X,04BRANCH,16,5X,15HUPSTREAM LEVEL=,F0,3,5X, 1479 110HUPSTREAM DIAMETER=,F6,33 1480 2005 F0RMAT (1X,14MCRSSDRAIN SET,15,3X,17MCR058DRAIN NUMBER,15, 1481 2007 F0RMAT (1X,14MCRSSDRAIN SET,15,3X,17MCR058DRAIN NUMBER,15, 1482 2007 F0RMAT (1X,7MMANH0LE,16,0H DIAM,NG,16,10H LEVEL NG,16, 1482 2007 F0RMAT (1X,20HNE CROSSDRAIN AT CHAINAGE,F16,3) 1483 2008 F00RMAT (1X,14H0UTFALL LEVEL=	1464	GA(IG)=DUS -
1466 IGEIG-1 1467 IF (N,LE,1) GB TB 32 1458 NABNA-1 1469 NC=NC=2 1470 GB TB 170 1471 2000 FORMAT (1MD////1X,13HENTERED TRACE) 1472 2001 FORMAT (1X,F11,3,9F12,3) 1473 2007 FORMAT (1X,F11,3,9F12,3) 1473 2007 FORMAT (1X,F11,3,9F12,3) 1473 2007 FORMAT (1X,SINTRALL DATAL LEVELS AND DIAMETERS STORED IN GA PROM, 1474 116,444 FG ,163 1475 2005 FORMAT (1X,20HNG FEASIBLE SOLUTION) 1476 116) 1477 2005 FORMAT (1X,0MBRANCH,16,5X,15HUPSTREAM LEVELS,FG,3,5X, 1479 1100HDSTREAM DIAMETERU,FG,3) 1479 2005 FORMAT (1X,14HCRESSDRAIN SET,15,3X,17HCRESSDRAIN NUMBER,15, 1480 2006 FORMAT (1X,14HCRESSDRAIN SET,15,3X,17HCRESSDRAIN NUMBER,15, 1481 13X,5HSTATE.16) 1482 2007 FORMAT (1X,7HMANHGLE,16,9H DIAM,NG,16,10H LEVEL NG.,16, 1483 2007 FORMAT (1X,26HNEN CRESSDRAIN AT CHAINAGE,F16,3) 1484 2008 FORMAT (1X,26HNEN CRESSDRAIN AT CHAINAGE,F16,3) 1483 2009 FORMAT (1X,14HOUTFALL LEVELS,P12,3,10H OUTFALL DIAMETERS,F12,3) 1484 2009 FORMAT (1X,14	1465	2K=1K-1
1487 IF (N,LE,1) GB TB 52 1468 NABNA-1 1490 NCBNC-2 1471 2000 FORMAT (1ND////IX.13HENTERED TRACE) 1472 2001 FORMAT (1X.F11.3.9F12.3) 1473 2002 FORMAT (1X.F11.3.9F12.3) 1474 2003 FORMAT (1X.F11.3.9F12.3) 1475 2003 FORMAT (1X.F11.3.9F12.3) 1476 116.9 1478 2003 FORMAT (1X.P3HCHEAPEST SOLUTION) 1478 2005 FORMAT (1X.20HNO FEASIBLE SOLUTION) 1476 116.3 1477 2005 FORMAT (1X.0HORANCH.16.5X.15HUPSTREAM LEVELS.F9.3.5X.10HEND STATES. 1478 2005 FORMAT (1X.0HORANCH.16.5X.15HUPSTREAM LEVELS.F9.3.5X.10HEND STATES. 1479 116HUPSTREAM DIAHETERS.F6.3.3 1480 2006 FORMAT (1X.14HCRESSDRAIN SET.IS.3X.17HCROSSDRAIN NUMBER.IS. 1481 13X.SHSTATE.IG.3 1482 2007 FORMAT (1X.7HHANHGLE.IG.9H OTAM.NGIG.10H LEVEL NGIG. 1483 19H DIAMETER.F6.3.6H LEVEL.F9.33 1484 2006 FORMAT (1X.20HNEN CROSSDRAIN AT CHAINAGE.F16.33) 1484 2009 FORMAT (1X.14HOUTFALL LEVELS.F12.3.19H OUTFALL DIAMETERS.F12.33) 1484 2009 FORMAT (1X.14HOUTFALL LEVELS.F12.3.19H OUTFALL DIAMETERS.F12.33)	1466	
1458 NABNA-1 1469 NCENC-2 1470 GB TØ 170 1471 2000 FØRMAT (1M0////1X.13HENTERED TRACE) 1472 2001 FØRMAT (1X.F11.3.9F12.3) 1473 2002 FØRMAT (1X.SINTRAIL DATAG LEVELS AND DIAMETERS STØRED IN GA FRØM, 1473 2003 FØRMAT (1X.SINTRAIL DATAG LEVELS AND DIAMETERS STØRED IN GA FRØM, 1474 170.4M TØ .16) 1475 2003 FØRMAT (1X.SINTRAIL DATAG LEVELS AND DIAMETERS STØRED IN GA FRØM, 1474 170.4M TØ .16) 1475 2003 FØRMAT (1X.20HNØ FEASIBLE SØLUTIØN) 1476 116) 1477 2004 FØRMAT (1X.0HBRANCH.16.5X.15HUPSTREAM LEVEL#.F9.3.5X.10HEND STATE#. 1479 110HUPSTREAM DIAMETER#.F6.33 1479 110HUPSTREAM DIAMETER#.F6.33 1480 2005 FØRMAT (1X.14HCRBSSDRAIN SET.IS.3X.17HCRBSBDRAIN NUMBER.IS. 1481 13X.5HSTATE.IG) 1482 2007 FØRMAT (1X.7HMANHØLE.IG.0H DIAM.NR.IG.10H LEVEL NØIG. 1483 19H DIAMETER.F6.3.0H LEVEL.F9.33 1484 2008 FØRMAT (1X.20HNEN CRØSDRAIN AT CHAINAGE.F16.3) 1484 2009 FØRMAT (1X.14HØUTFALL LEVEL#.P12.3.19H ØUTFALL DIAMETER#.F12.3) 1486 END	1467	1F (N.LE.1) 68 78 82
1409 NCENC=2 1470 GB TØ 170 1471 2000 FØRMAT (1M0////1X,13HENTERED TRACE) 1472 2001 FØRMAT (1X,F11.3.9F12,3) 1473 2007 FØRMAT (1X,S1HTRAIL DATAL LEVELS AND DIAMETERS STØRED IN GA FRØM, 1473 2007 FØRMAT (1X,S1HTRAIL DATAL LEVELS AND DIAMETERS STØRED IN GA FRØM, 1473 2007 FØRMAT (1X,S1HTRAIL DATAL LEVELS AND DIAMETERS STØRED IN GA FRØM, 1474 110,444 TØ ,163 1475 2003 FØRMAT (1X,20HNØ FEASIBLE SØLUTIØN) 1476 1160 1477 2005 FØRMAT (1X,64BRANCH,16,5X,15HUPSTREAM LEVEL#,FØ,3,5X, 1479 110HUPSTREAM DIAMETER#,F6,33 1479 110HUPSTREAM DIAMETER#,F6,33 1480 2005 FØRMAT (1X,14HCR\$SDRAIN SET,15,3X,17HCR\$SBDRAIN NUHBER,15, 1481 13X,5HSTATE.16) 1482 2007 FØRMAT (1X,7HMANHØLE.16,9H DIAM,NR,16,10H LEVEL NØ,16, 1483 19H DIAMETER,F6,3,0H LEVEL,FØ,33 1484 2008 FØRMAT (1X,20HNE CR&SDRAIN AT CHAINAGE,F16,3) 1483 2009 FØRMAT (1X,14HØUTFALL LEVEL#,P12,3,19H ØUTFALL DIAMETER#,F12,3) 1484 2008 FØRMAT (1X,14HØUTFALL LEVEL#,P12,3,19H ØUTFALL DIAMETER#,F12,3)	1458	NABNA-1
1470 GB TØ 170 1471 2000 FØRHAT (1HD////1X.13HENTERED TRACE) 1472 2001 FØRHAT (1X,F11.3,9F12.3) 1473 2002 FØRHAT (1X,SIHTRAIL DATAL LEVELS AND DIAMETERS STØRED IN GA FRØM, 1474 150.4M FØ ,163 1475 2002 FØRHAT (1X,20HNØ FEASIBLE SØLUTIØN CØSTS,F12.3,5X,10HEND STATES, 1476 116) 1477 2005 FØRHAT (1X,20HNØ FEASIBLE SØLUTIØN) 1478 2005 FØRHAT (1X,20HNØ FEASIBLE SØLUTIØN) 1479 1160 1479 100HPSTREAM DIAMETERN,FG.3) 1479 110HUPSTREAM DIAMETERN,FG.3) 1480 2005 FØRMAT (1X,14HCRRSSDRAIN SET,IS,3X,17HCRØSSDRAIN NUMBER,IS, 1480 2007 FØRHAT (1X,7HMANHØLE,IG,9H DIAM,NG.,IG,10H LEVEL NØ.,IG, 1481 13X,SHSTAFE.IG) 1482 2007 FØRHAT (1X.7HMANHØLE,IG,9H DIAM,NG.,IG,10H LEVEL NØ.,IG, 1483 19H DIAMETER,FG.3,6H LEVEL,FØ.3) 1484 2006 FØRHAT (1X.20HNEN CRØSDRAIN AT CHAINAGE,F16.3) 1483 2009 FØRHAT (1X.20HNEN CRØSDRAIN AT CHAINAGE,F16.3) 1484 2009 FØRHAT (1X.14HØUTFALL LEVELS,P12.3,19H ØUTFALL DIAMETERS,F12.3) 1484 2009 FØRHAT (1X.14HØUTFALL LEVELS,P12.3,19H ØUTFALL DIAMETERS,F12.3)	1469	
1471 2000 FORMAT (1H0////1X,13HENTERED TRACE) 1472 2001 FORMAT (1X,F11,3.9F12,3) 1473 2002 FORMAT (1X,F11,3.9F12,3) 1474 100,4M TO,10,9F12,3) 1475 2003 FORMAT (1X,F11,1.04TA4 LEVELS AND DIAMETERS STORED IN GA FROM, 1676 1474 100,4M TO,10,3M,23MCHEAPEST SOLUTION COSTS,F12,3,5X,10HEND STATEs, 1678 1475 2003 FORMAT (1X,20HNO FEASIBLE SOLUTION) 1476 116) 1477 2005 FORMAT (1X,0HORSANCH,16,5X,15HUPSTREAM LEVELS,F9,3,5X, 1679 1478 2005 FORMAT (1X,14HCROSSDRAIN SET,15,3X,17HCROSSDRAIN NUMBER,15, 1680 1480 2006 FORMAT (1X,14HCROSSDRAIN SET,15,3X,17HCROSSDRAIN NUMBER,15, 13X,5HSTATE,16) 1482 2007 FORMAT (1X,7HMANHGLE,16,0H 01AM,NG,16,10H LEVEL NO,16, 19H DIAMETER,F6,3,0H LEVEL,F0,3) 1483 19H DIAMETER,F6,3,0H LEVEL,F0,3) 1484 2006 FORMAT (1X,20HNE CROSSDRAIN AT CHAINAGE,F16,3) 1485 2009 FORMAT (1X,14HOUTFALL LEVELS,F12,3,19H OUTFALL DIAMETERS,F12,3) 1486 2009 FORMAT (1X,14HOUTFALL LEVELS,F12,3,19H OUTFALL DIAMETERS,F12,3)	1470	69 78 170
1472 2001 FORMAT (1x,F11.3,9F12,3) 1473 2007 FORMAT (1x,SINTRAIL DATAL LEVELS AND DIAMETERS STORED IN GA FROM, 1474 170,4M FC,160 1475 2003 FORMAT (1x,SINTRAIL DATAL LEVELS AND DIAMETERS STORED IN GA FROM, 1474 170,4M FC,160 1475 2003 FORMAT (1x,SINTRAIL DATAL LEVELS AND DIAMETERS STORED IN GA FROM, 1476 1160 1477 2004 FORMAT (1x,20NNG FEASIBLE SOLUTION) 1478 1160 1479 2005 FORMAT (1x,0MBRANCH,16,5x,15HUPSTREAM LEVELS,F9,3,5x, 1479 110MUPSTREAM DIAMETERS,F6,30 1479 110MUPSTREAM DIAMETERS,F6,33 1480 2005 FORMAT (1x,14HCRESSDRAIN SET,16,3x,17HCRESSDRAIN NUMBER,15, 1481 13x,5HSTATE,16) 1482 2007 FORMAT (1x,7MMANHOLE,16,0H DIAM,NR,16,10H LEVEL NG.,16, 1483 19H DIAMETER,F6,3,0H LEVEL,F0,33 1484 2006 FORMAT (1x,26HNEW CROSSDRAIN AT CHAINAGE,F16,3) 1484 2008 FORMAT (1x,14HOUTFALL LEVELS,P12,3,19H OUTFALL DIAMETERS,F12,3) 1485 2009 FORMAT (1x,14HOUTFALL LEVELS,P12,3,19H OUTFALL DIAMETERS,F12,3)	1471	2000 FORMAT (1H0/////1X+13HENTERED TRACE)
1473 2002 FORMAT (1x,51HTRAIL DATA: LEVELS AND DIAMETERS STORED IN GA FROM, 1674 174 170,444 TØ,163 1475 2003 FORMAT (1MO/14,33HECHEAPEST SOLUTION LOSTS,F12,3,5%,10HEND STATES, 166) 1477 2004 FORMAT (1%,20HNØ FEASIBLE SOLUTION) 1478 2005 FORMAT (1%,64BRANCH,16,5%,15HUPSTREAM LEVELS,F9,3,5%, 16HUPSTREAM DIAMETERS,F6,3) 1479 110HUPSTREAM DIAMETERS,F6,3) 1480 2006 FORMAT (1%,14HCRRSSDRAIN SET,10,3%,17HCRRSSDRAIN NUMBER,15, 13%,5HSTATE,16) 1482 2007 FORMAT (1%,7HMANHØLE,16,9H DIAM,NR,16,10H LEVEL NØ,,16, 19H DIAMETER,F6,3,0H LEVEL,F9,3) 1484 2008 FORMAT (1%,24HNE CROSSDRAIN AT CHAINAGE,F16,3) 1485 2009 FORMAT (1%,14HØUTFALL LEVELS,P12,3,19H ØUTFALL DIAMETERS,F12,3)	1472	2001 FORMAT (1x,F11.3,9F12,3)
1474 110,441 F0,163 1475 2003 FORMAT (140/14,234CHEAREST SOLUTION COSTS,F12,3,5%,104EMD STATES, 1476 1474 1163 1475 2005 FORMAT (1%,204NO FEASIBLE SOLUTION) 1476 1163 1477 2005 FORMAT (1%,204NO FEASIBLE SOLUTION) 1478 2005 FORMAT (1%,048RANCH,16,5%,154UPSTREAM LEVELS,F0,3,5%, 1679 1479 1104UPSTREAM DIAMETERS,F6,33 1480 2006 FORMAT (1%,1440CRESSDRAIN SET,15,3%,17MCR05SDRAIN NUMBER,15, 13%,5HSTATE.16) 1482 2007 FORMAT (1%,7MMANHOLE,16,9H DIAM,NG,16,10H LEVEL NO.,16, 19H DIAMETER,F6,3,6H LEVEL,F0,33 1483 2008 FORMAT (1%,204NEM CROSSDRAIN AT CHAINAGE,F16,33 1484 2008 FORMAT (1%,14HOUTFALL LEVELS,F12,3,19H OUTFALL DIAMETERS,F12,3) 1485 2009 FORMAT (1%,14HOUTFALL LEVELS,F12,3,19H OUTFALL DIAMETERS,F12,3)	1473	2002 FORMAT (1X,51HTRAIL DATAG LEVELS AND DIAMETERS STORED IN GA FROM,
1478 2003 FØRMAT (1H0/1%-23HCHEAPEST SØLUTIAN CØSTS,F12,3,5%,10HEND STATES, 1476 116) 1477 2004 FØRMAT (1%,20HNØ FEASIBLE SØLUTION) 1478 2005 FØRMAT (1%,0HBRANCH,16,5%,15HUPSTREAM LEVELS,F9,3,5%, 1479 116HUPSTREAM DIAHETERS,F6,3) 1480 2006 FØRMAT (1%,14HCRRSSDRAIN SET,15,3%,17HCRØSSDRAIN NUMBER,15, 1481 13%,5H3TATE,16) 1482 2007 FØRMAT (1%,7HMANHØLE,16,9H 01AM,NR,16,10H LEVEL NØ.,16, 1483 19H DIAMETER,F6,3,6H LEVEL,F9,3) 1484 2006 FØRMAT (1%,26HNEN CRØSSDRAIN AT CHAINAGE,F16,3) 1483 2009 FØRMAT (1%,14HØUTFALL LEVELS,F12,3,19H ØUTFALL DIAMETERS,F12,3) 1484 2009 FØRMAT (1%,14HØUTFALL LEVELS,F12,3,19H ØUTFALL DIAMETERS,F12,3)	1474	170,4H 70 ,IG)
1476 116) 1477 2004 FERMAT (1x,20HNE FEASIBLE SELUTION) 1478 2005 FERMAT (1x,6MERANCH,16,5x,15HUPSTREAM LEVELE,F9,3,5x, 1479 116HUPSTREAM DIAHETERE,F6,3) 1480 2005 FERMAT (1x,14HCRRSSDRAIN SET,15,3x,17HCRRSSDRAIN NUMBER,15, 1481 13x,5HSTATE,16) 1482 2007 FERMAT (1x,7HMANHELE,16,9H DIAM,NR,16,10H LEVEL NG,16, 1483 19H DIAMETER,F6,36H LEVEL,F9,33 1484 2008 FERMAT (1x,26HNEW CRESSDRAIN AT CHAINAGE,F16,3) 1485 2009 FERMAT (1x,14HEUTFALL LEVELE,P12,3,19H BUTFALL DIAMETERE,F12,3) 1486 END	1475	2003 FURHAT (INO/IX-23HCH2AFEST SOLUTION COSTS,F12-3-5X,10HEND STATES,
1477 2004 FURMAT (1x,20HN0 FEASIBLE SQLUTION) 1470 2005 FORMAT (1x,6HRANCH,16,5x,15HUPSTREAM LEVELU,F9,3,5x, 1479 110HUPSTREAM DIAMETERU,F6,3) 1480 2005 FORMAT (1x,14HCRRSSDRAIN SET,15,3x,17HCRRSSDRAIN NUMBER,15, 1481 13x,5HSTATE,16) 1482 2007 FORMAT (1x,7HMANHGLE,16,9H DIAM,NR,16,10H LEVEL NG,16, 1483 19H DIAMETER,F6,3,6H LEVEL,F0,33 1484 2005 FORMAT (1x,26HNEW CROSSDRAIN AT CHAINAGE,F16,3) 1485 2009 FORMAT (1x,14HOUTFALL LEVELU,F12,3,19H OUTFALL DIAMETERU,F12,3)	1476	116)
1470 2005 FØRMAT (1x,6MBRANCH,16,5x,15HUPSTREAM LEVEL#,F9,3,5x, 1479 110HUPSTREAM DIAMETER#,F6,3) 1480 2006 FØRMAT (1x,14HCRRSSDRAIN SET,15,3x,17HCRRSSDRAIN NUMBER,15, 1481 13x,5HSTATE.16) 1482 2007 FØRMAT (1x,7HMANHØLE.16,9H DIAM,NR.,16,10H LEVEL NØ.,16, 1483 19H DIAMETER,F6,3,6H LEVEL,F9,3) 1484 2008 FØRMAT (1x,26HNEN CRBSDRAIN AT CHAINAGE,F16,3) 1485 2009 FØRMAT (1x,14HØUTFALL LEVEL#,P12,3,19H ØUTFALL DIAMFTER#,F12,3) 1486 END	1477	2004 FURMAT (1x,20MNØ FEASIBLE SØLUTIØN)
1479 110HUPSTREAM DIAMETERW,FG,33 1480 2006 FØRMAT (1X,14HCRRSSDRAIN SET,IS,3X,17HCRRSSDRAIN NUMBER,IS, 1481 13X,5HSTATE.IG) 1482 2007 FØRMAT (1X,7HMANHØLE,IG,9H DIAM,NR,,IG,10H LEVEL NØ,,IG, 1483 19H DIAMETER,FG,36H LEVEL,FØ,33 1484 2008 FØRMAT (1X,2GHNEN CRØSSDRAIN AT CHAINAGE,FIG,3) 1485 2008 FØRMAT (1X,14HØUTFALL LEVEL#,F12,3,10H ØUTFALL DIAMETERW,F12,3) 1486 END	1478	2005 FORMAT (1x,6HBRANCH,16,5X,15HUPSTREAM LEVELS,P9,J,5X,
1480 2006 FØRMAT (1x,14HCRRSSDRÅIN SET,IS,3x,17HCRRSSDRÅIN NUMBER,IS, 1481 13x,5HSTATE,IG) 1482 2007 FØRMAT (1x,7HHANHØLE,IG,9H OTAM,NR,,IG,10H LEVEL NØ,,IG, 1483 19H DIAMETER,FG,36H LEVEL,F9,33 1484 2008 FØRMAT (1x,2GHNEN CRØSSDRAIN AT CHAINAGE,FIG,3) 1485 2008 FØRMAT (1x,2GHNEN CRØSSDRAIN AT CHAINAGE,FIG,3) 1485 2009 FØRMAT (1x,14HØUTFALL LEVEL#,P12,3,19H ØUTFALL DIAMETER#,F12,3) 1486 END	1479	116HUPSTREAM DIAMETERS, FG. 3)
13x,5HSTATE,16) 1482 2007 FBRMAT (1x,7HMANHBLE,16,9H DIAM,NR,,16,10H LEVEL NB,,16, 1483 19H DIAMETER,F6,36H LEVEL,F9,33 1484 2008 FBRMAT (1x,26HNEW CRBSDRAIN AT CHAINAGE,F16,3) 1485 2009 FBRMAT (1x,14HBUTFALL LEVEL#,F12,3,19H BUTFALL DIAMETER#,F12,3) 1485 2009 FBRMAT (1x,14HBUTFALL LEVEL#,F12,3,19H BUTFALL DIAMETER#,F12,3)	1480	2006 FORMAT (1X, 14HCRASSORAIN SET, 18, 3X, 17HCRASSORAIN NUMBER, 15,
1482 2007 F8RHAT (1Х,7НЙАННВЦЕ,16,9Н DIAM,NR,,16,10H LEVEL N8,,16, 1483 19Н DIAMETER,F6,3,6Н LEVEL,F9,33 1484 2008 F8RHAT (1Х,26HNEH CR88SDRAIN AT CHAINAGE,F16,3) 1485 2009 F8RHAT (1Х,14H8UTFALL LEVEL#,P12,3,19H BUTFALL DIAMFTER#,F12,3) 1486 END	1401	13x,5HSTATE, 16)
1483 19H DIAMETER, F6, 3, 6H LEVEL, F9, 33 1484 2008 F8RMAT (1x, 26HNEW CR88SDRAIN AT CHAINAGE, F16, 3) 1485 2009 F8RMAT (1x, 14H8UTFALL LEVEL#, F12, 3, 18H BUTFALL DIAMETER#, F12, 3) 1486 END	1462	2007 FORMAT (1X, 7HHANHOLE, 16, 9H DIAM, NR., 16, 10H LEVEL NO., 16,
1484 2008 FORMAT (1x, 26HNEW CROBSORATN AT CHAINAGE, F16, 3) 1485 2009 FORMAT (1x, 14HOUTFALL LEVELS, F12, 3, 10H OUTFALL DIAMPTERS, F12, 3) 1486 END END <td>1463</td> <td>19H DIAMETER, FG. J. 6H LEVEL, F9. J)</td>	1463	19H DIAMETER, FG. J. 6H LEVEL, F9. J)
1485 2009 FORMAT (1X,14HOUTPALL LEVELS, P12,3,10H OUTFALL DIAMFTERS, F12,3) 1486 END	1484	2008 FORMAT (1X, 26HNEN CROSSDRAIN AT CHAINAGE, FIG. 3)
1486 END	1485	2009 FORMAT (1X, 14HOUTPALL LEVELS, P12.3, 19H OUTFALL DIAMFTERS, F12.3)
	1486	END

END OF SEGNENT, LENGTH 1105, NAME TRACE

.

.

.

-

SUBRBUTINE BUTPUT (KE,GA,NKE,NGA) SUDHUTINE BUTFUT (RE, M, ME, NG, J DIMENSION KE(NKE), GA(NGA), GL(S), GFFSET(5) COMHON/DATA/NSET, NBRAN, NB, NPIPE, NXDSET, NITEM, KITEM, MK COMHON/PARAM/PDA19,20), MEND, JEND, MJ, ND, T. RP. ZD, DMAX, THIM, OFLTA COMMON/MHEREFIT(200,11), IDX(10,12), IDA, TDL, IDT, IDK, IDG, IDJ, 110P, TOR, TOM, TOR NRITE (8,2000) WRITE 1NTRODUCTORY BASIC DATA TO FORMATTED FILE DO 10 Jal. 0 2.00 IF (PDA(1,6),LT,0,D01) 60 70 11 10 CONTINUE I=10 11 N=I=1
wRITE (4,100D) N
D0 13 I=1,N
D0 12 J=1,20
IF (J.E0.2.0R.J.GE.6) PDA(I,J)=1000.0+PDA(I,J)
wRITE (4,1001) PDA(I,J)
IF (J.GE.6.AND.PDA(I,J).LT.D.001) G0 T0 13
12 CENTINUE 13 CONTINUE NEW MANY BRANCHES IN HEN NETWERKE NSET#1 1510 1511 1512 1513 KED NOU DØ 40 I=NK,NKE IF (KE(I),EG,IOX(NSET,7)) GØ TØ 20 IF (KE(I], VE,O] KEK+2+IOX(NSET,6)-1 00 T4 40 20 KeK+IDX(NSET,6)=1 NSET=NSET+1 40 CENTINUE NBRAENBRAN+K WRITE (4.1000) NORA NSET=0 NB#D NRO 1 3 2 2 IF (NB.GT.NBRAN) GO TO 300 C----MEMBER AP A CROSSDRAIN SETT IF (ID(N8,1).67,100) GØ TØ 100 NRENR+1 C----WRITE DETAILS OF THIS BRANCH TO FARMATTED FILE CALL DETAIL (KE,GA,NKE,NGA,NB,J,ID(NB,3),X,Y,AX,NR) C----BRT BUT UPSTREAM PIPES CALL UPBRAN (KE,NKE,NG) C----STORE NEW NUMBER FOR THIS BRANCH ISIDKOND-1 KE(1)=NR GB TR 50 C----BRANCH IS A MEMBER OF A CROSS DRAIN BET 100 NSETANSET+1 NRUNEIDX(NSET,6) TENK 120 ISIN 120 ISING C----FIND NUMBER OF DOWNSTREAM CROSSDRAIN NDXWKE(I) NDXWKE(I) DE 200 NG1, NRUN Commerfind Upstream and Dennstream Manhele Numbers IF (NUX,GT,G) GE TE 130 HH181 60 70 140 130 J=IDX(N8E7,7+N)+NUX=1 HHI=KE(J) 140 JEIDX (NSET, 7+N3+NDX=1 MH2=KE(J) C====FIND OLD BRANCH NUMBER NUMBRIDX(NSET,N) C=====WRITP DETAILS OF NEW BRANCH TO FILE NRENR+1 CALL DETAIL (KE,GA,NKE,NGA,NUMB,MH1,MH2,GL(N),BFFSET(4),AX,NH) Camarabert but upstream branch details IF (MH1, EQ. 1) GB TØ 160 IF (N, GT. 1) GØ TØ 150 KK#2 WRITE (6,1002) KK K=NR+2+NRUN+1 WRITE (4,1000) K WRITE (8,1003) K KaNR+1 WRITE (4.1000) « WRITE (6.1003) K 150 K#0 WRITE (4,1000) K JF (N,GT.1) WRITE (6,1002) K G0 T0 200

.

.

1070	(AC FALL LIPRAN (KENKENUMB)
1377	200 CANTINIE
10/0	C
1880	Ne NOINe 1
1000	Kendina 1
1381	
1554	
1584	NO IDFail
1584	WEITE (4.1000) NPIPE
1000	NOTTE (6.1004) NR. HEIPE
1887	
1.2.2.2	x = A = S (# F = S = T (K) = # F = S = T (K = 1) }
1.8.88	HETTE (4.1001) %
1500	HBTTF (6.1005) AX,877SET(K)
1801	METTE (6-1006) AX. 8FFSET(K=1)
(603	VeC.O
1501	WRITE (4,1001) V,V,AX,V,GL(K),V,GL(K-1),X
1594	1F (J.EG.1) GO TO 230
1595	KK#2
1596	WRITE (6.1002) KK
1597	LENR-2+J+1
1598	WRITE (4,1000) L
1599	WRITE (8,1003) L
1600	230 LENR=1
1401	WRITE (4,1000) L
1402	IF (J.EQ.1) WRITE (6.1002) J
1603	WRITE (6,1003) L
1604	L=0
1605	WRITE (4,1000) L
1606	250 CONTINUE
1807	
1909	TF (MUR, LT. IDI(NSET, 7)) GO TE 120
1904	Commercent and MEM BURNEM ADMER ARM CHREBDHATH BEA MAIN MEMORY
1610	
1011	KULDEIGI(NSET,1)
1412	KNF ## 10000###1+#R
1013	JEIDK+KSLD+1
1014	RE(J)SKNEW
1013	CUPDATE BRANCH COUNTER
1010	
101/	
1010	
1014	300 IEG
1820	NHITE (441000) I
1051	
1022	1000 FORMAT (1X, 110)
1623	1001 FORMAT (1%,F15,6)
1624	2000 FURHAT (IHO//20%,29H####ANEW NETWORK CREATED#####)
1625	1002 FERMAT (15%, 16, 19H UPSTREAM BRANCHES)
1626	1003 FURMAT (15%,112)
1627	1004 FURMAT (5x,6HBRANCH,16,6H TYPE,16)
1954	1000 FUNMAT (15%,20HUPSTREAM CHAINAGE =,F16,3,5%, PHAFFSTS,F12,3)
1958	IUUO PURMET (IDA, CUMUWNSTREAM CHAINAGEO, F16, 3, 5%, 7MBFFSF78, F12, 3)
1970	2 70

.

END OF SEGMENT, LENGTH 623. NAME OUTPUT

.

-

1631	SUBROUTINE DETAIL (KE,GA,NKE,NGA,NB,MH1,MH2,GL, BPPBE T,AK,NP)
1432	DIMENSION KE(NKE), GA(NGA)
1433	COMMEN/WHERE/ID(200,11), 19#(10,12), IDA, COL, IDT, IDA, IDD, LUJ,
1634	1 IDP, IDR, IDB
1635	
1836	
1637	
1038	WRITE LOIDOUT THE
1030	
1441	
1642	J#ID(NB.8)+MH2=1
1643	x2=GA(J)
1644	DISTAX2=X1
1645	WRITE (4,1001) DIST
1644	CHARGE ATCHMENT WIDTH (UNLY HURKS FOR CONSTANT WIDTH)
1647	I # I D (NB = #) + 1
1646	
1049	WEITE (A. 1001) WIDTH
1481	
1652	I=ID(N8,6)
1453	RFFSET=GA(I)
1454	WRITE (4,1001) #FF8ET
1655	ConcertaBBELUTE CHAINADE
1656	
1657	AXEGA (I)
1658	WRITE (4,1001) AK
1034	WRITE (0,1003) REPUTET
1000	7 A 2 4 7 2 7 4 4 7 4 4 4 7 4 4 4 7 4 4 4 4
1001	HOTTE (6.1004) AY. HEFSET
1663	CDIRFCTION INDICATOR
1664	INDEL
1665	J=I+1
1666	IF (GA(J),LT,GA(I)) IND==1
1667	IF (ABS(GÅ(J)=GÅ(I)).LT,D,001) IND=0
1668	WRITE (4,1000) IND
1669	CGRBUND LEVEL DATA
1670	I=ID(NB,10)=1
1071	J=ID(N8,11)=1
1672	
1073	DU 60 NHIAK
1675	
1676	IF (GA(J),LT,X)=0,001) GB TB 60
1677	1F (GA(J).GT.X2+0.001) de te 70
1678	X=6A(J)=X1
1679	WRITE (4,1001) GA(1),X
1660	GL#GA(I)
1001	BU CONTINUE
1483	20 HE 10HH
1684	1000 FRANKT (1%F15-6)
1665	TOOR FORMAT (5%,6HBRANCH, 16,6H TYPE, 16)
1686	1003 FORMAT (15X, 20HUPSTREAM CHAINAGE #, F16, 3, 5X, 7HOFFSET#, F12, 3)
1687	1004 FERMAT (15X,20HDBWNSTREAN CHAINAGE=,F16,3,5%,7HEFFSET#,F12,3)
1448	END

a a a a a ang atau ta a ta ang atau ang atau ang ta ang atau ang atau ang atau ang atau ang atau ang atau ang a

.

END OF SEGMENT, LENGTH 323, NAME DETAIL

•

1480	SUBRBUTINE UPBRAN (KE,NKE,NUMB)
1490	DIMENSION KE(NKE)
1491	COMMON/WHERE/ID (200,11), IOX (10,12), IDA, IDL, IDT, IDA, IOG, IOG, ID
1892	11DP, IDR, IDM, IDM
1401	NUBEID(NUHR, 2)
1404	WRITE (6,1001) NUB
1404	1F THUS. EQ. 01 68 TO 30
1404	Ja 20 CNUH8 - 5
1407	DA 20 KHI,NUB
1400	tal.et
1400	C
1000	KAI NAKE[]]
1700	FILL CARRESPONDING NEW BRANCH NUMBER(S)
1701	
1702	
1703	THE WENEW OT LODDIN GO TO 10
1704	NDT F (A. 1000) KNEW
1705	
1706	
1707	
1708	
1709	
1710	
1711	
1712	WATTE (4,1002) ANEM
1713	WRITE LOGIUGEJ HHEW
1714	ZO CONTINUE
1715	JU NERSU JA 1000 KNEW
1/10	RETIE (4) UUUJ KNEW
1717	NEIUNN Andread and 185
1710	INUU FURMAT IIIIII Iaal Varmat IIIIII
1719	1001 LAWA 1999 AND ALAN BUANFULS
1720	IDDE LANNAL (PROFIE)
1721	END

END		SEGHENT,	LENGTH	128.	NAME	UPBRAN
	•••					

.

1722	SUBROUTINE MESAGE (M,N)	
1723	WRITE (6,1000) N,N	
1724	\$T0P	
1725	1000 FORMAT (40X,17ManageError NUMBER,18,12H assa	,16)
1726	END	

- END OF SEGMENT, LENGTH 10, NAME HESAGE

.

•

- 269 -

SUBRAUTINE LEVELS (KE,GA,NKE,NGA,NG,NK) DIMENSION KE(NKE),GA(NGA),TDS(200),ZD(200) Camman/Data/NBET,NGRAN,NG,NPIPE,NKDSET,MITEM,KITEM,NK CBMMBN/PARAM/PDA(0,20),MEND,JEND,NJ,ND,T,RP,ZD,DMAX,DMIN,DELTA CBMMBN/PARAM/PDA(0,20),MEND,JEND,NJ,ND,T,RP,ZD,DMAX,DMIN,DELTA 1727 1728 1729 COMMONYUMERE/ID(200,11), IDX(10,12), IDA, IDL, IDT, IDK, IDG, IDD, IDJ, 11DP, IDR, IDM, IDB MRITE (6,1002) COST=0.0 1731 1732 1733 1734 1735 10=0 10 L00L0+1 NPIPE===0(ID(L0,1),10) 1730 1737 CHINEPDACHPIPE.1) RK =PDA(NPIPE,2) VHINEPDA(NPIPE,3) VHAXEPDA(NPIPE,4) 1739 1740 1741 1742 NEND=10(L4,3) NUSAI BOTOIDENTIPY THE UPSTRYAM BRANCHES AND HENCE U/S TIME OF FLOW AND LEVEL 1743 1744 NBReIDILS,2) 1746 TUSET 018TUS=0.0 1748 1749 1750 L1=ID(L0,10) GLUS=GA(L1) ZUS-GLUS-CHIN 1751 AREAUS=0,0 NGL=10(L0,4) 1752 IF (NSR, E0.0) 68 78 40 Haid(18,5)=1 1753 1755 08 30 I=1,N8# KuK+1 1758 1757 LEKE(K) TUSEAMAX1(TU8,TD5(L)) 1759 ZUSMAHINI (ZUS, ZD(L)) 1760 Jain4+L=1 1761 AREAUSEAREAUSOGA(J) 30 CONTINUE 1763 C----DEFINE PIPE DIAMETER, D/S H/H, CATCHNENT AREA 1764 40 HGEHG+1 1765 DIAMEGA(MG+1) ٤ 1766 MKEHK+1 1767 NDSEKE(MK+1) NIGENUS+1 AREA=AREAUS 1769 DØ 41 NIF=NIG,ND8 J=ID(L8,0)+NIF=1 41 AREA=AREA+GA(J) J=ID(L8,8)+ND8=1 DISTD8=GA(J) 1771 1773 1775 DIST=DISTOS+DISTUS 1776 C----FIND MINIHUM GRADIENT CONSISTENT WITH HINIHUM VELOCITY Commercial Minimum Gradient Consistent Mith Minimum Velocity Gminopda(NPIPE,5) Call Veloc (Gmin,diam,RK,V,B) IF (V,GT,VMIN) GR TO 44 Call Findg(diam,VMIN,RK,GMIN) Commercial Grade (diam,GMIN,RK,RP,Dist,AREA,TUS,SLOPE,Q,GMAX,VEL) Commercial Commercial Commercial Commercial Commercial L2=L1+NGL DF 50 I=L1,L2 Katanci 1777 1778 1779 17<u>80</u> 1781 1782 1783 1784 1785 1786 K#I+NGL 1787 IF (GA(K), GT_DISTDS=0,1) GO TO 60 50 CONTINUE 1789 60 GLD3#GA(1) . . . L2=1 1790 L2=1 ZD3=GLD3=CMIN ===15 Min SL&PE 3ALUTIØN FEASIBLE? IF (ZU&=SLØPE+DIST.GT.ZD3+0.DO1) GØ TØ 65 ZDS=ZU&=SLØPE+DIST 1791 C • 1793 1794 1795 1796 68 TS 70 , 65 SLEPE=(ZUS=ZDS)/DIST CALL VELOC (SLGPE, DIAM, RK, VEL, GMAX) IF (VEL, LT, VMAX+0, DOO1) GO TO 68 CALL FINDG (DIAM, VMAX, RK, GMAX) ZUS=ZDS+GMAX+0IST 1797 1799 1800 1801 SLOPESGHAX CALL VELOC(SLOPE,DIAM,RK,VEL,QMAK) 68 TIMEs(TUS+DIST/VEL)/60.0 1802 1803 CALL RAIN (RP, TIME, RI) QBAREA+RI/3.6E6 1804 1805

.

7

- ----

	C
1000	
1007	
1000	
1810	
1010	
1412	
1919	
4414	AD DIMAYAMAYI (ZUSASI BPF+ (GA (K)-DISTUBS-GATT)-CONIN. DZMAK)
1012	
1010	Compared with any appa of the section above straight line
101/	Construction and an inter an inter and interest and in a second and a second and a second and a second and a se
1919	
1019	
1950	
1951	DR TOO TEFSTFA
1955	
1853	
1454	GAREASCARE
1952	90 CALL SIZED (FUSINFIE)SFIESDIANS
1020	CUT CANTU CALLANDANCALCALANDACALANDACALANDACALANDACALANDA
1027	1997-1999 - 1997 - 18.8787.8788.81897.748.204.6148.6188.4878.64854.64
1950	WHITE (DIDUCT) CHINALACTICATE CONTROL CONT
1029	
1030	7/02/03/04/04/04/04
1411	
1434	
1835	
1836	NUSENDS
1137	IF (NDS.LT.NEND) GE TE 40
1838	MGENGAL
1839	МКФИК+1
1840	708/L03=7(#
1841	70(10)=705
1842	IF (LO.LT.NORAN) GO TO 10
1643	PETINGN
1844	1001 FORMAT (1H0.16.F9.3.F7.3.F7.4.4FA.3.F4.0.FA.3.2F7.4.FA.3.2F10.1)
1845	1002 FRAMAT (145//47.114HRBANCH IPAGTH DTAM BI DPE 1/5 BI D/S SI U/S
1846	I GL D/A GL APFA GROUND A. FI BH CAPACITY VII. CAST &
1847	
1848	
****	P. data

END OF SEGMENT, LENGTH 563, NAME LEVELS

 1849
 3UBRBUTINE GRADE (DIAM.SHIN, RK. AP., DIST. AREA, TUB., SL&PE.G., GFULL, V)

 1850
 C=====CALCULATES REQUIRED SLAPE OF A PIPE ACCORDING TO RATIONAL METHAD

 1851
 LJGICAL NPLUB, MINUS

 1852
 KoO

 1853
 NPLUSE, FALSE,

 1854
 MINUSE, FALSE,

 1855
 SL@PEGGHIN

 1856
 5 CALL VELOCISLOPE, DIAM, RK, V, QFULL)

 1857
 TIME=TUS+OIST/V

 1858
 5 CALL VELOCISLOPE, DIAM, RK, V, QFULL)

 1859
 GAAREARI/J, GEG

 1850
 IF (Q., T., GPULL, AND, K.EG., D) RETURN

 1861
 SL@PEGFFF(Q., DIAM, RK)

 1861
 SL@PEGFFF(Q., DIAM, RK)

 1861
 SL@PEFFF(Q., DIAM, RK, V, GPULL)

 1863
 IF (Q., II.M, RK)

 1864
 IO CALL VELOCISLOPE, DIAM, RK, V, GPULL)

 1865
 TIME=TUS+DIST/V

 1866
 TIME=TUS+DIST/V

 1866
 IF (A, IE.2) GØ TØ S

 1866
 IF (AAIN (RP, TIME/60, G, RI)

 1866
 IF (ASCIC-OFULL) AD, CO, 20

 1867
 DAREARI/J, GEG

 1868
 IF (MINUS) RETURN

 1870
 20 IF (MINUS) RETURN

END OF SEGMENT, LENGTH 176, NAME GRADE

1884	SUBROUTINE FINDG(DIAN, VV, RK, SLOPE)
1885	LEGICAL NPLUS, MINUS
1486	GEO. 7854+DIAM+DIAM+VV
1887	BLOPESFFF (Q.DIAN, RK)
1888	NPLIISE, FALSE.
1889	MTNUSE, FALSE,
1890	10 CALL VELOC (SLOPE-DIAN. SK. V.D)
1491	IF (AB8(VV=V).LT001) RETURN
1892	IF (VV+V) 20,20,30
1893	20 IF (NPLUS) RETURN
1894	5L0PE=SL0PE+0,999
1895	HINUS=, TRUE,
1896	GØ TØ 10
1897	30 IF (MINUS) RETURN
1898	SLOPE=SLOPE+1,001
1899	NPLUS#, TRUE,
1900	G8 TA 10
1901	END

•

•

.

•

END OF SEGMENT, LENGTH 80, NAME FINDS

.

END OF SEGMENT, LENGTH

· ·

.

FUNCTION FFF(0,0IAM,RK) FFF=0,02065+0+0/0IAM++SetAL8610(RK/S,7/0IAM+4,1385/18/DIAM# -11,141E+6)++0,89))++(+2) RETURN END

.

: